# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

**WASHINGTON, D.C. 20549** 

# FORM 8-K

CURRENT REPORT
Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

May 22, 2017
Date of Report (Date of earliest event reported)

# SONUS NETWORKS, INC.

(Exact Name of Registrant as Specified in its Charter)

**DELAWARE**(State or Other Jurisdiction of Incorporation)

**001-34115** (Commission File Number)

**04-3387074** (IRS Employer Identification No.)

#### 4 TECHNOLOGY PARK DRIVE, WESTFORD, MASSACHUSETTS 01886

(Address of Principal Executive Offices) (Zip Code)

(978) 614-8100

(Registrant's telephone number, including area code)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):

- x Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
- o Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
- o Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
- o Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company o

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. o

## Item 1.01. Entry into a Material Definitive Agreement.

# Merger Agreement

On May 23, 2017, Sonus Networks, Inc. ("Sonus" or the "Company") entered into an Agreement and Plan of Merger (the "Merger Agreement") with Solstice Sapphire Investments, Inc., a direct wholly-owned subsidiary of the Company ("NewCo"), Solstice Sapphire, Inc., a direct wholly-owned subsidiary of NewCo ("Cayman Merger Sub"), Green Sapphire Investments LLC, a direct wholly-owned subsidiary of NewCo ("Cayman Merger Sub"), Green Sapphire LLC, a direct wholly-owned subsidiary of NewCo ("GB Merger Sub"), GENBAND Holdings Company ("GENBAND"), GENBAND Inc. ("GB") and GENBAND II, Inc. ("GB II"), pursuant to which (i) Solstice Merger Sub will merge with and into the Company, with the Company surviving such merger as a wholly-owned subsidiary of NewCo, (ii) Cayman Merger Sub will merge with and into GB Merger Sub, with GB Merger Sub surviving such merger as a wholly-owned subsidiary of NewCo, (iii) GB II will merge with and into GB Merger Sub, with GB Merger Sub surviving such merger as a wholly-owned subsidiary of NewCo (such mergers in (i) through (iv) above, collectively, the "Mergers").

The Board of Directors of the Company has unanimously approved the Merger Agreement and the transactions contemplated thereby, and the Company has agreed to hold a stockholders meeting to submit the Merger Agreement to its stockholders for their consideration.

#### Effect of the Merger

In connection with and at the time of the closing of the Mergers (the "Effective Time"), each share of Sonus common stock, par value \$0.001 per share, issued and outstanding immediately prior to the Effective Time will be converted into the right to receive one share of NewCo common stock, par value \$0.0001 per share.

At the Effective Time, all outstanding shares of capital stock or other equity interests of GENBAND (other than those equity interests held by GB and GB II), GB and GB II issued and outstanding immediately prior to the Effective Time will be converted into (i) an aggregate number of shares of NewCo common stock equal to the number of shares of Sonus common stock issued and outstanding at the Effective Time, less the number of unvested Sonus restricted stock awards assumed by NewCo pursuant to the Merger Agreement, and (ii) the right to receive a three-year promissory note issued by NewCo having an aggregate principal amount of \$22.5 million and accruing interest at a rate of 7.5% per year if paid off in full within six months of the Effective Time or 10% thereafter. As a result of these transactions, the Company's stockholders, on the one hand, and the equityholders of GENBAND, GB and GB II, on the other hand, in the aggregate each will own approximately 50% of the outstanding shares of NewCo common stock following the Effective Time.

Substantially all of the outstanding Company stock options will vest in full and if not exercised prior to the Effective Time will be canceled. Restricted stock units ("RSUs") and performance stock units ("PSUs") granted by the Company that have not vested as of the Effective Time will be assumed by NewCo and become RSUs or PSUs (as applicable) of NewCo for an equivalent number of shares of NewCo common stock, subject to the same vesting and other terms as prior to the Effective Time. Shares of Sonus restricted stock that are not vested prior to the Mergers will convert into an equivalent number of shares of NewCo common stock, subject to the same vesting and other terms as prior to the Effective Time.

1

#### Registration and Listing of NewCo Common Stock

NewCo will file with the Securities and Exchange Commission (the "SEC") a registration statement in connection with the issuance of shares of NewCo common stock in the Mergers, which will include a joint proxy statement relating to the Company's, GENBAND's, GB's and GB II's respective stockholder or equityholder meeting to adopt the Merger Agreement. NewCo will apply to have its common stock listed on NASDAQ.

#### **Conditions to the Mergers**

The obligations of each of the Company and GENBAND are subject to specified conditions, including, among other things: (i) the approval of the respective mergers by the Company's stockholders and the stockholders of GENBAND, GB and GB II, (ii) the receipt of all antitrust approvals and clearances, (iii) the absence of any injunctions being entered or law being adopted that makes the Mergers illegal and (iv) the NewCo shares to be issued in the transactions being registered with the SEC and approved for listing on NASDAQ.

# Certain Other Terms of the Merger Agreement

The Merger Agreement contains customary representations and warranties from both the Company and GENBAND. It also contains customary covenants, including covenants providing for each of the parties to use its reasonable best efforts to cause the Mergers to be consummated, and covenants requiring the parties to carry on its business in the ordinary course of business consistent with past practice during the period between the execution of the Merger Agreement and the Effective Time and not to initiate, solicit, knowingly facilitate or knowingly encourage any inquiries, proposals or offers relating to alternate transactions or, subject to certain exceptions for the Company, to engage in any discussions or negotiations with respect to alternate transactions.

The Merger Agreement contains termination rights for each of the Company and GENBAND, including in the event that (i) the Mergers are made illegal or any governmental entity issues a nonappealable final order permanently enjoining the Mergers, or (ii) if the Mergers are not consummated within nine months, subject to extension for an additional three months if antitrust approvals have not been obtained (the "Outside Date"); or upon the failure of the Company's stockholders to approve the Mergers.

GENBAND may terminate the Merger Agreement if the Board of Directors of the Company changes its recommendation, recommends an alternative acquisition proposal or fails to affirm its recommendation of the Mergers after an alternative acquisition proposal is announced, or if the Company breaches its representations, warranties or covenants in a manner that would lead to the failure of a condition, subject to a 30-business day cure period.

The Company may terminate the Merger Agreement if the stockholders of GENBAND, GB or GB II do not approve the transaction, if the Company enters into a transaction for a superior proposal and concurrently pays to GENBAND the termination fee described below or if GENBAND, GB or GB II breaches its representations, warranties or covenants in a manner that would lead to the failure of a condition, subject to a 30-business day cure period.

The Merger Agreement provides that GENBAND would be entitled to receive a termination fee of \$14.5 million if GENBAND terminates the Merger Agreement due to a change in the recommendation of, or failure to affirm the recommendation by, the Board of Directors of the Company or following the termination of the Merger Agreement in certain circumstances if the Company enters into a definitive agreement in respect of another acquisition proposal (or consummates such a transaction).

2

The foregoing description of the Merger Agreement does not purport to be complete and is qualified in its entirety by reference to the full text of the Merger Agreement, which is filed as Exhibit 2.1 to this Current Report on Form 8-K and is incorporated herein by reference.

The Merger Agreement contains representations, warranties, covenants and other terms, provisions and conditions that the parties made to each other as of specific dates. The assertions embodied therein were made solely for purposes of the Merger Agreement, and may be subject to important qualifications and limitations agreed to by the parties in connection with negotiating their respective terms. Moreover, they may be subject to a contractual standard of materiality that may be different from what may be viewed as material to stockholders, or may have been used for the purpose of allocating risk between the

parties rather than establishing matters as facts. For the foregoing reasons, no person should rely on such representations, warranties, covenants or other terms, provisions or conditions as statements of factual information at the time they were made or otherwise.

#### Governance

The Merger Agreement provides that NewCo will have a board of directors consisting initially of nine directors, initially comprised of (i) two individuals who are independent directors and designated by GENBAND, (ii) three additional individuals who are designated by GENBAND (and who will not be required to be independent), (iii) NewCo's Chief Executive Officer, who will be the Company's current Chief Executive Officer or another individual designated by the Company, (iv) NewCo's Chairman, who will be the Company's current Chairman or another individual designated by the Company, and (v) two individuals who are independent directors and designated by the Company. At the Effective Time, certain GENBAND holders will enter into a stockholders agreement with NewCo containing certain voting obligations, transfer restrictions, standstill provisions and pre-emptive rights.

#### Item 5.03. Amendments to Articles of Incorporation or By-laws; Change in Fiscal Year.

On May 22, 2017, the Board of Directors of the Company adopted an amendment to the Company's Second Amended and Restated By-laws, as previously amended and restated on December 8, 2016 (the "By-law Amendment"). The By-law Amendment, among other things, designates the Court of Chancery of the State of Delaware as the sole and exclusive forum for any stockholder to bring any derivative, fiduciary duty and other intra-corporate claims against the Company, its directors, officers and other employees, including any claims challenging the Mergers, unless the Company otherwise consents in writing to an alternate forum.

The foregoing description of the By-law Amendment does not purport to be complete and is qualified in its entirety by reference to the full text of the By-law Amendment. A copy of the By-law Amendment is filed as Exhibit 3.1 to this Current Report on Form 8-K and is incorporated herein by reference.

#### Item 8.01. Other Events.

On May 23, 2017, the Company and GENBAND held a joint investor conference call in connection with the announcement of the execution of the Merger Agreement. A transcript of the conference call is filed as Exhibit 99.1 to this Current Report on Form 8-K and is incorporated herein by reference.

Also on May 23, 2017, Raymond Dolan, the Company's President and Chief Executive Officer, sent emails to (i) the Company's employees, (ii) the Company's customers and partners, (iii) the Company's managers and (iv) the Company's sales team, which are filed as Exhibits 99.2, 99.3, 99.4 and 99.5, respectively, to this Current Report on Form 8-K and are incorporated herein by reference.

3

Also on May 23, 2017, the Company distributed a list of frequently asked questions ("FAQ") to the Company's sales managers. A copy of the FAQ is filed as Exhibit 99.6 to this Current Report on Form 8-K and is incorporated herein by reference.

## Item 9.01. Financial Statements and Exhibits.

(d) Exhibits

See the Exhibit Index attached to this Current Report on Form 8-K, which is incorporated herein by reference.

## IMPORTANT INFORMATION ABOUT THE TRANSACTION WILL BE FILED WITH THE SEC

In connection with the proposed transaction, the Company will cause NewCo to file with the SEC a Registration Statement on Form S-4 that will include a joint proxy statement of the Company and GENBAND and certain of its affiliates (the "GENBAND Parties") and a prospectus of NewCo and the Company and the parties may file with the SEC other relevant documents concerning the proposed transaction. The Company will mail the definitive joint proxy statement/prospectus to the Company stockholders and the GENBAND Party equity holders. THE COMPANY STOCKHOLDERS AND GENBAND PARTY EQUITY HOLDERS ARE URGED TO READ THE REGISTRATION STATEMENT AND JOINT PROXY STATEMENT/PROSPECTUS REGARDING THE PROPOSED TRANSACTION WHEN IT BECOMES AVAILABLE AND ANY OTHER RELEVANT DOCUMENTS FILED WITH THE SEC, AS WELL AS ANY AMENDMENTS OR SUPPLEMENTS TO THOSE DOCUMENTS, BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION. Investors and security holders may obtain a free copy of the joint proxy statement/prospectus (when available) and other filings containing information about the Company at the SEC's website at www.sec.gov. The joint proxy statement/prospectus (when available) and the other filings may also be obtained free of charge from the Company's Investor Relations website (http://investors.sonusnet.com/) or by requesting them from the Company corporate secretary at Sonus Networks, Inc., 4 Technology Park Drive, Westford, Massachusetts 01886, Attention: Corporate Secretary.

The Company, NewCo, the GENBAND Parties and certain of their respective directors and executive officers, under the SEC's rules, may be deemed to be participants in the solicitation of proxies of the Company's stockholders in connection with the proposed transaction. Information about the directors and executive officers of the Company and their ownership of Company common stock is set forth in the proxy statement for the Company's 2017 annual meeting of stockholders, as filed with the SEC on Schedule 14A on April 28, 2017.

Additional information regarding the interests of those participants and other persons who may be deemed participants in the transaction may be obtained by reading the joint proxy statement/prospectus regarding the proposed transaction when it becomes available. Free copies of this document may be obtained as described in the preceding paragraphs.

# NO OFFERS OR SOLICITATIONS

This communication shall not constitute an offer to sell or the solicitation of an offer to sell or the solicitation of an offer to buy any securities, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offer of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the Securities Act of 1933, as amended.

#### CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS

This communication contains "forward-looking statements"—that is, statements related to future, not past, events. In this context, forward-looking statements often address expected future business and financial performance and financial condition, and often contain words such as "expect," "anticipate," "intend," "plan," "believe," "seek," "see," "will," "would," or "target." Forward-looking statements by their nature address matters that are, to different degrees, uncertain, such as statements about the consummation of the proposed transaction and the expected performance of the combined companies. Uncertainties that could cause actual results to be materially different than those expressed in the Company's and the GENBAND Parties' forward-looking statements include the failure to consummate the proposed transaction or to make or take any filing or other action required to consummate such transaction in a timely matter or at all. These or other uncertainties may cause actual future results to be materially different from those expressed in the Company's and the GENBAND Parties' forward-looking statements. These risks and uncertainties include, but are not limited to, economic, competitive, legal, governmental and technological factors. Accordingly, there is no assurance that the expectations of the Company or any GENBAND Party will be realized. Many factors could cause actual results to differ materially from these forward-looking statements with respect to the proposed transaction, including risks relating to the completion of the proposed transaction on anticipated terms and timing, including obtaining equity holder and regulatory approvals, anticipated tax treatment, unforeseen liabilities, future capital expenditures, revenues, expenses, earnings, synergies, economic performance, indebtedness, financial condition, losses, future prospects, business and management strategies for the management, expansion and growth of the new combined company's operations, the ability of the Company and the GENBAND Parties to integrate the business successfully and to achieve anticipated synergies, potential litigation relating to the proposed transaction, and the risk that disruptions from the proposed transaction will harm the Company's or the GENBAND Parties' business. While the list of factors presented here is considered representative, no such list should be considered to be a complete statement of all potential risks and uncertainties. Unlisted factors may present significant additional obstacles to the realization of forward looking statements. Consequences of material differences in results as compared with those anticipated in the forward-looking statements could include, among other things, business disruption, operational problems, financial loss, legal liability to third parties and similar risks, any of which could have a material adverse effect on the Company's or the GENBAND Parties consolidated financial condition, results of operations or liquidity. Neither the Company nor any GENBAND Party assumes any obligation to provide revisions to any forward-looking statements should circumstances change, except as otherwise required by securities and other applicable laws.

5

#### **SIGNATURE**

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: May 23, 2017 SONUS NETWORKS, INC.

By: /s/ Jeffrey M. Snider

Jeffrey M. Snider
Senior Vice President, Chief Admir

Senior Vice President, Chief Administrative Officer,

General Counsel and Secretary

6

# Exhibit Index

- 2.1 Agreement and Plan of Merger, dated as of May 23, 2017, between Sonus Networks, Inc., Solstice Sapphire Investments, Inc., Solstice Sapphire, Inc., Green Sapphire LLC, GENBAND Holdings Company, GENBAND Inc. and GENBAND II, Inc.\*
- 3.1 Amendment to By-Laws, effective as of May 22, 2017.
- 99.1 Transcript of joint investor conference call held by Sonus Networks, Inc. and GENBAND Holdings Company on May 23, 2017.
- 99.2 Email from Raymond Dolan, President and Chief Executive Officer of Sonus Networks, Inc., to employees of Sonus Networks, Inc. dated May 23, 2017.
- 99.3 Email from Raymond Dolan, President and Chief Executive Officer of Sonus Networks, Inc., to customers and partners of Sonus Networks, Inc., dated May 23, 2017.
- 99.4 Email from Raymond Dolan, President and Chief Executive Officer of Sonus Networks, Inc., to managers of Sonus Networks, Inc. dated May 23, 2017.
- 99.5 Email from Raymond Dolan, President and Chief Executive Officer of Sonus Networks, Inc., to the sales team of Sonus Networks, Inc. dated May 23, 2017.
- 99.6 Frequently Asked Questions distributed to sales managers of Sonus Networks, Inc., dated May 23, 2017.

<sup>\*</sup> Certain schedules and exhibits to this agreement have been omitted pursuant to Item 601(b)(2) of Regulation S-K and the Company agrees to furnish supplementally a copy of any omitted schedule and/or exhibit upon request.

## **EXECUTION VERSION**

## AGREEMENT AND PLAN OF MERGER

# BY AND AMONG

# SONUS NETWORKS, INC.

# SOLSTICE SAPPHIRE INVESTMENTS, INC.,

# SOLSTICE SAPPHIRE, INC.,

# GREEN SAPPHIRE INVESTMENTS LLC,

## GREEN SAPPHIRE LLC,

# GENBAND HOLDINGS COMPANY,

# GENBAND INC.,

#### AND

# GENBAND II, INC.

## **DATED AS OF MAY 23, 2017**

# TABLE OF CONTENTS

		Page
ARTICLE 1 C	CERTAIN DEFINED TERMS	2
ARTICLE 2 THE MERGER		18
2.1	Mergers	18
2.2	Closing	19
2.3	Effective Time	19
2.4	Effects of the Mergers	19
2.5	Directors and Officers	20
2.6	Company Post-Merger Operations	20
ARTICLE 3 C	CONVERSION OF SECURITIES	21
3.1	Effect of Mergers on Capital Stock	21
3.2	Exchange of Certificates	24
3.3	Treatment of Equity Awards and Equity Plans	27
3.4	Adjustment of Exchange Ratios	29
3.5	Rule 16b-3 Approval	29
3.6	Withholding	30
3.7	Appraisal Rights	30
ARTICLE 4 REPRESENTATIONS AND WARRANTIES OF THE SOLSTICE PARTIES		31
4.1	Existence; Good Standing; Corporate Authority	31
4.2	Authorization, Validity and Effect of Agreements	31
4.3	Capitalization	32
4.4	Significant Subsidiaries	33
4.5	No Violation	33
4.6	No Conflict	34
4.7	SEC Documents	34
4.8	Litigation	36
4.9	Absence of Certain Changes	36
4.10	Taxes	37
4.11	Employee Benefit Plans	39
4.12	Labor Matters	41
4.13	Environmental Matters	42
4.14	Intellectual Property	42

4.15	Title to Properties	44
4.16	Insurance	45
4.17	No Brokers	45
4.18	Opinion of Financial Advisor	45
4.19	Solstice Board Recommendation	45

4.19 Solstice Board Recommendation

	4.20	Vote Required	46
	4.21	Certain Approvals	46
	4.22	Relationships with Related Parties	46
	4.23	Internal Controls	47
	4.24	Certain Contracts	47
	4.25	Foreign Corrupt Practices Act	49
	4.26	Export Control	50
	4.27	Ownership and Operations of the Company and Merger Subs	50
	4.28	No Additional Representations	50
ARTIC		PRESENTATIONS AND WARRANTIES OF THE GLASS PARTIES	51
	5.1	Existence; Good Standing; Corporate Authority	51
	5.2	Authorization, Validity and Effect of Agreements	51
	5.3	Capitalization	51
	5.4	Subsidiaries	53
	5.5	No Violation	53
	5.6	No Conflict	54
	5.7	Financial Statements	55
	5.8	Litigation	55
	5.9	Absence of Certain Changes	56
	5.10	Taxes	56
	5.11	Employee Benefit Plans	59
	5.12	Labor Matters	61
	5.13	Environmental Matters	61
	5.14	Intellectual Property	62
	5.15	Title to Properties	64
	5.16	Insurance	64
	5.17	No Brokers	64
	5.18	Solstice Stock Ownership	65
	5.19	Glass Party Board Recommendations	65
	F 20	Vita Descrived	C.F.
	5.20	Vote Required	65
	5.21	Certain Approvals	65
	5.22	Relationships with Related Parties	65
	5.23	Internal Controls	66
	5.24	Certain Contracts	66
	5.25	Foreign Corrupt Practices Act	68
	5.26	Export Control	69
A DITT	5.27	No Additional Representations	69
ARTIC		OVENANTS	69
	6.1	Conduct of Solstice's Businesses	69
	6.2	Conduct of the Glass Parties' Businesses	72
	6.3	No Solicitation By Solstice	75 70
	6.4	No Solicitation By Glass Parties	78
	6.5	Meeting of Solstice Stockholders	79
	6.6	Registration Statement	80
	6.7	Glass Stockholder Meetings	81
	6.8	Filings; Reasonable Best Efforts	82
	6.9	Takeover Law	83
	6.10	Inspection	83
	6.11	Publicity	84
	6.12	Listing Application	84
	6.13	Expenses  Divertises' and Officers' Independing and Incomes	84
	6.14	Directors' and Officers' Indemnification and Insurance	85
	6.15	Taxes	87
	6.16	Employee Benefits	88
	6.17	Notification	90
	6.18	Security Holder Litigation	90
	6.19	FIRPTA Compliance	90
	6.20	280G Vote	91
	6.21	Allocation Schedule	91
	6.22	Glass Indebtedness Tormination of 401(k) Plan	91
л DTI	6.23	Termination of 401(k) Plan NDITIONS	92 92
AKIIC	7.1	Conditions to Each Party's Obligation to Effect the Transactions	92
	/.1	Conditions to Lacti Fairy's Congation to Effect the Transactions	92
	7.2	Conditions to Obligation of the Solstice Parties to Effect the Transactions	93
	7.2	Conditions to Obligation of the Glass Parties to Effect the Transactions	94
ARTIC		RMINATION	95
	_		

8.1

8.2

Termination by Mutual Consent

Termination by Solstice or Glass

95

0.2		0.6
8.3 8.4	Termination by Solstice Termination by Glass	96 96
8.5	Effect of Termination	97
8.6	Extension; Waiver	98
	NERAL PROVISIONS	98
9.1	Survival of Representations and Warranties	98
9.2	Notices	99
9.3	Assignment; Binding Effect; Benefit	99
9.4	Entire Agreement	100
9.5	Amendments	100
9.6	Governing Law	100
9.7	Counterparts	100
9.8	Headings	
		101
9.9	Interpretation	101
9.10	Waivers	101
9.11	Incorporation of Disclosure Letters and Exhibits	102
9.12	Severability	102
9.13	Enforcement of Agreement	102
Exhibits:		
Exhibit A	Voting Agreement	
Exhibit B	Form of Stockholders Agreement	
Exhibit C	Form of Registration Rights Agreement	
Exhibit D	Form of Promissory Note	
Exhibit E	Form of Amended and Restated Solstice Certificate of Incorporation	
Exhibit F	Form of Amended and Restated Glass Memorandum and Articles of Association	
Exhibit G	Form of Amended and Restated Certificate of Incorporation of the Company	
Exhibit H	Form of Amended and Restated By-Laws of the Company	
Exhibit I	Form of Solstice Officer's Certificate	
Exhibit J	Form of Glass Officer's Certificate	
Exhibit K	Form of GB Officer's Certificate	
Exhibit L	Form of GB II Officer's Certificate	
Schedules:		
Schedule 7.1(b)	Certain Non-U.S. Antitrust Laws	

## AGREEMENT AND PLAN OF MERGER

Schedule 7.2(f)

Specified Agreements

This AGREEMENT AND PLAN OF MERGER, dated as of May 23, 2017 (this "<u>Agreement</u>"), by and among Sonus Networks, Inc., a Delaware corporation ("<u>Solstice</u>"), Solstice Sapphire Investments, Inc., a Delaware corporation and direct wholly-owned subsidiary of Solstice (the "<u>Company</u>"), Solstice Sapphire, Inc., a Delaware corporation and a direct wholly-owned subsidiary of the Company ("<u>Solstice Merger Sub</u>"), Green Sapphire Investments LLC, a Delaware limited liability company and a direct wholly-owned subsidiary of the Company ("<u>Cayman Merger Sub</u>"), Green Sapphire LLC, a Delaware limited liability company and a direct wholly-owned subsidiary of the Company ("<u>GB Merger Sub</u>"), Genband Holdings Company, a Cayman Islands exempted company limited by shares ("<u>Glass</u>"), GENBAND Inc., a Delaware corporation ("<u>GB</u>"), and GENBAND II, Inc., a Delaware corporation ("<u>GB</u> <u>II</u>"). Capitalized terms have the respective meanings ascribed to them in <u>Article 1</u>.

## RECITALS

WHEREAS, the Board of Directors of each of Solstice, Glass and the Glass Corporate Members have determined that it is consistent with and in furtherance of their respective long-term business strategies and in the best interests of their respective companies and their respective holders of capital stock or other equity interests to combine their respective businesses upon the terms and conditions set forth in this Agreement;

WHEREAS, the Company is a newly formed Subsidiary of Solstice created for the purpose of consummating the transactions contemplated by this Agreement;

WHEREAS, in furtherance of the foregoing, at the Effective Time, upon the terms and subject to the conditions of this Agreement, (i) Solstice Merger Sub will merge with and into Solstice, with Solstice surviving such merger as a wholly-owned subsidiary of the Company, (ii) Cayman Merger Sub will merge with and into Glass, with Glass surviving such merger as a direct or indirect wholly-owned subsidiary of the Company, (iii) GB will merge with and into GB Merger Sub, with GB Merger Sub surviving such merger as a direct wholly-owned subsidiary of the Company and (iv) GB II will merge with and into GB Merger Sub, with GB Merger Sub surviving such merger as a wholly-owned subsidiary of the Company;

WHEREAS, as of immediately after the Mergers and the Recontribution, (a) the Company will own directly all of the issued and outstanding shares of capital stock of Solstice Surviving Corporation and all of the issued and outstanding limited liability company interests of GB Surviving Entity, and will own directly, or, through GB Surviving Entity, indirectly, all of the shares of capital stock of Glass Surviving Corporation and (b) the holders of Solstice Common Stock as of immediately prior to the Effective Time, the Glass Stockholders as of immediately prior to the Effective Time and the Glass Members (including Glass Equityholders who became Glass Members prior to the Effective Time), collectively, will own all of the issued and outstanding shares of Company Common Stock;

WHEREAS, the parties intend that as a result of the Mergers and the utilization of the Company as a holding company: (a) Solstice Surviving Corporation, Glass Surviving Corporation and GB Surviving Entity will constitute separate Subsidiaries of the Company so

1

that each enterprise will continue to be solely responsible for its respective liabilities and contingent liabilities and (b) the assets of each of Solstice Surviving Corporation, Glass Surviving Corporation and GB Surviving Entity will not be exposed to creditor claims associated with the liabilities of the other;

WHEREAS, this Agreement shall constitute a plan of reorganization within the meaning of Treasury Regulations Section 1.368-2(g) and the parties intend that the Solstice Merger will qualify as a reorganization under Section 368(a) of the Code or as a transaction that, taken together with the other Mergers, will qualify as a transfer of property to the Company described in Section 351 of the Code;

WHEREAS, as an inducement to and condition of the parties' willingness to enter into this Agreement, concurrently with the execution and delivery of this Agreement, certain Glass Members, certain Glass Stockholders, the Company and Solstice are entering into the voting agreement attached hereto as <a href="Exhibit A">Exhibit A</a> (the "Voting Agreement"); and

WHEREAS, as an inducement to and condition of the parties' willingness to complete the Transactions, (a) each of the OEP Glass Members will enter into the stockholders agreement with the Company dated as of the Closing Date in the form attached hereto as <a href="Exhibit B">Exhibit B</a> (the "Stockholders Agreement"), (b) each of the OEP Glass Members will enter into the registration rights agreement with the Company dated as of the Closing Date in the form attached hereto as <a href="Exhibit C">Exhibit C</a> (the "Registration Rights Agreement"), and (c) the Company will duly execute and deliver to the OEP Glass Members and if applicable the other Glass Members, the promissory note in the form attached hereto as <a href="Exhibit D">Exhibit D</a> (the "Promissory Note").

NOW, THEREFORE, in consideration of the foregoing and of the representations, warranties, covenants and agreements contained in this Agreement, the parties hereto agree as follows:

## ARTICLE 1

#### **CERTAIN DEFINED TERMS**

The following words and phrases shall have the respective meanings set forth below:

"401(k) Plan" has the meaning set forth in Section 6.23.

"Affected Employees" has the meaning set forth in Section 6.16(b).

"Affiliate" means, with respect to any Person, any other Person directly or indirectly controlling, controlled by or under common control with such Person. As used in this definition and otherwise in this Agreement, "control" (including, with its correlative meanings, "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of management or policies of a Person, whether through the ownership of securities or partnership or other ownership interests, by Contract or otherwise. Solely for purposes of Section 5.21, Section 5.22, Section 5.23 and Section 7.2(h) of this

2

Agreement, JPMorgan Chase & Co. and its controlled Affiliates shall be deemed to be Affiliates of each of the OEP Glass Members.

" $\underline{Agreement}$  " has the meaning set forth in the Introduction to this Agreement.

"Allocation Schedule" means a schedule, prepared by Glass in a format reasonably acceptable to Solstice, setting forth, for each Glass Equityholder the following information: (a) such Glass Equityholder's name and address; (b) the number of shares of capital stock or other equity interests of any Glass Party held by such Glass Equityholder as of the time immediately prior to the Effective Time; (c) the number of shares of capital stock or other equity interests issuable upon conversion of any convertible securities of any Glass Party held by such Glass Equityholder as of the time immediately prior to the Effective Time; (d) the number of shares of capital stock or other equity interests of any Glass Party subject to equity or equity-linked awards settled in capital stock or other equity interests of any Glass Positis Interests) held by such Glass Equityholder as of the time immediately prior to the Effective Time (and, if applicable, the exercise price or measurement price thereof); (e) the number of shares of capital stock or other equity interests of any Glass Party held by such Glass Equityholder as of the time immediately prior to the Effective Time but after giving effect to the issuance, conversion or cancellation of any shares of capital stock or other equity interests in connection with the Transactions; (f) the number of shares of Company Common Stock to be issued to such Glass Equityholder pursuant to this Agreement; and (g) the amount of cash payable to such Glass Equityholder in respect of shares of capital stock or other equity interests subject to any equity-linked awards of any Glass Party settled in cash.

- "Antitrust Laws" has the meaning set forth in Section 6.8(b).
- "Applicable Law" means, with respect to any Person, any foreign, supranational, federal, state, provincial or local law (statutory, common or otherwise), constitution, treaty, convention, ordinance, code, rule, regulation, order, injunction, judgment, decree, ruling or other similar requirement enacted, adopted, promulgated or applied by a Governmental Authority that is binding upon or applicable to such Person.
  - "Appraisal Statutes" shall mean Section 262 of the DGCL and Section 238 of the Companies Act.
  - "Assumed Company Option" has the meaning set forth in Section 3.3(a).
  - "Assumed Company RSUs" has the meaning set forth in Section 3.3(b).
  - "Board of Directors" means, with respect to any Person, the board of directors (or comparable governing body) of such Person.
  - "Book-Entry Shares" has the meaning set forth in Section 3.2(b).
- "Business Day" means any day on which national banking institutions in New York are open to the public for conducting business and are not required or authorized to close.
  - "Cayman Merger" has the meaning set forth in Section 2.1.

- "Cayman Merger Sub" has the meaning set forth in the Introduction to this Agreement.
- "Cayman Plan of Merger" has the meaning set forth in Section 2.3.
- "Certificates" means Designated Certificates and Glass Certificates.
- "Certificates of Merger" means, collectively, the Cayman Plan of Merger, the Solstice Certificate of Merger, the Glass Certificate of Merger, the GB Certificate of Merger and the GB II Certificate of Merger.
  - "Closing" has the meaning set forth in Section 2.2.
  - "Closing Date" has the meaning set forth in Section 2.2.
  - "Code" means the Internal Revenue Code of 1986, as amended.
  - "Collective Bargaining Agreement" has the meaning set forth in Section 4.12(a).
  - "Companies Law" has the meaning set forth in Section 2.1.
  - "Company" has the meaning set forth in the Introduction to this Agreement.
  - "Company By-laws" has the meaning set forth in Section 2.6(c).
  - "Company Certificate of Incorporation" has the meaning set forth in Section 2.6(c).
  - "Company Certificates" has the meaning set forth in Section 3.2(a).
  - "Company Chairman Designee" has the meaning set forth in Section 2.6(a).
  - "Company Chief Executive Officer Designee" has the meaning set forth in Section 2.6(a).
  - "Company Common Stock" means the common stock, par value \$0.0001 per share, of the Company.
  - "Confidentiality Agreement" means the confidentiality agreement, dated as of December 20, 2016, between Glass and Solstice, as amended.
- "Contract" means any contract, agreement, note, bond, indenture, mortgage, guarantee, option, lease, license, sales or purchase order, warranty, commitment or other instrument, obligation or binding arrangement or understanding of any kind, whether written or oral.
  - "Copyrights" means copyrights, registrations and applications for registration thereof, including copyrights in Software.
  - "Costs" has the meaning set forth in Section 8.5(c).

4

"<u>Damages</u>" means any and all claims, debts, obligations and other liabilities, monetary damages, fines, fees, penalties, interest obligations, deficiencies, losses and expenses (including amounts paid in settlement, interest, court costs, costs of investigators, reasonable fees and expenses of attorneys, accountants, financial advisors and other experts, and other reasonable expenses of litigation, arbitration or other dispute resolution procedures).

- "Delaware Courts" has the meaning set forth in Section 9.6.
- "Designated Certificates" has the meaning set forth in Section 3.2(b).
- "DGCL" means the General Corporation Law of the State of Delaware.
- "<u>Dissenting Cayman Shares</u>" means shares of Glass held as of the Effective Time by a member who has given written notice of his decision to dissent in respect of the Cayman Merger and with respect to which appraisal shall have been duly demanded and perfected in accordance with Section 238 of the Companies Law and not effectively withdrawn or forfeited prior to the Effective Time.
- "<u>Dissenting Shares</u>" means shares of GB Stock or GB II Stock held as of the Effective Time by a stockholder who has not voted such shares in favor of the adoption of this Agreement and with respect to which appraisal shall have been duly demanded and perfected in accordance with Section 262 of the DGCL and not effectively withdrawn or forfeited prior to the Effective Time.
  - "DLLCA" means the Limited Liability Company Act of the State of Delaware.
- "<u>Documentation</u>" means printed, visual or electronic materials, reports, white papers, documentation, specifications, designs, flow charts, code listings, instructions, user manuals, frequently asked questions, release notes, recall notices, error logs, diagnostic reports, marketing materials, packaging, labeling, service manuals and other information describing the use, operation, installation, configuration, features, functionality, pricing, marketing or correction of a product, whether or not provided to end users.
  - "Effective Time" has the meaning set forth in Section 2.3.
- "Environmental Claims" means, in respect of any Person, any and all administrative, regulatory or judicial actions, suits, orders, decrees, demands, directives, claims, liens, proceedings or notices of noncompliance or violation by any Governmental Authority or other third party, alleging (i) liability with respect to the potential presence or Release of, or exposure to, any Hazardous Materials at any location, whether or not owned, operated, leased or managed by such Person, (ii) indemnification, cost recovery, compensation or injunctive relief resulting from the presence or Release of, or exposure to, any Hazardous Materials, or (iii) any other liability arising under Environmental Laws.
- "Environmental Laws" means all Applicable Laws relating to pollution, contamination, Hazardous Materials, natural resources, protection of the environment or human health or safety.

- "Environmental Permits" means all permits, licenses, registrations and other governmental authorizations required under applicable Environmental Laws.
  - "ERISA" means the Employee Retirement Income Security Act of 1974, as amended.
- "ERISA Affiliate" includes any enterprise that, with Solstice or any Glass Party as applicable, is, or at prior dates still relevant for liability, was part of a controlled group of corporations, a group of trades or business under common control or an affiliated service group, within the meaning of Section 414(b), (c), (m) or (o) of the Code.
  - "ESPP Options" has the meaning set forth in Section 3.3(f).
  - "Exchange Act" means the Securities Exchange Act of 1934, as amended.
  - "Exchange Agent" has the meaning set forth in Section 3.2(a).
  - "Exchange Fund" has the meaning set forth in Section 3.2(a).
  - "Excluded Employees" has the meaning set forth in Section 6.16(a).
  - "Excluded GB Shares" has the meaning set forth in Section 3.1(c)(i).
  - "Excluded GB II Shares" has the meaning set forth in Section 3.1(d)(i).
  - "Excluded Glass Shares" has the meaning set forth in Section 3.1(b)(i).
- "Excluded Shares" means, collectively, the Excluded Solstice Shares, the Excluded Glass Shares, the Excluded GB Shares and the Excluded GB II Shares.
  - "Excluded Solstice Shares" has the meaning set forth in Section 3.1(a)(i).
- "Export Control Laws" means all Applicable Laws relating to the import or export of goods, technology or services or trading embargoes or other trading restrictions, including the Arms Export Control Act, the International Traffic in Arms Regulations, the Export Administration Act, the Export Administration Regulations, the International Economic Emergency Powers Act and executive orders and regulations administered by the Office of Foreign Assets Control of the U.S. Department of the Treasury and comparable foreign laws, ordinances, rules and regulations.
  - "Foreign Corrupt Practices Act" means the Foreign Corrupt Practices Act of 1977, as amended.
  - "GAAP" has the meaning set forth in Section 4.7(a).

"GB Exchange Ratio" means which respect to a share of GB Stock, (a) the total number of shares of Company Common Stock that GB would be entitled to receive upon a Realization Event (as defined in the Glass Charter) assuming that (i) the total Realization Consideration (as defined in the Glass Charter) is equal to the Total Solstice Shares with an aggregate value equal to the VWAP on the day immediately prior to the Closing Date as determined by the Board of Directors of Glass in accordance with the Glass Charter and distributed pursuant to Article 5.2 of the Glass Charter and (ii) that the aggregate value of the Promissory Note had already been distributed pursuant to Article 5.2 of the Glass Charter as Realization Consideration in respect of a Realization Event, divided by (b) the total number of shares of GB Stock issued and outstanding immediately prior to the Effective Time (other than the Excluded GB Shares).

"GB II Exchange Ratio" means which respect to a share of GB II Stock, (a) the total number of shares of Company Common Stock that GB II would be entitled to receive upon a Realization Event (as defined in the Glass Charter) assuming that (i) the total Realization Consideration (as defined in the Glass Charter) is equal to the Total Solstice Shares with an aggregate value equal to the VWAP on the day immediately prior to the Closing Date as determined by the Board of Directors of Glass in accordance with the Glass Charter and distributed pursuant to Article 5.2 of the Glass Charter and (ii) that the aggregate value of the Promissory Note had already been distributed pursuant to Article 5.2 of the Glass Charter as Realization Consideration in respect of a Realization Event, divided by (b) the total number of shares of GB II Stock issued and outstanding immediately prior to the Effective Time (other than the Excluded GB II Shares).

"GB Merger" has the meaning set forth in Section 2.1.

"GB Merger Sub" has the meaning set forth in the Introduction to this Agreement.

"GB Stock" means the common stock, par value \$0.01 per share, of GB.

"GB Surviving Entity" has the meaning set forth in Section 2.1.

"GB II" has the meaning set forth in the Introduction to this Agreement.

"GB II Certificate of Merger" has the meaning set forth in Section 2.3.

"GB II Merger" has the meaning set forth in Section 2.1.

"GB II Stock" means the common stock, par value \$0.001 per share, of GB II.

"Glass" has the meaning set forth in the Introduction to this Agreement.

"Glass Acquisition Proposal" means any proposal or offer made by any person other than Solstice or its Subsidiaries with respect to (a) a merger, consolidation, share exchange, business combination, reorganization, recapitalization, dissolution, liquidation or similar transaction involving any Glass Party, (b) any purchase of an equity interest (including by means of a tender or exchange offer) representing an amount greater than a 15% voting or economic interest in any Glass Party or (c) any purchase of assets, securities or ownership interests representing an amount greater than 15% of the consolidated assets (including stock of the Subsidiaries of any

7

Glass Party), consolidated net revenues or earnings before interest, Taxes, depreciation and amortization of any Glass Party and its Subsidiaries taken as a whole.

"Glass Certificate of Merger" has the meaning set forth in Section 2.3.

"Glass Certificates" has the meaning set forth in Section 3.2(c).

"Glass Charter" means the Amended and Restated Memorandum and Articles of Association, dated as of July 30, 2013, of Glass.

"Glass Closing Indebtedness" has the meaning set forth in Section 6.22.

"Glass Corporate Members" means, collectively, GB and GB II.

"Glass Corporate Mergers" means, collectively, the Glass Mergers other than the Cayman Merger.

"Glass Disclosure Letter" has the meaning set forth in the introductory paragraph to Article 5.

"Glass Equity Plans" means all equity compensation plans of any Glass Party or any of its Subsidiaries.

"Glass Equityholder" means the holder of any shares of capital stock or other equity interests of any Glass Party or the holder of any Glass Option, Glass Profits Interest or any other equity or equity-linked award issued by any Glass Party or otherwise with respect to any shares of capital stock or other equity interests of any Glass Party.

"Glass Facilities" means all real property owned, leased, or operated by any Glass Party or any of its Subsidiaries and any buildings, facilities, machinery, equipment, furniture, leasehold and other improvements, fixtures, vehicles, structures, any related capital items and other tangible property located on, in, under, or above the real property of any Glass Party or any of its Subsidiaries.

"Glass Financial Statements" has the meaning set forth in Section 5.7(a).

"Glass Foreign Plan" has the meaning set forth in Section 5.11(a).

"Glass Intellectual Property" means the Glass Owned Intellectual Property and the Glass Licensed Intellectual Property.

"Glass Leased Real Property" has the meaning set forth in Section 5.15(b).

"Glass Letter of Transmittal" has the meaning set forth in Section 3.2(c).

"Glass Licensed Intellectual Property" means all Intellectual Property that is licensed to any of the Glass Parties or any of their Subsidiaries by any third party.

8

"Glass Material Adverse Effect" means a Material Adverse Effect with respect to the Glass Parties, taken as a whole.

"Glass Material Contract" has the meaning set forth in Section 5.24(a).

"Glass Members" means the holders of Glass Shares (excluding, for the avoidance of doubt, any holders of profits interests of Glass) as of immediately prior to the Closing.

"Glass Mergers" means, collectively, the Cayman Merger, the GB Merger and the GB II Merger.

"Glass Most Recent Unaudited Financial Statements" has the meaning set forth in Section 5.7(a).

"Glass Options" means options to purchase equity interests of any Glass Party granted under any Glass Equity Plan.

"Glass Organizational Documents" has the meaning set forth in Section 5.1.

"Glass Owned Intellectual Property" means all Intellectual Property owned by any Glass Party or any of its Subsidiaries, in whole or in part.

"Glass Parties" means, collectively, Glass, GB and GB II.

"Glass Permits" has the meaning set forth in Section 5.5.

"Glass Plans" has the meaning set forth in Section 5.11(a).

"Glass Profits Interests" means profits interests in Glass.

"Glass Real Property Leases" has the meaning set forth in Section 5.15(b).

"Glass Register of Members" has the meaning set forth in Section 3.1(b)(i).

"Glass Registrations" means all Intellectual Property Registrations that are currently registered or filed in the name of any Glass Party or any of its Subsidiaries, alone or jointly with others. "Glass Registrations" does not include any Intellectual Property Registrations that have expired, been invalidated, been abandoned or which are not in force.

"Glass Regulatory Filings" has the meaning set forth in Section 5.6(b).

"Glass Shares" means the shares of capital stock or other equity interests of Glass.

"Glass Source Code" has the meaning set forth in Section 5.14(b).

"Glass Stockholder" means any holder of any capital stock of any Glass Corporate Member.

n

"Glass Stockholders Agreement" means the Fifth Amended and Restated Stockholders Agreement, dated as of July 30, 2013, of Glass.

"Glass Surviving Corporation" has the meaning set forth in Section 2.1.

"Glass' knowledge" or any word or phrase of similar import means the actual knowledge as of the date of this Agreement of David Walsh, Daryl Raiford and Jody Bishop.

"Governmental Authority" means any United States or foreign federal, state, county or local, or any supranational or non-U.S., government, political subdivision, governmental, legislative, regulatory or administrative authority, body, instrumentality, department, office agency, commission, self-regulatory organization, or any court, tribunal or judicial or arbitral body.

"<u>Hazardous Materials</u>" means (a) any substance that is listed, classified or regulated under any Environmental Laws; (b) any petroleum product or by-product, asbestos-containing material, lead-containing paint or plumbing, polychlorinated biphenyls, radioactive material, toxic molds, or radon; or (c) any other substance that is the subject of regulatory action, or that could give rise to liability, under any Environmental Laws.

"Held Glass Shares" has the meaning set forth in Section 3.1(b)(i).

"HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

"Indebtedness" means, with respect to any Person, without duplication, any (a) obligation of such Person with respect to any indebtedness for borrowed money (including all obligations for principal, interest, premiums, penalties, fees, expenses and breakage costs), (b) obligation of such Person with respect to any Indebtedness evidenced by any bond, debenture, note, mortgage, indenture or other debt instrument or debt security (including all obligations for principal, interest, premiums, penalties, fees, expenses and breakage costs), (c) commitments of such Person for which it assures a financial institution against loss (including contingent reimbursement obligations with respect to banker's acceptances or letters of credit), (d) liability of such Person with respect to commodity, interest rate or currency exchange swaps, forward Contracts, collars, caps or similar hedging obligations and (e) responsibility or liability of such Person directly or indirectly as obligor, guarantor, surety or otherwise of any of the foregoing.

"Independent Director" has the meaning set forth in the Stockholders Agreement.

"Initial Directors" has the meaning set forth in Section 2.6(a).

"Intellectual Property." means all intellectual property rights anywhere in the world, including: (a) Patent Rights; (b) Trademarks and all goodwill therein; (c) Copyrights; (d) Mask Works; (e) intellectual property rights in inventions and invention disclosures, statutory invention registrations, trade secrets, and intellectual property rights in know-how; and (f) internet domain names.

10

"<u>Intellectual Property Registrations</u>" means issued or registered Patent Rights and applications for the issuance or registrations of Patent Rights, registrations of and applications to register Trademarks, Copyright registrations, and Mask Works registrations.

"Intervening Event" means any material event, development or change in circumstances (i) that first becomes known to the Board of Directors of Solstice after the date of this Agreement to the extent any such event, development or change in circumstances was not reasonably foreseeable by the Board of Directors of Solstice as of the date of this Agreement or (ii) the consequences of which were not reasonably foreseeable by the Board of Directors of Solstice as of the date of this Agreement; provided, however, that in no event shall the following events, developments or changes in circumstances constitute an Intervening Event: (A) the receipt, existence or terms of a Solstice Acquisition Proposal or any matter relating thereto or consequence thereof; (B) any change in the price, or change in trading volume, of the Solstice Common Stock (provided, however, that the exception contained in this clause (B) shall not apply to the underlying causes giving rise to or contributing to such change or prevent any of such underlying causes from being taken into account in determining whether an Intervening Event has occurred); and (C) meeting or exceeding internal or analysts' expectations, projections or results of operations (provided, however, that the exception contained in this clause (C) shall not apply to the underlying causes giving rise to or contributing to such circumstances or prevent any of such underlying causes from being taken into account in determining whether an Intervening Event has occurred).

"Joint Proxy Statement/Prospectus" has the meaning set forth in Section 6.6(a).

"Latham" has the meaning set forth in Section 6.15(d).

"Letter of Transmittal" has the meaning set forth in Section 3.2(b).

"Liens" means all liens, pledges, security interests, claims, preferential purchase rights or other rights, interests or encumbrances.

"<u>made available</u>" means (a) with respect to information made available for purposes of <u>Article 4</u>, that such information was made available by Solstice to Representatives of Glass in the virtual data site available at https://global.merrillcorp.com and (b) with respect to information made available for purposes of <u>Article 5</u>, that such information was made available by Glass to Representatives of Solstice in the virtual data site available at https://genband.firmex.com/projects/41/documents, in each of cases (a) and (b) not less than three (3) days prior to the date of this Agreement.

"Mask Works" means mask works and registrations and applications for registration thereof and any other intellectual property rights in integrated circuit topographies under the laws of any jurisdiction.

"<u>Material Adverse Effect</u>" means, with respect to any specified Person, any change, effect, occurrence, state of facts or development that, individually or in the aggregate, materially and adversely affects the business, assets and liabilities (taken together), results of operations or financial condition (including capitalization) of such specified Person and its Subsidiaries on a

11

consolidated basis, except to the extent that such change, effect, occurrence, state of facts or development results from (a) general economic, regulatory or political conditions or changes therein in the United States, the Cayman Islands or the other countries in which such Person operates (except to the extent that any such matter shall have adversely affected the business of such Person in a manner that is materially disproportionate to the degree to which such matter shall have adversely affected similarly situated businesses); (b) financial or securities market fluctuations or conditions; (c) changes in, or events or conditions affecting, the industry in which such Person and its Subsidiaries operate generally (except to the extent that any such matter shall have adversely affected the business of such Person in a manner that is materially disproportionate to the degree to which such matter shall have adversely affected similarly situated businesses); (d) any failure to achieve any revenue, earnings or other projections provided, that the exception in this clause (d) shall not prevent or otherwise affect a determination that any fact, circumstance, event, change, effect or occurrence underlying such failure has resulted in, or contributed to, a Material Adverse Effect; (e) the announcement or pendency of the Transactions; (f) with respect to Solstice, any change in the market price or trading volume of the

Solstice Common Stock; provided, that the exception in this clause (f) shall not prevent or otherwise affect a determination that any other fact, circumstance, event, change, effect or occurrence underlying such change has resulted in, or contributed to, a Solstice Material Adverse Effect, or (g) any change in Applicable Law, regulation or GAAP (or authoritative interpretation thereof) (except to the extent that any such change shall have adversely affected such specified Person in a manner that is materially disproportionate to the degree to which such matter shall have adversely affected similarly situated businesses).

- "Mergers" means, collectively, the Solstice Merger and the Glass Mergers.
- "Merger Subs" means, collectively, Solstice Merger Sub, Cayman Merger Sub and GB Merger Sub.
- "Multiemployer Plan" has the meaning set forth in Section 4.11(a).
- "NASDAQ" has the meaning set forth in Section 3.2(f).
- "Nasdaq Rules" has the meaning set forth in the Stockholders Agreement.
- "Negotiation Period" has the meaning set forth in Section 6.3(e).
- "New Equity Plan" has the meaning set forth in Section 6.16(d).
- "OEP Glass Members" means the Glass Members that are Affiliates of JPMorgan Chase & Co and One Equity Partners, other than the Glass Corporate Members.

"Open Source Material" means any Software, Documentation or other material that (a) is distributed as "free software", "open source software" or under a similar licensing or distribution model, including the GNU General Public License (GPL), GNU Lesser General Public License (LGPL), Mozilla Public License (MPL), or (b) is distributed subject to a requirement that, as a condition of the modification, distribution or other use of such material, the licensee of such material grants, the right to any third party as a condition of the modification, distribution or

12

other use of such material, that any Software, Documentation or other material incorporated into, derived from or distributed with such material be (i) disclosed or distributed in source code form, (ii) licensed for the purpose of making derivative works or (iii) redistributable at no charge or minimal charge.

- "Option Vesting Date" has the meaning set forth in Section 3.3(a).
- "Outside Date" has the meaning set forth in Section 8.2(a).
- "<u>Patent Rights</u>" means all unexpired patents, pending patent applications (including provisional patent applications), utility models, design registrations and certificates of invention and other governmental grants for the protection of inventions or industrial designs (including all related continuations, continuations-in-part, divisionals, reissues and reexaminations).
  - "Payoff Letters" has the meaning set forth in Section 6.22.
  - "Plan Merger" has the meaning set forth in Section 6.23.
  - "Plan Termination" has the meaning set forth in Section 6.23.
- "Person" or "person" means any individual, corporation (including any non-profit corporation), general partnership, limited partnership, limited liability partnership, joint venture, estate, trust, company (including any limited liability company or joint stock company), firm or other enterprise, association, organization, entity or Governmental Authority.
  - "Pre-Closing Period" has the meaning set forth in Section 6.1(a).
  - "Promissory Note" has the meaning set forth in the Recitals to this Agreement.
  - "Qualified Glass Plan" has the meaning set forth in Section 5.11(b).
  - "Qualified Solstice Plan" has the meaning set forth in Section 4.11(b).
  - "Recontribution" has the meaning set forth in Section 3.1(e).
  - "Registration Rights Agreement" has the meaning set forth in the Recitals to this Agreement.
  - "Registration Statement" has the meaning set forth in Section 6.6(a).
  - "Regulation M-A Filing" has the meaning set forth in Section 6.6(c).
  - "Regulatory Conditions" has the meaning set forth in Section 8.2(a).
- "Release" means any spilling, leaking, pumping pouring, emitting, emptying, discharging, injecting, escaping, dumping, disposing, dispersing, leaching, or migrating into, onto, or through the environment or within or upon any building, structure, facility or fixture.
  - "Representatives" has the meaning set forth in Section 6.3(a).

- "Required Glass Stockholder Approvals" has the meaning set forth in Section 5.2.
- "Required Solstice Merger Sub Stockholder Approval" has the meaning set forth in Section 4.2.
- "Required Solstice Stockholder Approval" has the meaning set forth in Section 4.2.
- "Sarbanes-Oxley Act" means the Sarbanes-Oxley Act of 2002, as amended, or any successor statute, rules or regulations thereto.
- "SEC" means the United States Securities and Exchange Commission.
- "Securities Act" means the Securities Act of 1933, as amended.
- "Significant Subsidiary" means a significant subsidiary, as defined in Rule 1-02 of Regulation S-X of the SEC.
- "Signing Offering Period" has the meaning set forth in Section 3.3(f).
- "<u>Software</u>" means computer software code, applications, utilities, libraries, development tools, diagnostics, databases and embedded systems, whether in source code, interpreted code, object code or executable form.
  - "Solstice" has the meaning set forth in the Introduction to this Agreement.
- "Solstice Acquisition Proposal" means any proposal or offer made by any person other than Glass or its Subsidiaries with respect to (a) a merger, consolidation, share exchange, business combination, reorganization, recapitalization, dissolution, liquidation or similar transaction involving Solstice or involving any of its Subsidiaries representing, directly or indirectly, greater than 15% of the consolidated assets (including stock of the Subsidiaries of Solstice), consolidated net revenues or earnings before interest, Taxes, depreciation and amortization of Solstice and its Subsidiaries taken as a whole, (b) any purchase of an equity interest (including by means of a tender or exchange offer) representing, directly or indirectly, an amount greater than a 15% voting or economic interest in Solstice or (c) any purchase of assets, securities or ownership interests representing, directly or indirectly, an amount greater than 15% of the consolidated assets (including stock of the Subsidiaries of Solstice), consolidated net revenues or earnings before interest, Taxes, depreciation and amortization of Solstice and its Subsidiaries taken as a whole.
  - "Solstice Board Recommendation" has the meaning set forth in Section 4.19(a).
  - "Solstice Board Recommendation Notice" has the meaning set forth in Section 6.3(f).
  - "Solstice Certificate of Merger" has the meaning set forth in Section 2.3.
  - "Solstice Change of Recommendation" has the meaning set forth in Section 6.3(b).

- "Solstice Common Stock" means the common stock, par value \$0.001 per share, of Solstice.
- "Solstice Customer Offerings" means the products (including Software and Documentation) that Solstice or its Subsidiaries currently markets, distributes, makes available, sells or licenses to third parties.
  - "Solstice Cutoff Date" means the date the condition set forth in Section 7.1(a) is satisfied.
  - "Solstice Disclosure Letter" has the meaning set forth in the introductory paragraph to Article 4.
  - "Solstice ESPP" means Solstice's Amended and Restated 2000 Employee Stock Purchase Plan, as amended.
  - "Solstice Exchange Ratio" has the meaning set forth in Section 3.1(a)(ii).
- "Solstice Facilities" means all real property owned, leased, or operated by Solstice or any of its Subsidiaries and any buildings, facilities, machinery, equipment, furniture, leasehold and other improvements, fixtures, vehicles, structures, any related capital items and other tangible property located on, in, under, or above such real property of Solstice or any of its Subsidiaries.
  - "Solstice Foreign Plans" has the meaning set forth in Section 4.11(a).
  - "Solstice Intellectual Property" means the Solstice Owned Intellectual Property and the Solstice Licensed Intellectual Property.
  - "Solstice Leased Real Property" has the meaning set forth in Section 4.15(b).
  - "Solstice Licensed Intellectual Property" means all Intellectual Property that is licensed to Solstice or any of its Subsidiaries by any third party.
  - "Solstice Material Adverse Effect" means a Material Adverse Effect with respect to Solstice.
  - "Solstice Material Contract" has the meaning set forth in Section 4.24(a).
  - "Solstice Merger" has the meaning set forth in Section 2.1.

- "Solstice Merger-Related Proposals" has the meaning set forth in Section 4.2.
- "Solstice Merger Sub" has the meaning set forth in the Introduction to this Agreement.
- "Solstice Nominating Committee" has the meaning set forth in Section 2.6(a)(i).
- "Solstice Options" means options to purchase shares of Solstice Common Stock granted under any Solstice Stock Plan.

- "Solstice Owned Intellectual Property" means all Intellectual Property owned by Solstice or any of its Subsidiaries, in whole or in part.
- "Solstice Parties" means Solstice, the Company and the Merger Subs.
- "Solstice Permits" has the meaning set forth in Section 4.5.
- "Solstice Plans" has the meaning set forth in Section 4.11(a).
- "Solstice Preferred Stock" means the preferred stock, par value \$0.01 per share, of Solstice.
- "Solstice Real Property Leases" has the meaning set forth in Section 4.15(b).
- "Solstice Registrations" means all Intellectual Property Registrations that are currently registered or filed in the name of Solstice or any of its Subsidiaries, alone or jointly with others. "Solstice Registrations" does not include any Intellectual Property Registrations that have expired, been invalidated, been abandoned or which are not in force.
  - "Solstice Regulatory Filings" has the meaning set forth in Section 4.6(b).
  - "Solstice Reports" has the meaning set forth in Section 4.7(a).
- "Solstice Restricted Shares" means all shares of Solstice Common Stock issued pursuant to any Solstice Stock Plan and subject to forfeiture or repurchase restrictions as of immediately prior to the Effective Time.
- "Solstice RSUs" means restricted stock units representing the right to receive Solstice Common Stock upon satisfaction of specified vesting criteria, granted under any Solstice Stock Plan.
  - "Solstice Source Code" has the meaning set forth in Section 4.14(g).
  - "Solstice Stock Plans" means all stock option and equity compensation plans of Solstice or any of its Subsidiaries.
- "Solstice Superior Proposal" means a *bona fide* written Solstice Acquisition Proposal (except that references in the definition of the "Solstice Acquisition Proposal" to 15% shall be replaced by 50%) made after the date of this Agreement by any person other than Glass or its Subsidiaries, on terms that the Board of Directors of Solstice determines in good faith, after consultation with its outside legal counsel and financial advisors, and considering such factors as the Board of Directors of Solstice considers to be appropriate (including the timing and likelihood of consummation of such proposal), are more favorable to the Solstice stockholders than the Transactions, taking into account any change to the transaction proposed by Glass (including pursuant to Section 6.3(e)) or Section 6.3(f)).
  - "Solstice Superior Proposal Notice" has the meaning set forth in Section 6.3(e).

- "Solstice Surviving Corporation" has the meaning set forth in Section 2.1.
- "Solstice Termination Amount" means a cash amount equal to \$14,500,000.
- "Solstice's knowledge" or any word or phrase of similar import means the actual knowledge as of the date of this Agreement of Raymond P. Dolan, Susan Villare and Jeff Snider.
- "Specified Solstice Plan" means each of the Solstice Assumed Performance Technologies, Incorporated 2003 Omnibus Incentive Plan, the Solstice Assumed 2008 Network Equipment Technologies Stock Incentive Plan and the Solstice 2012 Amended Performance Technologies Incorporated Omnibus Incentive Plan.
  - "Stockholders Agreement" has the meaning set forth in the Recitals to this Agreement.
- "Subsidiary" when used with respect to any party means any corporation or other organization (including a limited liability company or a partnership), whether incorporated or unincorporated, of which such party directly or indirectly owns or controls at least a majority of the securities or other interests having by their terms ordinary voting power to elect a majority of the Board of Directors or others performing similar functions with respect to such corporation or other organization, or any organization of which such party is a general partner or managing member.
  - "Surviving Entities" means, collectively, the Solstice Surviving Corporation, the Glass Surviving Corporation, and the GB Surviving Entity.
  - "Surviving Glass Ordinary Stock" means the ordinary shares, par value \$0.00001 per share, of the Glass Surviving Corporation.

"Surviving Solstice Common Stock" means the common stock, par value \$0.0001 per share, of the Solstice Surviving Corporation.

"<u>Takeover Law</u>" means any "fair price", "moratorium", "control share acquisition" or other anti-takeover statute or regulation, including Section 203 of the DGCL.

"Tax" means any and all taxes, charges, fees, duties, contributions, levies or other similar assessments or liabilities, including, without limitation, income, gross receipts, corporation, branch profits, ad valorem, premium, value-added, net worth, capital stock, capital gains, documentary, recapture, alternative or add-on minimum, disability, registration, recording, excise, real property, personal property, sales, use, license, lease, service, service use, transfer, withholding, employment, unemployment, insurance, social security, national insurance, business license, business organization, environmental, workers compensation, payroll, profits, severance, stamp, occupation, windfall profits, customs duties, franchise, estimated and other taxes of any kind whatsoever imposed by the United States of America or any state, local or foreign government, or any agency or political subdivision thereof, and any interest, fines, penalties, assessments or additions to tax imposed with respect to or related to such items.

"Tax Returns" means any and all reports, returns (including information returns), declarations, claims for refund or statements relating to Taxes, including any schedule or

17

attachment thereto and any amendment thereof, filed with or submitted to, or required to be filed with or submitted to, any Governmental Authority in connection with the determination, assessment, collection or payment of Taxes or in connection with the administration, implementation or enforcement of or compliance with any legal requirement relating to any Tax.

"<u>Total Consideration</u>" means the aggregate number of shares of Company Common Stock, together with Promissory Note, to be issued in connection with the Glass Mergers.

"<u>Total Solstice Shares</u>" means that number of shares of Solstice Common Stock issued and outstanding immediately prior to the Effective Time (other than Excluded Solstice Shares and any Solstice Restricted Shares that are subject to <u>Section 3.3(c)</u>).

"<u>Trademarks</u>" means registered trademarks and service marks, logos, corporate names and doing business designations and all registrations and applications for registration of the foregoing, and common law trademarks, service marks and trade dress.

"Transactions" means, collectively, the transactions contemplated by this Agreement, including the Mergers.

"Uncertificated Glass Shares" has the meaning set forth in Section 3.2(c).

"VWAP" means the average, for each of the 5 consecutive trading days immediately prior to the date of determination, of the per share volume-weighted average prices as displayed under the heading "Bloomberg VWAP" on Bloomberg page "SONS" (or its equivalent successor if such page is not available) in respect of the period from the scheduled open of trading until the scheduled close of trading of the primary trading session on such trading day (or if such volume-weighted average price is unavailable, the market value of one share of Solstice Common Stock on such trading day determined, using a volume-weighted average method, by a nationally recognized independent investment banking firm retained for this purpose by Glass). The VWAP will be determined without regard to afterhours trading or any other trading outside of the regular trading session trading hours.

"Willful and Material Breach" means a material breach that is a consequence of an act undertaken by the breaching party with the knowledge that the taking of such act would, or would be reasonably expected to, cause a breach of this Agreement.

"WilmerHale" has the meaning set forth in Section 6.15(d).

## ARTICLE 2

## THE MERGER

2.1 Mergers. Upon the terms and subject to the conditions set forth in this Agreement, at the Effective Time, the following shall occur in the following order: (i) in accordance with the DGCL, Solstice Merger Sub shall be merged with and into Solstice (the "Solstice Merger") and the separate corporate existence of Solstice Merger Sub shall cease, and Solstice shall continue to exist as a direct wholly-owned subsidiary of the Company ("Solstice Surviving Corporation"), (ii) in accordance with the DLLCA and the Companies Law (2016

18

Revision) of the Cayman Islands ("Companies Law"), Cayman Merger Sub shall be merged with and into Glass (the "Cayman Merger") and the separate existence of Cayman Merger Sub shall cease, and Glass shall continue to exist as a direct or indirect wholly-owned subsidiary of the Company ("Glass Surviving Corporation"), (iii) in accordance with the DGCL and DLLCA, GB shall be merged with and into GB Merger Sub (the "GB Merger") and the separate corporate existence of GB shall cease, and GB Merger Sub shall continue to exist as a direct wholly-owned subsidiary of the Company ("GB Surviving Entity") and (iv) in accordance with the DGCL and DLLCA, GB II shall be merged with and into GB Merger Sub (the "GB II Merger") and the separate corporate existence of GB II shall cease, and GB Merger Sub shall continue to exist as GB Surviving Entity, as a direct wholly-owned subsidiary of the Company. The effects and consequences of each of the Mergers shall be as set forth in Section 2.4.

2.2 <u>Closing</u>. Unless this Agreement shall have been terminated pursuant to <u>Article 8</u>, the closing of the Solstice Merger (the "<u>Closing</u>") will occur at 10:00 a.m., local time, on the second Business Day after satisfaction or waiver of the conditions set forth in <u>Article 7</u> (other than those conditions which relate to actions to be taken at the Closing or conditions whose satisfaction is to be measured as of the Closing, but the Closing shall be subject to the satisfaction or waiver of such conditions) at the offices of WilmerHale in Boston, Massachusetts, or at such other time or place as Solstice and Glass shall agree (the day on which the Closing occurs being the "<u>Closing Date</u>").

- Effective Time. On the Closing Date, as soon as practicable after the Closing, Solstice shall cause to be duly filed with the Secretary of State of the State of Delaware a certificate of merger in customary form and substance for the Solstice Merger (the "Solstice Certificate of Merger"), Glass shall cause to be duly filed with the Secretary of State of the State of Delaware a certificate of merger in customary form and substance for the Cayman Merger (the "Glass Certificate of Merger") and Glass shall cause to be filed with the Registrar of Companies of the Cayman Islands a plan of merger and related documentation in customary form and substance for the Cayman Merger (the "Cayman Plan of Merger"), and GB Merger Sub shall cause to be duly filed with the Secretary of State of the State of Delaware a certificate of merger in customary form and substance for the GB Merger (the "GB Certificate of Merger"), and GB Merger Sub shall cause to be duly filed with the Secretary of State of the State of Delaware a certificate of merger in customary form and substance for the GB II Merger (the "GB II Certificate of Merger"), in each case executed in accordance with the applicable provisions of the DGCL, the DLLCA or the Company Law, as applicable, and the parties shall make all other filings or recordings required under the DGCL, the DLLCA or the Companies Law, as applicable. The parties shall coordinate the filings of the Certificates of Merger and the effective times of the Mergers set forth in the Certificates of Merger so that the effective times of the Mergers shall occur simultaneously on the Closing Date as in the order described herein, as soon as practicable after the Closing or at such later time as Solstice and Glass shall agree, subject to Applicable Law (such time when the Mergers become simultaneously effective being the "Effective Time").
- 2.4 <u>Effects of the Mergers</u>. The Solstice Merger shall have the effects set forth in Section 259 of the DGCL, each of the Glass Corporate Mergers shall have the effects set forth in Section 18-209 of the DLLCA, and the Cayman Merger shall have the effects set forth in Section 18-209 of the DLLCA and Section 237 of the Companies Law. At

the Effective Time, (a) the certificate of incorporation of Solstice shall be amended and restated to read in its entirety as set forth in Exhibit E hereto and, as so amended and restated, such certificate of incorporation shall be the certificate of incorporation of Solstice Surviving Corporation until (subject to Section 6.14) thereafter amended as provided therein or by Applicable Law, (b) the memorandum and articles of association of Glass shall be amended and restated to read in its entirety as set forth in Exhibit F hereto and, as so amended and restated, such memorandum and articles of association shall be the memorandum and articles of association of Glass Surviving Corporation until (subject to Section 6.14) thereafter amended as provided therein or by Applicable Law and (c) the certificate of formation and limited liability company agreement of GB Merger Sub shall be the certificate of formation and limited liability company agreement of GB Surviving Entity until (subject to Section 6.14) thereafter amended as provided therein or by Applicable Law. Subject to Section 6.14, the Company shall cause the by-laws of Solstice to be amended and restated in their entirety so that, immediately following the Effective Time, they are identical to the by-laws of Solstice Merger Sub, except that all references to the name of the Solstice Merger Sub therein shall be changed to refer to the name of Solstice Surviving Corporation.

2.5 <u>Directors and Officers</u>. The directors and managers, as applicable, of each Merger Sub immediately prior to the Effective Time shall be the directors and managers of the applicable Surviving Entity as of the Effective Time and until their successors are duly elected and qualified or designated, as applicable. The officers of each Merger Sub immediately prior to the Effective Time shall be the officers of the applicable Surviving Entity as of the Effective Time and until their successors are duly appointed. At the Effective Time, each of Solstice, Glass and the Glass Corporate Members shall, except as otherwise contemplated by the two preceding sentences, cause all officers and the members of its Board of Directors (or persons holding comparable positions) to tender his or her resignation as an officer or director (or comparable position) thereof, such resignation to be effective as of the Effective Time.

## 2.6 <u>Company Post-Merger Operations</u>.

- (a) Prior to the Closing, Solstice and Glass shall take, and Solstice shall cause the Company to take, all requisite action so that, at the Effective Time, (x) the entire Board of Directors of the Company comprises the Initial Directors, (y) the Company Chief Executive Officer Designee is the Chief Executive Officer of the Company, and (z) the Company Chairman Designee is the Chairman of the Board of Directors of the Company. For purposes of this Agreement, "Initial Directors" means, collectively:
- (i) two individuals designated in writing prior to the Closing by Glass each of whom, in the good faith determination of the nominating and corporate governance committee of Solstice's Board of Directors (the "Solstice Nominating Committee"), (A) qualifies as an Independent Director and (B) otherwise meets and complies with any qualification criteria theretofore adopted by the Solstice Nominating Committee, including the requirements of Applicable Law, the Nasdaq Rules and the SEC rules, that are consistent with the Stockholders Agreement;
- (ii) three individuals designated in writing prior to the Closing by Glass each of whom, in the good faith determination of the Solstice Nominating Committee,

20

meets and complies with any qualification criteria theretofore adopted by the Solstice Nominating Committee that are consistent with the Stockholders Agreement;

- (iii) Raymond P. Dolan (or if such individual is unable or unwilling to serve as an Initial Director and the Chief Executive Officer of the Company, the individual designated in writing prior to the Closing by Solstice, acting in accordance with the recommendation of the Solstice Nominating Committee, to replace such individual who is reasonably acceptable to Glass) (the "Company Chief Executive Officer Designee");
- (iv) Richard J. Lynch (or if such individual is unable or unwilling to serve in such capacity, the individual designated in writing prior to the Closing by Solstice, acting in accordance with the recommendation of the Solstice Nominating Committee, to replace such individual who is independent for purposes of applicable stock exchange and SEC rules and reasonably acceptable to Glass) (the "Company Chairman Designee"); and
- (v) two individuals designated in writing prior to the Closing by Solstice, acting in accordance with the recommendation of the Solstice Nominating Committee, each of whom is independent for purposes of applicable stock exchange and SEC rules).
- (b) From and after the Effective Time, each of the Initial Directors so designated shall serve as a director of the Company until such person's successor shall be duly elected and qualified or such person's earlier death, resignation or removal in accordance with Applicable Law, the Company

Certificate of Incorporation, the Company By-laws and the Stockholders Agreement.

(c) In furtherance of the foregoing, prior to the Closing, Solstice shall take, and shall cause the Company to take, all requisite action to cause (i) the certificate of incorporation of the Company to be amended and restated as set forth in Exhibit G (the "Company Certificate of Incorporation") and (ii) the by-laws of the Company to be amended and restated as set forth in Exhibit H (the "Company By-laws").

#### **ARTICLE 3**

#### CONVERSION OF SECURITIES

3.1 <u>Effect of Mergers on Capital Stock.</u> At the Effective Time, the following shall occur in the following order:

(a) At the Effective Time, by virtue of the Solstice Merger and without any action on the part of any party hereto or any holder of shares of Solstice Common Stock, or any capital stock of Solstice Merger Sub, the following shall occur:

(i) Each share of Solstice Common Stock that is owned by Solstice immediately prior to the Effective Time shall automatically be canceled and extinguished and shall cease to exist ("<u>Excluded Solstice Shares</u>"), and no consideration shall be delivered in exchange therefor.

21

- (ii) Each share of Solstice Common Stock issued and outstanding immediately prior to the Effective Time (other than Excluded Solstice Shares) shall be converted into the right to receive one (1) share of Company Common Stock (the "Solstice Exchange Ratio"). As of the Effective Time, all such shares of Solstice Common Stock shall cease to be outstanding and shall automatically be canceled and extinguished and shall cease to exist, and each holder of such shares of Solstice Common Stock shall cease to have any rights with respect thereto, except the right to receive the shares of Company Common Stock to be issued in consideration therefor in accordance with Section 3.2.
- (iii) Each share of common stock, par value \$0.0001 per share, of Solstice Merger Sub issued and outstanding immediately prior to the Effective Time shall be converted into one (1) share of Surviving Solstice Common Stock.
- (b) At the Effective Time, by virtue of the Cayman Merger and without any action on the part of any party hereto or any holder of Glass Shares or limited liability company interests of Cayman Merger Sub, the following shall occur:
- (i) Each Glass Share that is owned by Glass immediately prior to the Effective Time shall automatically be canceled and extinguished and shall cease to exist ("Excluded Glass Shares"), and the Glass Register of Members shall be amended accordingly ("Glass Register of Members") and no consideration shall be delivered in exchange therefor. Each Glass Share that is owned by any Glass Corporate Member or any of their respective Subsidiaries immediately prior to the Effective Time ("Held Glass Shares") shall not be converted into the right to receive any consideration in the Cayman Merger and shall be converted into one (1) share of Surviving Glass Ordinary Stock.
- (ii) Each Glass Share issued and outstanding immediately prior to the Effective Time (other than Excluded Glass Shares and Held Glass Shares) shall be converted into the right to receive a share of Company Common Stock and (B) that portion of the Promissory Note that the holder of such Glass Share is entitled to receive upon a Realization Event (as defined in the Glass Charter) assuming that the total Realization Consideration (as defined in the Glass Charter) is equal to (X) the Total Solstice Shares with an aggregate value equal to the VWAP on the day immediately prior to the Closing Date as determined by the Board of Directors of Glass, plus (Y) the aggregate value of the Promissory Note, in each case, as determined by the Board of Directors of Glass in accordance with the Glass Charter and distributed pursuant to Article 5.2 of the Glass Charter. As of the Effective Time, all such Glass Shares (other than the Held Glass Shares) shall cease to be outstanding, shall automatically be canceled and extinguished and shall cease to exist, and each holder of such Glass Shares shall cease to have any rights with respect thereto, except the right to receive (I) shares of Company Common Stock to be issued in consideration therefor in accordance with Section 3.2 and (II) amounts payable to the holder of such Glass Share pursuant to the Promissory Note and the Glass Register of Members shall be amended accordingly.
- (iii) All outstanding limited liability company interests of Cayman Merger Sub issued and outstanding immediately prior to the Effective Time shall be converted into one (1) share of Surviving Glass Ordinary Stock.

- (c) At the Effective Time, by virtue of the GB Merger and without any action on the part of any party hereto or any holder of shares of any GB Stock or limited liability company interests of GB Merger Sub, the following shall occur:
- (i) Each share of GB Stock that is owned by GB immediately prior to the Effective Time shall automatically be canceled and extinguished and shall cease to exist ("Excluded GB Shares"), and no consideration shall be delivered in exchange therefor.
- (ii) Each share of GB Stock issued and outstanding immediately prior to the Effective Time (other than Excluded GB Shares and Dissenting Shares) shall be converted into the right to receive a share of Company Common Stock multiplied by the GB Exchange Ratio, provided that if the GB Exchange Ratio is zero then no consideration shall be delivered in exchange for a share of GB Stock. As of the Effective Time, all such shares of GB Stock shall cease to be outstanding, shall automatically be canceled and extinguished and shall cease to exist, and each holder of such shares of GB Stock shall cease to have any rights with respect thereto, except the right to receive shares of Company Common Stock (if any) to be issued in consideration therefor in accordance with Section 3.2.
- (iii) Each limited liability company interest in GB Merger Sub issued and outstanding immediately prior to the Effective Time shall remain unchanged and continue to remain outstanding as a limited liability company interest in GB Surviving Entity.

- (d) At the Effective Time, by virtue of the GB II Merger and without any action on the part of any party hereto or any holder of shares of any GB II Stock or limited liability company interests of GB II Merger Sub, the following shall occur:
- (i) Each share of GB II Stock that is owned by GB II immediately prior to the Effective Time shall automatically be canceled and extinguished and shall cease to exist ("Excluded GB II Shares"), and no consideration shall be delivered in exchange therefor.
- (ii) Each share of GB II Stock issued and outstanding immediately prior to the Effective Time (other than Excluded GB II Shares and Dissenting Shares) shall be converted into the right to receive a share of Common Company Stock multiplied by the GB II Exchange Ratio, provided that if the GB II Exchange Ratio is zero then no consideration shall be delivered in exchange for a share of GB II Stock. As of the Effective Time, all such shares of GB II Stock shall cease to be outstanding, shall automatically be canceled and extinguished and shall cease to exist, and each holder of such shares of GB II Stock shall cease to have any rights with respect thereto, except the right to receive shares of Company Common Stock (if any) to be issued in consideration therefor in accordance with Section 3.2.
- (iii) Each limited liability company interest in GB Merger Sub issued and outstanding immediately prior to the Effective Time shall remain unchanged and continue to remain outstanding as a limited liability company interest in GB Surviving Entity.
- (e) Immediately after the Effective Time, Solstice Surviving Corporation shall sell to the Company, and the Company shall purchase from Solstice Surviving Corporation, all of the shares of Company Common Stock then held by Solstice Surviving Corporation for an

amount in cash equal to the aggregate par value of all such shares of Company Common Stock (the "Recontribution").

(f) For the avoidance of doubt, but subject to <u>Section 3.2(f)</u>, the total number of shares of Company Common Stock to be received by the holders of shares of GB Stock, GB II stock and Glass Shares in the Glass Mergers shall equal in the aggregate the number of Total Solstice Shares.

## 3.2 <u>Exchange of Certificates</u>.

- (a) Prior to the Effective Time, the Company will appoint the transfer agent for Solstice Common Stock to act as exchange agent (the "Exchange Agent") for the Mergers. Promptly after the Effective Time, the Company shall deposit or cause to be deposited with the Exchange Agent, for the benefit of the stockholders (or members), as applicable, of Solstice, Glass, GB and GB II entitled to receive shares of Company Common Stock pursuant to Section 3.1(a)(ii), Section 3.1(b)(ii), Section 3.1(c)(ii), and Section 3.1(d)(ii), as applicable, for exchange in accordance with this Article 3, certificates or non-certificated book entries representing the shares of Company Common Stock to be issued pursuant to such Sections (such certificates, whether represented in certificated or non-certificated book-entry form, as applicable, the "Company Certificates"). In addition, the Company shall make available to the Exchange Agent from time to time funds necessary for payments of cash in lieu of fractional shares pursuant to Section 3.2(f) or to pay dividends or distributions pursuant to Section 3.2(d). The Company Certificates and cash deposited with the Exchange Agent are referred to collectively as the "Exchange Fund."
- (b) Promptly after the Effective Time, the Company shall cause the Exchange Agent to mail to each holder of record of one or more certificates (such certificates, the "Designated Certificates") representing (immediately prior to the Effective Time) shares of Solstice Common Stock, GB Stock or GB II Stock (other than Excluded Shares): (i) one or more letters of transmittal (each a "Letter of Transmittal") which shall specify that delivery shall be effected, and risk of loss and title to the Designated Certificates shall pass, only upon delivery of the Designated Certificates to the Exchange Agent and shall be in such form and have such other provisions as the Company may reasonably specify and (ii) instructions for use in effecting the surrender of the Designated Certificates in exchange for Company Certificates and cash in lieu of fractional shares, if any, to the extent the holder of such Designated Certificates is entitled thereto pursuant to Section 3.1(a)(ii), Section 3.1(c)(ii) and/or Section 3.1(d)(ii). Upon surrender of a Designated Certificate for cancellation to the Exchange Agent together with a Letter of Transmittal, duly executed and completed in accordance with the instructions thereto, the holder of such Designated Certificate shall be entitled to receive in exchange therefor a Company Certificate representing that number of whole shares of Company Common Stock to which the holder thereof is entitled pursuant to Section 3.1(a)(ii), Section 3.1(c)(ii) and/or Section 3.1(d)(ii) in respect of the Designated Certificate surrendered, together with a check for the cash to be paid in lieu of fractional shares, if any, after giving effect to any required withholding Tax, and the Designated Certificate so surrendered shall forthwith be canceled. No interest will be paid or accrued on the cash in lieu of fractional shares, if any, payable to holders of Designated Certificates. In the event of a transfer of ownership of shares represented by a Designated Certificate which is not registered in the transfer re

24

a Company Certificate representing the proper number of shares of Company Common Stock, together with a check for the cash to be paid in lieu of fractional shares, if any, shall be issued to such transferee if the Designated Certificate is presented to the Exchange Agent, accompanied by all documents required to evidence and effect such transfer and to evidence that any applicable stock transfer Taxes have been paid or are not payable. Promptly after the Effective Time, the Company shall cause the Exchange Agent to mail to each holder of record of shares of Solstice Common Stock represented (immediately prior to the Effective Time) by book-entry on the records of Solstice or Solstice's transfer agent ("Book-Entry Shares") (other than Excluded Solstice Shares), on behalf of the Company, notice that such holder has become the holder of record of the number of shares of Company Common Stock into which such Book-Entry Shares shall have been converted pursuant to Section 3.1(a)(ii), together with a check for the cash to be paid in lieu of fractional shares, if any, after giving effect to any required withholding Tax.

(c) Promptly after the Effective Time, the Company shall cause the Exchange Agent to mail to each holder of record (immediately prior to the Effective Time) of Glass Shares (other than Excluded Shares and Held Glass Shares): (i) one or more letters of transmittal (each a "Glass Letter of Transmittal") which shall be in customary form for a Cayman Islands-incorporated company and which shall specify that delivery shall be effected, and risk of loss and title to Glass Shares represented by a certificate or certificates (the "Glass Certificates") shall pass, only upon delivery of the Glass Certificates to the Exchange Agent and in the case of Glass Shares not represented by a certificate or certificates ("Uncertificated Glass Shares") such other documents as may be required in exchange for the applicable Company Certificates (if any), and shall be in such form and have such other provisions as the Company may reasonably specify and (ii) instructions for use in effecting the surrender of the Glass Certificates (if any) in exchange for Company Certificates and cash in lieu of fractional shares, if any, to the extent the holder of such Certificates is entitled thereto pursuant to Section 3.1(b)(ii). Upon either (i) the surrender of a Glass Certificate for cancellation to the Exchange Agent; or with respect to Uncertificated Glass Shares, confirmation by Glass that the Uncertificated Glass

Shares have been cancelled, in each case together with a Glass Letter of Transmittal, duly executed and completed in accordance with the instructions thereto, the holder of such Glass Shares shall be entitled to receive in exchange therefor a Company Certificate representing that number of whole shares of Company Common Stock to which the holder thereof is entitled pursuant to Section 3.1(b)(ii) in respect of the Glass Shares to be converted pursuant thereto, together with a check for the cash to be paid in lieu of fractional shares, if any, after giving effect to any required withholding Tax, and any Glass Certificate so surrendered shall forthwith be canceled. No interest will be paid or accrued on the cash in lieu of fractional shares, if any, payable to holders of Glass Shares. In the event of a transfer of ownership of Glass Shares which is not registered in the transfer records of Glass, a Company Certificate representing the proper number of shares of Company Common Stock, together with a check for the cash to be paid in lieu of fractional shares, if any, shall be issued to such transfere if all documents required to evidence and effect such transfer and to evidence that any applicable share transfer Taxes have been paid or are not payable are presented to the Exchange Agent.

(d) Notwithstanding any other provisions of this Agreement, no dividends or other distributions declared or made after the Effective Time with respect to the shares of Company Common Stock with a record date after the Effective Time shall be paid to the holders

25

of Glass Shares or shares of Solstice Common Stock, GB Stock or GB II Stock immediately prior to the Effective Time represented by any Certificate until such Certificate is surrendered for exchange as provided herein. Subject to the effect of Applicable Laws, following surrender of any such Certificate, there shall be paid to the holder of the Company Certificates representing whole shares of Company Common Stock issued in exchange therefor, without interest, (i) at the time of such surrender, the amount of dividends or other distributions, if any, on shares of Company Common Stock with a record date after the Effective Time and theretofore payable with respect to such whole shares of Company Common Stock and not paid, less the amount of any withholding Taxes which may be required thereon, and (ii) at the appropriate payment date, the amount of dividends or other distributions, if any, with a record date after the Effective Time and a payment date subsequent to such surrender, payable with respect to such whole shares of Company Common Stock, less the amount of any withholding Taxes which may be required thereon.

- (e) At or after the Effective Time, there shall be no transfers on the stock transfer books or register of members, as applicable, of any Surviving Entity of the Glass Shares or the shares of Solstice Common Stock, GB Stock or GB II Stock that were outstanding immediately prior to the Effective Time. If, after the Effective Time, Certificates are presented to the Company or the applicable Surviving Entity, the presented Certificates shall be canceled and exchanged for Company Certificates and cash in lieu of fractional shares, if any, deliverable in respect thereof pursuant to this Agreement in accordance with the procedures set forth in this Article 3.
- (f) No fraction of a share of Company Common Stock will be issued in connection with the Mergers, but in lieu thereof each Person otherwise entitled to receive a fraction of a share of Company Common Stock pursuant to Section 3.1(a)(ii), Section 3.1(b)(iii), Section 3.1(c)(iii) or Section 3.1(d)(iii) will be entitled to receive, in accordance with the provisions of this Section 3.2(f), from the Exchange Agent or the Company an amount in cash equal to the product obtained by multiplying (i) the fractional share interest to which such holder would otherwise be entitled pursuant to such Section by (ii) the closing price of a share of Solstice Common Stock on The NASDAQ Stock Market LLC ("NASDAQ") on the last full trading day prior to the Effective Time.
- (g) Any portion of the Exchange Fund (including the proceeds of any investments thereof and any shares of Company Common Stock) that remains unclaimed twelve months after the Effective Time shall be delivered to the Company. Any former stockholder of Solstice, Glass, GB or GB II who has not theretofore complied with this <a href="Article 3">Article 3</a> shall thereafter look only to the Company for payment of their shares of Company Common Stock or cash in lieu of fractional shares and unpaid dividends and distributions on Company Common Stock deliverable in respect of each Certificate such former stockholder or former holder of Glass Shares holds as determined pursuant to this Agreement.
- (h) None of the Company, the Surviving Entities, the Exchange Agent or any other person shall be liable to any former holder of Glass Shares or shares of Solstice Common Stock, GB Stock or GB II Stock for any shares of Company Common Stock or cash properly delivered to a public official pursuant to applicable abandoned property, escheat or similar laws. If any Certificate shall not have been surrendered prior to seven (7) years after the Effective

26

Time (or immediately prior to such earlier date on which any shares of Company Common Stock or cash payable to the holder of such Certificate pursuant to this <a href="https://example.com/Article.3">Article 3</a> would otherwise escheat to or become the property of any Governmental Authority), any such shares of Company Common Stock and any such cash shall, to the extent permitted by Applicable Law, become the property of the Company, free and clear of all claims or interest of any person previously entitled thereto.

- (i) If any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming such Certificate to be lost, stolen or destroyed and, if required by the Company, the posting by such person of a bond in such reasonable amount as the Company may direct as indemnity against any claim that may be made against it with respect to such Certificate, the Exchange Agent will issue in exchange for such lost, stolen or destroyed Certificate the shares of Company Common Stock and cash in lieu of fractional shares, if any, and unpaid dividends and distributions on shares of Company Common Stock deliverable in respect thereof pursuant to this Agreement.
- (j) The Exchange Agent shall invest any cash included in the Exchange Fund as directed by the Company on a daily basis; provided, that no such investment or loss thereon shall affect any amounts payable in respect of Solstice Common Stock, Glass Shares, GB Stock or GB II Stock pursuant to <a href="Article 2">Article 2</a> and the other provisions of this <a href="Article 3">Article 3</a>. Any interest and other income resulting from such investments shall promptly be paid to the Company.

# 3.3 <u>Treatment of Equity Awards and Equity Plans.</u>

(a) Prior to the Effective Time, Solstice and the Board of Directors or Compensation Committee of Solstice shall take all steps necessary to (i) cause all Solstice Options outstanding as of the date that is five Business Days prior to the Closing Date to (the "Option Vesting Date") become vested in full as of the Option Vesting Date (to the extent not previously vested), (ii) with respect to any Solstice Options that became vested pursuant to Section 3.3(a)(i), permit the holders of such Solstice Options to exercise such awards on or prior to the date that is three Business Days prior to the Closing

Date, (iii) cancel, effective as of the end of the day that is three Business Days prior to the Closing Date all Solstice Options outstanding as of such time that have not been exercised, other than Solstice Options granted under the Specified Solstice Plans, and (iv) cause all Solstice Options to otherwise be treated in accordance with this Section 3.3(a). Any Solstice Option under a Specified Solstice Plan that is outstanding as of the end of the day that is three Business Days prior to the Closing Date not be permitted to be exercised between such time and the Effective Time and shall, as of the Effective Time, be assumed by the Company and converted into an option (an "Assumed Company Option") to purchase a number of shares of Company Common Stock (rounded down to the nearest whole share) that is equal to the product obtained by multiplying (i) the number of shares of Solstice Common Stock subject to such Solstice Option as of immediately prior to the Effective Time, by (ii) the Solstice Exchange Ratio, which Assumed Company Option shall have an exercise price per share of Company Common Stock equal to the quotient (rounded up to the nearest whole cent) obtained by dividing (x) the exercise price per share of Solstice Company Stock of such Solstice Option as of immediately prior to the Effective Time by (y) the Solstice Exchange Ratio. Each Assumed Company Option shall otherwise continue to have, and be subject to, the same terms and conditions (including the vesting arrangements and other terms

27

and conditions set forth in the applicable Solstice Stock Plan and the Solstice Option award agreement thereunder) as are in effect immediately prior to the Effective Time, except that the Company or the Board of Directors or Compensation Committee of the Company, as applicable, shall have any and all amendment and administrative authority with respect to such Assumed Company Option (subject, in the case of any amendment, to the required consent, if any, of the affected holder of such Assumed Company Option). The assumption of the Assumed Company Options pursuant to this 3.3(a) shall be undertaken in accordance with Sections 409A and 424 of the Code and the applicable Treasury Regulations thereunder.

- (b) Effective as of immediately prior to the Effective Time, (i) each Solstice RSU that is then vested, by its terms (after taking into account any vesting of the Solstice RSU that occurs by reason of the Transactions), will have been settled in shares of Solstice Common Stock and (ii) each Solstice RSU that is not then vested, by its terms (after taking into account any vesting of the Solstice RSU that occurs by reason of the Transactions), as a result of the consummation of the Transactions and settled in shares of Solstice Common Stock shall, at the Effective Time, be assumed by the Company and shall be converted into a new award of restricted stock units of the Company ("Assumed Company RSUs") covering a number of shares of Company Common Stock equal to the product (rounded down to the nearest whole share) of (i) the total number of shares of Solstice Common Stock then underlying such Solstice RSU award multiplied by (ii) the Solstice Exchange Ratio. Subject to the following sentence, each Assumed Company RSU shall otherwise continue to have, and be subject to, the same terms and conditions (including the vesting arrangements (and accelerated vesting arrangements) and other terms and conditions set forth in the applicable Solstice Stock Plan, the Solstice RSU award agreement thereunder and any other Solstice Plan set forth on Schedule 4.11(a) governing such Solstice RSU) as are in effect immediately prior to the Effective Time, except that the Company or the Board of Directors or Compensation Committee of the Company, as applicable, shall have any and all amendment and administrative authority with respect to such Assumed Company RSUs (subject, in the case of any amendment, to the required consent, if any, of the affected holder of such Assumed Company RSU).
- (c) Effective as of immediately prior to the Effective Time, each Solstice Restricted Share that will not be vested, by its terms (after taking into account any vesting of the Solstice Restricted Share that occurs by reason of the Transactions), as a result of the consummation of the Transactions shall, at the Effective Time, be converted into Company Common Stock pursuant to Section 3.1(a)(ii) and shall otherwise continue to have, and be subject to, the same terms and conditions (including the forfeiture and repurchase restrictions set forth in the applicable Solstice Stock Plan, Solstice Restricted Share award agreement thereunder and any other Solstice Plan set forth on Schedule 4.11(a) governing such Solstice Restricted Share) as are in effect immediately prior to the Effective Time, except that the Company or the Board of Directors or Compensation Committee of the Company, as applicable, shall have any and all amendment and administrative authority with respect to such terms and conditions (subject, in the case of any amendment, to the required consent of the affected holder thereof).
- (d) As soon as practicable following the execution of this Agreement, Solstice shall mail to each Person who is a holder of Solstice Options, Solstice Restricted Shares or Solstice RSUs a letter describing the treatment of and payment for such equity awards pursuant to this <u>Section 3.3</u>. On the Option Vesting Date, Solstice shall further notify each holder of a

28

Solstice Option, other than Assumed Company Options, regarding the treatment of his or her Solstice Option set forth in Section 3.3(a) in accordance with the terms and conditions of the applicable Solstice Stock Plan. The forms of such letter shall be provided to Glass for review and comment at least five (5) Business Days prior to distribution and Solstice shall reflect all reasonable comments of Glass thereon to the extent provided within three (3) Business Days following the receipt by Glass of such forms. Solstice acknowledges and agrees that, except as set forth in Sections 3.3(a) through (c), no Solstice Options, Solstice Restricted Shares or Solstice RSUs will be assumed by the Company or otherwise remain outstanding following the Effective Time.

- (e) The parties may agree to treat equity compensation held by Solstice employees subject to non-U.S. law in a manner other than that contemplated above in this Section 3.3 to the extent necessary or appropriate to take into account applicable non-U.S. law or Tax or employment considerations.
- (f) If the Effective Time occurs on or before the "Purchase Date" relating to the applicable "Offering Period" in effect as of the date of this Agreement (the "Signing Offering Period") (as such terms are defined in the Solstice ESPP) and there are options granted under the Solstice ESPP (the "ESPP Options") then outstanding with respect to the Signing Offering Period, the Board of Directors of Solstice shall, prior to the Effective Time, take such actions as are necessary or appropriate to provide for a New Purchase Date (as defined in the Solstice ESPP) for the Signing Offering Period in accordance with Section 18.2 of the Solstice ESPP. Each holder of an ESPP Option will receive notice of the New Purchase Date, and each ESPP Option shall be automatically exercised on the New Purchase Date, pursuant to the terms of Section 18.2 of the Solstice ESPP. If the Effective Time occurs after the Exercise Date relating to the Signing Offering Period, the Board of Directors of Solstice shall timely suspend the Solstice ESPP as of immediately following the Exercise Date relating to the Signing Offering Period. All shares of Solstice Common Stock due upon exercise of ESPP Options will be issued prior to the Effective Time and no ESPP Options will be outstanding as of the Effective Time.
- Adjustment of Exchange Ratios. If any of the Company, Solstice, Glass, GB and GB II changes the number of shares of capital stock or shares, as applicable, issued and outstanding as a result of a stock split, reverse stock split, stock dividend, recapitalization, reclassification, combination or exchange of shares or other similar transaction, in each case only to the extent permitted by Section 6.1 or Section 6.2, and the record date for such an event is on or subsequent to the date of this Agreement but prior to the Effective Time, the Solstice Exchange Ratio, the GB Exchange Ratio, the GB II Exchange Ratio and the number of shares of Company Common Stock issuable pursuant to each of the Mergers, and other items dependent thereon, shall be

appropriately adjusted to provide the holders of capital stock of Solstice, GB GB II and Glass Members, as the case may be, the same economic effect as contemplated by this Agreement prior to such event.

3.5 <u>Rule 16b-3 Approval</u>. Prior to the Effective Time, Solstice and the Company, and their respective Boards of Directors or committees thereof, shall take all actions necessary or appropriate to cause any dispositions of Solstice Common Stock (including derivative securities with respect to Solstice Common Stock) or acquisitions of Company Common Stock (including derivative securities with respect to Company Common Stock) resulting from the Transactions

29

by each individual who is or will be subject to the reporting requirements of Section 16(a) of the Exchange Act to be exempt from Section 16(b) of the Exchange Act under Rule 16b-3 promulgated under the Exchange Act in accordance with the terms and conditions set forth in no-action letters issued by the SEC in similar transactions.

3.6 <u>Withholding</u>. Each of the Surviving Entities, Glass, Solstice, the Company, any Subsidiary thereof and the Exchange Agent shall be entitled to deduct and withhold from amounts otherwise payable by it under this Agreement such amounts as it is required to deduct and withhold with respect to the making of such payment under the Code, or any provision of state, local or foreign Tax law, and to collect any necessary Tax forms, including Forms W-8 or W-9, as applicable, or any similar information, from holders of Solstice Common Stock, Glass Stockholders, OEP Glass Members and any other recipients of payments hereunder. In the event that any amount is so deducted and withheld, and properly remitted, such amount will be treated for all purposes of this Agreement as having been paid to the Person to whom the payment from which such amount was withheld was made.

#### 3.7 <u>Appraisal Rights</u>.

- (a) In accordance with Section 262 of the DGCL, no appraisal rights shall be available to holders of any shares of Solstice Common Stock in connection with the Solstice Merger.
- (b) Notwithstanding anything to the contrary contained in this Agreement, Dissenting Shares and Dissenting Cayman Shares shall not be converted into or represent the right to receive any consideration in accordance with <u>Section 3.1(b)(ii)</u>, <u>Section 3.1(c)(ii)</u> or <u>Section 3.1(d)(ii)</u> but shall be entitled only to such rights as are granted by the applicable Appraisal Statute to a holder of Dissenting Shares or Dissenting Cayman Shares.
- (c) If any Dissenting Shares or Dissenting Cayman Shares shall lose their status as such (through failure to perfect or otherwise), then, as of the later of the Effective Time or the date of loss of such status, such shares shall thereupon be deemed to have been converted as of the Effective Time into the right to receive the consideration otherwise payable in respect thereof pursuant to this Agreement, without interest thereon, upon surrender of the Certificate formerly representing such shares and shall not thereafter be deemed to be Dissenting Shares or Dissenting Cayman Shares, as the case may be.
- (d) Each Glass Party shall give Solstice (i) prompt notice of any written demand for appraisal received by any Glass Party pursuant to the applicable Appraisal Statute, any withdrawal of any such demand and any other demand, notice or instrument delivered to any Glass Party prior to the Effective Time pursuant to the applicable Appraisal Statute that relates to such demand and (ii) the opportunity to participate in all negotiations and proceedings with respect to any such demand, notice or instrument. No Glass Party shall settle or make any payment or settlement offer prior to the Effective Time with respect to any such demand, notice or instrument unless Solstice shall have given its written consent to such settlement, payment or settlement offer.

30

#### **ARTICLE 4**

## REPRESENTATIONS AND WARRANTIES OF THE SOLSTICE PARTIES

Except (a) as set forth in the applicable section of the disclosure letter delivered by Solstice to Glass on the date of this Agreement (the "Solstice Disclosure Letter") (it being agreed that any information disclosed in a section of the Solstice Disclosure Letter with respect to a corresponding Section of this Agreement shall be deemed to have been disclosed with respect to any other Section of this Agreement to the extent the applicability thereto is reasonably apparent from the face of the disclosure), or (b) other than with respect to Sections 4.1 and 4.3, as disclosed in the Solstice Reports (excluding the exhibits thereto) filed by Solstice with the SEC between December 31, 2014 and the date three (3) Business Days prior to the date hereof (other than in any "risk factor" section of the Solstice Reports, any disclosures in any section of the Solstice Reports designated as relating to forward-looking statements or any other disclosures included therein to the extent they are primarily predictive, cautionary or forward-looking in nature), the Solstice Parties, jointly and severally, hereby represent and warrant to the Glass Parties as follows:

- 4.1 <u>Existence</u>; Good Standing; Corporate Authority. Each Solstice Party is an entity duly organized, validly existing and in good standing under the laws of the State of Delaware. Each Solstice Party is duly qualified to do business as a foreign entity and is in good standing under the laws of any jurisdiction in which the character of the properties owned or leased by it therein or in which the transaction of its business makes such qualification necessary (to the extent the "good standing" concept is applicable in the case of any jurisdiction outside the United States), except where the failure to be so qualified would not, individually or in the aggregate, reasonably be expected to have a Solstice Material Adverse Effect. Each Solstice Party has all requisite corporate (or comparable) power and authority to own, operate and lease its properties and to carry on its business as now conducted. The copies of the certificates of incorporation and by-laws (or comparable governing documents) of each Solstice Party previously made available to Glass are true and correct and contain all amendments as of the date hereof.
- 4.2 <u>Authorization, Validity and Effect of Agreements</u>. Each Solstice Party has the requisite corporate (or comparable) power and authority to execute and deliver this Agreement and all other agreements and documents contemplated hereby to which it is or will be a party. The consummation by each Solstice Party of the Transactions has been duly authorized by all requisite corporate (or comparable) action on the part of such Solstice Party, other than (a) in the case of Solstice, (i) the adoption of this Agreement and approval of the Solstice Merger by the holders of a majority of the outstanding shares of Solstice Common Stock entitled to vote thereon at a stockholders' meeting duly called and held for such purpose and (ii) if applicable, the approval by Solstice's stockholders of the issuance of shares of Company Common Stock pursuant to the Transactions and any proposal related to the Solstice Merger that

"Required Solstice Stockholder Approval") and (b) in the case of Solstice Merger Sub, the adoption of this Agreement and the approval of the Solstice Merger by the Company, as Solstice Merger Sub's sole stockholder, which will be obtained by written consent and delivered to Glass immediately after the execution of this Agreement (the "Required Solstice Merger Sub Stockholder Approval"). This Agreement constitutes the valid and legally binding obligation of each Solstice Party, enforceable against such Solstice Party in accordance with its terms, subject to applicable bankruptcy, insolvency, moratorium or other similar laws relating to creditors' rights and general principles of equity.

#### 4.3 <u>Capitalization</u>.

- (a) The authorized capital stock of Solstice consists of 120,000,000 shares of Solstice Common Stock, and 5,000,000 shares of Solstice Preferred Stock. As of May 19, 2017, (i) there were 51,586,663 shares of Solstice Common Stock (including 2,058,354 Solstice Restricted Shares) issued and outstanding and no shares of Solstice Preferred Stock issued and outstanding, (ii) 5,445,878 shares of Solstice Common Stock were reserved for issuance upon exercise of outstanding Solstice Options, and (iii) 328,605 shares of Solstice Common Stock were reserved for issuance upon settlement of outstanding Solstice Restricted Stock Units, comprised of time-based Solstice Restricted Stock Units and performance-based Solstice Restricted Stock Units, with the issuance of shares of Solstice Common Stock pursuant to outstanding performance-based Solstice Restricted Stock Units, dependent upon the satisfaction of certain market conditions for the current and subsequent fiscal years. All issued and outstanding shares of Solstice Common Stock (A) are duly authorized, validly issued, fully paid, nonassessable and free of preemptive rights, (B) were not issued in violation of the terms of any agreement or other understanding binding upon Solstice and (C) were issued in compliance with the certificate of incorporation and by-laws of Solstice and all applicable federal and state securities laws, rules and regulations.
- (b) There are no outstanding subscriptions, options, warrants, calls, convertible securities or other similar rights, agreements, plans or commitments relating to the issuance of capital stock or other equity interests to which Solstice or any of its Subsidiaries is a party obligating Solstice or any of its Subsidiaries to (i) issue, transfer or sell any shares of capital stock or other equity interests of Solstice or any of its Subsidiaries or securities convertible into or exchangeable for such shares or equity interests, (ii) grant, extend or enter into any such subscription, option, warrant, call, convertible securities or other similar right, agreement or arrangement or (iii) redeem or otherwise acquire any such shares of capital stock or other equity interests (including securities or obligations convertible into or exchangeable or exercisable for any shares of capital stock or other equity interests), other than pursuant to this Agreement.
- (c) There are no stockholder agreements, voting trusts or other agreements or understandings to which Solstice or any of its Subsidiaries is a party or, to Solstice's knowledge, otherwise with respect to the voting of the capital stock or other equity interests of Solstice or any of its Subsidiaries.
- (d) A complete and correct list of all the outstanding Solstice Restricted Shares, Solstice Options, Solstice RSUs and any other awards granted under the Solstice Stock

32

Plans (including the grant date, the number of shares of Solstice Common Stock subject thereto, the exercise or purchase price (if any), the settlement date (if different than the vesting date(s)), the expiration date (if any) and any other individual terms not otherwise specified in the forms of award agreement) and the names of the holders thereof as of the date hereof has been made available to Glass. Solstice has made available to Glass a copy of each Solstice Stock Plan and the forms of awards agreement for grants of awards thereunder (and, where not included in such forms, a description of the terms and conditions of vesting (including any acceleration upon an acquisition, change in control or similar transaction) for each outstanding award). All outstanding Solstice Restricted Shares, Solstice Options and Solstice RSUs have been granted pursuant to individual award agreements in the forms of award agreements made available to Glass and/or filed as exhibits with the SEC.

- (e) The shares of Company Common Stock issuable pursuant to the Mergers, when issued, sold and delivered in accordance with the terms of this Agreement, will be duly authorized, validly issued, fully paid and nonassessable and, assuming the accuracy of the representations and warranties of the Glass Stockholders and the OEP Glass Members in their respective Letters of Transmittal, will be issued in compliance with all applicable federal and state securities laws, rules and regulations.
- Significant Subsidiaries. Each of Solstice's Significant Subsidiaries is a corporation or limited liability company duly organized, validly existing and in good standing (where applicable) under the laws of its jurisdiction of incorporation or organization, has the corporate (or comparable) power and authority to own, operate and lease its properties and to carry on its business as it is now being conducted, and is duly qualified to do business and is in good standing (where applicable) in each jurisdiction in which the ownership, operation or lease of its property or the conduct of its business requires such qualification (to the extent the "good standing" concept is applicable in the case of any jurisdiction outside the United States), except for jurisdictions in which such failure to be so qualified or to be in good standing would not, individually or in the aggregate, reasonably be expected to have a Solstice Material Adverse Effect. All of the outstanding shares of capital stock or other securities of, or other ownership interests in, each of Solstice's Significant Subsidiaries are duly authorized, validly issued, fully paid and nonassessable and are owned, directly or indirectly, by Solstice free and clear of all Liens. There are no existing options, rights of first refusal, conversion rights, preemptive rights, calls, puts, commitments, arrangements or obligations of any character, including voting agreements, proxies or similar arrangements relating to the issued or unissued capital stock or other securities of, or other ownership interests in, any Subsidiary of Solstice. Solstice does not directly or indirectly own any equity or similar interest in, or any interest convertible into or exchangeable or exercisable for any equity or similar interest in, any corporation, partnership, joint venture or other business association or entity that directly or indirectly conducts any activity which is material to Solstice and its Subsidiaries taken as a whole. The copies of Solstice's Subsidiaries' certificates of incorporation, by-laws and similar governing documents previously made available to Glass are true and correct and contain all amendments as of the date hereof. Exhibit 21.1 to Solstice's Annual Report on Form 10-K for the year ended December 31, 2016 filed with the SEC prior to the date hereof sets forth for each Significant Subsidiary of Solstice, its name and jurisdiction of incorporation or organization.

4.5 No Violation. Neither Solstice nor any of its Subsidiaries is, or has received notice or is otherwise aware that it would be with the passage of time, in violation of any term, condition or provision of (a) the certificate of incorporation or by-laws of Solstice, (b) the certificate of incorporation, by-laws or comparable governing documents of any Subsidiary of Solstice, (c) any loan or credit agreement, note, bond, mortgage, indenture, Contract, agreement, joint venture, lease, license, franchise, Solstice Permit or other instrument or (d) any Applicable Law, except, in the case of matters described in clauses (c) or (d), as would not, individually or in the aggregate, reasonably be expected to have a Solstice Material Adverse Effect. Solstice and its Subsidiaries hold all permits, licenses, variances, exemptions, orders, franchises and approvals of all Governmental Authorities necessary for the lawful conduct of their respective businesses (the "Solstice Permits"), except where the failure so to hold would not, individually or in the aggregate, reasonably be expected to have a Solstice Material Adverse Effect. Solstice and its Subsidiaries are in compliance with the terms of the Solstice Permits, except where the failure so to comply would not, individually or in the aggregate, reasonably be expected to have a Solstice Material Adverse Effect. No investigation by any Governmental Authority with respect to Solstice or any of its Subsidiaries is pending or, to the knowledge of Solstice, threatened, other than those the outcome of which would not, individually or in the aggregate, reasonably be expected to have a Solstice Material Adverse Effect.

#### 4.6 No Conflict.

- (a) Neither the execution and delivery by any Solstice Party of this Agreement nor the consummation by such Solstice Party of the Transactions in accordance with the terms hereof will: (i) conflict with or result in a breach of any provisions of or create any rights in favor of any other party under the certificate of incorporation or by-laws of Solstice; (ii) conflict with or result in a breach of any provisions of or create any rights in favor of any other party under the certificate of incorporation, by-laws or similar governing documents of any Subsidiary of Solstice; (iii) violate, or conflict with, or result in a breach of any provision of, or constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, or result in the termination or in a right of termination or cancellation of, or give rise to a right of purchase under, or accelerate the performance required by, or result in the creation of any Lien upon any of the properties of Solstice or its Subsidiaries under, or result in being declared void, voidable, or without further binding effect, or otherwise result in a detriment to Solstice or any of its Subsidiaries under any of the terms, conditions or provisions of, any loan or credit agreement, note, bond, mortgage, indenture, deed of trust, license, franchise, Solstice Permit, lease, Contract, agreement, joint venture or other instrument or obligation to which Solstice or any of its Subsidiaries is a party, or by which Solstice or any of its Subsidiaries or any of their properties is bound or affected; or (iv) contravene or conflict with or constitute a violation of any provision of any Applicable Laws, except, in the case of matters described in clauses (iii) or (iv), as would not, individually or in the aggregate, reasonably be expected to have a Solstice Material Adverse Effect and except in the case of clause (iv) for the matters described in clauses (ii) and (iii) of Section 4.6(b) below.
- (b) Neither the execution and delivery by any Solstice Party of this Agreement nor the consummation by such Solstice Party of the Transactions in accordance with the terms hereof will require any consent, approval or authorization of, or filing or registration with, any Governmental Authority, other than (i) the filings provided for in <a href="Article 2">Article 2</a> and (ii)

34

filings, consents, approvals and authorization required under applicable Antitrust Laws, the Exchange Act, the Securities Act or other applicable U.S. state securities and "blue sky" laws or the rules of NASDAQ ((i) and (ii) collectively, the "Solstice Regulatory Filings"), and listing on NASDAQ of the Company Common Stock to be issued pursuant to the Mergers, or pursuant to stock options or convertible or exchangeable securities of Solstice or Glass, except for any consent, approval or authorization the failure of which to obtain and for any filing or registration the failure of which to make would not prevent or materially delay the consummation of the Mergers or otherwise prevent Solstice from performing its obligations under this Agreement and would not, individually or in the aggregate, reasonably be expected to have a Solstice Material Adverse Effect.

(c) Other than as contemplated by <u>Section 4.6(b)</u> and except for the Required Solstice Stockholder Approval and the Required Solstice Merger Sub Stockholder Approval, no consents, assignments, waivers, authorizations or other certificates are necessary in connection with the Transactions to provide for the continuation in full force and effect of all of the Solstice Material Contracts or for any Solstice Party to consummate the Transactions, except where the failure to receive such consents, assignments, waivers, authorizations or other certificates would not, individually or in the aggregate, reasonably be expected to have a Solstice Material Adverse Effect.

#### 4.7 <u>SEC Documents</u>.

Since December 31, 2014, Solstice has filed all forms, reports and documents with the SEC that have been required to be filed by it under Applicable Laws prior to the date hereof, and Solstice will timely file prior to the Effective Time all forms, reports and documents with the SEC that are required to be filed by it under Applicable Laws prior to such time (all such forms, reports and documents, together with all exhibits and schedules thereto, the "Solstice Reports"). Each of the consolidated balance sheets included in or incorporated by reference into the Solstice Reports (including related notes and schedules) complied as to form in all material respects with the applicable accounting requirements and the published rules and regulations of the SEC with respect thereto and fairly presents in all material respects the consolidated financial position of Solstice and its Subsidiaries (or such entities as indicated in such balance sheet) as of its date, and each of the consolidated statements of operations, cash flows and changes in stockholders' equity included in or incorporated by reference into the Solstice Reports (including any related notes and schedules) fairly presents in all material respects the results of operations, cash flows or changes in stockholders' equity, as the case may be, of Solstice and its Subsidiaries (or such entities as indicated in such balance sheet) for the periods set forth therein (subject, in the case of unaudited statements, to (x) such exceptions as may be permitted by Form 10-Q of the SEC and (y) normal, recurring year-end audit adjustments which are not material in the aggregate), in each case in accordance with U.S. generally accepted accounting principles ("GAAP") consistently applied during the periods involved, except as may be noted therein. Except as and to the extent set forth on the consolidated balance sheet of Solstice and its Subsidiaries included in the most recent Solstice Report filed prior to the date of this Agreement that includes such a balance sheet, including all notes thereto, neither Solstice nor any of its Subsidiaries has any liabilities or obligations of any nature (whether accrued, absolute, contingent or otherwise) that would be required to be reflected on, or reserved against in, a consolidated balance sheet of Solstice or in the notes thereto prepared in accordance with

GAAP consistently applied, other than liabilities or obligations that may have arisen in the ordinary course of business since the date of such most recent Solstice Report and which are not and would not, individually or in the aggregate, reasonably be expected to have a Solstice Material Adverse Effect.

- As of its filing date (or, if amended or superseded by a filing prior to the date of this Agreement, on the date of the last such amendment or superseding filing) (i) each Solstice Report complied, or will comply, as the case may be, as to form in all material respects with the applicable requirements of the Securities Act, the Exchange Act, the Sarbanes-Oxley Act and the Dodd-Frank Act of 2010, as the case may be, and the applicable rules and regulations promulgated thereunder, each as in effect on the date such Solstice Report was or will be filed, and (ii) each Solstice Report did not, and will not, as the case may be, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading. True and correct copies of all Solstice Reports filed prior to the date hereof have been furnished to Glass or are publicly available in the Electronic Data Gathering, Analysis and Retrieval (EDGAR) database of the SEC. Solstice has made available to Glass complete and correct copies of all material correspondence between the SEC, on the one hand, and Solstice, on the other hand, occurring since December 31, 2014. As of the date hereof, there are no material outstanding or unresolved comments in comment letters from the SEC staff with respect to any of the Solstice Reports. To the knowledge of Solstice, as of the date hereof, none of the Solstice Reports is the subject of ongoing SEC review, outstanding SEC comment or outstanding SEC investigation. None of Solstice's Subsidiaries is required to file any forms, reports, registrations, statements or other documents with the SEC. No executive officer of Solstice has failed to make the certifications required of him or her under Section 302 or 906 of the Sarbanes-Oxley Act with respect to any Solstice Report, except as disclosed in certifications filed with the Solstice Reports. Neither Solstice nor any of its executive officers has received notice from any Governmental Authority challenging or questioning the accuracy, completeness, form or manner of filing of such certifications. Except as previously disclosed in the Solstice Reports since the enactment of the Sarbanes-Oxley Act, Solstice and each of its officers and, to the knowledge of Solstice, each of its directors, have been and are in compliance in all material respects with (A) the applicable provisions of the Sarbanes-Oxley Act and the rules and regulations promulgated thereunder and (B) the applicable listing and corporate governance rules and regulations of NASDAQ.
- 4.8 <u>Litigation</u>. There are no actions, suits or proceedings pending against Solstice or any of its Subsidiaries or, to Solstice's knowledge, threatened against Solstice or any of its Subsidiaries, at law or in equity, or before or by any Governmental Authority, that would reasonably be expected to have, individually or in the aggregate, a Solstice Material Adverse Effect. There are no outstanding judgments, decrees, injunctions, awards or orders against Solstice or any of its Subsidiaries except as would not, individually or in the aggregate, reasonably be expected to have a Solstice Material Adverse Effect.
- 4.9 <u>Absence of Certain Changes</u>. Since December 31, 2016, each of Solstice and its Subsidiaries has conducted their business according to their usual, regular and ordinary course consistent with past practice and there has not been any event that has had or would be reasonably expected to have, individually or in the aggregate, a Solstice Material Adverse Effect.

36

Without limiting the generality of the foregoing, since December 31, 2016, neither Solstice nor any of its Subsidiaries has taken any action (or has failed to take any action) that would require the consent of Glass under <u>Section 6.1</u> if taken (or failed to be taken) after the date of this Agreement and prior to the Effective Time.

# 4.10 <u>Taxes</u>.

- (a) Each of Solstice and its Subsidiaries is classified as set forth in <u>Section 4.10(a)</u> of the Solstice Disclosure Letter for all U.S. federal and applicable state income Tax purposes.
- (b) Each of Solstice and its Subsidiaries has properly filed on a timely basis all material Tax Returns that it was required to file, and all such Tax Returns are true, correct and complete in all material respects. Except as would not reasonably be expected to have, individually or in the aggregate, a Solstice Material Adverse Effect, each of Solstice and its Subsidiaries has paid on a timely basis all Taxes, whether or not shown on any Tax Return, that were due and payable. Except as would not reasonably be expected to have, individually or in the aggregate, a Solstice Material Adverse Effect, the unpaid Taxes of Solstice and each of its Subsidiaries (A) for taxable periods through the date of the most recent balance sheet included in the financial statements included in the Solstice Reports do not exceed the accruals and reserves for Taxes (excluding accruals and reserves for deferred Taxes established to reflect timing differences between book and Tax income) set forth on the most recent balance sheet included in the financial statements included in the Solstice Reports and (B) for taxable periods through the Closing Date, will not exceed the reserve as adjusted for the passage of time through the Closing Date in accordance with GAAP. All material unpaid Taxes of Solstice and each of its Subsidiaries for all taxable periods commencing after the date of the most recent balance sheet included in the financial statements included in the Solstice Reports arose in the ordinary course of business.
- (c) All material Taxes that Solstice or any of its Subsidiaries is or was required by Applicable Law to withhold or collect have been duly withheld or collected and, to the extent required, have been properly paid to the appropriate Governmental Authority, and each of Solstice and any of its Subsidiaries has complied with all information reporting and backup withholding requirements in all material respects, including the maintenance of required records with respect thereto, in connection with amounts paid to any employee, independent contractor, creditor, or other third party.
- (d) Neither Solstice nor any of its Subsidiaries is or has ever been a member of an affiliated group with which it has filed (or been required to file) consolidated, combined, unitary or similar Tax Returns, other than a group of which the common parent is Solstice. With the exceptions of customary commercial leases or contracts that are not primarily related to Taxes entered into in the ordinary course of business and liabilities thereunder, neither Solstice nor any of its Subsidiaries (i) has any material liability under Treasury Regulation Section 1.1502-6 (or any comparable or similar provision of federal, state, local or foreign law), as a transferee or successor, pursuant to any contractual obligation, or otherwise for any Taxes of any Person other than Solstice or any of its Subsidiaries, or (ii) is a party to or bound by any material Tax indemnity, Tax sharing, Tax allocation or similar agreement.

of its Subsidiaries relating to Taxes for all taxable periods ending on or after December 31, 2011, and (ii) complete and correct copies of all material agreements, rulings, settlements or other Tax documents with or from any Governmental Authority relating to Tax incentives of Solstice or any of its Subsidiaries.

- (f) No examination, suit, proceeding, investigation or audit or other similar action of or relating to any material Tax Return of Solstice or any of its Subsidiaries by any Governmental Authority is currently in progress or, to the knowledge of Solstice, threatened or contemplated. No deficiencies for material Taxes of Solstice or any of its Subsidiaries have been claimed, proposed or assessed in writing by any Governmental Authority. Neither Solstice nor any of its Subsidiaries has been informed in writing by any jurisdiction in which Solstice or any Subsidiary does not file a Tax Return that the jurisdiction believes that Solstice or any of its Subsidiaries was required to file any Tax Return that was not filed or is subject to Tax in such jurisdiction with respect to Taxes that would be the subject of such Tax Return. Neither Solstice nor any of its Subsidiaries has (i) waived any statute of limitations with respect to Taxes or agreed to extend the period for assessment or collection of any Taxes, which waiver or extension is still in effect, (ii) requested any extension of time within which to file any Tax Return, which Tax Return has not yet been filed, or (iii) executed or filed any power of attorney with any taxing authority, which is still in effect.
- (g) Neither Solstice nor any of its Subsidiaries has made any payment, is obligated to make any payment, or is a party to any agreement, contract, arrangement or plan that could obligate it to make any payment that may be treated as an "excess parachute payment" under Section 280G of the Code (without regard to Sections 280G(b)(4) and 280G(b)(5) of the Code) in connection with the Transactions.
- (h) Neither Solstice nor any of its Subsidiaries will be required to include any material item of income in, or exclude any material item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of (i) any adjustments under Section 481 of the Code (or any similar adjustments under any provision of the Code or the corresponding foreign, state or local Tax law) due to a change in method of accounting made prior to Closing, (ii) deferred intercompany gain or any excess loss account described in Treasury Regulations under Section 1502 of the Code (or any corresponding provision of state, local or foreign Tax law) that arose prior to Closing, (iii) closing agreement as described in Section 7121 of the Code (or any corresponding or similar provision of state, local or foreign Tax law) executed prior to the Closing, (iv) installment sale or open transaction disposition made prior to the Closing, (v) prepaid amount received prior to the Closing, or (vi) any election made pursuant to Section 108(i) of the Code prior to the Closing.
- (i) Neither Solstice nor any of its Subsidiaries has been a United States real property holding corporation within the meaning of Section 897(c)(2) of the Code during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code.

38

- (j) Neither Solstice nor any of its Subsidiaries (nor any of their predecessors) has distributed to its stockholders or security holders stock or securities of a controlled corporation, nor has stock or securities of Solstice or any of its Subsidiaries been distributed, in a transaction to which Section 355 or Section 361 of the Code applies in the two years prior to the date of this Agreement.
- (k) There are no material liens or other encumbrances with respect to Taxes upon any of the assets of Solstice or any of its Subsidiaries, other than with respect to Taxes not yet due and payable.
- (l) Neither Solstice nor any of its Subsidiaries is a stockholder in a "passive foreign investment company" within the meaning of Section 1297 of the Code.
- (m) Neither Solstice nor any of its Subsidiaries has engaged in a "reportable transaction" as set forth in Treasury Regulation Section 1.6011-4(b) or a "listed transaction" as set forth in Treasury Regulation Section 301.6111-2(b)(2) or any analogous provision of state or local law. Each of Solstice and its Subsidiaries has disclosed on its federal income Tax Returns all positions taken therein that could give rise to a substantial understatement of federal income Tax within the meaning of Section 6662 of the Code.
- (n) Neither Solstice nor any of its Subsidiaries has taken any action, has failed to take any action, or knows of any fact or circumstance that, in each case, could reasonably be expected to prevent (a) the Solstice Merger from qualifying as a reorganization under Section 368(a) of the Code or as a transaction that, taken together with the other Mergers, qualifies as a transfer of property to the Company described in Section 351 of the Code; or (b) the Cayman Merger, taken together with the other Mergers, from qualifying as a transfer of property to the Company described in Section 351 of the Code. Neither Solstice nor any of its Subsidiaries has any plan or intention to cause or permit the Company to liquidate, merge, transfer all or substantially all of the assets of, or otherwise dissolve Solstice Surviving Corporation, Glass Surviving Corporation or GB Surviving Entity.

# 4.11 <u>Employee Benefit Plans</u>.

(a) Section 4.11(a) of the Solstice Disclosure Letter sets forth a list of each material Solstice Plan. "Solstice Plans" means all employee or director benefit plans, programs, policies, agreements or other arrangements, including any employee welfare plan within the meaning of Section 3(1) of ERISA, any employee pension benefit plan within the meaning of Section 3(2) of ERISA (in each case whether or not such plan is subject to ERISA), any employment, individual consulting or other compensation agreements and any bonus, incentive, equity or equity-based compensation, deferred compensation, vacation, stock purchase, stock option, severance, employment, change of control, retirement, salary continuation, disability plan, health or life insurance or fringe benefit plan, program or agreement, in each case that are sponsored, maintained or contributed to by Solstice or any of its Subsidiaries for the benefit of current or former employees, directors or consultants of Solstice or its Subsidiaries or to which Solstice or any of its Subsidiaries has any obligation or liability (contingent or otherwise), directly or through ERISA Affiliates; provided, however, that Solstice Plans shall not include any Solstice Foreign Plan or any "multiemployer plan" within the meaning of Section 4001(a)(3)

and which would have been treated as a Solstice Plan had it been a United States plan, program or Contract. With respect to each material Solstice Plan, Solstice has made available to Glass a true, correct and complete copy of the following (where applicable): (i) each writing constituting a part of such Solstice Plan, including all plan documents, trust agreements, and other funding vehicles; (ii) the most recent annual report (Form 5500 Series) and accompanying schedule, if any; (iii) the current summary plan description and any material modifications thereto, if any; (iv) the most recent annual financial report, if any; (v) the most recent actuarial report, if any, and (vi) the most recent determination, opinion or advisory letter from the Internal Revenue Service, if any. Solstice has provided a copy of each material Solstice Foreign Plan or summaries of the material terms thereof.

- (b) With respect to each Solstice Plan that is intended to be a "qualified plan" within the meaning of Section 401(a) of the Code ("Qualified Solstice Plans"), the Internal Revenue Service has issued a favorable determination, opinion, or advisory letter with respect to such Qualified Solstice Plan and the related trust that has not been revoked, and to the knowledge of Solstice, no circumstances exist and no events have occurred that would reasonably be expected to adversely affect the qualified status of any Qualified Solstice Plan or the related trust.
- (c) All material contributions required to be made to any Solstice Plan by Applicable Law or by any plan document or other contractual undertaking, and all premiums due or payable with respect to insurance policies funding any Solstice Plan, before the date of this Agreement, have been made or paid in full in all material respects on or before the final due date thereof.
- (d) Solstice and its Subsidiaries have complied, and to the knowledge of Solstice are now in compliance, in all material respects, with all provisions of ERISA, the Code and all laws and regulations applicable to the Solstice Plans. To the knowledge of Solstice, each Solstice Plan has been operated in material compliance with its terms. Neither Solstice nor any of its ERISA Affiliates maintains, contributes to, or has or has had any liability with respect to any employee benefit plan subject to Title IV or Section 302 of ERISA (including any Multiemployer Plan). No Solstice Plan subject to ERISA holds securities issued by Solstice, any of its Subsidiaries, or its ERISA Affiliates.
- (e) Except for health continuation coverage as required by Section 4980B of the Code or Part 6 of Title I of ERISA or in connection with and during a limited period in which severance is paid (not to exceed two years), neither Solstice nor any of its Subsidiaries has promised or provided life, health, medical or other welfare benefits to former employees or beneficiaries or dependents thereof.

40

- (f) Neither the execution and delivery of this Agreement, stockholder approval of the Transactions, nor the consummation of the Transactions, either alone or in combination with another event, will result in, cause the accelerated vesting, funding of any amounts to a rabbi trust or delivery of, or increase the amount or value of, any payment or benefit to any employee, officer, director or consultant of Solstice or any of its Subsidiaries. Neither Solstice nor any of its Subsidiaries is a party to, nor is Solstice or any of its Subsidiaries otherwise obligated under, any plan, policy, agreement or arrangement that provides for the gross-up or reimbursement of Taxes imposed under Section 409A or 4999 of the Code (or any corresponding provisions of state or local Applicable Law relating to Tax).
- (g) There are no pending or to the knowledge of Solstice threatened claims (other than claims for benefits in the ordinary course), lawsuits or arbitrations which have been asserted or instituted against the Solstice Plans or any Solstice Foreign Plan, any fiduciaries thereof with respect to their duties to the Solstice Plans or the assets of any of the trusts under any of the Solstice Plans which would reasonably be expected to result in any material liability of Solstice or any of its Subsidiaries to any Solstice Plan, the United States Department of Treasury, the United States Department of Labor, or any current or former participants of such Solstice Plans.
- (h) Each Solstice Plan that is a "nonqualified deferred compensation plan" within the meaning of Section 409A(d)(1) or 457A of the Code and any award thereunder, in each case that is subject to Section 409A or 457A of the Code, has been operated in compliance in all material respects with Section 409A and, where applicable, 457A of the Code.
- (i) All material Solstice Foreign Plans subject to the laws of any jurisdiction outside of the United States (i) have been maintained in all material respects in accordance with all applicable legal requirements; (ii) if they are intended to qualify for special tax treatment meet all requirements for such treatment in all material respects; and (iii) if they are required to be funded and/or book-reserved are so funded and/or book reserved, as applicable, in all material respects.

## 4.12 Labor Matters.

(a) Neither Solstice nor any of its Subsidiaries is a party to, or bound by, or as of the date of this Agreement negotiating, any collective bargaining agreement, labor union Contract, or trade union agreement (each a "Collective Bargaining Agreement"), nor are the employees of Solstice and its Subsidiaries covered by any works council, employee representative agreement or information or consultation agreement. There is no material labor strike, slowdown, work stoppage or lockout pending or, to the knowledge of Solstice, threatened against Solstice or any of its Subsidiaries, and neither Solstice nor any of its Subsidiaries has experienced any material labor dispute since December 31, 2014. To the knowledge of Solstice, there are no material organizational efforts with respect to the formation of a collective bargaining unit presently being made or threatened involving employees of Solstice or any of its Subsidiaries, and there have not been any such material organizational efforts since December 31, 2014.

41

(b) Except as would not reasonably be expected to have, individually or in the aggregate, a Solstice Material Adverse Effect, (i) since December 31, 2014, Solstice and its Subsidiaries have been in compliance with any (A) Applicable Law respecting employment and employment practices, terms and conditions of employment and wages and hours, including any such law respecting employment discrimination, employee classification, workers' compensation, family and medical leave, the Immigration Reform and Control Act and occupational safety and health requirements, and (B) employment agreement or other agreement establishing terms of employment; and (ii) no claims, controversies, investigations, audits or suits are pending or, to the knowledge of Solstice, threatened in writing, with respect to such laws or agreements, either by private individuals or by Governmental Authorities. Neither Solstice nor any of its Subsidiaries has incurred, and no circumstances exist under which either Solstice or any of its Subsidiaries is reasonably likely to incur, any material liability (contingent or otherwise) arising from the misclassification of employees as consultants or independent contractors, or from the misclassification of consultants or independent contractors as employees.

4.13 Environmental Matters. Except as would not, individually or in the aggregate, reasonably be expected to have a Solstice Material Adverse Effect: (a) Solstice and each of its Subsidiaries are and have been in compliance in all respects with all applicable Environmental Laws; (b) Solstice and each of its Subsidiaries have obtained all Environmental Permits necessary for their operations as currently conducted and are in compliance with all such Environmental Permits, have applied for issuance or reissuance of Environmental Permits in a timely fashion, and have no knowledge of any reason that would preclude renewal, issuance or reissuance of Environmental Permits; (c) there are no Environmental Claims pending or, to the knowledge of Solstice, threatened against Solstice or any of its Subsidiaries; (d) neither Solstice nor any of its Subsidiaries is party to any agreement, order, judgment, or decree by or with any Governmental Authority or third party imposing any liability or obligation on Solstice or any of its Subsidiaries under any Environmental Law; (e) neither Solstice nor any of its Subsidiaries has retained or assumed, either contractually or by operation of law, any liability or obligation that would reasonably be expected to form the basis of any Environmental Claim against, or any liability under any Environmental Law on, Solstice or any of its Subsidiaries; and (f) Solstice has made available to Glass copies of any environmental assessments, reports, audits, studies, analyses, tests or monitoring possessed by, or reasonably available to, Solstice or any of its Subsidiaries pertaining to compliance with, or liability under, Environmental Laws relating to Solstice or any of its Subsidiaries, the Solstice Facilities, any real property formerly owned, leased or operated by Solstice or any of its Subsidiaries.

#### 4.14 <u>Intellectual Property</u>.

(a) Section 4.14(a) of the Solstice Disclosure Letter lists all Solstice Registrations, in each case, enumerating specifically the applicable filing or registration number, title, jurisdiction in which filing was made or from which registration issued, date of filing or issuance, as applicable. To Solstice's knowledge, all assignments of Solstice Registrations to Solstice or any of its Subsidiaries have been properly executed and recorded, except for such deficiencies as would not materially affect the enforceability thereof. To Solstice's knowledge, all issuance, renewal, maintenance and other payments that are or have become due with respect

42

to the Solstice Registrations have been timely paid by or on behalf of Solstice or the relevant Subsidiary.

- (b) To Solstice's knowledge, no action or proceeding is pending or threatened in which any Person is challenging the validity, enforceability, registration, ownership, or use of any Solstice Registrations (excluding, for the avoidance of doubt, office actions issued by a Governmental Authority with respect to applications for Solstice Registrations).
- (c) There are no third party joint owners of any Solstice Registrations, and Solstice or one of its Subsidiaries is the sole and exclusive owner of all the Solstice Registrations.
- (d) Neither the execution and delivery by any Solstice Party of this Agreement nor the consummation by such Solstice Party of the Transactions in accordance with the terms hereof will result in the loss, termination, or impairment of any rights in the Solstice Intellectual Property, except as would not, individually or in the aggregate, reasonably be expected to have a Solstice Material Adverse Effect.
- (e) Solstice and its Subsidiaries have taken commercially reasonable measures to maintain in confidence and protect the proprietary nature of all material trade secrets and confidential information comprising a part of the Solstice Intellectual Property. To Solstice's knowledge, there has been no: (i) unauthorized disclosure of any third party proprietary or confidential information in the possession, custody or control of Solstice or any of its Subsidiaries, or (ii) breach of Solstice's or any of its Subsidiaries' security or information privacy procedures, in each case which, individually or in the aggregate, would reasonably be expected to have a Solstice Material Adverse Effect.
- (f) To Solstice's knowledge, (i) neither Solstice nor any of its Subsidiaries is infringing, misappropriating, diluting, or otherwise violating any Intellectual Property of any third party, and (ii) no Person (including any current or former employee or consultant of Solstice or any of its Subsidiaries) is infringing, diluting, misappropriating, or otherwise violating any of the Solstice Owned Intellectual Property. Section 4.14(f) of the Solstice Disclosure Letter lists any written complaint, claim or notice, or written threat of any of the foregoing (including any written notification that a license under any patent is or may be required), that, to Solstice's knowledge, has been received by Solstice or any of its Subsidiaries in the past two (2) years alleging (or requesting or demanding indemnification for) any such infringement, violation or misappropriation of third party intellectual property.
- (g) Neither Solstice nor any of its Subsidiaries, nor any of their employees, has licensed, distributed or disclosed the source code, other than as required by Open Source Material license agreements, for any Software included in the products or services offered by Solstice or any of its Subsidiaries, or other confidential information constituting, embodied in or pertaining to such Software (collectively, "Solstice Source Code") to any person, other than escrow agents and third party contractors engaged by or on behalf of Solstice or its Subsidiaries to develop, modify, maintain or otherwise work with the Solstice Source Code, which contractors are contractually obligated to maintain the confidentiality of the Solstice Source Code, and Solstice and its Subsidiaries have taken commercially reasonable physical and

43

electronic security measures to prevent disclosure of such Solstice Source Code. To Solstice's knowledge, no event has occurred, and no circumstance or condition exists, that (with or without notice or lapse of time, or both) will, or would reasonably be expected to, nor will the consummation of the Transactions, result in the disclosure or release of Solstice Source Code by Solstice, any of its Subsidiaries or escrow agent(s) or any other person to any third party.

- (h) Other than commercially reasonable protections against unauthorized use, to Solstice's knowledge, the products and services offered by Solstice or any of its Subsidiaries, and the Software and internal computer systems used by Solstice and its Subsidiaries do not contain any disabling device, virus, worm, back door, Trojan horse or other disruptive or malicious code that may or are intended to materially impair their intended performance or otherwise permit unauthorized access to, hamper, delete or damage any computer system, software, network or data.
- (i) Each employee of Solstice or any of its Subsidiaries and each independent contractor of Solstice or any of its Subsidiaries who has contributed to the development of material Intellectual Property on behalf of Solstice or any of its Subsidiaries has executed a valid and binding written agreement assigning such Intellectual Property to or otherwise vesting ownership of such Intellectual Property in Solstice or its relevant Subsidiary. All of the

agreements referenced in the preceding sentence will continue to be in full force and effect immediately following the Effective Time in accordance with the terms thereof as in effect immediately prior to the Effective Time.

(j) To Solstice's knowledge, no material support, funding, resources or assistance from any Governmental Authority has been received by Solstice or any of its Subsidiaries during the six years prior to the date of this Agreement in connection with the development, design, testing, modification, manufacture, use, sale, reproduction, marketing, distribution, support or maintenance of any of the products or services offered, or contemplated to be offered, by Solstice or any of its Subsidiaries.

#### 4.15 Title to Properties.

- (a) Neither Solstice nor any of its Subsidiaries owns any real property.
- (b) Solstice or one of its Subsidiaries has a good and valid leasehold interest in all real property leased or purported to be leased, or otherwise occupied or permitted to be occupied, by any Solstice Party or any of its Subsidiaries, whether as sublessor, tenant, subtenant or otherwise (the "Solstice Leased Real Property" and the lease, sublease or other occupancy agreement, the "Solstice Real Property Leases"). Section 4.15(b) of the Solstice Disclosure Letter sets forth a complete and accurate list of all Solstice Leased Real Property, the location of the premises and each document comprising the Solstice Real Property Leases, including all amendments thereto and guarantees thereof. No Solstice Party nor any Subsidiary thereof has assigned, transferred, conveyed, mortgaged, deeded in trust or otherwise encumbered any interest in the leasehold or subleasehold. All Solstice Leased Real Property is supplied with utilities and other services necessary for the operation of said facilities as currently operated. Each Solstice Real Property Lease is in full force and effect and will continue to be legal, valid,

44

binding, enforceable and in full force and effect immediately following the Closing in accordance with the terms thereof as in effect prior to the Closing.

- (c) Solstice or one of its Subsidiaries owns or leases all of the material personal property reflected on the most recent balance sheet included in the financial statements included in the Solstice Reports, free and clear of all Liens, except to the extent disposed of in the ordinary course of business since the date of such balance sheet. The Solstice Parties and their respective Subsidiaries have good and valid title to, or in the case of leased properties and assets, valid leasehold interests or other comparable contractual rights in, all material tangible properties and assets, real, personal and mixed, necessary for the conduct of their respective businesses, free and clear of all Liens.
- (d) To the knowledge of Solstice, the Solstice Leased Real Property and the equipment of Solstice and its Subsidiaries used in the operations of their respective businesses are (i) suitable for the uses to which they are currently employed, (ii) in good operating condition and repair, subject to normal wear and tear, (iii) regularly and properly maintained, (iv) not obsolete, dangerous or in need of renewal or replacement, except for renewal or replacement in the ordinary course of business, and (v) free from any material defects or deficiencies.
- 4.16 <u>Insurance</u>. Solstice and its Subsidiaries maintain insurance coverage reasonably adequate for the operation of their respective businesses (taking into account the cost and availability of such insurance).
- 4.17 No Brokers. Neither Solstice nor any Subsidiary thereof has entered into any Contract with any person which may result in the obligation of Solstice, the Company or any Subsidiary thereof or any Glass Party to pay any finder's fees, brokerage or agent's commissions or other like payments in connection with the negotiations leading to this Agreement or the consummation of the Transactions, except that Solstice has retained Evercore Partners to provide financial advice with respect to the Transactions and to render the opinion referred to in Section 4.18. Solstice's arrangements with such investment banks have been disclosed in writing to Glass prior to the date hereof.
- 4.18 <u>Opinion of Financial Advisor</u>. The Board of Directors of Solstice has received the written opinion of Evercore Partners to the effect that, as of the date of this Agreement, the Solstice Exchange Ratio is fair, from a financial point of view, to the holders of shares of Solstice Common Stock, and the Total Consideration, taking into account a significant portion of such Total Consideration is in the form of Company Common Stock, is fair, from a financial point of view, to Solstice, a copy of which opinion will be made available to Glass promptly following receipt thereof.

## 4.19 Solstice Board Recommendation.

(a) The Board of Directors of Solstice has, and not subsequently rescinded or modified (except as expressly permitted hereunder): (i) approved this Agreement and the Transactions; (ii) determined that, as of the date of this Agreement, this Agreement and the Transactions, including the Solstice Merger, are in the best interests of the stockholders of Solstice and declared the advisability of this Agreement; and (iii) recommended that such

45

stockholders vote to adopt this Agreement and approve the Solstice Merger and the Solstice Merger-Related Proposals (the "Solstice Board Recommendation") and directed that such matters be submitted for consideration by Solstice's stockholders at a meeting of Solstice's stockholders.

- (b) The Board of Directors of Solstice Merger Sub has: (i) approved this Agreement and the Transactions; (ii) determined that, as of the date of this Agreement, this Agreement and the Transactions, including the applicable Merger, are in the best interests of the Company, as the sole stockholder of such Merger Sub, and declared the advisability of this Agreement; and (iii) recommended that the Company, as sole stockholder of such Merger Sub, vote to adopt this Agreement and approve the applicable Merger and directed that such matters be submitted for approval by action by written consent of the Company, as sole stockholder of such Merger Sub. The Company, in its capacity as the sole member of the Cayman Merger Sub, and in its capacity as the sole member of GB Merger Sub, has approved this Agreement.
- 4.20 <u>Vote Required</u>. The Required Solstice Stockholder Approval is the only vote of the holders of any class or series of Solstice capital stock necessary to approve this Agreement, the Solstice Merger and the Transactions.

4.21 <u>Certain Approvals.</u> Assuming the accuracy of the representations and warranties of the Glass Parties in <u>Section 5.18</u>, Solstice's Board of Directors has taken any and all necessary and appropriate action to render inapplicable to this Agreement and the Transactions the provisions of Section 203 of the DGCL. To the knowledge of Solstice, no other Takeover Law is applicable to this Agreement or any of the Transactions. Solstice is not a party to any "stockholder rights plan" or similar anti-takeover plan or device.

#### 4.22 Relationships with Related Parties.

- (a) No officer or director of Solstice or any of its Subsidiaries or any Affiliate of any of the foregoing, (a) has any interest in any property (real, personal, or mixed and whether tangible or intangible), used in or pertaining to the business of Solstice or any of its Subsidiaries as currently conducted or contemplated to be conducted or (b) is a party to any Contract (except for Solstice Plans) with Solstice or any of its Subsidiaries, including with respect to compensation or remuneration to be paid to such officer or director or Affiliate in connection with this Agreement or the Transactions.
- (b) Neither Solstice nor any of its Subsidiaries is indebted, directly or indirectly, to any Person who is an Affiliate, officer or director of Solstice or any of its Subsidiaries in any amount whatsoever, other than for salaries for services rendered or reimbursable business expenses, nor is any such Affiliate, officer or director indebted to Solstice or any of its Subsidiaries, except for advances made to employees of Solstice or any of its Subsidiaries in the ordinary course of business consistent with past practice to meet reimbursable business expenses reasonably anticipated to be incurred by such obligor.

46

#### 4.23 <u>Internal Controls</u>.

- (a) Solstice and its Subsidiaries maintain disclosure controls and procedures (as such terms are defined in Rule 13a-15 under the Exchange Act) that satisfy the requirements of Rule 13a-15 under the Exchange Act. Such disclosure controls and procedures are effective to ensure that all material information (financial and non-financial) concerning Solstice (including its Subsidiaries) is made known on a timely basis to the chief executive officer and the chief financial officer of Solstice by others within those entities. To the knowledge of Solstice, there has not been any fraud that involves management or other employees who have a significant role in Solstice's internal controls over financial reporting.
- (b) Solstice maintains a system of internal control over financial reporting (as such term is defined in Rule 13a-15 under the Exchange Act) designed to provide reasonable assurance regarding the reliability of its financial reporting and the preparation of its financial statements for external purposes in accordance with GAAP, and includes policies and procedures that: (i) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect its transactions and dispositions of its assets; (ii) provide reasonable assurance that transactions are recorded as necessary to permit preparation of its financial statements in accordance with GAAP, and that its receipts and expenditures are being made only in accordance with authorizations of its management and directors; and (iii) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use or disposition of its assets that could have a material effect on its financial statements.
- (c) Since December 31, 2014, Solstice has not identified or been made aware of any significant deficiencies or material weaknesses in the design or operation of internal controls over financial reporting that are reasonably likely to adversely affect Solstice's ability to record, process, summarize and report financial information of Solstice and its Subsidiaries on a consolidated basis.
- (d) Neither Solstice nor any of its Subsidiaries is a party to, or has any commitment to become a party to, any joint venture, partnership agreement or any similar Contract (including any Contract or arrangement relating to any transaction, arrangement or relationship between or among Solstice or any of its Subsidiaries, on the one hand, and any unconsolidated Affiliate, including any structured finance, special purpose or limited purpose entity or Person, on the other hand (such as any arrangement described in Section 303(a)(4) of Regulation S-K under the Securities Act)) where the purpose or effect of such arrangement is to avoid disclosure of any material transaction involving Solstice or any of its Subsidiaries in Solstice's consolidated financial statements.

#### 4.24 Certain Contracts.

(a) <u>Section 4.24(a)</u> of the Solstice Disclosure Letter sets forth, as of the date hereof, a complete and accurate list of all Solstice Material Contracts. As used in this Agreement, "<u>Solstice Material Contract</u>" means (i) any Contract pursuant to which Solstice and its Subsidiaries is reasonably likely to spend, in the aggregate, more than \$500,000 with respect to any such Contract during the current fiscal year or during the next fiscal year, (ii) any non-competition or other agreement that prohibits or otherwise restricts, in any material respect, Solstice or any of its Subsidiaries from freely engaging in any business material to Solstice and its Subsidiaries, taken as a whole, anywhere in the world, (iii) any Contract to which Solstice or

47

any of its Subsidiaries is a party involving research, development or the license of any Solstice Intellectual Property (other than non-exclusive licenses of Solstice Intellectual Property made in the ordinary course of business), (iv) any Contract to which Solstice or any of its Subsidiaries is a party granting a right of first refusal, or right of first offer or comparable right with respect to any material Solstice Intellectual Property, (v) any Contract to which Solstice or any of its Subsidiaries is a party relating to a material joint venture, partnership or other material arrangement involving a sharing of profits, losses, costs or liabilities with another person, (vi) any Contract which would be binding on an Affiliate of Solstice (other than one of its Subsidiaries) or any Glass Party or a controlled Affiliate of any Glass Party and (vii) any "material contract" (as such term is defined in Item 601(b)(10) of Regulation S-K of the SEC) with respect to Solstice and its Subsidiaries filed with the SEC on or following January 1, 2017 or included in the Exhibit Index filed with Solstice's Form 10-K filed on the SEC's EDGAR system on February 27, 2017. Solstice has made available to Glass a complete and accurate copy of each Solstice Material Contract. Each Solstice Material Contract is in full force and effect and is enforceable in accordance with its terms, subject to the effects of bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar laws relating to or affecting creditors' rights generally and general equitable principles (whether considered in a proceeding in equity or at law). Neither Solstice nor any of its Subsidiaries nor, to Solstice's knowledge, any other party or any Solstice Material Contract is in material violation of or in material default under (nor does there exist any condition, which, upon the passage of time or the giving of notice or both, would cause such a violation of or default under), nor will the consummation of the Transactions result in any

- (b) Other than such agreements that Solstice has filed with or furnished to the SEC that are available in unredacted form on the SEC's EDGAR system, Section 4.24(b) of the Solstice Disclosure Letter sets forth a complete and accurate list of each agreement to which Solstice or any of its Subsidiaries is a party or bound with any Affiliate of Solstice (other than any person that is a direct or indirect wholly-owned Subsidiary of Solstice). Complete and accurate copies of all the agreements listed in Section 4.24(b) of the Solstice Disclosure Letter have heretofore been made available to Glass. Neither Solstice nor any of its Subsidiaries has entered into any transaction with any Affiliate of Solstice or any of its Subsidiaries or any transaction that has not been included in any Solstice Reports filed prior to the date hereof and that would be subject to disclosure pursuant to Item 404 of Regulation S-K.

  (c) There is no non-competition or other similar agreement, commitment, judgment, injunction or order to which Solstice or any of its
- (c) There is no non-competition or other similar agreement, commitment, judgment, injunction or order to which Solstice or any of its Subsidiaries is a party or is subject that has or would reasonably be expected to result in the effect of prohibiting or impairing the conduct of the business of Solstice or any of its Subsidiaries as currently conducted. Neither Solstice nor any of its Subsidiaries has entered into (or is otherwise bound by) any agreement under which it is now, or following the Effective Time any Glass Party or any Affiliate of any Glass Party (including Solstice or any of its Subsidiaries) would be, restricted from selling, licensing or otherwise distributing any of their respective technology or products, or providing services to, customers or potential customers or any class of customers, in any geographic area, during any period of time or any segment of the market or line of business, provided, that the

foregoing shall not include any obligations of any Glass Party or any of Affiliate of any Glass Party solely arising under agreements entered into prior to the Effective Time by any Glass Party or any of its Affiliates (excluding Solstice and its Subsidiaries).

- (d) Neither Solstice nor any of its Subsidiaries is a party to any agreement under which a third party would be entitled to receive a license or any other right to Intellectual Property of any Glass Party or any Affiliate of any Glass Party (excluding Solstice and its Subsidiaries) following the Closing, provided, that the foregoing shall not include any obligations of any Glass Party or any Affiliate of any Glass Party arising solely under agreements entered into prior to the Effective Time by any Glass Party or any Affiliate of any Glass Party (excluding Solstice and its Subsidiaries).
- 4.25 <u>Foreign Corrupt Practices Act</u>. Except for such matters as would not, individually or in the aggregate, reasonably be expected to have a Solstice Material Adverse Effect, as of the date of this Agreement:
- (a) In connection with Solstice's and its Subsidiaries' compliance with the Foreign Corrupt Practices Act, there have been no voluntary disclosures under the Foreign Corrupt Practices Act.
- (b) No Governmental Authority has notified Solstice or any of its Subsidiaries in writing of any actual or alleged violation or breach of the Foreign Corrupt Practices Act or any other similar Applicable Law.
- (c) Neither Solstice nor any of its Subsidiaries has undergone or is undergoing any audit, review, inspection, investigation, survey or examination of records, in each case conducted by a Governmental Authority and relating to Solstice's or its Subsidiaries' compliance with the Foreign Corrupt Practices Act or any other similar Applicable Law, and to Solstice's knowledge, there is no basis for any such audit, review, inspection, investigation, survey or examination of records by a Governmental Authority.
- (d) Neither Solstice nor any of its Subsidiaries has been or is now under any administrative, civil or criminal charge or indictment or, to Solstice's knowledge, investigation, alleging noncompliance with the Foreign Corrupt Practices Act or any other similar Applicable Law, nor, to Solstice's knowledge, is there any basis for any such charge, indictment or investigation.
- (e) Neither Solstice nor any of its Subsidiaries has been or is now a party to any administrative or civil litigation alleging noncompliance with the Foreign Corrupt Practices Act or any other similar Applicable Law, nor, to Solstice's knowledge, is there any basis for any such proceeding.
- (f) Neither Solstice nor any of its Subsidiaries, nor any of their Affiliates, directors, officers and employees nor any other Person acting on behalf of any of them has made any offer, payment, promise to pay, or authorization for the payment of any money, or any offer, gift, promise to give, or authorization of the giving of anything in value, which would cause Solstice or any of its Subsidiaries, or any of its or their Affiliates, directors, officers or

49

employees or any other Person acting on behalf of any of them, to have violated or be in violation of the Foreign Corrupt Practices Act or any other similar Applicable Law.

#### 4.26 Export Control.

- (a) Solstice and each of its Subsidiaries is and has been for the past five (5) years in material compliance with all Export Control Laws, including all applicable regulations pertaining to the disclosure of controlled technical information to foreign persons wherever located and/or the provision of access to such technical information by such foreign persons, and has maintained a program to facilitate such compliance, including training, technology assessment and classification, transaction screening, license compliance tracking, export clearance and recordkeeping measures. Neither Solstice nor any of its Subsidiaries has received any notice from any person alleging that Solstice or any of its Subsidiaries is not in compliance with, or has liability under, such Export Control Laws.
- (b) Solstice and each of its Subsidiaries has obtained and complied in all material respects with all licenses, agreements, authorizations, license exceptions or exemptions required for their respective exports of articles or technology or provision of services.
- (c) During the five (5) years prior to the date of this Agreement, neither Solstice nor any of its Subsidiaries has conducted or initiated any internal investigation, made any mandatory or voluntary disclosure, declined to make a voluntary disclosure with respect to known violation of Export Control Laws or failed to make any mandatory report or disclosure to any Governmental Authority pursuant to Export Control Laws.

- 4.27 <u>Ownership and Operations of the Company and Merger Subs</u>. Each of the Company and the Merger Subs was formed solely for the purpose of engaging in the Transactions, has engaged in no other business activities and has conducted its operations only as contemplated by this Agreement. All of the issued and outstanding capital stock of the Company is owned beneficially and of record by Solstice, and all of the issued and outstanding capital stock or limited liability company interests, as applicable, of each Merger Sub are owned beneficially and of record by the Company.
- 4.28 <u>No Additional Representations.</u> Notwithstanding anything contained in this Agreement to the contrary, each Solstice Party acknowledges and agrees that no Glass Party nor any other Person has made or is making any representations or warranties relating to any Glass Party whatsoever, express or implied, beyond those expressly given by the Glass Parties in <u>Article 5</u> hereof or given in any Letter of Transmittal or the Voting Agreement, including any implied representation or warranty as to the accuracy or completeness of any information regarding any Glass Party furnished or made available to any Solstice Party or any of its Representatives. Without limiting the generality of the foregoing, each Solstice Party acknowledges that no representations or warranties are made with respect to any projections, forecasts, estimates, budgets or prospect information that may have been made available to any Solstice Party or any of its Representatives. No Solstice Party has relied on any representations or warranties relating to the Glass Parties in determining to enter into this Agreement, except as expressly given by the Glass Parties in <u>Article 5</u> hereof or given in the Voting Agreement.

#### ARTICLE 5

#### REPRESENTATIONS AND WARRANTIES OF THE GLASS PARTIES

Except as set forth in the applicable section of the disclosure letter delivered by Glass to Solstice on the date of this Agreement (the "Glass Disclosure Letter") (it being agreed that any information disclosed in a section of the Glass Disclosure Letter with respect to a corresponding Section of this Agreement shall be deemed to have been disclosed with respect to any other Section of this Agreement to the extent the applicability thereto is reasonably apparent from the face of the disclosure), the Glass Parties, severally, and not jointly, hereby represent and warrant to Solstice as follows:

- Existence; Good Standing; Corporate Authority. Each Glass Party is an entity duly organized or incorporated, validly existing and in good standing under the laws of its jurisdiction of organization or incorporation. Each Glass Party is duly qualified to do business as a foreign entity and is in good standing under the laws of any jurisdiction in which the character of the properties owned or leased by it therein or in which the transaction of its business makes such qualification necessary (to the extent the "good standing" concept is applicable in the case of any jurisdiction outside the United States), except where the failure to be so qualified would not, individually or in the aggregate, reasonably be expected to have a Glass Material Adverse Effect. Each Glass Party has all requisite corporate (or comparable) power and authority to own, operate and lease its properties and to carry on its business as now conducted. The copies of the certificate of incorporation or formation, memorandum and articles of association, by-laws, operating agreement and similar formation or governing documents and instruments of each Glass Party (collectively, the "Glass Organizational Documents") previously made available to Solstice are true and correct and contain all amendments as of the date hereof.
- 5.2 <u>Authorization, Validity and Effect of Agreements</u>. Each Glass Party has the requisite corporate (or comparable) power and authority to execute and deliver this Agreement and all other agreements and documents contemplated hereby to which it is a party. The consummation by each Glass Party of the Transactions has been duly authorized by all requisite corporate (or comparable) action on the part of such Glass Party, other than the adoption of this Agreement and the approval of the applicable Merger by the holders of the requisite majority of the issued and outstanding shares of capital stock of each Glass Party at a meeting duly called and held for such purpose, or the delivery of written consents or proxies in accordance with the Glass Stockholders Agreement and, in the case of Glass, the Companies Law (collectively, the "Required Glass Stockholder Approvals"). This Agreement constitutes the valid and legally binding obligation of each Glass Party, enforceable against such Glass Party in accordance with its terms, subject to applicable bankruptcy, insolvency, moratorium or other similar laws relating to creditors' rights and general principles of equity.

#### 5.3 <u>Capitalization</u>.

(a) <u>Section 5.3</u> of the Glass Disclosure Letter sets forth a description of the authorized capital stock of each Glass Party. The issued and outstanding equity interests of each Glass Party consist solely of equity interests set forth on <u>Section 5.3</u> of the Glass Disclosure Letter, which are owned of record by the holder identified in <u>Section 5.3</u> of the Glass Disclosure

51

Letter and in the respective amounts set forth opposite the name of each such holder on Section 5.3 of the Glass Disclosure Letter. All such issued and outstanding equity interests (i) are duly authorized, validly issued, fully paid and non-assessable (as applicable) and free of preemptive rights (other than as set forth in the Glass Stockholders Agreement), (ii) were not issued in violation of the terms of any agreement or other understanding binding upon any Glass Party and (iii) were issued in compliance with the applicable Glass Organizational Documents and all applicable federal and state securities laws, rules and regulations. The Transactions, when consummated in accordance with the terms of this Agreement, will constitute a Realization Event (as defined the Glass Charter).

- (b) Except as set forth in the Glass Stockholders Agreement or as described in the list delivered pursuant to Section 5.3(d), there are no outstanding subscriptions, options, warrants, calls, convertible securities or other similar rights, agreements, plans or commitments relating to the issuance of capital stock or other equity interests to which any Glass Party or any of its Subsidiaries is a party obligating any Glass Party or any of its Subsidiaries to (i) issue, transfer or sell any shares of capital stock or other equity interests of any Glass Party or any of its Subsidiaries or securities convertible into or exchangeable for such shares or equity interests, (ii) grant, extend or enter into any such subscription, option, warrant, call, convertible securities or other similar right, agreement or arrangement or (iii) redeem or otherwise acquire any such shares of capital stock or other equity interests (including securities or obligations convertible into or exchangeable or exercisable for any shares of capital stock or other equity interests).
- (c) Except for the Glass Stockholders Agreement, there are no stockholder agreements, voting trusts or other agreements or understandings to which any Glass Party or any of its Subsidiaries is a party or, to Glass's knowledge, otherwise with respect to the voting of the limited liability company interests or other equity interests of any Glass Party or any of its Subsidiaries.

- (d) A complete and correct list of all outstanding Glass Options and any other awards granted under the Glass Equity Plans (including the grant date, the number of shares of Glass Stock, GB Stock and GB II Stock subject thereto, and exercise or purchase price (if any), the settlement date (if different than the vesting date(s)), the expiration date (if any) and any other individual terms not otherwise specified in the forms of award agreement) and the names of the holders thereof as of the date hereof has been made available to Solstice. Glass has made available to Solstice a copy of each Glass Equity Plan and the forms of awards agreement for grants of awards thereunder (and, where not included in such forms, a description of the terms and conditions of vesting (including any acceleration upon an acquisition, change in control or similar transaction) for each outstanding award). All outstanding Glass Options and other equity or equity-linked awards of any Glass Party have been granted pursuant to individual award agreements in the forms of award agreements made available to Solstice.
- (e) <u>Section 5.3(e)</u> of the Glass Disclosure Letter sets forth a complete and correct list of all the outstanding Glass Profits Interests and the names of the holders thereof.
- (f) The Allocation Schedule, when delivered to Solstice pursuant to <u>Section 6.21</u>, will set forth a true, correct and complete summary of the allocation of the shares of Company Common Stock pursuant to the Glass Mergers. Such allocation set forth on the

Allocation Schedule will comply with the terms of the Glass Organizational Documents, the Glass Equity Plans, the Glass Profits Interests and any other equity or equity-linked award granted by any Glass Party or otherwise with respect to any shares of capital stock or other equity interests of any Glass Party.

- 5.4 Subsidiaries. Section 5.4 of the Glass Disclosure Letter sets forth a true and correct list of all of the Subsidiaries of each Glass Party, listing for each such Subsidiary its name, its jurisdiction of organization, its authorized capital stock or other equity interests, its authorized capital stock or other equity interests and the current and record beneficial owner of such shares or interest. Each such Subsidiary is duly organized, validly existing and in good standing (where applicable) under the laws of its jurisdiction of incorporation or organization, has the corporate (or comparable) power and authority to own, operate and lease its properties and to carry on its business as it is now being conducted, and is duly qualified to do business and is in good standing (where applicable) in each jurisdiction in which the ownership, operation or lease of its property or the conduct of its business requires such qualification (to the extent the "good standing" concept is applicable in the case of any jurisdiction outside the United States), except for jurisdictions in which such failure to be so qualified or to be in good standing would not, individually or in the aggregate, reasonably be expected to have a Glass Material Adverse Effect. All of the outstanding shares of capital stock or other securities of, or other ownership interests in, each such Subsidiary are duly authorized, validly issued, fully paid and nonassessable, and are owned, directly or indirectly, by a Glass Party free and clear of all Liens. There are no existing options, rights of first refusal, conversion rights, preemptive rights, calls, puts, commitments, arrangements or obligations of any character, including voting agreements, proxies or similar arrangements relating to the issued or unissued capital stock or other securities of, or other ownership interests in, any such Subsidiary. No Glass Party directly or indirectly owns any equity or similar interest in, or any interest convertible into or exchangeable or exercisable for any equity or similar interest in, any corporation, partnership, joint venture or other business association or entity that directly or indirectly conducts any activity which is material to the Glass Parties and their respective Subsidiaries taken as a whole. The copies of each such Subsidiary's certificates of incorporation, by-laws and similar governing documents previously made available to Solstice are true and correct and contain all amendments as of the date hereof.
- 5.5 No Violation. No Glass Party nor any Subsidiary of any Glass Party is, or has received notice or is otherwise aware that it would be with the passage of time, in violation of any term, condition or provision of (a) the Glass Organizational Documents; (b) the certificate of incorporation, by-laws or comparable governing documents of any Subsidiary of any Glass Party, (c) any loan or credit agreement, note, bond, mortgage, indenture, Contract, agreement, joint venture, lease, license, franchise, Glass Permit or other instrument or (d) any Applicable Law, except, in the case of matters described in clauses (c) or (d), as would not, individually or in the aggregate, reasonably be expected to have a Glass Material Adverse Effect. The Glass Parties and their respective Subsidiaries hold all permits, licenses, variances, exemptions, orders, franchises and approvals of all Governmental Authorities necessary for the lawful conduct of their respective businesses (the "Glass Permits"), except where the failure so to hold would not, individually or in the aggregate, reasonably be expected to have a Glass Material Adverse Effect. The Glass Parties and their respective Subsidiaries are in compliance with the terms of the Glass

53

Permits, except where the failure so to comply would not, individually or in the aggregate, reasonably be expected to have a Glass Material Adverse Effect. No investigation by any Governmental Authority with respect to any Glass Party or any of its Subsidiaries is pending or, to the knowledge of Glass, threatened other than those the outcome of which would not, individually or in the aggregate, reasonably be expected to have a Glass Material Adverse Effect.

# 5.6 No Conflict.

- (a) Neither the execution and delivery by any Glass Party of this Agreement nor the consummation by any Glass Party of the Transactions in accordance with the terms hereof will: (i) conflict with or result in a breach of any provisions of or create any rights in favor of any other party under the applicable Glass Organizational Documents; (ii) conflict with or result in a breach of any provisions of or create any rights in favor of any other party under the certificate of incorporation, by-laws or similar governing documents of any Subsidiary of any Glass Party; (iii) violate, or conflict with, or result in a breach of any provision of, or constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, or result in the termination or in a right of termination or cancellation of, or give rise to a right of purchase under, or accelerate the performance required by, or result in the creation of any Lien upon any of the properties of any Glass Party or its Subsidiaries under, or result in being declared void, voidable, or without further binding effect, or otherwise result in a detriment to any Glass Party or any of its Subsidiaries under any of the terms, conditions or provisions of, any loan or credit agreement, note, bond, mortgage, indenture, deed of trust, license, franchise, Glass Permit, lease, Contract, agreement, joint venture or other instrument or obligation to which any Glass Party or any of its Subsidiaries is a party, or by which any Glass Party or any of its Subsidiaries or any of their properties is bound or affected; or (iv) contravene or conflict with or constitute a violation of any provision of any Applicable Laws, except, in the case of matters described in clauses (iii) or (iv), as would not, individually or in the aggregate, reasonably be expected to have a Glass Material Adverse Effect and except in the case of clause (iv) for the matters described in clauses (ii) and (iii) of Section 5.6(b) below.
- (b) Neither the execution and delivery by any Glass Party of this Agreement nor the consummation by such Glass Party of the Transactions in accordance with the terms hereof will require any consent, approval or authorization of, or filing or registration with, any Governmental Authority, other than (i) the filings provided for in <a href="Article 2">Article 2</a> and (ii) filings, consents, approvals and authorization required under applicable Antitrust Laws, the Exchange Act, the Securities Act or other applicable U.S. state securities and "blue sky" laws ((i) and (ii) collectively, the "Glass Regulatory Filings"),

and listing on NASDAQ of the Company Common Stock to be issued pursuant to the Mergers, or pursuant to stock options or convertible or exchangeable securities of Solstice or any Glass Party, except for any consent, approval or authorization the failure of which to obtain and for any filing or registration the failure of which to make would not prevent or materially delay the consummation of the Mergers or otherwise prevent any Glass Party from performing its obligations under this Agreement and would not, individually or in the aggregate, reasonably be expected to have a Glass Material Adverse Effect.

(c) Other than as contemplated by <u>Section 5.6(b)</u>, and except for the Required Glass Stockholder Approvals, no consents, assignments, waivers, authorizations or other certificates are necessary in connection with the Transactions to provide for the continuation in

54

full force and effect of all Glass Material Contracts or for Glass to consummate the Transactions, except where the failure to receive such consents, assignments, waivers, authorizations or other certificates would not, individually or in the aggregate, reasonably be expected to have a Glass Material Adverse Effect.

#### 5.7 <u>Financial Statements</u>.

- (a) Section 5.7(a) of the Glass Disclosure Letter sets forth the following financial statements (the "Glass Financial Statements"):
  (i) the audited consolidated balance sheet of Glass and its Subsidiaries as of December 31, 2016 and as of December 31, 2015, and the related statements of income, changes in equity and cash flows for the years ending December 31, 2016 and December 31, 2015 and (ii) the unaudited consolidated balance sheet of Glass and its Subsidiaries as of March 31, 2017, and the related unaudited statements of operations and cash flows, respectively, for the three-month period ended on such date (the "Glass Most Recent Unaudited Financial Statements"). Each of the Glass Financial Statements has been prepared in accordance with GAAP applied on a basis consistent with prior periods and fairly presents in all material respects the consolidated financial condition of Glass and its Subsidiaries as of its respective date and the consolidated results of operations and stockholders' equity, or cash flows, as the case may be, of Glass and its Subsidiaries for the period covered thereby, subject, in the case of the Glass Most Recent Unaudited Financial Statements, to the absence of footnote disclosure and to normal, recurring end-of-period adjustments which are, in the aggregate, not material.
- (b) The financial books and records of the Glass Parties and their respective Subsidiaries have been maintained in accordance with customary business practices and fairly and accurately reflect, in all material respects, on a basis consistent with past periods and throughout the periods involved, (i) the consolidated financial position of the Glass Parties and their respective Subsidiaries and (ii) all transactions of the Glass Parties and their respective Subsidiaries, including all transactions between any Glass Party or any of its Subsidiaries, on the one hand, and any holder of any Glass Party, on the other hand. Glass has not received any advice or notification from its independent accountants that any Glass Party has used any improper accounting practice that would have the effect of not reflecting or incorrectly reflecting in the books and records of any Glass Party or any of its Subsidiaries any properties, assets, liabilities, revenues, expenses, equity accounts or other accounts.
- (c) Except as reflected in the Glass Most Recent Unaudited Financial Statements, no Glass Party nor any Subsidiary of any Glass Party has any liabilities of any nature (whether accrued, contingent, absolute or otherwise) that would be required to be reflected on, or reserved against in, a consolidated balance sheet of any Glass Party or in the notes thereto prepared in accordance with GAAP consistently applied, except for liabilities that may have arisen in the ordinary course of business since March 31, 2017 and which are not, and would not reasonably be expected to have, a Glass Material Adverse Effect.
- 5.8 <u>Litigation</u>. There are no actions, suits or proceedings pending against any Glass Party or any of its Subsidiaries or, to Glass's knowledge, threatened against any Glass Party or any of its Subsidiaries, at law or in equity, or before or by any Governmental Authority, that would reasonably be expected to have, individually or in the aggregate, a Glass Material Adverse

55

Effect. There are no outstanding judgments, decrees, injunctions, awards or orders against any Glass Party or any of its Subsidiaries except as would not, individually or in the aggregate, reasonably be expected to have a Glass Material Adverse Effect.

5.9 <u>Absence of Certain Changes</u>. Since December 31, 2016, each of the Glass Parties and their respective Subsidiaries has conducted its business in the ordinary course consistent with past practice, and there has not been any event that has had or would be reasonably expected to have, individually or in the aggregate, a Glass Material Adverse Effect. Without limiting the generality of the foregoing, since December 31, 2016, no Glass Party nor any of their respective Subsidiaries has taken any action (or has failed to take any action) that would require the consent of Solstice under <u>Section 6.2</u> if taken (or failed to be taken) after the date of this Agreement and prior to the Effective Time.

## 5.10 <u>Taxes</u>.

- (a) Each of the Glass Parties and their respective Subsidiaries is classified as set forth in <u>Section 5.10(a)</u> of the Glass Disclosure Letter for all U.S. federal and applicable state income Tax purposes.
- (b) Each of the Glass Parties and each of their respective Subsidiaries has properly filed on a timely basis all material Tax Returns that it was required to file, and all such Tax Returns are true, correct and complete in all material respects. Except as would not reasonably be expected to have, individually or in the aggregate, a Glass Material Adverse Effect, each of the Glass Parties and each of their respective Subsidiaries has paid on a timely basis all Taxes, whether or not shown on any Tax Return, that were due and payable. Except as would not reasonably be expected to have, individually or in the aggregate, a Glass Material Adverse Effect, the unpaid Taxes of each of the Glass Parties and their respective Subsidiaries (A) for taxable periods through the date of the most recent balance sheet included in the Glass Financial Statements (or, in the case of the Glass Corporate Members, the most recent balance sheet included in the financial books and records of such Glass Corporate Member) do not exceed the accruals and reserves for Taxes (excluding accruals and reserves for deferred Taxes established to reflect timing differences between book and Tax income) set forth on such most recent balance sheet and (B) for taxable periods through the Closing Date, will not exceed the reserve as adjusted for the passage of time through the Closing Date in accordance with GAAP. All material unpaid Taxes of each of the Glass Parties and their respective Subsidiaries for all taxable periods commencing after the date of the most recent balance sheet included in the Glass Financial Statements (or, in the case of the Glass Corporate Members, the most recent balance sheet included in the financial books and records of such Glass Corporate Member) arose in the ordinary course of business.

- (d) Neither Glass nor any of its Subsidiaries is or has ever been a member of an affiliated group with which it has filed (or been required to file) consolidated, combined, unitary or similar Tax Returns, other than a group of which the common parent is Glass. At all times since January 1, 2012, none of the Glass Corporate Members nor any of their respective Subsidiaries is or has been a member of an affiliated group with which it has filed (or been required to file) consolidated, combined, unitary or similar Tax Returns. With the exceptions of customary commercial leases or contracts that are not primarily related to Taxes entered into in the ordinary course of business and liabilities thereunder, none of the Glass Parties nor any of their respective Subsidiaries (i) has any liability under Treasury Regulation Section 1.1502-6 (or any comparable or similar provision of federal, state, local or foreign law), as a transferee or successor, pursuant to any contractual obligation, or otherwise for any Taxes of any Person other than the Glass Parties or any of their respective Subsidiaries, or (ii) is a party to or bound by any Tax indemnity, Tax sharing, Tax allocation or similar agreement, other than the Glass Organizational Documents or similar documents with respect to Subsidiaries of the Glass Parties.
- (e) Glass has made available to Solstice (i) complete and correct copies of all private letter rulings, material closing agreements, material settlement agreements, pending ruling requests and any similar documents submitted by, received by, or agreed to by or on behalf of any of the Glass Parties or any of their respective Subsidiaries relating to Taxes for all taxable periods ending on or after December 31, 2011, and (ii) complete and correct copies of all material agreements, rulings, settlements or other Tax documents with or from any Governmental Authority relating to Tax incentives of any of the Glass Parties or any of their respective Subsidiaries.
- (f) No examination, suit, proceeding, investigation or audit or other similar action of or relating to any material Tax Return of any of the Glass Parties or any of their respective Subsidiaries by any Governmental Authority is currently in progress or, to the knowledge of the Glass Parties, threatened or contemplated. No deficiencies for material Taxes of any of the Glass Parties or any of their respective Subsidiaries have been claimed, proposed or assessed in writing by any Governmental Authority. None of the Glass Parties nor any of their respective Subsidiaries has been informed in writing by any jurisdiction in which such Glass Party or any of its Subsidiaries does not file a Tax Return that the jurisdiction believes that such Glass Party or any of its Subsidiaries was required to file any Tax Return that was not filed or is subject to Tax in such jurisdiction with respect to Taxes that would be subject of such Tax Return. None of the Glass Parties nor any of their respective Subsidiaries has (i) waived any statute of limitations with respect to Taxes or agreed to extend the period for assessment or collection of any Taxes, which waiver or extension is still in effect, (ii) requested any extension of time within which to file any Tax Return, which Tax Return has not yet been filed, or (iii) executed or filed any power of attorney with any taxing authority, which is still in effect.
- (g) None of the Glass Parties nor any of their respective Subsidiaries has made any payment, is obligated to make any payment, or is a party to any agreement, contract, arrangement or plan that could obligate it to make any payment that may be treated as an "excess parachute payment" under Section 280G of the Code (without regard to Sections 280G(b)(4) and 280G(b)(5) of the Code) in connection with the Transactions.

- (h) None of the Glass Parties nor any of their respective Subsidiaries will be required to include any material item of income in, or exclude any material item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of (i) any adjustments under Section 481 of the Code (or any similar adjustments under any provision of the Code or the corresponding foreign, state or local Tax law) due to a change in method of accounting made prior to Closing, (ii) deferred intercompany gain or any excess loss account described in Treasury Regulations under Section 1502 of the Code (or any corresponding provision of state, local or foreign Tax law) that arose prior to Closing, (iii) closing agreement as described in Section 7121 of the Code (or any corresponding or similar provision of state, local or foreign Tax law) executed prior to the Closing, (iv) installment sale or open transaction disposition made prior to the Closing, (v) prepaid amount received prior to the Closing, or (vi) any election made pursuant to Section 108(i) of the Code prior to the Closing.
- (i) None of the Glass Parties nor any of their respective Subsidiaries has been a United States real property holding corporation within the meaning of Section 897(c)(2) of the Code during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code.
- (j) None of the Glass Parties nor any of their respective Subsidiaries (nor any of their predecessors) has distributed to its shareholders or security holders stock or securities of a controlled corporation, nor has stock or securities of any of the Glass Parties or any of their respective Subsidiaries been distributed, in a transaction to which Section 355 or Section 361 of the Code applies in the two years prior to the date of this Agreement.
- (k) There are no material liens or other encumbrances with respect to Taxes upon any of the assets of any of the Glass Parties or any of their respective Subsidiaries, other than with respect to Taxes not yet due and payable.
- (l) None of the Glass Parties nor any of their respective Subsidiaries is a stockholder in a "passive foreign investment company" within the meaning of Section 1297 of the Code.
- (m) None of the Glass Parties nor any of their respective Subsidiaries has engaged in a "reportable transaction" as set forth in Treasury Regulation Section 1.6011-4(b) or a "listed transaction" as set forth in Treasury Regulation Section 301.6111-2(b)(2) or any analogous provision of state or local law. Each of the Glass Parties and each of their respective Subsidiaries has disclosed on its federal income Tax Returns all positions taken therein that could give rise to a substantial understatement of federal income Tax within the meaning of Section 6662 of the Code.
- (n) None of the Glass Parties nor any of their respective Subsidiaries has taken any action, has failed to take any action, or knows of any fact or circumstance that, in each case, could reasonably be expected to prevent (i) the Solstice Merger from qualifying as a reorganization under Section 368(a) of the Code or as a transaction that, taken together with the other Mergers, qualifies as a transfer of property to the Company described in

Company to liquidate, merge, transfer all or substantially all of the assets of, or otherwise dissolve Solstice Surviving Corporation, Glass Surviving Corporation or GB Surviving Entity.

## 5.11 <u>Employee Benefit Plans</u>.

- Section 5.11(a) of the Glass Disclosure Letter sets forth a list of each material Glass Plan. "Glass Plans" means all employee or director benefit plans, programs, policies, agreements or other arrangements, including any employee welfare plan within the meaning of Section 3(1) of ERISA, any employee pension benefit plan within the meaning of Section 3(2) of ERISA (in each case whether or not such plan is subject to ERISA), any employment, individual consulting or other compensation agreements and any bonus, incentive, equity or equity-based compensation, deferred compensation, vacation, stock purchase, stock option, severance, employment, change of control, retirement, salary continuation, disability plan, health or life insurance or fringe benefit plan, program or agreement, in each case that are sponsored, maintained or contributed to by any Glass Party or any of its Subsidiaries for the benefit of current or former employees, directors or consultants of any Glass Party or its Subsidiaries or to which any Glass Party or any of its Subsidiaries has any obligation or liability (contingent or otherwise), directly or through ERISA Affiliates; provided, however, that Glass Plans shall not include any Glass Foreign Plan or any Multiemployer Plan. For purposes of this Agreement, (x) individual employment offer letters, employment agreements, severance arrangements or other individually negotiated compensation arrangements shall be deemed to be "material" Glass Plans only if they are not terminable on less than 60 days' notice or they require payment of severance payments in excess of \$50,000 upon any termination of employment; and (y) the term "Glass Foreign Plan" shall refer to each material plan, program or Contract that is subject to or governed by the laws of any jurisdiction other than the United States, and which would have been treated as a Glass Plan had it been a United States plan, program or Contract. With respect to each material Glass Plan, Glass has made available to Solstice a true, correct and complete copy of the following (where applicable): (i) each writing constituting a part of such Glass Plan, including all plan documents, trust agreements, and other funding vehicles; (ii) the most recent annual report (Form 5500 Series) and accompanying schedule, if any; (iii) the current summary plan description and any material modifications thereto, if any; (iv) the most recent annual financial report, if any; (v) the most recent actuarial report, if any, and (vi) the most recent determination, opinion or advisory letter from the Internal Revenue Service, if any. Glass has made available to Solstice a copy of each material Glass Foreign Plan or summaries of the material terms thereof.
- (b) With respect to each Glass Plan that is intended to be a "qualified plan" within the meaning of Section 401(a) of the Code ("Qualified Glass Plans"), the Internal Revenue Service has issued a favorable determination, opinion, or advisory letter with respect to such Qualified Glass Plan and the related trust that has not been revoked, and to the knowledge of Glass, no circumstances exist and no events have occurred that would reasonably be expected to adversely affect the qualified status of any Qualified Glass Plan or the related trust.
- (c) All material contributions required to be made to any Glass Plan by Applicable Law or by any plan document or other contractual undertaking, and all premiums due or payable with respect to insurance policies funding any Glass Plan, before the date of this

59

Agreement, have been made or paid in full in all material respects on or before the final due date thereof.

- (d) The Glass Parties and their respective Subsidiaries have complied, and to the knowledge of Glass are now in compliance, in all material respects, with all provisions of ERISA, the Code and all laws and regulations applicable to the Glass Plans. To the knowledge of Glass, each Glass Plan has been operated in material compliance with its terms. No Glass Party nor any of its ERISA Affiliates maintains, contributes to, or has or has had any liability with respect to any employee benefit plan subject to Title IV or Section 302 of ERISA (including any Multiemployer Plan). No Glass Plan subject to ERISA holds securities issued by any Glass Party, any of its Subsidiaries, or its ERISA Affiliates.
- (e) Except for health continuation coverage as required by Section 4980B of the Code or Part 6 of Title I of ERISA or in connection with and during a limited period in which severance is paid (not to exceed two years), no Glass Party nor any of its Subsidiaries has promised or provided life, health, medical or other welfare benefits to former employees or beneficiaries or dependents thereof.
- (f) Neither the execution and delivery of this Agreement, shareholder approval of the Transactions, nor the consummation of the Transactions, either alone or in combination with another event, will result in, cause the accelerated vesting, funding of any amounts to a rabbi trust or delivery of, or increase the amount or value of, any payment or benefit to any employee, officer, director or consultant of any Glass Party or any of its Subsidiaries. No Glass Party nor any of its Subsidiaries is a party to, nor is any Glass Party or any of its Subsidiaries otherwise obligated under, any plan, policy, agreement or arrangement that provides for the gross-up or reimbursement of Taxes imposed under Section 409A or 4999 of the Code (or any corresponding provisions of state or local Applicable Law relating to Tax).
- (g) There are no pending or to the knowledge of Glass threatened claims (other than claims for benefits in the ordinary course), lawsuits or arbitrations which have been asserted or instituted against the Glass Plans or any Glass Foreign Plan, any fiduciaries thereof with respect to their duties to the Glass Plans or the assets of any of the trusts under any of the Glass Plans which would reasonably be expected to result in any material liability of any Glass Party or any of its Subsidiaries to any Glass Plan, the United States Department of Treasury, the United States Department of Labor, or any current or former participants of such Glass Plans.
- (h) Each Glass Plan that is a "nonqualified deferred compensation plan" within the meaning of Section 409A(d)(1) or 457A of the Code and any award thereunder, in each case that is subject to Section 409A or 457A of the Code, has been operated in compliance in all material respects with Section 409A and, where applicable, 457A of the Code.
- (i) All material Glass Foreign Plans subject to the laws of any jurisdiction outside of the United States (i) have been maintained in all material respects in accordance with all applicable legal requirements; (ii) if they are intended to qualify for special tax treatment meet all requirements for such treatment in all material respects; and (iii) if they are required to be funded and/or book-reserved are so funded and/or book reserved, as applicable, in all material respects.

### 5.12 <u>Labor Matters</u>.

- (a) No Glass Party nor any of its Subsidiaries is a party to, or bound by, or as of the date of this Agreement negotiating, any Collective Bargaining Agreement, nor are the employees of any Glass Party or any of its Subsidiaries covered by any works council, employee representative agreement or information or consultation agreement. There is no material labor strike, slowdown, work stoppage or lockout pending or, to the knowledge of Glass, threatened against any Glass Party or any of its Subsidiaries, and no Glass Party nor any of its Subsidiaries has experienced any material labor dispute since December 31, 2014. To the knowledge of Glass, there are no material organizational efforts with respect to the formation of a collective bargaining unit presently being made or threatened involving employees of any Glass Party or any of its Subsidiaries, and there have not been any such material organizational efforts since December 31, 2014.
- (b) Except as would not reasonably be expected to have, individually or in the aggregate, a Glass Material Adverse Effect, (i) since December 31, 2014, the Glass Parties and their respective Subsidiaries have been in compliance with any (A) Applicable Law respecting employment and employment practices, terms and conditions of employment and wages and hours, including any such law respecting employment discrimination, employee classification, workers' compensation, family and medical leave, the Immigration Reform and Control Act and occupational safety and health requirements, and (B) employment agreement or other agreement establishing terms of employment; and (ii) no claims, controversies, investigations, audits or suits are pending or, to the knowledge of Glass, threatened in writing, with respect to such laws or agreements, either by private individuals or by Governmental Authorities. No Glass Party nor any of its Subsidiaries has incurred, and no circumstances exist under which a Glass Party or any of its Subsidiaries is reasonably likely to incur, any material liability (contingent or otherwise) arising from the misclassification of employees as consultants or independent contractors, or from the misclassification of consultants or independent contractors as employees.
- Environmental Matters. Except as would not, individually or in the aggregate, reasonably be expected to have a Glass Material Adverse Effect: (a) each Glass Party and each of its Subsidiaries are and have been in compliance in all respects with all applicable Environmental Laws; (b) each Glass Party and each of its Subsidiaries have obtained all Environmental Permits necessary for their operations as currently conducted and are in compliance with all such Environmental Permits, have applied for issuance or reissuance of Environmental Permits in a timely fashion, and have no knowledge of any reason that would preclude renewal, issuance or reissuance of Environmental Permits; (c) there are no Environmental Claims pending or, to the knowledge of Glass, threatened against any Glass Party or any of its Subsidiaries; (d) none of the Glass Parties nor any of their respective Subsidiaries is party to any agreement, order, judgment, or decree by or with any Governmental Authority or third party imposing any liability or obligation on any Glass Party or any of its Subsidiaries under any Environmental Law; (e) none of the Glass Parties nor any their respective Subsidiaries has retained or assumed, either contractually or by operation of law, any liability or obligation that would reasonably be expected to form the basis of any Environmental Claim against, or any liability under any Environmental Law on, any Glass Party or any of its Subsidiaries; and (f) Glass has made available to Solstice copies of any environmental assessments, reports, audits, studies, analyses, tests or monitoring possessed by, or reasonably available to, any Glass Party or any of its

61

Subsidiaries pertaining to compliance with, or liability under, Environmental Laws relating to any Glass Party or any of its Subsidiaries, the Glass Facilities, any real property formerly owned, leased or operated by any Glass Party or any of its Subsidiaries.

# 5.14 <u>Intellectual Property</u>.

- (a) Section 5.14(a) of the Glass Disclosure Letter lists all Glass Registrations, in each case, enumerating specifically the applicable filing or registration number, title, jurisdiction in which filing was made or from which registration issued, date of filing or issuance, as applicable. To Glass's knowledge, all assignments of Glass Registrations to any Glass Party or any of its Subsidiaries have been properly executed and recorded, except for such deficiencies as would not materially affect the enforceability thereof. To Glass's knowledge, all issuance, renewal, maintenance and other payments that are or have become due with respect to the Glass Registrations have been timely paid by or on behalf of the relevant Glass Party or Subsidiary thereof.
- (b) To Glass's knowledge, no action or proceeding is pending or threatened in which any Person is challenging the validity, enforceability, registration, ownership, or use of any Glass Registrations (excluding, for the avoidance of doubt, office actions issued by a Governmental Authority with respect to applications for Glass Registrations).
- (c) There are no third party joint owners of any Glass Registrations, and a Glass Party or one of its Subsidiaries is the sole and exclusive owner of all Glass Registrations.
- (d) Neither the execution and delivery by any Glass Party of this Agreement nor the consummation by such Glass Party of the Transactions in accordance with the terms hereof will result in the loss, termination, or impairment of any rights in the Glass Intellectual Property, except as would not, individually or in the aggregate, reasonably be expected to have a Glass Material Adverse Effect.
- (e) The Glass Parties and their respective Subsidiaries have taken commercially reasonable measures to maintain in confidence and protect the proprietary nature of all material trade secrets and confidential information comprising a part of the Glass Intellectual Property. To Glass's knowledge, there has been no: (i) unauthorized disclosure of any third party proprietary or confidential information in the possession, custody or control of any Glass Party or any of its Subsidiaries, or (ii) breach of any Glass Party's or any of its Subsidiaries' security or information privacy procedures, in each case which, individually or in the aggregate, would reasonably be expected to have a Glass Material Adverse Effect.
- (f) To Glass' knowledge, (i) no Glass Party nor any of their respective Subsidiaries is infringing, misappropriating, diluting, or otherwise violating any Intellectual Property of any third party, and (ii) no Person (including any current or former employee or consultant of any Glass Party or any of its Subsidiaries) is infringing, diluting, misappropriating, or otherwise violating any of the Glass Owned Intellectual Property. Section 5.14(f) of the Glass Disclosure Letter lists any written complaint, claim or notice, or written threat of any of the foregoing (including any written notification that a license under any patent is or may be required), that to Glass's knowledge, has been received by any Glass Party or any of its

Subsidiaries in the past two (2) years alleging (or requesting or demanding indemnification for) any such infringement, violation or misappropriation of third party intellectual property.

- None of the Glass Parties and their respective Subsidiaries, nor any of their employees, has licensed, distributed or disclosed the source code, other than as required by Open Source Material license agreements, for any Software included in the products or services offered by any Glass Party or any of its Subsidiaries, or other confidential information constituting, embodied in or pertaining to such Software (collectively, "Glass Source Code") to any person, other than escrow agents, and third party contractors engaged by or on behalf of the Glass Parties or their respective Subsidiaries to develop, modify, maintain or otherwise work with the Glass Source Code, which contractors are contractually obligated to maintain the confidentiality of the Glass Source Code, and the Glass Parties and their respective Subsidiaries have taken commercially reasonable physical and electronic security measures to prevent disclosure of such Glass Source Code. To Glass's knowledge, no event has occurred, and no circumstance or condition exists, that (with or without notice or lapse of time, or both) will, or would reasonably be expected to, nor will the consummation of the Transactions, result in the disclosure or release of Glass Source Code by any Glass Party, any of its Subsidiaries or escrow agent(s) or any other person to any third party.
- (h) Other than commercially reasonable protections against unauthorized use, to Glass's knowledge, the products and services offered by any Glass Party or any of its Subsidiaries, and the Software and internal computer systems used by any Glass Party or any of its Subsidiaries do not contain any disabling device, virus, worm, back door, Trojan horse or other disruptive or malicious code that may or are intended to materially impair their intended performance or otherwise permit unauthorized access to, hamper, delete or damage any computer system, software, network or data.
- (i) Each employee of any Glass Party or any of their respective Subsidiaries and each independent contractor of any Glass Party or any of their respective Subsidiaries who has contributed to the development of material Intellectual Property on behalf of any Glass Party or any of their respective Subsidiaries has executed a valid and binding written agreement assigning such Intellectual Property to or otherwise vesting ownership of such Intellectual Property in any Glass Party or its relevant Subsidiary. All of the agreements referenced in the preceding sentence will continue to be in full force and effect immediately following the Effective Time in accordance with the terms thereof as in effect immediately prior to the Effective Time.
- (j) To Glass's knowledge, no material support, funding, resources or assistance from any Governmental Authority has been received by any Glass Party or any of their respective Subsidiaries during the six years prior to the date of this Agreement in connection with the development, design, testing, modification, manufacture, use, sale, reproduction, marketing, distribution, support or maintenance of any of the products or services offered, or contemplated to be offered, by any Glass Party or any of their respective Subsidiaries. The Glass Parties and their respective Subsidiaries are in material compliance with all of the applicable terms of any such support, funding, resources or assistance, and are in material compliance with all Applicable Laws in connection therewith, including in respect of Patent

63

Rights arising out of, relating to, or developed in connection with any such support, funding, resources or assistance.

# 5.15 <u>Title to Properties</u>.

- (a) No Glass Party nor any of its Subsidiaries owns any real property.
- (b) Glass or one of its Subsidiaries has a good and valid leasehold interest in all real property leased or purported to be leased, or otherwise occupied or permitted to be occupied, by any Glass Party or any of its Subsidiaries, whether as sublessor, tenant, subtenant or otherwise (the "Glass Leased Real Property." and the lease, sublease or other occupancy agreement, the "Glass Real Property Leases"). Section 5.15(b) of the Glass Disclosure Letter sets forth a complete and accurate list of all Glass Leased Real Property, the location of the premises and each document comprising the Glass Real Property Leases, including all amendments thereto and guarantees thereof. No Glass Party nor any Subsidiary thereof has assigned, transferred, conveyed, mortgaged, deeded in trust or otherwise encumbered any interest in the leasehold or subleasehold. All Glass Leased Real Property is supplied with utilities and other services necessary for the operation of said facilities as currently operated. Each Glass Real Property Lease is in full force and effect, except as would not, individually or in the aggregate, reasonably be expected to have a Glass Material Adverse Effect, and will continue to be legal, valid, binding, enforceable and in full force and effect immediately following the Closing in accordance with the terms thereof as in effect prior to the Closing.
- (c) Glass or one of its Subsidiaries owns or leases all of the material personal property reflected on the most recent balance sheet included in the Glass Financial Statements, free and clear of all Liens, except to the extent disposed of in the ordinary course of business since the date of such balance sheet. The Glass Parties and their respective Subsidiaries have good and valid title to, or in the case of leased properties and assets, valid leasehold interests or other comparable contractual rights in, all material tangible properties and assets, real, personal and mixed, necessary for the conduct of their respective businesses, free and clear of all Liens.
- (d) To the knowledge of Glass, the Glass Leased Real Property and the equipment of the Glass Parties and their respective Subsidiaries used in the operations of their respective businesses are (i) suitable for the uses to which they are currently employed, (ii) in good operating condition and repair, subject to normal wear and tear, (iii) regularly and properly maintained, (iv) not obsolete, dangerous or in need of renewal or replacement, except for renewal or replacement in the ordinary course of business, and (v) free from any material defects or deficiencies.
- 5.16 <u>Insurance</u>. The Glass Parties and their respective Subsidiaries maintain insurance coverage reasonably adequate for the operation of their respective businesses (taking into account the cost and availability of such insurance).
- 5.17 No Brokers. No Glass Party nor any Subsidiary thereof has entered into any Contract with any person which may result in the obligation of any Glass Party, any Subsidiary thereof, the Company or Solstice to pay any finder's fees, brokerage or agent's commissions or other like payments in connection with the negotiations leading to this Agreement or the

consummation of the Transactions, except that Glass has retained Guggenheim Partners to provide financial advice with respect to the Transactions. Glass' arrangements with such investment banks have been disclosed in writing to Solstice prior to the date hereof.

- 5.18 <u>Solstice Stock Ownership</u>. No Glass Party nor any Subsidiary thereof owns any shares of capital stock of Solstice or any other securities convertible into or otherwise exercisable to acquire capital stock of Solstice. No Glass Party or any of their "affiliates" or "associates" is, or any time during the last three (3) years has been, an "interested stockholder" of Solstice as such terms are defined in Section 203 of the DGCL.
- 5.19 <u>Glass Party Board Recommendations</u>. The Board of Directors of each Glass Party has, by unanimous written consent of directors, and not subsequently rescinded or modified: (i) approved this Agreement and the Transactions; (ii) determined that, as of the date of this Agreement, this Agreement and the Transactions, including the applicable Merger, are in the best interests of its members or stockholders (as applicable) and declared the advisability of this Agreement; and (iii) recommended that such members or stockholders (as applicable) vote to adopt this Agreement and approve the applicable Glass Merger and directed that such matters be submitted for approval at a meeting of the members or stockholders of such Glass Party.
- 5.20 <u>Vote Required</u>. The Required Glass Stockholder Approvals are the only votes of the holders of any class or series of capital stock of any Glass Party necessary to approve this Agreement, the Glass Mergers and the Transactions.
- 5.21 <u>Certain Approvals</u>. No Takeover Law or anti-takeover provision in the Glass Organizational Documents is applicable to the Transactions or this Agreement. No Glass Party is a party to any "stockholder rights plan" or similar anti-takeover plan or device.

# 5.22 <u>Relationships with Related Parties</u>.

- (a) No officer, director, or record or beneficial owner of more than 5% of the equity interests of any Glass Party or any of its Subsidiaries nor any Affiliate of any of the foregoing (b) has any interest in any property (real, personal, or mixed and whether tangible or intangible), used in or pertaining to the business of any Glass Party or any of its Subsidiaries as currently conducted or contemplated to be conducted or (c) is a party to any Contract (except for Glass Plans) with any Glass Party or any of its Subsidiaries, including with respect to compensation or remuneration to be paid to such officer, director, record or beneficial owner of more than 5% of the equity interests in any Glass Party, or Affiliate in connection with this Agreement or the Transactions.
- (b) Except as set forth in Section 5.22(b) of the Glass Disclosure Letter, none of the Glass Parties nor any their respective Subsidiaries is indebted, directly or indirectly, to any Person who is an Affiliate, record or beneficial owner of more than 5% of the equity interests of any Glass Party, officer or director of any Glass Party or any of its Subsidiaries in any amount whatsoever, other than for salaries for services rendered or reimbursable business expenses, nor is any such Affiliate, record or beneficial owner of more than 5% of the equity interests of any Glass Party, officer or director indebted to any Glass Party or any of its Subsidiaries, except for advances made to employees of any Glass Party or any of its Subsidiaries in the ordinary course

65

of business consistent with past practice to meet reimbursable business expenses reasonably anticipated to be incurred by such obligor.

### 5.23 <u>Internal Controls</u>.

- (a) The Glass Parties maintain a system of internal control over financial reporting (as such term is defined in Rule 13a-15 under the Exchange Act) sufficient to provide reasonable assurance that: (i) transactions are executed in accordance with management's general or specific authorizations; (ii) transactions are recorded as necessary to permit preparation of financial statements in conformity with GAAP and to maintain asset accountability; (iii) access to assets is permitted only in accordance with management's general or specific authorization; and (iv) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences.
- (b) Since December 31, 2014, no Glass Party has identified or been made aware of any significant deficiencies or material weaknesses in the design or operation of internal controls over financial reporting that are reasonably likely to adversely affect the ability of any Glass Party to record, process, summarize and report financial information of the Glass Parties and their respective Subsidiaries on a consolidated basis.
- (c) None of the Glass Parties and their respective Subsidiaries is a party to, or has any commitment to become a party to, any joint venture, partnership agreement or any similar Contract (including any Contract or arrangement relating to any transaction, arrangement or relationship between or among any Glass Party or any of its Subsidiaries, on the one hand, and any unconsolidated Affiliate, including any structured finance, special purpose or limited purpose entity or Person, on the other hand (such as any arrangement described in Section 303(a)(4) of Regulation S-K under the Securities Act)) where the purpose or effect of such arrangement is to avoid disclosure of any material transaction involving any Glass Party or any of its Subsidiaries in Glass's consolidated financial statements.

# 5.24 Certain Contracts.

(a) Section 5.24(a) of the Glass Disclosure Letter sets forth, as of the date hereof, a complete and accurate list of all Glass Material Contracts. As used in this Agreement, "Glass Material Contract" means (i) any Contract pursuant to which any Glass Party or any of its Subsidiaries is reasonably likely to spend, in the aggregate, more than \$500,000 with respect to any such Contract during the current fiscal year or during the next fiscal year, (ii) any non-competition or other agreement that prohibits or otherwise restricts, in any material respect, any Glass Party or any of its Subsidiaries from freely engaging in any business material to the Glass Parties and their respective Subsidiaries, taken as a whole, anywhere in the world, (iii) any Contract to which any Glass Party or any of its Subsidiaries is a party involving research, development or the license of any Glass Intellectual Property (other than non-exclusive licenses of Glass Intellectual Property made in the ordinary course of business), (iv) any Contract to which any Glass Party or any of its Subsidiaries is a party granting a right of first refusal, or right of first offer or comparable right with respect to any material Glass Intellectual Property, (v) any Contract to which any Glass Party or any of its Subsidiaries is a party relating to a material joint

venture, partnership or other material arrangement involving a sharing of profits, losses, costs or liabilities with another person, (vi) any Contract which would be binding on an Affiliate of any Glass Party (other than one of its Subsidiaries) or Solstice or an Affiliate of Solstice and (vii) any "material contract" (as such term is defined in Item 601(b)(10) of Regulation S-K of the SEC as it would be applicable to any Glass Party if it were an SEC reporting company). Glass has made available to Solstice a complete and accurate copy of each Glass Material Contract. Each Glass Material Contract is in full force and effect and is enforceable in accordance with its terms, subject to the effects of bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other similar laws relating to or affecting creditors' rights generally and general equitable principles (whether considered in a proceeding in equity or at law). None of the Glass Parties and their respective Subsidiaries nor, to Glass's knowledge, any other party to any Glass Material Contract is in material violation of or in material default under (nor does there exist any condition, which, upon the passage of time or the giving of notice or both, would cause such a violation of or default under), nor will the consummation of the Transactions result in any material violation of or material default under (x) any loan or credit agreement, note, bond, mortgage or indenture to which it is a party or by which it or any of its properties or assets is bound or (y) any Glass Material Contract.

- (b) Section 5.24(b) of the Glass Disclosure Letter sets forth a complete and accurate list of each agreement to which any Glass Party or any of its Subsidiaries is a party or bound with any Affiliate of any Glass Party (other than any person that is a Glass Party or a direct or indirect whollyowned Subsidiary of a Glass Party). Complete and accurate copies of all the agreements listed in Section 5.24(b) of the Glass Disclosure Letter have heretofore been made available to Solstice. None of the Glass Parties and their respective Subsidiaries has entered into any transaction with any Affiliate of any Glass Party or any of its Subsidiaries (other than any person that is a Glass Party or a direct or indirect wholly-owned Subsidiary of a Glass Party) that would be subject to disclosure pursuant to Item 404 of Regulation S-K as it would be applicable to any Glass Party if it were an SEC reporting company, other than the transactions set forth on Section 5.24(b) of the Glass Disclosure Letter.
- (c) There is no non-competition or other similar agreement, commitment, judgment, injunction or order to which any Glass Party or any of its Subsidiaries is a party or is subject that has or would reasonably be expected to result in the effect of prohibiting or impairing the conduct of the business of any Glass Party or any of its Subsidiaries as currently conducted. None of the Glass Parties and their respective Subsidiaries has entered into (or is otherwise bound by) any agreement under which it is now, or following the Effective Time Solstice or any of Solstice's Affiliates (including the Glass Parties and their respective Subsidiaries) would be, restricted from selling, licensing or otherwise distributing any of their respective technology or products, or providing services to, customers or potential customers or any class of customers, in any geographic area, during any period of time or any segment of the market or line of business, provided, that the foregoing shall not include any obligations of Solstice or any of Solstice's Affiliates solely arising under agreements entered into prior to the Effective Time by Solstice or any of Solstice's Affiliates (excluding the Glass Parties and their respective Subsidiaries).
- (d) None of the Glass Parties and their respective Subsidiaries is a party to any agreement under which a third party would be entitled to receive a license or any other right

67

to Intellectual Property of Solstice or any of Solstice's Affiliates (excluding any Glass Party or any of their respective Subsidiaries) following the Closing, provided, that the foregoing shall not include any obligations of Solstice or any of the Solstice's Affiliates arising solely under agreements entered into prior to the Effective Time by Solstice or any of Solstice's Affiliates (excluding the Glass Parties and their respective Subsidiaries).

- 5.25 <u>Foreign Corrupt Practices Act</u>. Except for such matters as would not, individually or in the aggregate, reasonably be expected to have a Glass Material Adverse Effect, as of the date of this Agreement:
- (a) In connection with the Glass Parties' and their respective Subsidiaries' compliance with the Foreign Corrupt Practices Act, there have been no voluntary disclosures under the Foreign Corrupt Practices Act.
- (b) No Governmental Authority has notified any Glass Party or any of its Subsidiaries in writing of any actual or alleged violation or breach of the Foreign Corrupt Practices Act or any other similar Applicable Law.
- (c) None of the Glass Parties and their respective Subsidiaries has undergone or is undergoing any audit, review, inspection, investigation, survey or examination of records, in each case conducted by a Governmental Authority and relating to any Glass Party's or any of its Subsidiaries' compliance with the Foreign Corrupt Practices Act or any other similar Applicable Law, and to Glass's knowledge, there is no basis for any such audit, review, inspection, investigation, survey or examination of records by a Governmental Authority.
- (d) None of the Glass Parties and their respective Subsidiaries has been or is now under any administrative, civil or criminal charge or indictment or, to Glass's knowledge, investigation, alleging noncompliance with the Foreign Corrupt Practices Act or any other similar Applicable Law nor, to Glass's knowledge, is there any basis for any such charge, indictment or investigation.
- (e) None of the Glass Parties and their respective Subsidiaries has been or is now a party to any administrative or civil litigation alleging noncompliance with the Foreign Corrupt Practices Act or any other similar Applicable Law nor, to Glass's knowledge, is there any basis for any such proceeding.
- (f) No Glass Party or any Subsidiary thereof, nor any of their Affiliates, stockholders, members, directors, officers and employees nor any other Person acting on behalf of any of them has made any offer, payment, promise to pay, or authorization for the payment of any money, or any offer, gift, promise to give, or authorization of the giving of anything in value, which would cause any Glass Party or any of its Subsidiaries, or any of its or their Affiliates, stockholders, members, directors, officers or employees or any other Person acting on behalf of any of them, to have violated or be in violation of the Foreign Corrupt Practices Act or any other similar Applicable Law.

68

### 5.26 Export Control.

(a) Each of the Glass Parties and their respective Subsidiaries is and has been for the past five (5) years in material compliance with all Export Control Laws, including all applicable regulations pertaining to the disclosure of controlled technical information to foreign persons wherever located and/or the provision of access to such technical information by such foreign persons, and has maintained a program to facilitate such compliance, including

training, technology assessment and classification, transaction screening, license compliance tracking, export clearance and recordkeeping measures. None of the Glass Parties and their respective Subsidiaries has received any notice from any person alleging that any Glass Party or any of its Subsidiaries is not in compliance with, or has liability under, such Export Control Laws.

- (b) Each of the Glass Parties and their respective Subsidiaries has obtained and complied in all material respects with all licenses, agreements, authorizations, license exceptions or exemptions required for their respective exports of articles or technology or provision of services.
- (c) During the five (5) years prior to the date of this Agreement, none of the Glass Parties and their respective Subsidiaries has conducted or initiated any internal investigation, made any mandatory or voluntary disclosure, declined to make a voluntary disclosure with respect to known violation of Export Control Laws or failed to make any mandatory report or disclosure to any Governmental Authority pursuant to Export Control Laws.
- 5.27 <u>No Additional Representations.</u> Notwithstanding anything contained in this Agreement to the contrary, each Glass Party acknowledges and agrees that no Solstice Party nor any other Person has made or is making any representations or warranties relating to any Solstice Party whatsoever, express or implied, beyond those expressly given by the Solstice Parties in <u>Article 4</u> hereof, including any implied representation or warranty as to the accuracy or completeness of any information regarding any Solstice Party furnished or made available to any Glass Party or any of its Representatives. Without limiting the generality of the foregoing, each Glass Party acknowledges that no representations or warranties are made with respect to any projections, forecasts, estimates, budgets or prospect information that may have been made available to any Glass Party or any of its Representatives. No Glass Party has relied on any representations or warranties relating to any Solstice Party in determining to enter into this Agreement, except as expressly given by the Solstice Parties in <u>Article 4</u> hereof.

### **ARTICLE 6**

#### COVENANTS

### 6.1 Conduct of Solstice's Businesses.

(a) During the period from the date of this Agreement until the Effective Time or earlier termination of this Agreement in accordance with <a href="Article 8">Article 8</a> (the "Pre-Closing Period"), except as expressly set forth in the Solstice Disclosure Letter or as expressly required or permitted by any other provision of this Agreement, unless Glass has consented in writing thereto (which consent shall not be unreasonably withheld, delayed or conditioned), each

69

Solstice Party shall, and each shall cause each of its Subsidiaries to, conduct its operations in the ordinary course and shall use its reasonable best efforts to preserve intact their respective business organizations and goodwill, keep available the services of their respective officers and employees and maintain satisfactory relationships with those persons having business relationships with them.

- (b) Without limiting anything in Section 6.1(a), during the Pre-Closing Period, except as expressly set forth in the Solstice Disclosure Letter or as expressly required or permitted by any other provision of this Agreement, unless Glass has consented in writing thereto (which consent shall not be unreasonably withheld, delayed or conditioned), no Solstice Party shall, and each shall cause its Subsidiaries not to:
- (i) amend its certificate of incorporation, by-laws or comparable organizational documents or effect any split or reverse split of any of its shares of capital stock or other equity interests;
- (ii) (A) except pursuant to the exercise of options, warrants, conversion rights and other contractual rights or upon the settlement of restricted stock units in each case existing on the date hereof, issue any shares of its capital stock or other equity securities (including under the New Equity Plan) (other than to Solstice or one of its wholly-owned Subsidiaries) except for grants to new hires in the ordinary course of business of Solstice Restricted Shares or Solstice RSUs; (B) grant, confer or award any option, warrant, conversion right or other right to acquire any shares of its capital stock except for grants to new hires in the ordinary course of business of Solstice Restricted Shares or Solstice RSUs (including under the New Equity Plan); (C) increase any compensation or benefits of employees, officers, or directors, except for non-material increases in compensation and benefits in the ordinary course of business consistent with past practice, or enter into or amend any employment, severance, change in control, retention, or similar plans, arrangements or agreements with any of its present or future officers or directors, except (1) offer letters that do not provide for severance or notice of termination with new employees consistent with past practice or (2) amendments to the extent required by Applicable Law, or (D) adopt any plan, program, policy, agreement or other arrangement that, if in existence on the date hereof would have constituted a Solstice Plan (including any stock option, stock benefit or stock purchase plan) or amend (except to the extent required by Applicable Law) any existing Solstice Plan in any material respect, or (E) enter into, amend, modify or terminate any Collective Bargaining Agreement;
- (iii) (A) declare, set aside or pay any dividend or make any other distribution or payment with respect to any shares of its capital stock (except for dividends and distributions from Subsidiaries of Solstice to Solstice or any of its Subsidiaries) or (B) except to the extent otherwise permitted pursuant to Section 6.1(b)(ii), redeem, purchase or otherwise acquire any shares of its capital stock or capital stock of any of its Subsidiaries or any option, warrant, conversion right or other right to acquire such shares, or make any commitment for any such action;
- (iv) sell, lease, encumber or otherwise dispose of (by merger or otherwise), or enter into a contract to sell, lease, encumber or otherwise dispose of (by merger or otherwise), any of its tangible assets (including capital stock of Subsidiaries) or properties,

70

except for (A) sales of inventory or products in the ordinary course of business and (B) sales of surplus or obsolete equipment;

(v) sell, assign, transfer, license, abandon, permit to lapse, or otherwise dispose of any Solstice Intellectual Property, except for (i) non-exclusive licenses in the ordinary course of business, and (ii) the abandonment in the ordinary course of business of Solstice Registrations that are not material to the business of Solstice or any of its Subsidiaries and that are reasonably determined by Solstice to be no longer economical to maintain;

- (vi) enter into any joint venture outside of the ordinary course of business, enter into any partnership or make any election to treat any venture with a third party as a partnership, make or commit to make capital expenditures, or acquire or agree to acquire by merging or consolidating with, or by purchasing a substantial equity interest in or a substantial portion of the assets of, or by any other manner, any business or any corporation, partnership, association or other business organization or division thereof or otherwise acquire or agree to acquire any assets or securities, except that Solstice and its Subsidiaries may make and commit to make capital expenditures and such acquisitions that do not involve payments by it and its Subsidiaries that, in the aggregate, exceed (x) amounts set forth in the Solstice capital expenditures budget provided to the Glass Parties prior to the date hereof or (y) \$500,000 in the aggregate;
- (vii) except as may be required as a result of a change in Applicable Law or in generally accepted accounting principles, change any of the accounting principles or practices used by it;
- (viii) fail to use their reasonable efforts to maintain with financially responsible insurance companies insurance in such amounts and against such risks and losses as are customary for such party;
- (ix) except as otherwise permitted by <u>Section 6.18</u>, settle, or agree to settle any action, suit, litigation, investigation or proceeding pending or threatened before any arbitrator, court or other Governmental Authority for settlement amounts in excess of \$250,000 individually or \$1,000,000 in the aggregate;
- (x) waive, release or assign any claims or rights having a value in excess of \$250,000 individually or \$1,000,000 in the aggregate;
- (xi) make or change any material Tax election, change an annual accounting period, file any amended income or other material Tax Return except as required by Applicable Law, enter into any closing agreement with respect to a material amount of Taxes, waive or extend any statute of limitations with respect to Taxes, settle or compromise any material Tax liability, claim or assessment, surrender any right to claim a material refund of Taxes or take any other similar action relating to the filing of any Tax Return or the payment of any material amount of Taxes;
- (xii) (A) incur any Indebtedness for borrowed money in excess of \$1,000,000 in the aggregate, (B) except in the ordinary course of business, enter into any

71

material lease (whether such lease is an operating or capital lease) or create any material mortgages, liens, security interests or other encumbrances on the property of Solstice or any of its Subsidiaries in connection with any Indebtedness thereof, or (C) make any loans, advances or capital contributions to, or investments in, any other Person, other than to Solstice or any of its Subsidiaries and other than for cash management purposes in the ordinary course of business;

- (xiii) except in the ordinary course of business, enter into, amend, modify, extend or terminate any Solstice Material Contact, or waive any rights or claims thereunder;
- (xiv) subject to <u>Sections 6.3</u>, <u>6.5</u> and <u>6.8</u>, take any action that would reasonably be expected to delay materially or adversely affect the ability of any of the parties hereto to (1) obtain any consent, authorization, order or approval of any Governmental Authority or the expiration of any applicable waiting period required to consummate the Transactions or (2) consummate the Transactions;
- (xv) cause the acceleration of rights, benefits or payments under any Solstice Plans other than any such acceleration resulting from the consummation of the Transactions and the transactions expressly provided for herein under the terms of the Solstice Plans as of the date hereof;
- (xvi) except in the ordinary course of business, enter into forward sales contracts, fixed price contracts, fixed price swaps, collars, options or other hedging arrangements, except as permitted by such programs currently in effect and approved by Solstice's Board of Directors as of the date hereof;
- (xvii) adopt a plan of complete or partial liquidation, dissolution, merger, consolidation, restructuring, recapitalization or other reorganization of Solstice or any of its Subsidiaries (other than the Transactions and the Mergers and other than a merger among wholly-owned Subsidiaries of Solstice), or enter into any agreement with respect to the voting of its capital stock or other securities held by Solstice or any of its Subsidiaries; or
  - (xviii) agree in writing to take any of the foregoing actions;

*provided*, *however*, that, notwithstanding any other provision of this Agreement, prior to the Closing, neither party shall manage or interfere with the other party's conduct of business in the ordinary course.

# 6.2 Conduct of the Glass Parties' Businesses.

(a) During the Pre-Closing Period, except as expressly set forth in the Glass Disclosure Letter or as expressly required or permitted by any other provision of this Agreement, unless Solstice, has consented in writing thereto (which consent shall not be unreasonably withheld, delayed or conditioned), each Glass Party shall, and each shall cause each of its Subsidiaries to, conduct its operations in the ordinary course and use its reasonable best efforts to preserve intact their respective business organizations and goodwill, keep available the services

72

 $of their respective \ of ficers \ and \ employees \ and \ maintain \ satisfactory \ relationships \ with \ those \ persons \ having \ business \ relationships \ with \ them.$ 

(b) Without limiting anything in <u>Section 6.2(a)</u>, during the Pre-Closing Period, except as expressly set forth in the Solstice Disclosure Letter or as expressly required or permitted by any other provision of this Agreement, unless Solstice has consented in writing thereto (which consent shall

not be unreasonably withheld, delayed or conditioned), no Glass Party shall, and each shall cause its Subsidiaries not to:

(i) amend its certificate of incorporation, by-laws or comparable organizational documents or effect any split or reverse split

of any of its shares of capital stock or other equity interests;

- (ii) (A) except pursuant to the exercise of settlement of any equity awards and other contractual rights in each case existing on the date hereof, issue any shares of capital stock or other equity interests (other than to a Glass Party or one of their wholly-owned Subsidiaries); (B) grant, confer or award any option, warrant, conversion right or other right to acquire any Glass Shares or shares of capital stock of any of its Subsidiaries; (C) increase any compensation or benefits of employees, officers, or directors, except for non-material increases in compensation and benefits in the ordinary course of business consistent with past practice, or enter into or amend any employment, severance, change in control, retention, or similar plans, arrangements or agreements with any of its present or future officers or directors, except (1) offer letters that do not provide for severance or notice of termination with new employees consistent with past practice or (2) amendments to the extent required by Applicable Law, or (D) adopt any new plan, program, policy, agreement or other arrangement that, if in existence on the date hereof would have constituted a Glass Plan (including any stock option, stock benefit or stock purchase plan) or amend (except to the extent required by Applicable Law) any existing Glass Plan in any material respect, or (E) enter into, amend, modify or terminate any Collective Bargaining Agreement;
- (iii) (A) declare, set aside or pay any dividend or make any other distribution or payment with respect to any Glass Shares or equity interests of any of its Subsidiaries, (B) pay any management or similar fee, (C) repay any of the loans made to Glass prior to the date hereof by any Glass Member or the controlling person of any Glass Member, or (D) redeem, purchase or otherwise acquire any Glass Shares or equity interests of any of its Subsidiaries or any option, warrant, conversion right or other right to acquire such Glass Shares or shares, or make any commitment for any such action, except as required under any Glass Equity Plan disclosed in the Glass Disclosure Letter;
- (iv) sell, lease, encumber or otherwise dispose of (by merger or otherwise), or enter into a contract to sell, lease, encumber or otherwise dispose of (by merger or otherwise), any of its tangible assets (including capital stock of Subsidiaries) or properties, except for (A) sales of inventory or products in the ordinary course of business and (B) sales of surplus or obsolete equipment;
- (v) sell, assign, transfer, license, abandon, permit to lapse, or otherwise dispose of any Glass Intellectual Property, except for (i) non-exclusive licenses in the

73

ordinary course of business, and (ii) the abandonment in the ordinary course of business of Glass Registrations that are not material to the business of Glass or any of its Subsidiaries and that are reasonably determined by Glass to be no longer economical to maintain;

- (vi) enter into any joint venture outside the ordinary course of business, enter into any partnership or make any election to treat any venture with a third party as a partnership, make or commit to make capital expenditures, or acquire or agree to acquire by merging or consolidating with, or by purchasing a substantial equity interest in or a substantial portion of the assets of, or by any other manner, any business or any corporation, partnership, association or other business organization or division thereof or otherwise acquire or agree to acquire any assets or securities, except that Glass and its Subsidiaries may make and commit to make capital expenditures and such acquisitions that do not involve payments by it and its Subsidiaries that, in the aggregate, exceed (x) amounts set forth in the Glass Parties' capital expenditures budget provided to Solstice prior to the date hereof or (y) \$500,000 in the aggregate.
- (vii) except as may be required as a result of a change in Applicable Law or in generally accepted accounting principles, change any of the accounting principles or practices used by it;
- (viii) fail to use their reasonable efforts to maintain with financially responsible insurance companies insurance in such amounts and against such risks and losses as are customary for such party;
- (ix) except as otherwise permitted by <u>Section 6.18</u>, settle, or agree to settle any action, suit, litigation, investigation or proceeding pending or threatened before any arbitrator, court or other Governmental Authority for settlement amounts in excess of \$250,000 individually or \$1,000,000 in the aggregate;
- (x) waive, release or assign any claims or rights having a value in excess of \$250,000 individually or \$1,000,000 in the aggregate;
- (xi) make or change any material Tax election, change an annual accounting period, file any amended income or other material Tax Return except as required by Applicable Law, enter into any closing agreement with respect to a material amount of Taxes, waive or extend any statute of limitations with respect to Taxes, settle or compromise any material Tax liability, claim or assessment, surrender any right to claim a material refund of Taxes or take any other similar action relating to the filing of any Tax Return or the payment of any material amount of Taxes;
- (xii) (A) incur any Indebtedness for borrowed money in excess of \$1,000,000 in the aggregate, (B) except in the ordinary course of business, enter into any material lease (whether such lease is an operating or capital lease) or create any material mortgages, liens, security interests or other encumbrances on the property of any Glass Party or any of its Subsidiaries in connection with any Indebtedness thereof, or (C) make any loans, advances or capital contributions to, or investments in, any other Person, other than to any Glass

74

Party or any of its Subsidiaries and other than for cash management purposes in the ordinary course of business;

(xiii) except in the ordinary course of business, enter into, amend, modify, extend or terminate any Glass Material Contact, or waive any rights or claims thereunder;

- (xiv) subject to <u>Section 6.8</u>, take any action that would reasonably be expected to delay materially or adversely affect the ability of any of the parties hereto to (A) obtain any consent, authorization, order or approval of any Governmental Authority or the expiration of any applicable waiting period required to consummate the Transactions or (B) consummate the Transactions;
- (xv) cause the acceleration of rights, benefits or payments under any Glass Plans other than any such acceleration resulting from the consummation of the Transactions and the transactions expressly provided for herein under the terms of the Glass Plans as of the date hereof;
- (xvi) except in the ordinary course of business, enter into forward sales contracts, fixed price contracts, fixed price swaps, collars, options or other hedging arrangements, except as permitted by such programs currently in effect and approved by Glass's Board of Directors as of the date hereof;
- (xvii) adopt a plan of complete or partial liquidation, dissolution, merger, consolidation, restructuring, recapitalization or other reorganization of any Glass Party or any of its Subsidiaries (other than the Transactions and the Mergers and other than a merger among wholly-owned Subsidiaries of Glass), or enter into any agreement with respect to the voting of its capital stock or other securities held by Glass or any of its Subsidiaries; or
  - (xviii) agree in writing to take any of the foregoing actions;

*provided*, *however*, that, notwithstanding any other provision of this Agreement, prior to the Closing, neither party shall manage or interfere with the other party's conduct of business in the ordinary course.

# 6.3 No Solicitation By Solstice.

(a) Subject to Section 6.3(c) through Section 6.3(f), each Solstice Party agrees that, during the Pre-Closing Period, neither it nor any of its Subsidiaries shall, and that it shall cause its and its Subsidiaries' respective directors, officers, employees, agents, investment bankers, attorneys, accountants and other representatives ("Representatives") not to, directly or indirectly, (i) maintain, initiate or solicit or knowingly encourage or facilitate (including by way of furnishing information which has not been previously publicly disseminated) any inquiries with respect to, or the making of any proposal which constitutes, or would reasonably be expected to lead to, a Solstice Acquisition Proposal, (ii) engage in any discussions or negotiations concerning, or provide any confidential information or data regarding Solstice or its Subsidiaries to any person relating to, a Solstice Acquisition Proposal, (iii) approve or

75

recommend, or propose publicly to approve or recommend, any Solstice Acquisition Proposal, (iv) approve or recommend, or propose publicly to approve or recommend, or execute or enter into, any letter of intent, agreement in principle, merger agreement, acquisition agreement, option agreement or other similar agreement relating to any Solstice Acquisition Proposal, (v) take any action to make the provisions of any takeover statute inapplicable to any transactions contemplated by a Solstice Acquisition Proposal, or (vi) propose publicly or agree to do any of the foregoing.

- (b) Subject to Section 6.3(c) through Section 6.3(f), during the Pre-Closing Period, neither the Board of Directors of Solstice nor any committee thereof shall, directly or indirectly, (i) withdraw, withhold, modify or qualify, or publicly propose to withdraw, withhold, modify or qualify, in a manner adverse to Glass, the Solstice Board Recommendation, (ii) approve, adopt or recommend, or publicly propose to approve, adopt or recommend, any Solstice Acquisition Proposal, (iii) in the event of the commencement of a tender offer or exchange offer for any outstanding shares of Solstice's capital stock, fail to recommend against acceptance of such tender offer or exchange offer by the Solstice stockholders (including, subject to Section 6.3(c), by taking no position or a neutral position with respect to any such offer) within ten Business Days of the commencement thereof, or (iv) recommend that the Solstice stockholders not adopt this Agreement or approve the Solstice Merger or the Solstice Merger-Related Proposals (any action described in clauses (i)-(iv) above being referred to as an "Solstice Change of Recommendation").
- (c) Nothing contained in this Agreement shall prevent Solstice or the Board of Directors of Solstice from (i) complying with its disclosure obligations under Rule 14d-9 and 14e-2 promulgated under the Exchange Act, or issuing a "stop, look and listen" statement pending disclosure of its position (none of which, in and of itself, shall be deemed to constitute a Solstice Change of Recommendation") or (ii) making any disclosure to Solstice's stockholders if the Board of Directors of Solstice determines in good faith, after consultation with its outside legal counsel, that the failure to make such disclosure would be inconsistent with its fiduciary duties or Applicable Law.
- (d) Notwithstanding the limitations set forth in Section 6.3(a) and Section 6.3(b), until the earlier of the receipt of the Required Solstice Stockholder Approval and any termination of this Agreement pursuant to Article 8, if after the date of this Agreement Solstice receives a written bona fide Solstice Acquisition Proposal, that was not, directly or indirectly, solicited, initiated or knowingly encouraged in material violation of this Section 6.3, and that the Board of Directors of Solstice has determined in good faith, after consultation with its outside legal counsel and financial advisors (i) constitutes a Solstice Superior Proposal or (ii) could reasonably be expected to result in a Solstice Superior Proposal, then Solstice, its Subsidiaries and its and their respective Representatives may, subject to Section 6.3(h), in response to such Solstice Acquisition Proposal take the following actions, to the extent that the Board of Directors of Solstice has determined in good faith, after consultation with its outside legal counsel, that a failure to take such actions would be inconsistent with its fiduciary duties to Solstice's stockholders under Applicable Law: (A) furnish nonpublic information to the third party making such Solstice Acquisition Proposal, if, and only if, prior to furnishing such information, Solstice receives from the third party an executed confidentiality agreement with provisions no less restrictive to such third party with respect to the use or disclosure of nonpublic information than

76

the Confidentiality Agreement and (B) engage in discussions or negotiations with the third party with respect to such Solstice Acquisition Proposal.

(e) Notwithstanding anything in this Agreement to the contrary but subject to compliance with this Section 6.3, nothing contained in this Agreement shall prevent Solstice or the Board of Directors of Solstice from, at any time prior to, but not after, the receipt of the Required Solstice Stockholder Approval, in response to the receipt of a written Solstice Acquisition Proposal after the date of this Agreement that did not result from a breach of this Section 6.3, (i) terminating this Agreement pursuant to Section 8.3(b) in order to enter into a definitive written agreement with respect to such Solstice Acquisition Proposal or (ii) effecting a Solstice Change of Recommendation, if and only if, prior to taking any of the actions described in clauses (i) or (ii),

(A) the Board of Directors of Solstice determines in good faith, after consultation with its outside legal counsel and financial advisors, that such Solstice Acquisition Proposal constitutes a Solstice Superior Proposal, (B) Glass shall have received written notice (the "Solstice Superior Proposal Notice") of Solstice's or of the Board of Directors of Solstice, as applicable, intention to take such action at least four (4) Business Days prior to the taking of such action by Solstice or the Board of Directors of Solstice, as applicable, which notice shall include the information with respect to such Solstice Acquisition Proposal that is specified in Section 6.3(h) (to the extent not previously provided), (iii) during the four (4) Business Days following the receipt by Glass of the Solstice Superior Proposal Notice (the "Negotiation Period"), Solstice, if requested by Glass, shall have made its Representatives available to negotiate with Glass regarding any revisions to the terms of the Transactions proposed by Glass in response to such Solstice Acquisition Proposal, and (iv) at the end of the four (4) Business Day period described in the foregoing clause (iii) the Board of Directors of Solstice continues to believe, in good faith after consultation with its outside legal counsel and financial advisors, and after taking into account any modifications to the terms of the Transactions that are proposed in a written offer by Glass after its receipt of the Solstice Superior Proposal Notice, that such Solstice Acquisition Proposal continues to constitute a Solstice Superior Proposal and that failure to take such action would be inconsistent with the Solstice directors' fiduciary duties under Applicable Law. Any material amendment or modification to such Solstice Acquisition Proposal prior to a Solstice Change of Recommendation shall require a new Solstice Superior Proposal Notice (it being understood, however, that the Negotiation Period for such Solstice Acquisition Proposal shall not be extended more than once pursu

(f) Notwithstanding anything in this Agreement to the contrary, nothing contained in this Agreement shall prevent the Board of Directors of Solstice from, at any time prior to, but not after, the receipt of the Required Solstice Stockholder Approval, effecting a Solstice Change of Recommendation (other than in response to the receipt or making of a Solstice Acquisition Proposal, which shall be subject to Section 6.3(e)) if the Board of Directors of Solstice determines that an Intervening Event has occurred and is continuing and if, prior to taking such action, the Board of Directors of Solstice determines in good faith, after consultation with its outside legal counsel, that failure to take such action would be inconsistent with the Solstice directors' fiduciary duties under Applicable Law; provided, however that (i) Glass shall have received written notice (a "Solstice Board Recommendation Notice") of the Board of

77

Directors of Solstice's intention to take any such action at least four (4) Business Days prior to the taking of such action by the Board of Directors of Solstice specifying in reasonable detail the circumstances related to such determination, (ii) during such notice period, Solstice, if requested by Glass, shall have made available its Representatives to negotiate with Glass to revise the terms of the Transactions and (iii) the Board of Directors of Solstice continues to believe, after taking into account any modifications to the terms of the Transactions that are proposed by Glass in a written offer after its receipt of such Solstice Board Recommendation Notice, that failing to take such action would be inconsistent with the Solstice directors' fiduciary duties under Applicable Law.

- G) Solstice agrees that it and its Subsidiaries shall, and that they shall cause their respective Representatives to, (i) immediately cease and cause to be terminated any existing activities, discussions or negotiations with any person (other than the parties) conducted prior to the date of this Agreement with respect to any Solstice Acquisition Proposal and (ii) request each third party that has heretofore executed a confidentiality agreement that relates to a Solstice Acquisition Proposal (other than Glass) to return or destroy all confidential information regarding Solstice or its Subsidiaries heretofore furnished to such third party by Solstice or on its behalf. Solstice agrees that it and its Subsidiaries will take the necessary steps to promptly inform its and its Subsidiaries' Representatives of the obligations undertaken in this Section 6.3.
- (h) During the Pre-Closing Period, Solstice shall promptly orally notify Glass of any request for information or any inquiries, proposals or offers relating to a Solstice Acquisition Proposal indicating, in connection with such notice, the name of such person making such request, inquiry, proposal or offer and the material terms and conditions of any proposals or offers and Solstice shall provide to Glass written notice of any such inquiry, proposal or offer within 24 hours of such event and copies of any written or electronic correspondence to or from any person making a Solstice Acquisition Proposal (or its Representatives). Solstice shall keep Glass informed orally, as soon as is reasonably practicable, of the status of any Solstice Acquisition Proposal, including with respect to the status and material terms of any such proposal or offer and whether any such proposal or offer has been withdrawn or rejected and Solstice shall provide to Glass written notice of any such withdrawal or rejection and copies of any written proposals or requests for information within 24 hours. Solstice also agrees to provide any information to Glass (not previously provided or made available to Glass) that it is providing to another person pursuant to this Section 6.3 at substantially the same time it provides such information to such other person. All information provided to Glass under this Section 6.3 shall be kept confidential by Glass in accordance with the terms of the Confidentiality Agreement.

# 6.4 No Solicitation By Glass Parties.

(a) Each Glass Party agrees that, during the Pre-Closing Period, neither it nor any of its Subsidiaries shall, and that it shall cause its and its Subsidiaries' respective Representatives not to, directly or indirectly, (i) maintain, initiate or solicit or knowingly encourage or facilitate (including by way of any inquiries with respect to, or the making of, any proposal which constitutes, or would reasonably be expected to lead to) a Glass Acquisition Proposal, (ii) engage in any negotiations concerning, or provide any confidential information or data regarding Glass or its Subsidiaries to any person relating to, a Glass Acquisition Proposal, (iii) approve or recommend, or propose publicly to approve or recommend, any Glass

78

Acquisition Proposal, (iv) approve or recommend, or propose publicly to approve or recommend, or execute or enter into, any letter of intent, agreement in principle, merger agreement, acquisition agreement, option agreement or other similar agreement relating to any Glass Acquisition Proposal, (v) take any action to make the provisions of any takeover statute inapplicable to any transactions contemplated by a Glass Acquisition Proposal, or (vi) propose publicly or agree to do any of the foregoing.

(b) Glass agrees that it and its Subsidiaries shall, and that they shall cause their respective Representatives to, (i) immediately cease and cause to be terminated any existing activities, discussions or negotiations with any person (other than the parties) conducted prior to the date of this Agreement with respect to any Glass Acquisition Proposal and (ii) request each third party that has heretofore executed a confidentiality agreement that relates to a Glass Acquisition Proposal (other than Solstice) to return or destroy all confidential information regarding Glass or its Subsidiaries heretofore furnished to such third party by Glass or on its behalf. Glass agrees that it and its Subsidiaries will take the necessary steps to promptly inform its and its Subsidiaries' Representatives of the obligations undertaken in this Section 6.4.

(c) During the Pre-Closing Period, Glass shall promptly orally notify Solstice of any request for information or any inquiries, proposals or offers relating to a Glass Acquisition Proposal indicating, in connection with such notice, the name of such person making such request, inquiry, proposal or offer and the material terms and conditions of any proposals or offers and Glass shall provide to Solstice written notice of any such inquiry, proposal or offer within 24 hours of such event and copies of any written or electronic correspondence to or from any person making a Glass Acquisition Proposal (or its Representatives). Glass shall keep Solstice informed orally, as soon as is reasonably practicable, of the status of any Glass Acquisition Proposal, including with respect to the status and material terms of any such proposal or offer and whether any such proposal or offer has been withdrawn or rejected and Glass shall provide to Solstice written notice of any such withdrawal or rejection and copies of any written proposals or requests for information within 24 hours.

### 6.5 <u>Meeting of Solstice Stockholders.</u>

(a) Solstice will take all action necessary in accordance with Applicable Law and its certificate of incorporation and by-laws to convene a meeting of its stockholders as promptly as practicable after the Registration Statement has been declared effective to consider and vote upon the adoption of this Agreement and the approval of the Solstice Merger and the Solstice Merger-Related Proposals. Such meeting shall not be postponed or adjourned by Solstice for more than five (5) Business Days in the aggregate from the originally scheduled date without the prior written consent of Glass, except Solstice may postpone or adjourn such meeting, without such consent (i) to solicit additional proxies for the purpose of obtaining the Solstice Stockholder Approval, (ii) in the absence of a quorum and/or (iii) to allow reasonable additional time for the filing and/or mailing of any supplemental or amended disclosure that Solstice has determined after consultation with outside legal counsel is necessary under Applicable Law and for such supplemental or amended disclosure to be disseminated and reviewed by Solstice's stockholders prior to such meeting. Solstice shall, upon the reasonable request of Glass, advise Glass at least on a daily basis on each of the last seven (7) Business

79

Days prior to the date of such meeting as to the aggregate tally of proxies received by Solstice with respect to the Required Solstice Stockholder Approval.

(b) Solstice, acting through its Board of Directors, shall recommend to its stockholders that they vote in favor of the adoption of this Agreement and the approval of the Solstice Merger and the Solstice Merger-Related Proposals; provided that the Board of Directors of Solstice may at any time prior to the receipt of the Required Solstice Stockholder Approval, to the extent permitted by Section 6.3 and unless this Agreement is terminated in accordance with Article 8 and subject to compliance with Article 8, make a Solstice Change of Recommendation. Subject to Section 6.3, Solstice shall use its commercially reasonable efforts to (A) solicit from its stockholders proxies in favor of the adoption of this Agreement and the approval of the Solstice Merger and the Solstice Merger-Related Proposals and (B) take all other action necessary or advisable to secure the Required Solstice Stockholder Approval.

Notwithstanding anything in this Agreement to the contrary, unless this Agreement is terminated in accordance with Article 8 and subject to compliance with Article 8, Solstice, regardless of whether the Board of Directors of Solstice has approved, endorsed or recommended a Solstice Acquisition Proposal or has made a Solstice Change of Recommendation or any Solstice Acquisition Proposal shall have been proposed or announced or otherwise submitted to Solstice or its Subsidiaries or their respective Representatives, Solstice will submit this Agreement for approval by the stockholders of Solstice at such meeting.

# 6.6 <u>Registration Statement.</u>

- Solstice and Glass shall cooperate to prepare, and Solstice shall use its reasonable best efforts to cause the Company to file with the SEC as soon as practicable after the date hereof, a Registration Statement on Form S-4 under the Securities Act (the "Registration Statement") with respect to the shares of Company Common Stock issuable pursuant to the Mergers. A portion of the Registration Statement shall also serve as the proxy statement with respect to the meetings of the stockholders of each of Solstice and the Glass Parties in connection with this Agreement and the Transactions (the "Joint Proxy Statement/Prospectus"). Solstice and the Glass Parties and their respective counsel shall be given an adequate opportunity to review and comment on the Joint Proxy Statement/Prospectus and the Registration Statement, and Solstice and the Glass Parties shall give due consideration to all reasonable comments of the other parties or their counsel thereon. Each of Solstice and the Glass Parties shall cause the Joint Proxy Statement/Prospectus and the Registration Statement to comply as to form in all material respects with the applicable provisions of the Securities Act, the Exchange Act and the rules and regulations thereunder. Each of Solstice and the Glass Parties shall, and Solstice shall cause the Company to, use its reasonable best efforts to have the Registration Statement declared effective by the SEC as promptly as practicable and to keep the Registration Statement effective as long as is necessary to consummate the Transactions. Solstice shall cause the Company to use its reasonable best efforts to obtain, prior to the effective date of the Registration Statement, all necessary state or foreign securities law or "blue sky" permits or approvals required to carry out the Transactions, provided that the costs and expenses of obtaining such permits or approvals shall be shared equally by Solstice and the Glass Parties.
- (b) Solstice shall cause the Joint Proxy Statement/Prospectus to be mailed to its stockholders and the stockholders of the Glass Parties as promptly as practicable after the

80

Registration Statement is declared effective by the SEC and in any event within five (5) days thereafter.

- (c) Each of the Solstice Parties and the Glass Parties agrees that the information provided by it for inclusion in the Joint Proxy Statement/Prospectus and each amendment or supplement thereto, at the time of mailing thereof and at the time of the meetings of stockholders of Solstice and the Glass Parties, or, in the case of information provided by it for inclusion in the Registration Statement or any amendment or supplement thereto, at the time it is filed or becomes effective, will not include an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. In addition, the information supplied or to be supplied by or on behalf of any party hereto for inclusion in any filing pursuant to Rule 165 and Rule 425 under the Securities Act or Rule 14a-12 under the Exchange Act (each, a "Regulation M-A Filing") shall not, at the time any such Regulation M-A Filing is filed with the SEC, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading.
- (d) Each of the Solstice Parties and the Glass Parties agrees that if it shall become aware, prior to the time of Solstice's or any Glass Party's stockholders' meeting, of any information that would cause any of the statements in the Joint Proxy Statement/Prospectus or in the Registration

Statement to be false or misleading with respect to any material fact, or to omit to state any material fact necessary in order to make the statements made therein not false or misleading, to promptly inform the other parties thereof and to take the necessary steps to correct the Joint Proxy Statement/Prospectus.

- (e) Solstice and Glass will advise each other, promptly after Solstice or Glass, as the case may be, receives notice thereof, of the time when the Registration Statement has become effective or any supplement or amendment has been filed, of the issuance of any stop order or the suspension of the qualification of the Company Common Stock for offering or sale in any jurisdiction, of the initiation or threat of any proceeding for any such purpose, or of any request by the SEC for the amendment or supplement of the Joint Proxy Statement/Prospectus or the Registration Statement or comments thereon and responses thereto or requests by the SEC for additional information.
- (f) Solstice and Glass will promptly provide each other copies of all written correspondence received from the SEC and summaries of all oral comments received from the SEC in connection with the Transactions. Solstice and Glass will promptly provide each other with drafts of all correspondence intended to be sent to the SEC in connection with the Transactions and allow each such party the reasonable opportunity to comment thereon prior to delivery to the SEC.

## 6.7 Glass Stockholder Meetings.

(a) Each Glass Party will take all action necessary in accordance with Applicable Law and its certificate of incorporation and bylaws (or other governing documents) to convene a meeting of its stockholders as promptly as practicable after the Registration

81

Statement has been declared effective to consider and vote upon the adoption of this Agreement, the Cayman Plan of Merger and the approval of the applicable Glass Merger. Such meetings shall not be postponed or adjourned by the Glass Parties for more than five (5) Business Days in the aggregate from the originally scheduled date without the prior written consent of Solstice, except any Glass Party may postpone or adjourn its meeting, without such consent (i) to solicit additional proxies for the purpose of obtaining the required approval for the applicable Glass Merger, (ii) in the absence of a quorum and/or (iii) to allow reasonable additional time for the filing and/or mailing of any supplemental or amended disclosure that the applicable Glass Party has determined after consultation with outside legal counsel is necessary under Applicable Law and for such supplemental or amended disclosure to be disseminated and reviewed by such Glass Party's stockholders prior to such meeting.

(b) Each Glass Party, acting through its Board of Directors, shall recommend to its stockholders that they vote in favor of the adoption of this Agreement and the approval of the applicable Glass Merger, which recommendation shall be included in the Joint Proxy Statement/Prospectus.

## 6.8 <u>Filings; Reasonable Best Efforts.</u>

- (a) Subject to the terms hereof, including Section 6.8(b) and Section 6.8(c), each of the Glass Parties and the Solstice Parties shall use its reasonable best efforts to (i) take, or cause to be taken, all actions, and do, or cause to be done, and to assist and cooperate with the other parties in doing, all things necessary, proper or advisable to consummate and make effective the Transactions as promptly as practicable, (ii) as promptly as practicable, obtain from any Governmental Authority or any other third party any consents, licenses, permits, waivers, approvals, authorizations, or orders required to be obtained or made by such party or any of its Subsidiaries in connection with the authorization, execution and delivery of this Agreement and the consummation of the Transactions, (iii) as promptly as practicable (and, in the case of the Pre-Merger Notification and Report Form filed pursuant to the HSR Act, in any event within ten (10) Business Days) after the date of this Agreement, make all necessary filings, and thereafter make any other required submissions, with respect to this Agreement and the Transactions required under (A) the Securities Act and the Exchange Act, and any other applicable federal or state securities laws, (B) the HSR Act and any related governmental request thereunder, and (C) any other Applicable Law and (iv) execute or deliver any additional instruments necessary to consummate the Transactions and to fully carry out the purposes of this Agreement. The parties hereto shall cooperate with each other in connection with the making of all such filings and shall use their respective reasonable best efforts to furnish to each other all information required for any application or other filing to be made pursuant to the rules and regulations of any Applicable Law in connection with the Transactions. For the avoidance of doubt, the parties agree that nothing contained in this Section 6.8(a) shall modify or affect their respective rights and responsibilities under Section 6.8(b) or Section 6.
- (b) Subject to the terms hereof, each party shall, and each shall cause each of its Subsidiaries, to cooperate with the other parties and to use its reasonable best efforts to obtain any government clearances or approvals, or expirations or terminations of waiting periods, required for the consummation of the Transactions under the HSR Act and any other federal, state or foreign law or, regulation or decree designed to prohibit, restrict or regulate actions for

82

the purpose or effect of monopolization or restraint of trade (collectively "Antitrust Laws"), and to respond to any government requests for information under any Antitrust Law. The parties hereto will consult and cooperate with one another, and consider in good faith the views of one another, in connection with any analyses, appearances, presentations, memoranda, briefs, arguments, opinions and proposals made or submitted by or on behalf of any party hereto in connection with proceedings under or relating to any Antitrust Law. Except as prohibited by Applicable Law, each of the parties shall keep the other parties and their respective counsel informed of any substantive communication received by such party from, or given by such party to any Governmental Authority, in each case regarding any of the Transactions; and permit the other parties and their respective counsel to review any substantive communication given by it to, and consult with each other in advance of any meeting or conference with any such Governmental Authority.

(c) Each party agrees to cooperate and use its reasonable best efforts to vigorously contest and resist any action or proceeding, including administrative or judicial action or proceeding, and to have vacated, lifted, reversed or overturned any decree, judgment, injunction or other order (whether temporary, preliminary or permanent) that is in effect and that restricts, prevents or prohibits consummation of the Mergers and the Transactions contemplated by this Agreement, including reasonably pursuing administrative and judicial appeal; provided, however, that notwithstanding the foregoing or anything else to the contrary set forth in this Agreement, no party shall be required by this Agreement to agree to any material divestitures, licenses, hold separate arrangements or similar matters, including material covenants affecting business operating practices, except to the extent any such divestitures, licenses, arrangements, matters or covenants would not reasonably be expected to be material to the Glass Parties and Solstice, taken as a whole.

- (d) No party shall make any settlement offers or negotiate any consent decree, consent order or undertaking with any Governmental Authority relating to the Transactions contemplated by this Agreement without the prior consent of the other party, which shall not be unreasonably withheld, delayed or conditioned. No party may accept any settlement, consent decree, consent order or enter into any undertaking proposed by any Governmental Authority, without the prior written consent of the other party, which shall not be unreasonably withheld, delayed or conditioned.
- 6.9 Takeover Law. None of the parties shall take any action that would cause the Transactions to be subject to the requirements of any Takeover Law. If any Takeover Law shall become applicable to the Transactions, each of the Company, Solstice and the Glass Parties, the members of Boards of Directors and the Glass Members shall grant such approvals and take such actions as are necessary so that the Transactions may be consummated as promptly as practicable on the terms contemplated hereby, and otherwise act to eliminate or minimize the effects of such Takeover Law on the Transactions.
- 6.10 <u>Inspection</u>. From the date hereof to the Effective Time, each of Solstice and Glass shall allow all Representatives of Solstice or Glass, as the case may be, access at all reasonable times upon reasonable notice to the records and files, correspondence, audits and properties, as well as to all information relating to commitments, contracts, titles and financial position, or otherwise pertaining to the business and affairs of Solstice and Glass and their

83

respective Subsidiaries, including inspection of such properties; <u>provided</u> that no investigation pursuant to this <u>Section 6.10</u> shall affect any representation or warranty given by any party hereunder; <u>provided further</u> that notwithstanding the provision of information or investigation by any party, no party shall be deemed to make any representation or warranty except as expressly set forth in this Agreement. Notwithstanding the foregoing, no party shall be required to provide any information which it reasonably believes it may not provide to the other party by reason of Applicable Law, rules or regulations, which such party reasonably believes constitutes information protected by attorney/client privilege, or which it is required to keep confidential by reason of contracts or agreements with third parties. The parties hereto will make reasonable and appropriate substitute disclosure arrangements under circumstances in which the restrictions of the preceding sentence apply. Each of Solstice and Glass agrees that it will not, and will cause its respective Representatives not to, use any information obtained pursuant to this <u>Section 6.10</u> for any purpose unrelated to the consummation of the Transactions. All information provided by a party to the other party hereunder shall be subject to the terms of the Confidentiality Agreement. If any person is injured as a result of any inspection or testing of any property of Solstice or any of its Subsidiaries, Glass shall defend, indemnify and hold harmless Solstice and its Subsidiaries from any Damages, incurred in connection with any such injury, other than any injury attributable to any gross negligence or willful misconduct of Glass or any of its Subsidiaries.

- 6.11 <u>Publicity.</u> Except in connection with and following a Solstice Change in Recommendation or Solstice Superior Proposal as permitted in accordance with <u>Section 6.3</u> and in connection with any disclosure of any information concerning this Agreement or the Transactions in connection with any dispute between the parties regarding this Agreement or the Transactions, the parties will consult with each other and will mutually agree upon any press releases or public announcements pertaining to this Agreement or the Transactions and shall not issue any such press releases or make any such public announcements prior to such consultation and agreement, except as may be required by Applicable Law or by obligations pursuant to any listing agreement with any national securities exchange.
- 6.12 <u>Listing Application</u>. Solstice and Glass shall cooperate with each other and use their reasonable best efforts to cause the Company to promptly prepare and submit to NASDAQ a listing application covering the Company Common Stock to be issued pursuant to the Transactions, and shall use their reasonable best efforts to obtain, prior to the Effective Time, approval for such listing, subject to official notice of issuance.
- 6.13 <u>Expenses</u>. If the Transactions are not consummated, but subject to <u>Section 8.5</u>, all Costs incurred in connection with this Agreement and the Transactions shall be paid by the party incurring such expenses; <u>provided</u>, <u>however</u>, each of Solstice and Glass shall share equally the costs, expenses and filing fees for the Solstice Regulatory Filings and the Glass Regulatory Filings made in connection with the Transactions and for printing and distributing the Registration Statement and the Joint Proxy Statement/Prospectus.

84

### 6.14 <u>Directors' and Officers' Indemnification and Insurance</u>.

The Company shall (A) indemnify, hold harmless and advance expenses to all past and present directors and officers of Solstice and its Subsidiaries (in all of their capacities) (x) to the same extent such persons are indemnified or have the right to advancement of expenses as of the date of this Agreement by Solstice pursuant to Solstice's certificate of incorporation, by-laws and indemnification agreements, if any, in existence on the date hereof with any directors and officers of Solstice and its Subsidiaries and (y) without limitation to clause (x), to the fullest extent permitted by Applicable Law, in each case for acts or omissions in their capacities as directors or officers occurring at or prior to the Effective Time (including for acts or omissions occurring in connection with the approval of this Agreement and the consummation of the Transactions), (B) include and cause to be maintained in effect in Solstice Surviving Corporation's (or any successor's) certificate of incorporation and by-laws after the Effective Time provisions regarding elimination of liability of directors, indemnification of officers and directors, and advancement of expenses which are, in the aggregate, no less advantageous to the intended beneficiaries than the corresponding provisions contained in Solstice's certificate of incorporation and by-laws, (C) periodically advance to any such indemnitee its legal and other expenses (including the cost of any investigation and preparation incurred in connection therewith), subject to clause (c) of this Section 6.14, and subject to the providing by such indemnitee of an undertaking to reimburse all amounts so advanced in the event of a final non-appealable determination by a court of competent jurisdiction that such indemnitee is not entitled thereto and (D) cause to be maintained for a period of six years after the Effective Time the current policies of directors' and officers' liability insurance and fiduciary liability insurance ("D&O Insurance") maintained by Solstice (provided that Solstice Surviving Corporation (or any successor) may substitute therefor one or more policies of at least the same coverage and amounts containing terms and conditions which are, in the aggregate, no less advantageous to the insured) with respect to claims arising from facts or events that occurred on or before the Effective Time; provided, however, that in no event shall the Company or Solstice Surviving Corporation be required pursuant to this Section 6.14(a) to expend in any one year an amount in excess of 300% of the last annual premium paid by Solstice for such insurance prior to the date hereof, the amount of such annual premium being set forth in Section 6.14(a) of the Solstice Disclosure Letter; provided, further, that if the annual premiums of such insurance coverage exceed such amount, Solstice Surviving Corporation shall be obligated to obtain a policy with the greatest coverage available for a cost not exceeding such amount. Solstice may extend coverage, effective as of the Effective Time, under Solstice's D&O Insurance by obtaining a six-year

"tail" policy prior to the Closing on terms and conditions no less advantageous to the covered persons than Solstice's existing D&O Insurance, and such "tail" policy shall satisfy the provisions of this <u>Section 6.14(a)</u>. If such "tail" policy has been obtained by Solstice prior to the Closing, Solstice Surviving Corporation shall maintain such policies in full force and effect and continue to honor Solstice's obligations thereunder. From and after the Effective Time, the Company will guarantee the obligations of Solstice Surviving Corporation under this <u>Section 6.14(a)</u>.

(b) The Company shall (A) indemnify, hold harmless and advance expenses to all past and present directors and officers of each Glass Party and its Subsidiaries (in all of their capacities) (x) to the same extent such persons are indemnified or have the right to

85

advancement of expenses as of the date of this Agreement by any Glass Party pursuant to the Glass Organizational Documents and indemnification agreements, if any, in existence on the date hereof with any directors and officers of any Glass Party and its Subsidiaries and (y) without limitation to clause (x), to the fullest extent permitted by Applicable Law, in each case for acts or omissions in their capacities as directors or officers occurring at or prior to the Effective Time (including for acts or omissions occurring in connection with the approval of this Agreement and the consummation of the Transactions), (B) include and cause to be maintained in effect in the applicable Surviving Corporation's (or any successor's) memorandum and articles of association or other organizational documents after the Effective Time provisions regarding elimination of liability of directors, indemnification of officers and directors, and advancement of expenses which are, in the aggregate, no less advantageous to the intended beneficiaries than the corresponding provisions contained in the Glass Organizational Documents, (C) periodically advance to any such indemnitee its legal and other expenses (including the cost of any investigation and preparation incurred in connection therewith), subject to the following clause (c) of this Section 6.14, and subject to the providing by such indemnitee of an undertaking to reimburse all amounts so advanced in the event of a final non-appealable determination by a court of competent jurisdiction that such indemnitee is not entitled thereto and (D) cause to be maintained for a period of six years after the Effective Time the current D&O Insurance maintained by Glass (provided that any Surviving Corporation (or any successor) may substitute therefor one or more policies of at least the same coverage and amounts containing terms and conditions which are, in the aggregate, no less advantageous to the insured) with respect to claims arising from facts or events that occurred on or before the Effective Time; provided, however, that in no event shall the Company or any Surviving Corporation be required pursuant to this Section 6.14(b) to expend in any one year an amount in excess of 300% of the last annual premium paid by Glass for such insurance prior to the date hereof, the amount of such annual premium being set forth in <u>Section 6.14(b)</u> of the Glass Disclosure Letter; <u>provided</u>, <u>further</u>, that if the annual premiums of such insurance coverage exceed such amount, the applicable Surviving Corporation shall be obligated to obtain a policy with the greatest coverage available for a cost not exceeding such amount. Glass may extend coverage, effective as of the Effective Time, under Glass's D&O Insurance by obtaining a six-year "tail" policy prior to the Closing on terms and conditions no less advantageous to the covered persons than Glass's existing D&O Insurance, and such "tail" policy shall satisfy the provisions of this Section 6.14(b). If such "tail" policy has been obtained by Glass prior to the Closing, the applicable Surviving Corporation shall maintain such policies in full force and effect and continue to honor Glass's obligations thereunder. From and after the Effective Time, the Company will guarantee the obligations of the Surviving Entities under this Section 6.14(b).

(c) No Surviving Corporation shall be obligated to pay the fees and expenses of more than one counsel for all indemnitees in any single claim except to the extent that, in the opinion of independent legal counsel selected by the indemnitee, which counsel shall be reasonably acceptable to the Company, representation of two or more of such indemnitees would present a conflict of interest under applicable standards of conduct in the legal profession. No Surviving Corporation shall be liable for any settlement effected without its written consent, which consent shall not unreasonably be withheld.

86

(d) The obligations of the Company and the Surviving Entities under this Section 6.14 shall not be terminated or modified in such a manner as to adversely affect any indemnitee to whom this Section 6.14 applies without the consent of such affected indemnitee (it being expressly agreed that the indemnitees to whom this Section 6.14 applies shall be third party beneficiaries of this Section 6.14). In the event that the Company, any Surviving Corporation or any of their respective successors or assigns (A) consolidates with or merges into any other person and shall not be the continuing or surviving corporation or entity in such consolidation or merger or (B) transfers all or substantially all its properties and assets to any person, then and in each case, proper provision shall be made so that the successors and assigns of such person or entity assume the indemnification obligations set forth in this Section 6.14.

### 6.15 <u>Taxes</u>.

- (a) So long as the opinions of counsel referred to in <u>Sections 7.2(d)</u> and <u>7.3(d)</u> are given, each party hereto agrees to report for U.S. federal income Tax purposes the exchange of Glass Shares for Company Common Stock pursuant to the Cayman Merger, taken together with the exchange of Solstice Common Stock for Company Common Stock pursuant to the Solstice Merger, as a transaction qualifying as a transfer of property to the Company described under Section 351 of the Code, except as otherwise required as a result of a final determination of a taxing authority; <u>provided</u>, <u>however</u>, the parties may treat the Solstice Merger as a reorganization within the meaning of Section 368(a) of the Code.
- (b) None of Solstice, the Glass Parties, the Company, or any of their respective Subsidiaries shall (i) take any action, or fail to take any reasonable action, as a result of which the exchange of Solstice Common Stock for Company Common Stock pursuant to the Solstice Merger would reasonably be expected to fail to qualify either as a reorganization under Section 368(a) of the Code or as a transaction that, taken together with the other Mergers, qualifies as a transfer of property to the Company described in Section 351 of the Code, or (ii) take any action, or fail to take any reasonable action, as a result of which the exchange of Glass Shares for Company Common Stock pursuant to the Cayman Merger, taken together with the other Mergers, would reasonably be expected to fail to qualify as a transaction qualifying as a transfer of property to the Company described in Section 351 of the Code. Each of Solstice and Glass shall use its reasonable best efforts to obtain the opinion of counsel referred to in Sections 7.2(d) and 7.3(d).
- (c) The parties shall cooperate to prepare and file any Internal Revenue Service Forms 8937 as required pursuant to Section 6045B of the Code in connection with the Transactions. The parties agree to furnish to each other, upon request, such information and assistance relating to Taxes as is reasonably necessary for the filing of all Tax Returns, the making of any election relating to Taxes, the preparation for any audit by any Governmental Authority and the prosecution or defense of any claim, suit or proceeding relating to any Tax.
- (d) Solstice shall deliver to Wilmer Cutler Pickering Hale and Dorr LLP ("<u>WilmerHale</u>") and Latham and Watkins LLP ("<u>Latham</u>") an officer's certificate, dated as of the Closing Date and signed by an officer of Solstice substantially in the form of <u>Exhibit I</u> (allowing for such amendments as

dated as of the Closing Date and signed by an officer of such Glass Party substantially in the forms of <u>Exhibit J</u>, <u>Exhibit K</u> and <u>Exhibit L</u>, respectively (allowing for such amendments as counsel to Glass or counsel to Solstice deems necessary) to enable WilmerHale to render the opinion described in <u>Section 7.2(d)</u> and Latham to render the opinion described in <u>Section 7.3(d)</u>.

### 6.16 <u>Employee Benefits</u>.

- (a) From and after the Effective Time, the Company shall honor, and shall cause each Surviving Corporation to honor, in accordance with their terms, all employment and severance agreements that constitute Solstice Plans or Glass Plans and all benefits and obligations under the other Solstice Plans and Glass Plans, as in effect on the date hereof (or as amended thereafter as required by the terms thereof as in effect as of the date of the Agreement, as contemplated hereby or with the prior written consent of Solstice and Glass). Subject to the previous sentence, no provision of this Agreement shall be construed as a limitation on the right of the Company or any Surviving Corporation to amend or terminate any of the foregoing agreements, plans and arrangements to the extent permitted by the terms thereof and Applicable Law, and no provision of this Agreement shall be construed to create a right in any employee, director or beneficiary that such employee, director or beneficiary would not otherwise have under any of the foregoing agreements, plans and arrangements. Section 6.16 shall not apply to persons employed outside the United States or covered by collective bargaining or other labor agreements (such non-U.S. employees and collectively bargained employees, the "Excluded Employees"). The foregoing notwithstanding, the Company agrees to comply with the requirements of Applicable Law with respect to the treatment of Excluded Employees and to honor, or to cause each Surviving Corporation or any of their respective Subsidiaries following the Effective Time to honor, in accordance with their terms all Solstice Plans, Solstice Foreign Plans, Glass Plans, Glass Foreign Plans and all Collective Bargaining Agreements or other labor agreements to which Solstice, Glass, or any of their Subsidiaries is a party or otherwise bound and which apply to the Excluded Employees, provided that such plans or agreements may be amended, terminated or suspended in accordance with their terms and Applicable Law.
- (b) The Company shall, and shall cause each Surviving Corporation to, give individuals who were employed by the Company, any Surviving Corporation or their respective Subsidiaries as of the Effective Time (the "Affected Employees") full credit for purposes of eligibility, vesting and determination of the level of benefits (other than benefit accrual under a defined benefit pension plan) under any employee benefit plans or arrangements maintained by the Company, any Surviving Corporation or any of their respective Subsidiaries as of the Effective Time for such Affected Employees' service with Glass or Solstice or any Subsidiary of Glass or Solstice to the same extent recognized by Glass or Solstice immediately prior to the Effective Time; provided, however, that no Affected Employee shall receive a duplication of benefits with respect to the same period of service.
- (c) The Company shall use commercially reasonable efforts to, and shall cause each Surviving Corporation to use commercially reasonable efforts to, (i) waive all limitations as to preexisting conditions, exclusions and waiting periods with respect to participation and coverage requirements applicable to the Affected Employees under any medical or dental benefit plans that the Affected Employees may be eligible to participate in after the Effective Time, other than limitations or waiting periods that are already in effect with

88

respect to such employees and that have not been satisfied as of the Effective Time under any medical or dental benefit plan maintained for the Affected Employees immediately prior to the Effective Time, and (ii) to the extent permitted by the applicable medical or dental plan, provide each Affected Employee with credit for the year in which the Effective Time occurs for any co-payments and deductibles paid prior to the Effective Time in satisfying any applicable deductible or out-of-pocket requirements under any medical and dental plans that the Affected Employees are eligible to participate in after the Effective Time

- (d) Solstice agrees to cause the Company to adopt or assume, prior to or as of the Effective Time, an equity compensation plan (the "New Equity Plan") providing for the granting of stock options and other equity-based awards to the employees of the Company, each Surviving Corporation and their respective Subsidiaries. The New Equity Plan shall authorize a share reserve for the grant of awards covering a number of shares of Company Common Stock as agreed by Solstice and Glass and Solstice and Glass agree to cause the Company to allocate a portion of such share reserve consisting of a number of shares of Company Common Stock equal to the sum of (i) the aggregate number of shares of Company Common Stock that would be issued at Closing pursuant to the Assumed Company Options if each such option was "net exercised" on the Closing Date using the VWAP on the day immediately prior to the Closing Date (and for the avoidance of doubt, without any shares of Company Common Stock being withheld to pay income, payroll or other Taxes) and (ii) the aggregate number of shares of Company Common Stock covered by equity awards under the Solstice Stock Plans that are assumed by the Company pursuant to Sections 3.3(b) and 3.3(c) for grants to employees of the Glass Surviving Corporation and its Subsidiaries. The New Equity Plan will be in a form agreed to by Solstice and Glass and no awards under the New Equity Plan may be made on or prior to the Effective Time without the consent of Glass. The Company will adopt, and Solstice agrees that, as the stockholder of the Company, it will approve the adoption of, the New Equity Plan prior to the Effective Time.
- (e) Prior to the Effective Time, each of Solstice and Glass shall cause the Solstice Plans and Solstice Foreign Plans and the Glass Plans and Glass Foreign Plans to be amended as necessary to the extent permitted by Applicable Law so that (i) no Affected Employee who, immediately prior to the Effective Time, was an employee of Solstice or one of its Subsidiaries shall be eligible to participate in a Glass Plan or Glass Foreign Plan as of the Effective Time and (ii) no Affected Employee who, immediately prior to the Effective Time, was an employee of Glass or one of its Subsidiaries shall be eligible to participate in a Solstice Plan or Solstice Foreign Plan as of the Effective Time.
- (f) Glass and Solstice and their respective Subsidiaries have taken or caused to be taken all action necessary such that neither the execution of this Agreement nor the consummation of any of the Transactions shall constitute an event that requires the funding of any rabbi or similar trust with respect to any Solstice Plan, Solstice Foreign Plan, Glass Plan or Glass Foreign Plan.
- (g) This <u>Section 6.16</u> is included for the sole benefit of the parties hereto and their respective transferees and permitted assigns and does not and shall not, except between the parties hereto, create any right in or obligation to any Person, including any Affected Employee, any person previously employed by the parties, any participant in any Solstice Plan, Glass Plan,

or successor or replacement plan or arrangement or any beneficiary or trustee thereof. Nothing contained in this Agreement (express or implied) (i) is intended to require the Company or its Affiliates to establish or maintain or cause to be established or maintained any specific employee benefit plan or other compensatory arrangement for any length of time, (ii) is intended to create such a plan or arrangement or amend any of the foregoing, (iii) is intended to confer upon any individual any right to employment or other service relationship or continued employment or service relationship for any period of time, or any right to a particular term or condition of employment or service, or (iv) shall be construed to indicate existence of employment relations between Company and any of its service providers (including contractors and consultants). No Affected Employee or other current or former employee of any of the parties or their Affiliates, including any beneficiary or dependent thereof, or any other Person not a party to this Agreement, shall be entitled to assert any claim (directly or as a third party beneficiary) against Solstice, Glass, the Company, the Surviving Entities or any of their respective Affiliates under this Section 6.16.

- 6.17 <u>Notification</u>. Solstice, Glass and each of the Glass Members shall give prompt notice to the other of (i) any representation or warranty made by it or him or contained in this Agreement becoming untrue or inaccurate in any material respect and (ii) the failure by it or him to comply with or satisfy in any material respect any covenant, condition or agreement to be complied with or satisfied by it under this Agreement; <u>provided</u>, <u>however</u>, that no such notification shall affect the representations, warranties, covenants or agreements of the parties or the conditions to the obligations of the parties under this Agreement.
- 6.18 <u>Security Holder Litigation</u>. Each of Solstice and Glass shall provide the other party the opportunity to participate in the defense of any litigation brought by stockholders or any holder of other securities of Solstice or shareholders or any holder of other securities of Glass or any Glass Corporate Member or in the name of Solstice or Glass against Solstice or Glass, as applicable, and/or their respective directors or officers related to this Agreement or any of the Transactions; <u>provided</u>, <u>however</u>, that no party shall compromise, settle, come to an arrangement regarding or agree to compromise, settle or come to an arrangement regarding any litigation arising or related to this Agreement or any of the Transactions, or consent to the same, without the prior written consent of the other party (not to be unreasonably withheld, conditioned of delayed), except that no such consent shall be required for any such settlement, compromise or arrangement that would not result in the payment by such party of any amount in excess of \$1,000,000 over and above any retention or deductible under any applicable insurance policy.
- 6.19 <u>FIRPTA Compliance</u>. On or prior to the Closing Date, (i) each of GB and GB II shall deliver or cause to be delivered to the Company a signed original of a certification that the shares of their respective stock are not United States real property interests as defined in Section 897(c) of the Code, together with a notice to the Internal Revenue Service (which shall be filed by the Company with the Internal Revenue Service following the Closing Date), in accordance with the Treasury Regulations under Sections 897 and 1445 of the Code and (ii) Glass shall deliver to the Company a signed original of a certification pursuant to Treasury Regulation Section 1.1445-11T(d)(2). If any of GB, GB II or Glass has not provided such certification and notice, as applicable, to the Company on or before the Closing Date, the Company shall be permitted to withhold from the payments to be made pursuant to this Agreement any required withholding Tax under Section 1445 of the Code.

- 280G Vote. Not less than three (3) Business Days prior to the Closing Date, each of the Glass Parties shall submit to a stockholder vote, in a manner that satisfies the stockholder approval requirements under Section 280G(b)(5)(B) of the Code and the Treasury Regulations promulgated thereunder, the right of any "disqualified individual" (as defined in Section 280G(c) of the Code) of such Glass Party that executed any required waiver or consent to receive all or a portion of payments (or other benefits) contingent on the consummation of the Transactions (within the meaning of Section 280G(b)(2)(A) (i) of the Code) to the extent necessary so that, upon stockholder approval, no payment received by such "disqualified individual" shall be a "parachute payment" under Section 280G(b) of the Code (determined without regard to Section 280G(b)(4) of the Code). Such vote shall establish each such disqualified individual's right to the payment or other compensation that, absent waiver and stockholder approval in accordance with Section 280G of the Code, could constitute a "parachute payment" under Section 280G(b) of the Code, and each of the Glass Parties shall use commercially reasonable efforts to obtain any required waivers or consents from the disqualified individuals of such Glass Party prior to the vote. In addition, each of the Glass Parties shall provide adequate disclosure to their respective stockholders that hold voting capital stock of all material facts concerning all payments to any such disqualified individual that, but for such vote, could be deemed a "parachute payment" under Section 280G of the Code in a manner that satisfies Section 280G(b)(5)(B)(ii) of the Code and Treasury Regulations promulgated thereunder. At least five (5) Business Days prior to the vote, Solstice and its counsel shall be given the right to review and comment on all documents required to be delivered to each of the Glass Parties' respective stockholders in connection with such vote and any required disqualified individual waivers or consents (or a form thereof), and each of the Glass Parties shall reflect all reasonable comments of Solstice thereon to the extent provided within three (3) Business Days following the receipt by Solstice or its counsel of such document, waiver, or consent, as applicable. Prior to soliciting the vote, each of the Glass Parties shall provide to Solstice and its counsel copies of all waivers or consents executed by the disqualified individuals of such Glass Party in connection with the vote, and promptly following the vote, Solstice and its counsel shall be provided copies of all documents executed by each of the Glass Parties' respective stockholders in connection with the vote. For the avoidance of doubt, if, absent any action under this Section 6.20, no payment to any disqualified individual of a Glass Party would constitute a "parachute payment" under Section 280G(b) of the Code, such Glass Party will have no obligations to take any actions pursuant to this Section 6.20.
  - 6.21 <u>Allocation Schedule</u>. Not less than one (1) day prior to the Closing Date, Glass shall deliver to Solstice the Allocation Schedule.
- Glass Indebtedness. Prior to the Closing Date, Glass shall obtain and deliver to Solstice executed payoff letters (the "Payoff Letters"), to be effective upon the Closing, from each lender of Indebtedness of Glass and any of its Subsidiaries as of immediately prior to the Closing (the "Glass Closing Indebtedness"). Subject to the satisfaction or waiver of the conditions set forth in Section 7.1 and Section 7.2, at the Closing and on behalf of the applicable Glass Party, the Company shall pay, in cash by wire transfer of immediately available funds, the Glass Closing Indebtedness in the amounts set forth in the Payoff Letters delivered pursuant to this Section 6.22 and pursuant to wire instructions provided by Glass to Solstice prior to the Closing Date.

6.23 <u>Termination of 401(k) Plan</u>. Prior to the Closing Date, Glass and Solstice will review the advisability of either (a) termination of one or more of the existing Glass Plans or Solstice Plans intended to comply with Section 401(k) of the Code (each a "401(k) Plan") to take effect prior to the Closing Date (a "<u>Plan Termination</u>") or (b) merger of the 401(k) Plans of the Parties as soon as practicable following the Effective Time (a "<u>Plan Merger</u>"). In the event the Parties agree to effect a Plan Termination, the Board of Directors of either Glass or Solstice, as applicable, will adopt resolutions for the foregoing, having first provided the other party with the form of resolutions and obtained its approval not less than two (2) Business Days preceding the Closing Date and the party whose 401(k) Plan is being terminated will provide other Party with evidence that such 401(k) Plan has been terminated pursuant to resolution of the applicable Board of Directors prospectively adopted and effective not later than the day immediately preceding the Closing Date. In the event Glass and Solstice agree to effect a Plan Merger, the Company shall and shall cause the Surviving Corporations to take all steps necessary or appropriate to merge the 401(k) Plans as soon as reasonably practicable after the Effective Time (or such later date agreed between the Parties).

### ARTICLE 7

## **CONDITIONS**

- 7.1 <u>Conditions to Each Party's Obligation to Effect the Transactions.</u> The respective obligations of each party to effect the Transactions shall be subject to the fulfillment or waiver by each of Solstice and Glass (subject to Applicable Law) at or prior to the Closing Date of the following conditions:
  - (a) Each of the Required Solstice Stockholder Approval and the Required Glass Stockholder Approvals shall have been obtained.
- (b) Any mandatory waiting period under the HSR Act or the Antitrust Laws described on <u>Schedule 7.1(b)</u> shall have expired or terminated, and any approvals required or deemed advisable by the parties under such Antitrust Laws shall have been obtained.
- (c) None of the parties hereto shall be subject to any decree, order or injunction of any Governmental Authority of competent jurisdiction which prohibits or makes unlawful the consummation of any of the Transactions.
- (d) The Registration Statement shall have become effective and no stop order with respect thereto shall be in effect and no proceedings for that purpose shall have been commenced or threatened by the SEC.
- (e) The shares of Company Common Stock to be issued pursuant to the Transactions shall have been authorized for listing on the NASDAQ, subject to official notice of issuance.
- (f) The Company Certificate of Incorporation shall have been filed with the Secretary of State of the State of Delaware and shall be effective in accordance with the DGCL.

92

- 7.2 <u>Conditions to Obligation of the Solstice Parties to Effect the Transactions</u>. The obligations of the Solstice Parties to effect the Transactions shall be subject to the fulfillment or waiver (subject to Applicable Law) by Solstice at or prior to the Closing Date of the following conditions:
- (a) Each Glass Party shall have performed in all material respects its covenants and agreements contained in this Agreement required to be performed on or prior to the Closing Date; and Solstice shall have received a certificate of such Glass Party, executed on its behalf by its Chief Executive Officer or Chief Financial Officer, dated the Closing Date, certifying to such effect.
- (b) Each of the OEP Glass Members shall have performed in all material respects its covenants and agreements contained in the Voting Agreement required to be performed on or prior to the Closing Date.
- (c) The representations and warranties (other than those in Sections 5.1, 5.2, 5.3(a), (b) and (c), 5.17, 5.18, 5.19, 5.20 and 5.21) of the Glass Parties contained in this Agreement shall be true and correct in all respects (disregarding any materiality and Glass Material Adverse Effect qualifiers contained therein) as of the date hereof and as of the Closing Date (except for representations and warranties made as of a specified date, which need be true and correct only as of the specified date), except for failures of such representations and warranties to be true and correct, individually or in the aggregate, that would not reasonably be expected to have a Glass Material Adverse Effect; the representations and warranties of the Glass Parties set forth in Sections 5.3(a), (b) and (c) shall be true and correct in all respects (other than *de minimis* inaccuracies) as of the date hereof and as of the Closing Date; the representations and warranties of the Glass Parties set forth in Sections 5.1, 5.2, 5.17, 5.18, 5.19, 5.20 and 5.21 shall be true and correct in all material respects as of the date hereof and as of the Closing Date; and Solstice shall have received a certificate of each Glass Party, executed on its behalf by its Chief Executive Officer or Chief Financial Officer, dated the Closing Date, certifying to such effect.
- (d) Solstice shall have received an opinion of counsel in form and substance reasonably satisfactory to Solstice, on the basis of certain facts, representations and assumptions set forth in such opinion, dated as of the Closing Date, to the effect that the exchange of Solstice Common Stock for Company Common Stock in the Solstice Merger will qualify as a reorganization under Section 368(a) of the Code or as a transaction that, taken together with the other Mergers, will qualify as a transfer of property to the Company described in Section 351 of the Code. In rendering such opinion, such counsel shall be entitled to receive and rely upon the representations of officers of Solstice and the Glass Parties contained in the certificates delivered pursuant to Section 6.15(d). The condition set forth in this Section 7.2(d) shall not be waivable after receipt of the affirmative vote of stockholders referred to in Section 7.1(a), unless further stockholder approval is obtained with appropriate disclosure.
  - (e) Since the date of this Agreement, no Glass Material Adverse Effect shall have occurred.
  - (f) The agreements identified on <u>Schedule 7.2(f)</u> shall have been terminated in its entirety.

- (h) Each Person contemplated to be a party to the Stockholders Agreement and the Registration Rights Agreement, excluding any Solstice Party but expressly including each OEP Glass Member and any Affiliate of One Equity Partners or any such OEP Glass Member that receives shares of Company Common Stock in connection with the Mergers, shall have executed and delivered such agreement to Solstice and each such agreement shall be in full force and effect as of the Effective Time.
- 7.3 <u>Conditions to Obligation of the Glass Parties to Effect the Transactions</u>. The obligation of the Glass Parties to effect the Transactions shall be subject to the fulfillment or waiver (subject to Applicable Law) by Glass at or prior to the Closing Date of the following conditions:
- (a) Each Solstice Party shall have performed in all material respects its covenants and agreements contained in this Agreement required to be performed on or prior to the Closing Date; and Glass shall have received a certificate of such Solstice Party, executed on its behalf by its Chief Executive Officer or Chief Financial Officer, dated the Closing Date, certifying to such effect.
- (b) The representations and warranties (other than those in Sections 4.1, 4.2, 4.3(a), (b) and (c), 4.17, 4.19, 4.20, 4.21 and 4.27) of the Solstice Parties contained in this Agreement shall be true and correct (disregarding any materiality and Solstice Material Adverse Effect qualifiers contained therein) as of the date hereof and as of the Closing Date (except for representations and warranties made as of a specified date, which need be true and correct only as of the specified date), except for failures of such representations and warranties to be true and correct, individually or in the aggregate, that would not reasonably be expected to have a Solstice Material Adverse Effect; the representations and warranties of the Solstice Parties set forth in Section 4.3(a), (b) and (c) shall be true and correct in all respects (other than *de minimis* inaccuracies as of the date hereof or the Closing Date); the representations and warranties of the Solstice Parties set forth in Sections 4.1, 4.2, 4.17, 4.19, 4.20, 4.21 and 4.27 shall be true and correct in all material respects as of the date hereof and as of the Closing Date; and Glass shall have received a certificate of each Solstice Party, executed on its behalf by its Chief Executive Officer or Chief Financial Officer, dated the Closing Date, certifying to such effect.
  - (c) Since the date of this Agreement, no Solstice Material Adverse Effect shall have occurred.
- (d) Glass shall have received an opinion of counsel in form and substance reasonably satisfactory to Glass, on the basis of certain facts, representations and assumptions set forth in such opinion, dated as of the Closing Date, to the effect that for U.S. federal income Tax purposes the exchange of Glass Shares for Company Common Stock pursuant to the Cayman Merger, taken together with the other Mergers, qualifies as a transfer of property to the Company described in Section 351 of the Code. In rendering such opinion, such counsel shall be entitled to receive and rely upon the representations of officers of Solstice and the Glass Parties as to such matters contained in the certificates delivered pursuant to Section 6.15(d).

94

(e) Each Solstice Party contemplated to be a party to the Stockholders Agreement, the Registration Rights Agreement and the Promissory Note shall have executed and delivered such agreement to Glass and each such agreement shall be in full force and effect as of the Effective Time.

### **ARTICLE 8**

# TERMINATION

- 8.1 <u>Termination by Mutual Consent</u>. This Agreement may be terminated at any time prior to the Effective Time (whether before or after stockholder approval) by the mutual written consent of Solstice and Glass.
- 8.2 <u>Termination by Solstice or Glass</u>. This Agreement may be terminated at any time prior to the Effective Time (whether before or after stockholder approval) by action of the Board of Directors of Solstice or Glass if:
- the Transactions shall not have been consummated by 11:59 p.m., New York City time, on February 23, 2018 (the "Outside Date"); provided that the Outside Date may be extended by either Solstice or Glass, by written notice to the other party, to May 31, 2018, in the event that all conditions set forth in Article 7 (other than the conditions set forth in Section 7.1(b) (the "Regulatory Conditions")) have been or are reasonably capable of being satisfied at the time of such extension and the Regulatory Conditions are reasonably capable of being satisfied on or prior to May 31, 2018; provided, further, that the right to terminate this Agreement pursuant to this clause (a) shall not be available to any party whose failure to perform or observe in any material respect any of its obligations under this Agreement in any manner shall have been the primary cause of, or resulted in, the failure of any Merger to occur on or before such date; or
- (b) a meeting of Solstice's stockholders contemplated by Section 6.5(a) shall have been held and the Solstice stockholders shall have voted upon, and shall have not approved this Agreement as contemplated by Section 7.1(a); provided, however, that the right to terminate this Agreement pursuant to this clause (b) shall not be available to any party whose failure to perform or observe any of its obligations under this Agreement in any manner shall have been the primary cause of, or resulted in, the failure to obtain the approval of the Solstice stockholders as contemplated by Section 7.1(a); or
- (c) there shall be any Applicable Law that makes consummation of any of the Transactions illegal or otherwise prohibited, or if any judgment, injunction, order or decree of a competent Governmental Authority enjoining any of the parties from consummating the Transactions shall have been entered and such judgment, injunction, order or decree shall have become final and nonappealable; <u>provided</u>, that the party seeking to terminate pursuant to this clause (c) shall have complied with <u>Section 6.8</u> and with respect to other matters not covered by <u>Section 6.8</u> shall have used its reasonable best efforts, subject to the limitations set forth in Section 6.8, to remove such injunction, order or decree.

- 8.3 <u>Termination by Solstice</u>. This Agreement may be terminated by action of the Board of Directors of Solstice:
- (a) whether before or after stockholder approval, if there has been a breach by any Glass Party or any Glass Member of any representation, warranty, covenant or agreement set forth in this Agreement or the Voting Agreement or if any such representation or warranty of any Glass Party or any Glass Member shall have become untrue, in either case such that the conditions set forth in <u>Sections 7.2(a)</u>, <u>7.2(b)</u>, <u>7.2(c)</u> or <u>7.2(d)</u> would not be satisfied (other than the delivery of closing certificates) (assuming for purposes of this <u>Section 8.3(a)</u> that the references in <u>Sections 7.2(a)</u>, <u>7.2(b)</u>, <u>7.2(b)</u>, <u>7.2(c)</u> or

7.2(d) to "Closing Date" mean the date of termination pursuant to this Section 8.3(a)) and such breach or failure of a representation or warranty to be true is not curable, or, if curable, is not cured within 30 days after written notice of such breach or failure of a representation or warranty to be true is given to Glass by Solstice; provided, however, that the right to terminate this Agreement pursuant to this Section 8.3(a) shall not be available to Solstice if any Solstice Party, at such time, is in material breach of any representation, warranty, covenant or agreement set forth in this Agreement such that the conditions set forth in Sections 7.3(a) or Section 7.3(b) shall not be satisfied (other than the delivery of closing certificates) and Glass would then be entitled to terminate this Agreement under Section 8.4(a) (without giving effect to the proviso in Section 8.4(a) or the 30-day advance notice period); or

- (b) if prior to the Solstice Cutoff Date, Solstice has received a Solstice Superior Proposal, provided that Solstice has complied in all material respects with its obligations under <u>Section 6.3</u> and, in connection with the termination of this Agreement, Solstice pays to Glass in immediately available funds the Solstice Termination Amount to be paid pursuant to <u>Section 8.5(a)</u>; or
- (c) if a meeting of stockholders of any Glass Party contemplated by Section 6.7 shall have been held and such stockholders shall have voted upon, and shall have not approved this Agreement as contemplated by Section 7.1(a); provided, however, that the right to terminate this Agreement pursuant to this clause (c) shall not be available to Solstice if Solstice's failure to perform or observe any of its obligations under this Agreement in any manner shall have been the primary cause of, or resulted in, the failure to obtain the approval of such stockholders as contemplated by Section 7.1(a);
  - 8.4 <u>Termination by Glass</u>. This Agreement may be terminated by action of the Board of Directors of Glass:
- (a) whether before or after stockholder approval, if there has been a breach by any Solstice Party of any representation, warranty, covenant or agreement set forth in this Agreement or if any such representation or warranty of any Solstice Party shall have become untrue, in either case such that the conditions set forth in Sections 7.3(a) or 7.3(b) would not be satisfied (other than the delivery of closing certificates) (assuming for purposes of this Section 8.4(a) that the references in Sections 7.3(a) and 7.3(b) to "Closing Date" mean the date of termination pursuant to this Section 8.4(a)) and such breach or failure of a representation or warranty to be true is not curable, or, if curable, is not cured within 30 days after written notice of such breach or failure of a representation or warranty to be true is given to Solstice by Glass;

96

provided, however, that the right to terminate this Agreement pursuant to this Section 8.4(a) shall not be available to Glass if any Glass Party or any OEP Glass Member, at such time, is in material breach of any representation, warranty, covenant or agreement set forth in this Agreement or the Voting Agreement such that the conditions set forth in Sections 7.2(a), Section 7.2(b), Section 7.2(c) or Section 7.2(d) (other than the delivery of closing certificates) shall not be satisfied and Solstice would then be entitled to terminate this Agreement under Section 8.3(a) (without giving effect to the proviso in Section 8.3(a) or the 30-day advance notice period); or

(b) prior to the Solstice stockholders meeting (including adjournments or postponement) contemplated by Section 6.5(a), in the event that (x) the Board of Directors of Solstice makes a Solstice Change of Recommendation or (y) after the date hereof a Solstice Acquisition Proposal was publicly announced or disclosed, the Solstice Board fails to affirm the Solstice Board Recommendation within 10 Business Days after receipt of a Glass written request to do so.

### 8.5 <u>Effect of Termination</u>.

- (a) Simultaneously with any termination of this Agreement by Solstice pursuant to <u>Section 8.3(b)</u>, Solstice shall pay to Glass the Solstice Termination Amount in accordance with <u>Section 8.3(b)</u>.
- (b) If at any time after the date of this Agreement, (i) Glass terminates this Agreement pursuant to Section 8.4(b), (ii) this Agreement is terminated by Glass or Solstice pursuant to Section 8.2(b) at a time when this Agreement was terminable pursuant to Section 8.4(b) or (iii)(A) a Solstice Acquisition Proposal shall have been publicly announced or disclosed, (B)(I) this Agreement is terminated by Glass or Solstice pursuant to Section 8.2(b), (as long as, in the case of Section 8.2(b), the Solstice Acquisition Proposal was publicly announced or disclosed and not withdrawn at the time of the Solstice stockholder meeting) and (C) within 12 months after such termination, Solstice enters into a definitive agreement in respect of a Solstice Acquisition Proposal or consummates a Solstice Acquisition Proposal, then Solstice shall pay Glass the Solstice Termination Amount within five (5) Business days after (x) the date of termination in the event of a termination described in the foregoing clauses (i) or (ii) or (y) the earlier of the date of consummation of the Solstice Acquisition Proposal or execution of a definitive agreement with respect thereto in the event of a termination described in the foregoing clause (iii) (provided, that, for purpose of clause (iii) of this Section 8.5(b), the term "Solstice Acquisition Proposal" shall have the meaning assigned to such term in Article 1, except that the reference to "greater than 15% shall be deemed to be a reference to "greater than 50%").
- (c) In the event of termination of this Agreement and the abandonment of the Transactions pursuant to this Article 8, all obligations of the parties hereto shall terminate, except the obligations of the parties pursuant to this Section 8.5 and Section 6.13, and except for the provisions of Article 9 and the obligations of the parties under the Confidentiality Agreement, which shall survive such termination; provided, however, that (i) a party may recover Damages caused by a Willful and Material Breach of this Agreement by another party of any of its representations, warranties, covenants or other agreements set forth in this Agreement and Solstice may recover from Glass Damages caused by a Willful and Material Breach of the

97

Voting Agreement by any Glass Party or OEP Glass Member, and (ii) in addition to other remedies at law or equity for such Willful and Material Breach, the party so found to have committed a Willful and Material Breach (or Glass in the case of a Willful and Material Breach of the Voting Agreement by any Glass Party or OEP Glass Member) shall indemnify, hold harmless and reimburse the other parties for their respective reasonable out-of-pocket costs, fees and expenses of their counsel, accountants, financial advisors and other experts and advisors as well as fees and expenses incident to negotiation, preparation and execution of this Agreement and related documentation and stockholders' meetings and consents (collectively, "Costs"), provided, however, that, upon payment by Solstice of the Solstice Termination Amount, in full, no Solstice Party shall any longer be required to indemnify and hold harmless the other parties for their respective Costs pursuant to this Section 8.5(c). The parties acknowledge and agree that a party's Damages hereunder in the event of a breach by another party shall not be limited to reimbursement of expenses or out-of-pocket costs, and may include the benefit of the bargain lost by a party's stockholders or members, taking into consideration all relevant matters.

- (d) Any payment of the Solstice Termination Amount shall be made in cash by wire transfer of immediately available funds to an account designated by Glass in writing and any payments of Costs shall be made in cash by wire transfer of immediately available funds to an account designated by the recipient of such Costs.
- (e) The parties acknowledge and agree that in no event shall any Solstice Party be required to pay the Solstice Termination Amount on more than one occasion. In the event the Solstice Termination Amount are paid in accordance with this <u>Section 8.5</u>, except in the case of fraud by a Solstice Party, notwithstanding <u>Section 8.5(c)</u> hereof, such payment shall be liquidated damages (and not a penalty) and shall be the sole and exclusive remedy of the Glass Parties and their respective Subsidiaries, stockholders and Representatives with respect to the termination, event or breach giving rise to that payment.
- 8.6 <u>Extension; Waiver</u>. At any time prior to the Effective Time, each party may (by action taken by its Board of Directors), to the extent legally allowed, (a) extend the time for the performance of any of the obligations or other acts of the other parties hereto, (b) waive any inaccuracies in the representations and warranties made to such party contained herein or in any document delivered pursuant hereto and (c) waive compliance with any of the agreements or conditions for the benefit of such party contained herein. Any agreement on the part of a party hereto to any such extension or waiver shall be valid only if set forth in an instrument in writing signed on behalf of such party.

### **ARTICLE 9**

### GENERAL PROVISIONS

9.1 <u>Survival of Representations and Warranties</u>. None of the representations and warranties in this Agreement or in any document delivered in connection with this Agreement shall survive the consummation of the Transactions.

98

- 9.2 Notices. All notices and other communications hereunder shall be in writing and shall be deemed to have been duly delivered and received hereunder (a) four Business Days after being sent by registered or certified mail, return receipt requested, postage prepaid, (b) one Business Day after being sent for next Business Day delivery, fees prepaid, via a reputable nationwide overnight courier service, or (c) immediately upon delivery by hand or by facsimile or email (with a written or electronic confirmation of delivery), if sent during normal business hours of the recipient, then on the recipient's next Business Day, in each case to the intended recipient as set forth below:
  - (a) if to any Glass Party, to:

Genband Holdings Company 3605 E Plano Pkwy Plano, Texas 75074 Attention: David Walsh

Email: david.walsh@genband.com Facsimile: (972) 521-5837

with a copy (which shall not constitute notice) to:

Latham & Watkins LLP 885 Third Avenue New York, New York 10022 Attention: David S. Allinson Email: david.allinson@lw.com Facsimile: (212) 751-4864

(b) if to any Solstice Party, to:

Sonus Networks, Inc. 4 Technology Park Dr. Westford, Massachusetts 01886 Attention: General Counsel Email: jsnider@sonusnet.com Facsimile: (978) 614-8913

with a copy (which shall not constitute notice) to:

Wilmer Cutler Pickering Hale and Dorr LLP 60 State Street Boston, Massachusetts 02109 Attention: Jay E. Bothwick, Esq. and Joseph B. Conahan, Esq.

Email: jay.bothwick@wilmerhale.com; joseph.conahan@wilmerhale.com

Facsimile: (617) 526-5000

9.3 <u>Assignment; Binding Effect; Benefit</u>. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any of the parties hereto (whether

by operation of law or otherwise) without the prior written consent of the other parties. Subject to the preceding sentence, this Agreement shall be binding upon and shall inure to the benefit of and be enforceable by the parties hereto and their respective successors and assigns. Notwithstanding anything contained in this Agreement to the contrary, except for the provisions of <u>Sections 6.14</u>, nothing in this Agreement, expressed or implied, is intended to confer on any person other than the parties hereto or their respective successors and permitted assigns any rights, remedies, obligations or liabilities under or by reason of this Agreement.

- 9.4 <u>Entire Agreement</u>. This Agreement, the Exhibits, the Schedules, the Solstice Disclosure Letter, the Glass Disclosure Letter and any documents delivered by the parties in connection herewith and the Confidentiality Agreement constitute the entire agreement among the parties with respect to the subject matter hereof and supersede all prior agreements and understandings among the parties with respect thereto. No addition to or modification of any provision of this Agreement shall be binding upon any party hereto unless made in writing and signed by all parties hereto.
- 9.5 Amendments. This Agreement may be amended by the parties hereto, by action taken or authorized by their Boards of Directors, at any time before or after approval of matters presented in connection with the Mergers by the stockholders of Solstice and the other parties to such Mergers, but after any such stockholder approval, no amendment shall be made which by law requires the further approval of stockholders unless such amendment is subject to such further approval. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties hereto.
- 9.6 Governing Law. This Agreement and the rights and obligations of the parties hereto shall be governed by and construed in accordance with the laws of the State of Delaware, without giving effect to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware. Each of the parties hereby irrevocably agrees that any legal action or proceeding with respect to this Agreement, the Transactions, and the rights and obligations arising hereunder, or for recognition and enforcement of any judgment in respect of this Agreement, the Transactions, and the rights and obligations arising hereunder brought by the other party hereto or its successors or assigns, shall be brought and determined exclusively in the Delaware Court of Chancery and any state appellate court therefrom within the State of Delaware (or, if the Delaware Court of Chancery declines to accept jurisdiction over a particular matter, any state or federal court within the State of Delaware) (collectively with Delaware Court of Chancery, the "Delaware Courts"). Each of the parties hereto further agrees not to commence any litigation relating to this Agreement or the Transactions except in the Delaware Courts, waives any objection to the laying of venue of any such litigation in the Delaware Courts and agrees not to plead or claim in any Delaware Court that such litigation brought therein has been brought in an inconvenient forum. EACH PARTY TO THIS AGREEMENT IRREVOCABLY WAIVES THE RIGHT TO A TRIAL BY JURY IN CONNECTION WITH ANY MATTER ARISING OUT OF THIS AGREEMENT AND, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY DEFENSE OR OBJECTION IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY PROCEEDING UNDER THIS AGREEMENT BROUGHT IN THE DELAWARE COURTS AND ANY CLAIM THAT ANY PROCEEDING

100

### UNDER THIS AGREEMENT BROUGHT IN ANY SUCH COURT HAS BEEN BROUGHT IN AN INCONVENIENT FORUM.

- 9.7 <u>Counterparts</u>. This Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered shall be an original, but all such counterparts shall together constitute one and the same instrument. Each counterpart may consist of a number of copies hereof each signed by less than all, but together signed by all of the parties hereto.
- 9.8 <u>Headings</u>. Headings of the Articles and Sections of this Agreement are for the convenience of the parties only, and shall be given no substantive or interpretative effect whatsoever.
- 9.9 <u>Interpretation</u>. Except where expressly stated otherwise in this Agreement, the following rules of interpretation apply to this Agreement: (a) "either" and "or" are not exclusive and "include," "includes" and "including" are not limiting; (b) "hereof," "hereto," "hereby," "herein" and "hereunder" and words of similar import when used in this Agreement refer to this Agreement as a whole and not to any particular provision of this Agreement; (c) "date of this Agreement" refers to the date set forth in the initial caption of this Agreement; (d) "extent" in the phrase "to the extent" means the degree to which a subject or other thing extends, and such phrase does not mean simply "if"; (e) the descriptive headings and table of contents included herein are included for convenience only and shall not affect in any way the meaning or interpretation of this Agreement or any provision hereof; (f) definitions contained in this Agreement are applicable to the singular as well as the plural forms of such terms; (g) references to a contract or agreement mean such contract or agreement as amended or otherwise supplemented or modified from time to time; (h) references to a Person are also to its permitted successors and assigns; (i) references to an "Article," "Section," "Exhibit" or "Schedule" refer to an Article or Section of, or an Exhibit or Schedule to, this Agreement; (j) references to "\$" or otherwise to dollar amounts refer to the lawful currency of the United States; (k) references to a federal, state, local or foreign law include any rules, regulations and delegated legislation issued thereunder; and (1) references to accounting terms used and not otherwise defined herein have the meaning assigned to them under GAAP. The language used in this Agreement shall be deemed to be the language chosen by the parties hereto to express their mutual intent, and no rule of strict construction shall be applied against any party hereto. No summary of this Agreement prepared by any party shall affect the meaning or interpretation of this Agreement. If any date on which a party is required to make a payment or a delivery pursuant to the terms hereof is not a Business Day, then such party shall make such payment or delivery on the next succeeding Business Day. Time shall be of the essence in this Agreement.
- 9.10 <u>Waivers</u>. Except as provided in this Agreement, no action taken pursuant to this Agreement, including any investigation by or on behalf of any party, or delay or omission in the exercise of any right, power or remedy accruing to any party as a result of any breach or default hereunder by any other party shall be deemed to constitute a waiver by the party taking such action of compliance with any representations, warranties, covenants or agreements contained in this Agreement. The waiver by any party hereto of a breach of any provision hereunder shall not operate or be construed as a waiver of any prior or subsequent breach of the same or any other provision hereunder.

- 9.11 <u>Incorporation of Disclosure Letters and Exhibits</u>. The Solstice Disclosure Letter, the Glass Disclosure Letter and all Exhibits and Schedules hereto are hereby incorporated herein and made a part hereof for all purposes as if fully set forth herein.
- 9.12 <u>Severability</u>. If any provision (or part thereof) of this Agreement is invalid, illegal or unenforceable, that provision (or part thereof) will, to the extent possible, be modified in such a manner as to be valid, legal and enforceable but so as to retain most nearly the intent of the parties as expressed

herein, and if such a modification is not possible, that provision (or part thereof) will be severed from this Agreement, and in either case the validity, legality and enforceability of the remaining provisions (or parts thereof) of this Agreement will not in any way be affected or impaired thereby. If any provision (or part thereof) of this Agreement is so broad as to be unenforceable, the provision (or part thereof) shall be interpreted to be only so broad as is enforceable.

9.13 <u>Enforcement of Agreement</u>. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with its specific terms or if this Agreement was otherwise breached and that monetary damages, even if available, would not be an adequate remedy hereunder. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any Delaware Court without proof of actual damages and each party hereto waives any requirement for the securing or posting of any bond in connection with such remedy, this being in addition to any other remedy to which they are entitled at law or in equity. The parties further agree not to assert that a remedy of specific enforcement is unenforceable, invalid, contrary to Applicable Law or in equity for any reason, nor to assert that a remedy of monetary damages would provide an adequate remedy for such breach.

## [SIGNATURE PAGES FOLLOW]

102

IN WITNESS WHEREOF, the parties have executed this Agreement and caused the same to be duly delivered on their behalf on the day and year first written above.

SONUS NETWORKS, INC.

By: /s/ Raymond P. Dolan
Name: Raymond P. Dolan
Title: President and CEO

SOLSTICE SAPPHIRE INVESTMENTS, INC.,

By: /s/ Susan Villare
Name: Susan Villare
Title: President and Treasurer

SOLSTICE SAPPHIRE, INC.,

By: /s/ Susan Villare
Name: Susan Villare
Title: President and Treasurer

GREEN SAPPHIRE INVESTMENTS LLC,

By: /s/ Susan Villare
Name: Susan Villare
Title: President and Treasurer

GREEN SAPPHIRE LLC,

By: <u>/s/ Susan Villare</u>
Name: Susan Villare
Title: President and Treasurer

GENBAND HOLDINGS COMPANY,

By: /s/ David Walsh
Name: David Walsh

Title: President, CEO and Chairman

GENBAND INC.,

By: /s/ David Walsh

Name: David Walsh

Title: President, CEO and Chairman

GENBAND II INC.,

By: /s/ David Walsh
Name: David Walsh

Title: President, CEO and Chairman

EXHIBIT A AGREED FORM

### **VOTING AGREEMENT**

This VOTING AGREEMENT (this "<u>Agreement</u>") is made and entered into as of May 23, 2017, by and among Sonus Networks, Inc., a Delaware corporation ("<u>Solstice</u>"); Solstice Sapphire Investments, Inc., a Delaware corporation and wholly-owned subsidiary of Solstice (the "<u>Company</u>"); each member of GENBAND Holdings Company, a Cayman Islands exempted company limited by shares ("<u>Glass</u>"), named on <u>Annex 1</u> (the "<u>Glass Members</u>"); each stockholder of GENBAND Inc., a Delaware corporation ("<u>GB</u>"), named on <u>Annex 2</u> (the "<u>GB Stockholders</u>"); and each stockholder of GENBAND II, Inc., a Delaware corporation ("<u>GB II</u>"), named on <u>Annex 3</u> (the "<u>GB II Stockholders</u>" and collectively with the Glass Members and the GB Stockholders, the "<u>Equityholders</u>").

## RECITALS

WHEREAS, the Merger Agreement (as defined below) provides for, among other things, (a) the merger of Cayman Merger Sub with and into Glass, with Glass surviving such merger as a subsidiary wholly owned, in part directly and in part indirectly, by the Company (the "Glass Merger"); (b) the merger of GB with and into GB Merger Sub, with GB Merger Sub surviving such merger as a wholly owned subsidiary of the Company (the "GB Merger"); and (c) the merger of GB II with and into GB Merger Sub, with GB Merger Sub surviving such merger as a wholly owned subsidiary of the Company (the "GB II Merger" and collectively with the Glass Merger and the GB Merger, the "Glass Mergers"); and

WHEREAS, as an inducement to and condition of the parties to the Merger Agreement to enter into the Merger Agreement, the Equityholders are entering into this Agreement;

NOW, THEREFORE, in consideration of the foregoing and of the representations, warranties, covenants and agreements contained in this Agreement, the parties hereto agree as follows:

## 1. <u>Certain Definitions</u>.

(a) Capitalized terms used herein and not otherwise defined herein shall have the respective meanings ascribed to them in the Agreement and Plan of Merger, dated as of the date hereof (the "Merger Agreement"), among Solstice, the Company, Glass, GB and the other parties thereto. For all purposes of and under this Agreement, the following terms shall have the following respective meanings:

"Beneficial Ownership" by a Person of any securities means that such Person is a beneficial owner of such securities in accordance with Rule 13d-3 adopted by the SEC under the Exchange Act. The term "Beneficially Own" shall have a correlative meaning.

"Constructive Sale" means with respect to any security, a short sale with respect to such security, entering into or acquiring a derivative contract with respect to such security, entering into or acquiring a futures or forward contract to deliver such

security or entering into any other hedging or other derivative transaction that has the effect of either directly or indirectly materially changing the economic benefits or risks of ownership of such security.

"<u>Permitted Transfer</u>" means a Transfer by an Equityholder (or an Affiliate thereof) to an Affiliate of such Equityholder, provided that such transferee Affiliate agrees in writing to assume all of such transferring Equityholder's obligations hereunder in respect of the securities subject to such Transfer and to be bound by, and comply with, the terms of this Agreement, with respect to the Shares that are subject to such Transfer, to the same extent as such transferring Equityholder is bound hereunder.

"Shares" means, when used with respect to any Equityholder, (i) all shares of capital stock or other equity interests of any Glass Party Beneficially Owned or owned of record by such Equityholder as of the date hereof and (ii) all shares of capital stock or other equity interests of any Glass Party of which such Equityholder acquires Beneficial Ownership or record ownership during the period commencing with the execution and delivery of this Agreement and expiring on the termination of this Agreement in accordance with Section 7.

"<u>Transaction Documents</u>" means this Agreement, the Merger Agreement, the Stockholders Agreement and each agreement, document or instrument referred to therein or herein, including the Registration Rights Agreement (as defined in the Stockholders Agreement).

"Transfer" means, with respect to any security (including any Share), the direct or indirect assignment, sale, transfer, tender, exchange, pledge, hypothecation, or the grant, creation or suffrage of a Lien in or upon, or the gift, grant or placement in trust, or the Constructive Sale or other disposition of such security (excluding transfers by testamentary or intestate succession, by domestic relations order or other court order) or any right, title or interest therein (including any right or power to vote to which the holder thereof may be entitled, whether such right or

power is granted by proxy or otherwise), or the record ownership or Beneficial Ownership thereof, the offer to make such a sale, transfer, Constructive Sale or other disposition, and each agreement, arrangement or understanding, whether or not in writing, to effect any of the foregoing.

- 2. <u>Transfer and Voting Restrictions</u>. Each Equityholder hereby agrees as follows:
- (a) At all times during the period commencing with the execution and delivery of this Agreement and expiring upon the termination of this Agreement in accordance with Section 7, such Equityholder shall not, except by operation of the Glass Mergers, Transfer any of the Shares, unless such Transfer is a Permitted Transfer.
- (b) Except as otherwise permitted by this Agreement or by order of a court of competent jurisdiction, such Equityholder shall not commit any act that is intended to restrict or affect such Equityholder's legal power, authority and right to vote all of the Shares or otherwise prevent or disable such Equityholder from performing any of such Equityholder's obligations under this Agreement. Without limiting the generality of the foregoing, except for this

A-2

Agreement and as otherwise permitted by this Agreement, such Equityholder shall not enter into any voting agreement with any Person with respect to any of the Shares, grant any Person any proxy (revocable or irrevocable) or power of attorney with respect to any of the Shares, deposit any of the Shares in a voting trust or otherwise enter into any agreement or arrangement with any Person, in each case, limiting or affecting such Equityholder's legal power, authority or right to vote the Shares in favor of the adoption of the Merger Agreement and the approval of the Glass Mergers and the other transactions contemplated by the Transaction Documents (the "<u>Transactions</u>"), or amend or waive any rights under the Glass Stockholders Agreement. Without limiting the generality of the foregoing, except for this Agreement and as otherwise permitted by this Agreement, each Equityholder shall not enter into any agreement that would restrict or impede its rights to exercise rights over Shares held by other parties.

- (c) Such Equityholder understands and agrees that if such Equityholder attempts to Transfer, vote or provide any other person with the authority to vote any of the Shares other than in compliance with this Agreement, each Glass Party shall not, and such Equityholder hereby unconditionally and irrevocably instructs each Glass Party not to, (i) permit any such Transfer on its books and records, (ii) issue a new certificate representing any of the Shares, or (iii) record such vote.
  - 3. <u>Agreement to Vote Shares</u>. Each Equityholder hereby agrees as follows:
- (a) Until the termination of this Agreement in accordance with Section 7, at every meeting of the members or stockholders of any Glass Party called, and at every adjournment or postponement thereof, and on every action or approval by written consent of the members or stockholders of any Glass Party, such Equityholder (in such Equityholder's capacity as such) shall appear at the meeting or otherwise cause the Shares to be present thereat for purposes of establishing a quorum and, to the extent not voted by the Proxyholders, vote (i) in favor of the adoption of the Merger Agreement and the approval of the Transactions, (ii) against the approval or adoption of any proposal made in opposition to, or in competition with, the Transactions, and (iii) against any Glass Acquisition Proposal or any other action that is intended, or would reasonably be expected, to result in a breach of any covenant, representation or warranty or any other obligation or agreement of any Glass Party under the Merger Agreement or of such Equityholder under any Transaction Document to which such Equityholder is or will be a party or otherwise impede, interfere with, delay, postpone, discourage or adversely affect the consummation of the Transactions (each of (ii) and (iii), a "Competing Transaction").
- (b) If such Equityholder is the Beneficial Owner, but not the record holder, of the Shares, such Equityholder agrees to use commercially reasonable efforts to cause the record holder and any nominees to vote all of the Shares in accordance with <u>Section 3(a)</u>.
- (c) Such Equityholder shall use its reasonable best efforts to cause the equity holders of each Glass Party to adopt the Merger Agreement and approve the Transactions, and to cause the consummation of the Transactions.

A-3

- 4. <u>Grant of Irrevocable Proxy</u>. Each Equityholder hereby agrees as follows:
- (a) Solely in the event of a failure by such Stockholder to vote the Shares or otherwise comply with its obligations under Section 3, instruct nominees or record holders to vote the Shares, or grant a consent or approval in respect of such Shares in accordance with <u>Section 3</u>, as the case may be, such Equityholder hereby irrevocably appoints the Company, Solstice and each of their respective executive officers or other designees (the "<u>Proxyholders</u>"), as such Equityholder's proxy and attorney-in-fact (with full power of substitution and re-substitution), and grants to the Proxyholders full authority, for and in the name, place and stead of such Equityholder, to vote the Shares, to instruct nominees or record holders to vote the Shares, or grant a consent or approval in respect of such Shares in accordance with <u>Section 3</u> and, in the discretion of the Proxyholders with respect to any proposed adjournments or postponements of any meeting of members or stockholders of any Glass Party at which any of the matters described in <u>Section 3</u> is to be considered.
  - (b) Such Equityholder hereby revokes any proxies heretofore given by such Equityholder in respect of the Shares.
- (c) Such Equityholder hereby affirms that the irrevocable proxy set forth in this Section 4 is given in connection with the execution of the Merger Agreement, and that such irrevocable proxy is given to secure the performance of the duties of such Equityholder under this Agreement. Such Equityholder hereby further affirms that the irrevocable proxy is coupled with an interest, is intended to be irrevocable in accordance with the provisions of Section 212 of the Delaware General Corporation Law (to the extent applicable), and may under no circumstances be revoked during the term of this Agreement. The irrevocable proxy granted by such Equityholder herein is a durable power of attorney and shall survive the dissolution, bankruptcy or incapacity of such Equityholder.
- (d) The Proxyholders may not exercise this irrevocable proxy on any matter except as provided above. Such Equityholder may vote the Shares on all other matters.

- The Company may terminate this proxy at any time by written notice to such Equityholder. (e) Other Covenants. Each Equityholder hereby agrees as follows: Such Equityholder agrees to comply with Section 6.4 of the Merger Agreement as if it were a Glass Party and to execute and deliver at or prior to the Closing each Transaction Document contemplated by the Merger Agreement to be executed and delivered by such Equity holder at or prior to the Closing. Except with respect to information publicly disclosed by the Company or Solstice or in connection with any disclosure of any (b) information concerning this Agreement or the Transactions in connection with any dispute between the parties regarding this Agreement or the Transactions, such Equityholder may not issue any press release or make any public announcements pertaining to any Transaction Document or the Transactions without the prior written consent of the Company (which consent shall not be unreasonably withheld, conditioned
  - A-4

or delayed), except as may be required by Applicable Law or by obligations pursuant to any listing agreement with any national securities exchange.

- Subject to Section 8.5 of the Merger Agreement, if the Mergers are not consummated, all costs and expenses incurred in connection with any Transaction Document and the Transactions shall be paid by the party incurring such expenses.
- Such Equityholder shall give prompt notice to the Company of (a) any representation or warranty made by it or contained in this Agreement becoming untrue or inaccurate in any material respect and (b) the failure by such Equityholder to comply with or satisfy in any material respect any covenant, condition or agreement to be complied with or satisfied by such Equityholder under this Agreement; provided, however, that no such notification shall affect the representations, warranties, covenants or agreements of the parties under any Transaction Document.
- If such Equityholder is an OEP Glass Member, it agrees to report, for U.S. federal income Tax purposes, the exchange of its Glass Shares for Company Common Stock pursuant to the Glass Merger as a transaction qualifying under Section 351 of the Code, except as otherwise required as a result of a final determination of a taxing authority; provided, however, that such Equityholder may treat the Glass Merger as a reorganization within the meaning of Section 368(a) of the Code.
- Representations and Warranties of Each Equityholder. Each Equityholder, solely as to itself, represents and warrants to the Company and Solstice that the following representations and warranties are true and correct as of the date hereof and will be true and correct as of the Closing as if made at and as of the Closing:
- If such Equityholder is not a natural person, such Equityholder is an entity duly organized, validly existing and in good standing under the laws of its jurisdiction of organization. Such Equityholder has all requisite corporate (or comparable) power, authority and legal capacity (as applicable) to Beneficially Own or own of record the Shares Beneficially Owned or owned of record by such Equityholder.
- Such Equityholder has the requisite corporate (or comparable) power, authority and legal capacity, as applicable, to execute and deliver this Agreement and to perform its obligations hereunder. Without limiting the foregoing, except as otherwise provided in this Agreement, (i) such Equityholder has full power and authority to (A) make, enter into and carry out the terms of this Agreement and to grant the irrevocable proxy as set forth in Section 4 and (B) vote all of the Shares in the manner set forth in this Agreement without the consent or approval of, or any other action on the part of, any other Person (including any Governmental Authority) and (ii) such Equityholder has not entered into any voting agreement (other than this Agreement) with any Person with respect to any of the Shares, granted any Person any proxy (revocable or irrevocable) or power of attorney with respect to any of the Shares, deposited any of the Shares in a voting trust or entered into any arrangement or agreement with any Person, in each case, limiting or affecting such Equityholder's legal power, authority or right to vote the Shares on any matter provided herein (other than those arising under the Glass Organizational

A-5

Documents and the Glass Stockholders Agreement, applicable securities laws and this Agreement).

5.

- If such Equityholder is not a natural person, the consummation by such Equityholder of the Transactions has been duly authorized by all requisite corporate (or comparable) action on the part of such Equityholder. This Agreement constitutes the valid and legally binding obligation of such Equityholder, enforceable against such Equityholder in accordance with its terms, subject to applicable bankruptcy, insolvency, moratorium or other similar laws relating to creditors' rights and general principles of equity.
- Annex 1 hereto completely and accurately sets forth, as of the date hereof, the number of Glass Shares owned of record by such Equityholder and the number of Glass Shares otherwise Beneficially Owned by such Equityholder. Annex 2 hereto completely and accurately sets forth, as of the date hereof, the number of GB Shares owned of record by such Equityholder and the number of GB Shares otherwise Beneficially Owned by such Equityholder. Annex 3 hereto completely and accurately sets forth, as of the date hereof, the number of GB II Shares owned of record by such Equityholder and the number of GB II Shares otherwise Beneficially Owned by such Equityholder. Such Equityholder Beneficially Owns or owns of record, free and clear of all Liens (other than those arising under the Glass Organizational Documents and the Glass Stockholders Agreement, applicable securities laws and this Agreement), all Shares indicated as being Beneficially Owned or owned of record by such Equityholder on Annexes 1, 2 and 3. Except as set forth on Annexes 1, 2 and 3, such Equityholder does not own of record, have the power to vote or otherwise Beneficially Own any equity securities of any Glass Party.
- Neither the execution and delivery by such Equityholder of this Agreement nor the consummation by such Equityholder of the (e) Transactions in accordance with the respective terms of the applicable Transaction Documents, nor the performance by such Equityholder of its obligations hereunder will: (i) if such Equityholder is not a natural person, conflict with or result in a breach of any provisions of or create any rights in favor of any other party under the organizational documents of such Equityholder; (ii) violate, or conflict with, or result in a breach of any provision of, or constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, or result in the termination or in a right of termination or cancellation of, or give rise to a right of purchase under, or accelerate the performance required by, any note, bond, mortgage, indenture, deed of trust, lease, contract or agreement to which such Equityholder is a party; or (ii) contravene or conflict with or constitute a violation of any provision of any Applicable

	Laws binding upon or applicable to such Equityholder, except, in the case of matters described in clauses (ii) or (iii), as would not, individually or in the aggregate, reasonably be expected to have a Glass Material Adverse Effect or prevent or materially delay the performance by such Equityholder of its obligations under this Agreement.		
	(f) Neither the execution and delivery by such Equityholder of this Agreement nor the consummation by such Equityholder of the Transactions in accordance with the respective terms of the applicable Transaction Documents, nor the performance by such Equityholder of its obligations hereunder will require any consent, approval or authorization of, or filing or registration with, any Governmental Authority on the part of such Equityholder other		
	A-6		
than filings, consents, approvals and authorizations required under applicable Antitrust Laws, the Glass Regulatory Filings and except for any consent, approval or authorization the failure of which to obtain and for any filing or registration the failure of which to make would not, individually or in the aggregate, reasonably be expected to have a Glass Material Adverse Effect or prevent or materially delay the performance by such Equityholder of its obligations under the Transaction Documents to which it is or will be a party.			
	(g) Such Equityholder has not entered into any contract, arrangement or understanding with any person or firm which may result in		

- (g) Such Equityholder has not entered into any contract, arrangement or understanding with any person or firm which may result in the obligation of Solstice, the Company or any Glass Party to pay any finder's fees, brokerage or agent's commissions or other like payments in connection with the negotiations leading to this Agreement or the consummation of the Transactions (except that, for the avoidance of doubt, Glass has retained Guggenheim Partners to provide financial advice with respect to the Transactions).
- (h) Notwithstanding anything contained in this Agreement to the contrary, such Equityholder acknowledges and agrees that no Solstice Party nor any other Person has made or is making any representations or warranties relating to any Solstice Party whatsoever, express or implied, except as expressly provided in Section 8(a), including any implied representation or warranty as to the accuracy or completeness of any information regarding any Solstice Party furnished or made available to such Equityholder or any of its Representatives. Without limiting the generality of the foregoing, such Equityholder acknowledges that no representations or warranties are made with respect to any projections, forecasts, estimates, budgets or prospect information that may have been made available to any Equityholder or any Glass Party or any of their respective Representatives. Such Equityholder has not relied on any representations or warranties relating to any Solstice Party in determining to enter into this Agreement, except for those express representations and warranties on which such Equityholder is entitled to rely pursuant to Section 8(a).
- (i) Such Equityholder understands and acknowledges that Solstice and the Company are entering into the Merger Agreement in reliance upon such Equityholder's execution and delivery of this Agreement and the representations and warranties of such Equityholder contained herein.

### 7. Termination.

- (a) This Agreement (including the proxy granted pursuant to <u>Section 4</u>) shall automatically terminate upon the earliest of (i) the mutual agreement of the parties hereto, (ii) the Effective Time, and (iii) any termination prior to the Closing of the Merger Agreement in accordance with <u>Article 8</u> thereof.
- (b) In the event of termination of this Agreement pursuant to this <u>Section 7</u>, all obligations of the parties hereto shall terminate, except a party may recover Damages caused by a Willful and Material Breach of this Agreement by another party of any of its representations, warranties, covenants or other agreements set forth in this Agreement (but solely to the extent any such breach has been the primary cause of, or resulted in, the failure of the Closing of the Mergers).

A-7

(c) At any time prior to the Effective Time, each party may (in the case of any party that is not a natural person, by action taken by its Board of Directors), to the extent legally allowed, (i) extend the time for the performance of any of the obligations or other acts of the other parties hereto, (ii) waive any inaccuracies in the representations and warranties made to such party contained herein or in any document delivered pursuant hereto and (iii) waive compliance with any of the agreements or conditions for the benefit of such party contained herein. Any agreement on the part of a party hereto to any such extension or waiver shall be valid only if set forth in an instrument in writing signed on behalf of such party.

# 8. Miscellaneous Provisions.

- (a) <u>Survival of Representations and Warranties; Reliance on Certain Representations and Warranties of the Solstice Parties.</u> None of the representations and warranties of the parties set forth in this Agreement shall survive the Closing and the consummation of the Transactions. The covenants that by their terms are to be performed in whole or in part after the Closing, of the parties set forth in this Agreement shall survive the Closing and the consummation of the Transactions. Subject to Section 9.1 of the Merger Agreement, each Equityholder shall be entitled to rely on the representations and warranties of the Solstice Parties set forth in Article 4 of the Merger Agreement.
- (b) Notices. All notices and other communications hereunder shall be in writing and shall be deemed to have been duly delivered and received hereunder (i) four Business Days after being sent by registered or certified mail, return receipt requested, postage prepaid, (ii) one Business Day after being sent for next Business Day delivery, fees prepaid, via a reputable nationwide overnight courier service, or (iii) immediately upon delivery by hand or by facsimile or email (with a written or electronic confirmation of delivery), if sent during normal business hours of the recipient, or if not sent during normal business hours of the recipient, then on the recipient's next Business Day, in each case to the intended recipient as set forth below:
  - (A) if to any Equityholder, to:

the address set forth on its signature page hereto

with a copy (which shall not constitute notice) to:

Latham & Watkins LLP 885 Third Avenue New York, New York 10022 Attention: David S. Allinson

Facsimile: (212) 751-4864

(B) if to Solstice or the Company (or after the Closing) Glass, to:

4 Technology Park Drive Westford, MA 01886 Attention: Jeffrey M. Snider

with a copy (which shall not constitute notice) to:

A-8

Wilmer Cutler Pickering Hale and Dorr LLP 60 State Street Boston, MA 02109

Attention: Jay E. Bothwick

Joseph B. Conahan

Facsimile: (617) 526-5000

- (c) Assignment; Binding Effect; Benefit. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any of the parties hereto (whether by operation of law or otherwise) without the prior written consent of the other parties. Subject to the preceding sentence, this Agreement shall be binding upon and shall inure to the benefit of and be enforceable by the parties hereto and their respective successors and assigns. Notwithstanding anything contained in this Agreement to the contrary, nothing in this Agreement, expressed or implied, is intended to confer on any person other than the parties hereto or their respective successors and permitted assigns any rights, remedies, obligations or liabilities under or by reason of this Agreement.
- (d) <u>Entire Agreement</u>. This Agreement, the Annexes and the documents delivered pursuant to this Agreement constitute the entire agreement among the parties with respect to the subject matter hereof and supersede all prior agreements and understandings among the parties with respect thereto (it being understood that the other Transaction Documents shall remain in full force and effect in accordance with their terms). No addition to or modification of any provision of this Agreement shall be binding upon any party hereto unless made in writing and signed by all parties hereto.
- (e) <u>Amendments</u>. This Agreement may be amended by the parties hereto, by (in the case of a party that is not a natural person) action taken or authorized by their Boards of Directors, managers, governing committee or members, as applicable, at any time before or after approval of matters presented in connection with the Mergers by the stockholders of Solstice and the other parties to such Mergers, but after any such stockholder approval, no amendment shall be made which by law requires the further approval of stockholders of any party to the Merger Agreement unless such amendment is subject to such further approval. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties hereto.
- (f) Governing Law. This Agreement and the rights and obligations of the parties hereto shall be governed by and construed in accordance with the laws of the State of Delaware, without giving effect to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware. Each of the parties hereby irrevocably agrees that any legal action or proceeding with respect to this Agreement, the Transactions, and the rights and obligations arising hereunder, or for recognition and enforcement of any judgment in respect of this Agreement, the Transactions, and the rights and obligations arising hereunder brought by the other party hereto or its successors or assigns, shall be brought and determined exclusively in the Delaware Courts. Each of the parties hereto further agrees not to commence any litigation relating to this Agreement or the Transactions except in the Delaware Courts, waives any objection to the laying of venue of any such litigation in the Delaware Courts and agrees not to plead or claim in any Delaware Court that such litigation brought therein has been brought in an

A-9

inconvenient forum. EACH PARTY TO THIS AGREEMENT IRREVOCABLY WAIVES THE RIGHT TO A TRIAL BY JURY IN CONNECTION WITH ANY MATTER ARISING OUT OF THIS AGREEMENT AND, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY DEFENSE OR OBJECTION IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY PROCEEDING UNDER THIS AGREEMENT BROUGHT IN THE DELAWARE COURTS AND ANY CLAIM THAT ANY PROCEEDING UNDER THIS AGREEMENT BROUGHT IN ANY SUCH COURT HAS BEEN BROUGHT IN AN INCONVENIENT FORUM.

- (g) <u>Counterparts</u>. This Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered shall be an original, but all such counterparts shall together constitute one and the same instrument. Each counterpart may consist of a number of copies hereof each signed by less than all, but together signed by all of the parties hereto.
- (h) <u>Headings</u>. Headings of the Articles and Sections of this Agreement are for the convenience of the parties only, and shall be given no substantive or interpretative effect whatsoever.
- (i) <u>Interpretation</u>. Except where expressly stated otherwise in this Agreement, the following rules of interpretation apply to this Agreement: (i) "either" and "or" are not exclusive and "include," "includes" and "including" are not limiting; (ii) "hereof," "hereto," "hereby," "herein" and "hereunder" and words of similar import when used in this Agreement refer to this Agreement as a whole and not to any particular provision of this Agreement; (iii) "date of this Agreement" refers to the date set forth in the initial caption of this Agreement; (iv) "extent" in the phrase "to the extent" means the degree to which a subject or other thing extends, and such phrase does not mean simply "if"; (v) the descriptive headings included herein are included for convenience only and shall not affect in any way the meaning or interpretation of this Agreement or any provision hereof; (vi) definitions contained in this Agreement are applicable to the singular as well as the plural forms of such terms; (vii) references to a contract or agreement mean such contract or

(ix) references to an "Article," "Section" or "Annex" refer to an Article, Sec dollar amounts refer to the lawful currency of the United States; (xi) reference delegated legislation issued thereunder; and (xii) references to accounting terunder GAAP. The language used in this Agreement shall be deemed to be thrule of strict construction shall be applied against any party hereto. No summinterpretation of this Agreement. If any date on which a party is required to then such party shall make such payment or delivery on the next succeeding	rms used and not otherwise defined herein have the meaning assigned to them ne language chosen by the parties hereto to express their mutual intent, and no mary of this Agreement prepared by any party shall affect the meaning or make a payment or a delivery pursuant to the terms hereof is not a Business Day, Business Day. Time shall be of the essence in this Agreement.			
(j) <u>Waivers</u> . Except as provided in this Agreement, the behalf of any party, or delay or omission in the exercise of any right, power of the control of the	no action taken pursuant to this Agreement, including any investigation by or on or remedy accruing to any party as a result of any breach or			
	A-10			
default hereunder by any other party shall be deemed to constitute a waiver by warranties, covenants or agreements contained in this Agreement. The waive be construed as a waiver of any prior or subsequent breach of the same or an	er by any party hereto of a breach of any provision hereunder shall not operate or			
(k) <u>Incorporation of Annexes</u> . The Annexes hereto a set forth herein.	are hereby incorporated herein and made a part hereof for all purposes as if fully			
thereof) will, to the extent possible, be modified in such a manner as to be va as expressed herein, and if such a modification is not possible, that provision validity, legality and enforceability of the remaining provisions of (or parts the	this Agreement is invalid, illegal or unenforceable, that provision (or part alid, legal and enforceable but so as to retain most nearly the intent of the parties a (or part thereof) will be severed from this Agreement, and in either case the hereof) this Agreement will not in any way be affected or impaired thereby. If any ble, the provision (or part thereof) shall be interpreted to be only so broad as is			
(m) Enforcement of Agreement. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with its specific terms or if this Agreement was otherwise breached and that monetary damages, even if available, would not be an adequate remedy hereunder. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any Delaware Court without proof of actual damages and each party hereto waives any requirement for the securing or posting of any bond in connection with such remedy, this being in addition to any other remedy to which they are entitled at law or in equity. The parties further agree not to assert that a remedy of specific enforcement is unenforceable, invalid, contrary to Applicable Law or in equity for any reason, nor to assert that a remedy of monetary damages would provide an adequate remedy for such breach.				
(SIGNATURE	PAGES FOLLOW)			
	A-11			
IN WITNESS WHEREOF, the parties have executed this and year first written above.	Agreement and caused the same to be duly delivered on their behalf on the day  Solstice Sapphire Investments, Inc.			
	By: Name: Susan Villare Title: President and Treasurer			
Signature Page	to Voting Agreement			
	Sonus Networks, Inc.			
	By: Name: Raymond P. Dolan			
Signature Page	Title: President and CEO  to Voting Agreement			
	[EQUITYHOLDER]			
	By:			

	Title:	
Notice Information:		
Address:		
Phone:		
Fax:		
Email:		
	Signature Page to Voting Agreement	
	Approx 1	
	Annex 1  Glass Shares	
	Glass Snares	
	Annex 2	
	GB Shares	
	Annex 3	
	GB II Shares	
		EXHIBIT B AGREED FORM
	PRINCIPAL STOCKHOLDERS AGREEMENT	
	BY AND AMONG	
	SOLSTICE SAPPHIRE INVESTMENTS, INC.	
	AND	
	THE OEP STOCKHOLDERS	
	[·], 2017	
	TABLE OF CONTENTS	
		Page

B-1

B-1

B-2

B-6

B-6

B-6

B-10

SECTION I.

1.1

1.2

1.3

SECTION II.

2.1 2.2 DEFINITIONS

**Defined Terms** 

**Board of Directors** 

Committees of the Board of Directors

Effectiveness

Drafting Conventions; No Construction Against Drafter

BOARD MATTERS; CHIEF EXECUTIVE OFFICER

Name:

2.3	Additional Management Provisions	B-11
2.4	Certain Transactions	B-12
2.5	Removal of Chairman and CEO	B-12
SECTION III.	STANDSTILL PROVISIONS	B-13
3.1	Standstill	B-13
3.2	Nonapplicability to Certain Affiliates	B-14
3.3	Nonintervention by Company	B-15
SECTION IV.	TRANSFER RESTRICTIONS	B-15
4.1	Transfer Restrictions	B-15
4.2	Change of Control Transactions	B-16
4.3	Legend	B-16
SECTION V.	PREEMPTIVE RIGHTS	B-16
5.1	Preemptive Rights	B-16
SECTION VI.	MISCELLANEOUS PROVISIONS	B-17
6.1	Reliance	B-17
6.2	Amendment and Waiver; Actions of the Board	B-17
6.3	Notices	B-18
6.4	Counterparts	B-19
6.5	Remedies; Severability	B-19
6.6	Entire Agreement	B-19
6.7	Termination	B-19
6.8	Governing Law	B-19
6.9	Successors and Assigns; Beneficiaries	B-20
6.10	Consent to Jurisdiction; Specific Performance; Waiver of Jury Trial	B-20
6.11	Further Assurances; Company Logo	B-20
6.12	Competitive Opportunity	B-20
6.13	Recapitalization, Exchange, Etc. Affecting the Shares	B-21
6.14	No Recourse	B-21

B-i

### **EXHIBITS AND SCHEDULES**

Exhibit A: Form of Joinder Agreement

Exhibit B: Registration Rights Agreement

Schedule 1: Specified Person

B-ii

# PRINCIPAL STOCKHOLDERS AGREEMENT

This Principal Stockholders Agreement (this "<u>Agreement</u>") is made as of [·], 2017 by and among Solstice Sapphire Investments, Inc., a Delaware corporation (the "<u>Company</u>"), Heritage PE (OEP) II, L.P., a Cayman Islands exempted limited partnership ("<u>OEP II</u>"), Heritage PE (OEP) III, L.P., a Cayman Islands exempted limited partnership ("<u>OEP III</u>", and together with OEP II, the "<u>Initial OEP Stockholders</u>"), and any other stockholder who from time to time becomes party to this Agreement by execution of a joinder agreement substantially in the form of <u>Exhibit A</u> (a "<u>Joinder Agreement</u>").

## RECITALS

- A. On the Effective Date, the Company will issue shares of Common Stock to the Initial OEP Stockholders pursuant to the Merger Agreement, subject to the terms and conditions set forth therein.
- B. The parties hereto desire to enter into this Agreement to agree upon the respective rights and obligations after the Effective Time with respect to the securities of the Company then or thereafter issued and outstanding and to be held by the parties to this Agreement and certain matters with respect to their investment in the Company.

### **AGREEMENT**

Now therefore, in consideration of the foregoing, and the mutual agreements and covenants contained herein, the receipt and sufficiency of which are hereby acknowledged, the parties agree as follows:

# SECTION I. DEFINITIONS

- 1.1 <u>Drafting Conventions; No Construction Against Drafter.</u>
- (a) Except where expressly stated otherwise in this Agreement, the following rules of interpretation apply to this Agreement: (i) "either" and "or" are not exclusive and "include," "includes" and "including" are not limiting; (ii) "hereof," "hereby," "herein" and "hereunder"

and words of similar import when used in this Agreement refer to this Agreement as a whole and not to any particular provision of this Agreement; (iii) "date of this Agreement" refers to the date set forth in the initial caption of this Agreement; (iv) "extent" in the phrase "to the extent" means the degree to which a subject or other thing extends, and such phrase does not mean simply "if"; (v) the headings and table of contents included herein are included for convenience only and shall not affect in any way the meaning or interpretation of this Agreement or any provision hereof; (vi) definitions contained in this Agreement are applicable to the singular as well as the plural forms of such terms; (vii) references to a contract or agreement mean such contract or agreement as amended or otherwise supplemented or modified from time to time; (viii) references to a Person are also to its permitted successors and assigns; (ix) references to an "Article," "Section," "Exhibit" or "Schedule" refer to an Article or Section of, or an Exhibit or Schedule to, this Agreement; (x) references to "\$" or otherwise to dollar amounts refer to the lawful currency of the United States; and (xi) references to a federal, state, local or foreign law include any rules, regulations and delegated legislation issued thereunder. If any date on which a party is required to make a payment or a delivery or take an action, in each case, pursuant to the terms hereof is not a Business Day, then such party shall make such payment or delivery or take such action on the next succeeding Business Day. Time shall be of the essence in this Agreement. Unless specified otherwise, the words "party" and "parties" refer only to a party named in this Agreement or one who joins this Agreement as a party pursuant to the terms hereof.

B-1

- (b) The language used in this Agreement shall be deemed to be the language chosen by the parties hereto to express their mutual intent, and no rule of strict construction shall be applied against any party hereto. No summary of this Agreement prepared by any party shall affect the meaning or interpretation of this Agreement.
- 1.2 <u>Defined Terms</u>. Capitalized terms used herein and not otherwise defined herein shall have the respective meanings set forth in the Merger Agreement. The following capitalized terms, as used in this Agreement, shall have the meanings set forth below.

"Affiliate" shall mean with respect to any specified Person, any other Person which, directly or indirectly, controls, is controlled by or is under common control with the specified Person, including if the specified Person is a private equity fund, (i) any general partner of the specified Person and (ii) any investment fund now or hereafter managed by, or which is controlled by or is under common control with, one or more general partners of the specified Person; provided, however, that, for purposes of this Agreement, (A) neither the Company nor any of its Subsidiaries shall be deemed to be an Affiliate of the OEP Stockholders, (B) no OEP Stockholder shall be deemed to be an Affiliate of the Company or any of its Subsidiaries, (C) each OEP Stockholder shall be deemed to be affiliates of each of the OEP Stockholders, provided, however, no other affiliates of JPMorgan Chase & Co. and its controlled Affiliates of the OEP Stockholders other than those Persons described in clauses (i) or (ii). For the purposes of this definition, "control" (including, with its correlative meanings, the terms "controlled by" and "under common control with"), as used with respect to any Person, shall mean the possession, directly or indirectly, of the power to direct, or cause the direction of the management and policies of such Person, whether through the ownership of securities, by contract or otherwise.

"Baseline Amount" shall mean, as of a particular date, the lesser of (i) the number of voting Shares issued to the Initial OEP Stockholders on the Effective Date pursuant to the Merger Agreement, and (ii) the weighted average number of voting Shares held in the aggregate by the Stockholders in the two hundred fifty (250) Business Days prior to such date.

"Beneficial Ownership" by a Person of any securities means that such Person is a beneficial owner of such securities in accordance with Rule 13d-3 adopted by the SEC under the Exchange Act (provided that, for purposes of calculating "Beneficial Ownership" with respect to the restrictions set forth under Sections 3.1 and 4.1 and notwithstanding anything to the contrary in Rule 13d-3, a Person shall additionally be deemed to Beneficially Own any Common Stock or other securities that may be acquired by such Person upon the conversion, exchange or exercise of any warrants, options, rights or other securities convertible into Common Stock or other securities of the Company, whether such acquisition may be made within sixty (60) days or a longer period); provided, however, that, for purposes of this Agreement, the OEP Stockholders shall not be deemed to Beneficially Own any Shares or other securities issued to any Investor Designee by the Company in his capacity as such; and provided, further, that, for purposes of calculating Beneficial Ownership by a Person, Shares Beneficially Owned by such Person shall not be double-counted with Shares Beneficially Owned by such Person's Affiliates and any Group in which such Person is a member. The term "Beneficially Own" shall have a correlative meaning.

"Bylaws" shall mean the Company's bylaws in effect as of the Effective Time, as amended from time to time.

"Charter" shall mean the Company's certificate of incorporation in effect as of the Effective Time, as amended from time to time.

B-2

"Change of Control Transaction" shall mean any of the following occurring after the Effective Date: (i) a recapitalization, merger, share exchange, business combination or similar extraordinary transaction or series of related transactions as a result of which, the Persons that Beneficially Own the voting Shares of the Company (immediately prior to the consummation of such transaction or series of related transactions) would cease to (immediately after consummation of such transaction or series of related transactions) Beneficially Own voting Shares entitling them to vote a majority or more of the voting Shares in the elections of Directors at any annual or special meeting (or, if the Company is not the surviving or resulting entity); (ii) a sale of all or substantially all of the assets the Company (determined on a consolidated basis) in one transaction or series of related transactions; or (iii) the acquisition (by purchase, merger or otherwise) by any Person of Beneficial Ownership of voting Shares of the Company entitling that Person (together with its Affiliates and any Group in which such Person is a member) to vote a majority of the voting Shares, except any acquisition in the open market by any OEP Stockholder of voting Shares permitted by Section 3.1(b)(i).

"Common Stock" shall mean the common shares, par value \$0.001 per share, of the Company.

"Company" shall have the meaning set forth in the preamble and shall include any successor thereto.

"Company Information" shall mean the following Confidential Information: (i) financial information, financial projections and other financial estimates, (ii) Confidential Information shared by an OEP Stockholder as part of the general portfolio information of such OEP Stockholder that does not identify the Company; (iii) Confidential Information that is aggregated as part of such OEP Stockholder's normal internal reporting or review procedures, including those of its parent entities; (iv) valuation projections and such other summary financial ratios and/or multiples calculated by an OEP Stockholder by

reference to Confidential Information (without directly incorporating such Confidential Information), and (v) the number and type of Shares to be distributed in connection with a proposed or planned in-kind distribution and the value of such Shares at the time of distribution.

"Confidential Information" shall mean all information relating to the Company or the business, products, condition (financial or other), operations, assets, liabilities, results of operations, cash flows or prospects of the Company (whether prepared by the Company, its advisors or otherwise) that is delivered, disclosed or furnished by or on behalf of the Company on or after the date hereof, regardless of the manner in which it is delivered, disclosed or furnished.

"Director" shall mean a member of the Board of Directors.

"Effective Date" shall mean the Closing Date.

"Exchange Act" shall mean the Securities Exchange Act of 1934, as amended, and the rules and regulations thereunder.

"Group" shall mean, with respect to a Person, such Person together with any syndicate or group deemed to be a "person" under Section 13(d)(3) of the Exchange Act.

"Independent Director" shall mean, regardless of whether designated by the OEP Stockholders, a person nominated for or appointed to the Board of Directors who, as of the time of determination: (i) is independent for purposes of the Nasdaq Rules and the SEC rules; (ii) is not a current or former director, officer, partner or other senior-level employee (or other employee or consultant within the past 5 years) of, and does not otherwise then receive (and has not at any time received) any material compensation

B-3

from, any OEP Stockholder or any of its Affiliates (but excluding (a) service for or compensation received from any former portfolio company of such OEP Stockholder or its Affiliate, and (b) compensation received in connection with a bona fide underwritten public offering performed on behalf of such OEP Stockholder or its Affiliate); and (iii) does not then serve (and has not at any time within the two (2) years prior to such time served) as a director, officer, employee or consultant of, and does not otherwise then receive (and has not at any time within the two (2) years prior to such time otherwise received) any compensation from, any portfolio company of any OEP Stockholder or any of its Affiliates, or any Person (or Group in which such Person is a member) that owns 15% or more of the total then issued and outstanding Shares of the Company, provided, however, the restrictions in (ii) and (iii) above shall not apply to (1) any person who (x) has served (but is not then serving) as an independent director of any portfolio company of any OEP Stockholder or any of its Affiliates in which such OEP Stockholder and its Affiliates, collectively, do not have (and have not had during the tenure of such person's service as an independent director) a controlling interest or is serving as an independent director on the Board of Genband Inc. as of the date of the Merger Agreement, (y) was not nominated or designated by such OEP Stockholder or any of its Affiliates to serve on the board of directors of such portfolio company, and (z) would otherwise be independent under clause (ii) and (iii) but for his or her past service as an independent director of such portfolio company or (2) the person named on Schedule 1 hereto to the extent such person would otherwise be independent under clause (ii) and (iii) but for his past service as a consultant to any OEP Stockholder or any its Affiliates or any portfolio company thereof; provided further, the Nominating and Corporate Governance Committee may at any time waive the restricti

"Initial Directors" shall have the meaning set forth in the Merger Agreement.

"Merger Agreement" shall mean the Agreement and Plan of Merger by and among Genband Holdings Company, Sonus Networks, Inc., the Company and the other parties thereto, dated as of May 23, 2017.

"Nasdaq Rules" shall mean the Nasdaq Stock Market Rules or other rules of a national securities exchange upon which the Company's Common Stock is listed or to which it is then subject.

"Necessary Action" shall mean, with respect to a specified result, all actions necessary or desirable to cause such result, including (i) attending meetings in person or by proxy for purposes of obtaining a quorum, (ii) voting or providing (or causing the voting or providing of) a written consent or proxy with respect to all Shares then Beneficially Owned, (iii) causing the adoption of resolutions and amendments to the organizational documents of the Company, (iv) executing agreements and instruments and (v) making, or causing to be made, with governmental, administrative or regulatory authorities, all filings, registrations or similar actions that are required to achieve such result.

"New Shares" shall mean any voting Shares of the Company or any of its subsidiaries, including Common Stock, whether authorized or not by the Board of Directors or any committee of the Board of Directors, and rights, options, or warrants to purchase any voting Shares, and securities of any type whatsoever that are, or may become, convertible into any voting Shares; provided, however, that the term "New Shares" shall not include: (i) Shares issued to employees, consultants, officers and directors of the Company, pursuant to any arrangement approved by the Board of Directors or the Compensation Committee of the Board of Directors; (ii) Shares issued as consideration in the acquisition of another business or assets of another Person by the Company by merger or purchase of the assets or shares, reorganization or otherwise; (iii) Shares issued pursuant to any rights or agreements, including convertible securities, options and warrants, provided, that either (x) the Company shall have complied with Section 5.1 with respect to the initial sale or grant by the Company of such rights or agreements or (y) such rights

B-4

or agreements existed prior to the Effective Date (it being understood that any modification or amendment to any such pre-existing right or agreement subsequent to the Effective Date with the effect of increasing the percentage of the Company's fully-diluted Shares underlying such rights agreement shall not be included in this clause (iii)); (iv) Shares issued in connection with any stock split, stock dividend, recapitalization, reclassification or similar event by the Company; (v) warrants issued to a lender in a bona fide debt financing; (vi) Shares registered under the Securities Act that are issued in an underwritten public offering; (vii) any right, option, or warrant to acquire any security convertible into the securities excluded from the definition of New Shares pursuant to clauses (i) through (vi) above; (viii) any issuance by a subsidiary of the Company to the Company or a wholly-owned subsidiary of the Company; and (ix) any issuance as to which the OEP Majority Interest (on behalf of the OEP Stockholders) elects to waive the rights set forth in Section 5.1.

"OEP Majority Interest" shall mean, at any given time, the OEP Stockholders holding a majority of the outstanding Shares held at that specified time by all OEP Stockholders.

"OEP Stockholders" shall mean (i) the Initial OEP Stockholders and (ii) any Permitted Transferee of any Initial OEP Stockholder described in clause (i) of the definition of "Permitted Transferee" (x) which is issued Shares or becomes the Beneficial Owner of any Shares or is Transferred any Shares by any other Person and (y) which becomes a party hereto by executing a Joinder Agreement; provided, however, that no Shares Beneficially Owned by any Investor Designee or officer or employee of the Company or its subsidiaries shall be deemed to be Beneficially Owned by the OEP Stockholders for the purposes of Sections II, III and IV of this Agreement.

"Permitted Transferee" shall mean, with respect to any OEP Stockholder, (i) any Affiliate of such OEP Stockholder or (ii) any direct or indirect member or general or limited partner of such OEP Stockholder that is the Transferee of Shares pursuant to a pro rata distribution of Shares by such OEP Stockholders to its partners or members, as applicable (or any subsequent Transfer of such Shares by the Transferee to another Permitted Transferee), in each case that becomes a party to this Agreement by executing a Joinder Agreement.

"Person" shall mean an individual, corporation, partnership, limited liability company, joint venture, association, trust, unincorporated organization, government (or agency or political subdivision thereof) or any other entity.

"Registration Rights Agreement" shall mean the Registration Rights Agreement, dated as of the date hereof, attached hereto as Exhibit B.

"Regulatory Requirement" shall mean any set of facts or circumstances arising after the date hereof that has resulted, or, based on the advice of legal counsel, would reasonably be expected by JPMorgan Chase & Co. to result, in the Beneficial Ownership by JPMorgan Chase & Co. or its Affiliates of any voting Shares causing (i) a material violation of Applicable Law by JPMorgan Chase & Co. or its Affiliates, (ii) a limitation under Applicable Law that will materially impair the ability of JPMorgan Chase & Co. or any of its Affiliates to operate in the ordinary course business or engage in their respective ordinary course business activities, or (iii) a requirement under Applicable Law that such voting Shares be Transferred to a third Person.

"SEC" shall mean the Securities and Exchange Commission.

"Securities Act" shall mean the Securities Act of 1933, as amended, and the rules and regulations thereunder.

B-5

"Shares" shall mean, at any time, (i) shares of Common Stock and (ii) any other voting equity securities now or hereafter issued by the Company, together with any options thereon and any other shares of stock or other equity securities issued or issuable with respect thereto (whether by way of a stock dividend, stock split or in exchange for or in replacement or upon conversion of such shares or otherwise in connection with a combination of shares, recapitalization, merger, consolidation or other corporate reorganization).

"Stockholders" shall mean the OEP Stockholders and any other stockholders who from time to time become party to this Agreement by execution of a Joinder Agreement.

"Transfer" shall mean any direct or indirect transfer, donation, sale, assignment, pledge, hypothecation, grant of a security interest in or other disposal or attempted disposal (whether by merger, consolidation or otherwise by operation of law) of all or any portion of a security, any interest or rights in a security, or any rights under this Agreement; provided, however, that any Transfer of equity securities of any Person, including as a result of a change of control of such Person, that Beneficially Owns any equity securities of any OEP Stockholder shall not, by itself, be deemed a Transfer of Shares for the purposes of this Agreement, unless the equity securities of any OEP Stockholder constitute such Person's primary asset or such Person was formed in contemplation of such Transfer.

"Transferee" shall mean the recipient of a Transfer.

1.3 <u>Effectiveness.</u> This Agreement, and all rights and obligations hereunder, shall become effective upon the Effective Time. Upon any termination of the Merger Agreement prior to the Effective Time, this Agreement shall terminate automatically and shall be of no force or effect.

# SECTION II. BOARD MATTERS; CHIEF EXECUTIVE OFFICER

- 2.1 Board of Directors. From and after the Effective Time:
- (a) <u>Board Composition</u>. For as long as the OEP Stockholders in the aggregate continue to Beneficially Own at least ten percent (10%) of the Shares issued to the Initial OEP Stockholders on the Effective Date pursuant to the Merger Agreement, the board of directors of the Company (the "<u>Board of Directors</u>") shall be composed as follows:
  - (i) The authorized number of directors on the Board of Directors shall be established and remain at nine (9) or such other number approved by the Board of Directors, including the approval of a majority of the Independent Directors.
  - (ii) The OEP Stockholders holding an OEP Majority Interest shall have the right (but not the obligation) to designate as Directors, and the individuals nominated for election as Directors by or at the direction of the Board or a duly-authorized committee thereof shall include, five (5) designees of the OEP Stockholders (the "Investor Designees"), at least two (2) of whom must, in the good faith determination of the Nominating and Corporate Governance Committee, qualify as Independent Directors; provided, that (A) from and after the first time that the OEP Stockholders in the aggregate Beneficially Own less than eighty percent (80%) of the Shares issued to the Initial OEP Stockholders on the Effective Date pursuant to the Merger Agreement, the number of Investor Designees to be designated by the OEP Majority Interest (on behalf of the OEP Stockholders) shall be reduced to four (4) Directors, at least one (1) of whom must, in the good faith determination of the Nominating and Corporate Governance Committee, qualify as an Independent Director; (B) from and after the first time that the OEP Stockholders in the aggregate Beneficially Own less than seventy percent (70%) of the Shares issued to the Initial OEP Stockholders on the Effective Date pursuant to the Merger Agreement, the number of

Investor Designees to be designated by the OEP Majority Interest (on behalf of the OEP Stockholders) shall be reduced to three (3) Directors, at least one (1) of whom must, in the good faith determination of the Nominating and Corporate Governance Committee, qualify as an Independent Director; (C) from and after the first time that the OEP Stockholders in the aggregate Beneficially Own less than fifty percent (50%) of the Shares issued to the Initial OEP Stockholders on the Effective Date pursuant to the Merger Agreement, the number of Investor Designees to be designated by the OEP Majority Interest (on behalf of the OEP Stockholders) shall be reduced to two (2) Directors, neither of whom must qualify as an Independent Director; (D) from and after the first time that the OEP Stockholders in the aggregate Beneficially Own less than thirty percent (30%) of the Shares issued to the Initial OEP Stockholders on the Effective Date pursuant to the Merger Agreement, the number of Investor Designees to be designated by the OEP Majority Interest (on behalf of the OEP Stockholders) shall be reduced to one (1) Director, who need not qualify as an Independent Director; and (E) from and after the first time that the OEP Stockholders in the aggregate Beneficially Own less than ten percent (10%) of the Shares issued to the Initial OEP Stockholders on the Effective Date pursuant to the Merger Agreement, the OEP Stockholders shall have no right to designate any members of the Board of Directors.

- (iii) The Nominating and Corporate Governance Committee shall designate as Directors (A) the Company's then serving Chief Executive Officer (the "CEO Director"), who initially shall be Raymond P. Dolan, and (B) such additional number of designees as constitutes the full Board of Directors.
- (b) <u>Obligation to Vote</u>. For as long as the OEP Stockholders have a right to designate any members of the Board of Directors pursuant to Section 2.1(a):
  - (i) The Company shall take all Necessary Actions within its control to cause the individuals designated in accordance with Section 2.1(a) to be nominated for election to the Board of Directors, shall solicit proxies in favor thereof, and at each meeting of the stockholders of the Company at which Directors are to be elected, shall recommend that the stockholders of the Company elect to the Board of Directors each such individual nominated for election at such meeting.
  - (ii) Each OEP Stockholder shall take all Necessary Actions within its control to vote either (a) all shares affirmatively in favor of the election of each Person nominated to serve as a Director by the Board of Directors (including those designated in accordance with Section 2.1(a)), or (b) in the same proportion as the Shares not Beneficially Owned by OEP Stockholders are voted affirmatively in favor of, or to withhold authority with respect to the election of, each Person nominated to serve as a Director by the Board (including those designated in accordance with Section 2.1(a)).
  - (iii) The Company and each OEP Stockholder shall take all Necessary Actions within its control to (A) effect or cause any removal required pursuant to Section 2.1(f), subject, in the case of a removal pursuant to clause (A) or (D) of Section 2.1(f)(i), to the prior direction or approval of the Nominating and Corporate Governance Committee, and (B) cause an appropriate successor to be elected or appointed to fill such vacancy pursuant to Section 2.1(a)(ii).

B-7

## (c) Nominee Qualifications.

- (i) Each Director shall, at the time of his or her nomination or appointment as a Director and at all times thereafter until such individual ceases to serve as a Director, meet and comply with, in the good faith determination of the Nominating and Corporate Governance Committee, any qualification criteria adopted by the Nominating and Corporate Governance Committee, including without limitation the requirements of Applicable Law, the Nasdaq Rules, the SEC rules and corporate governance policies adopted by the Board of Directors that are consistent with the terms set forth herein.
- (ii) In addition to the criteria set forth in Section 2.1(c)(i), each Independent Director shall, at the time of his or her nomination or appointment as a Director and at all times thereafter until such individual ceases to serve as a Director, qualify as an Independent Director in the good faith determination of the Nominating and Corporate Governance Committee.
- (d) <u>Initial Designees</u>. The initial Investor Designees pursuant to the provisions of Section 2.1(a)(ii) shall be the Initial Directors determined pursuant to Section 2.6(a)(i) of the Merger Agreement and Section 2.6(a)(ii) of the Merger Agreement. The initial designees pursuant to the provisions of Section 2.1(a)(iii) shall be the Initial Directors determined pursuant to Section 2.6(a)(iv) of the Merger Agreement, the Company Chief Executive Officer Designee (as defined in the Merger Agreement) and the Company Chairman Designee (as defined in the Merger Agreement), who shall serve as the initial Chairman of the Board of Directors with power and authority set forth in the Charter and the Bylaws and as otherwise delegated or agreed by the Board of Directors from time to time.
- (e) <u>Procedures for Election</u>. Except as set forth herein, each Director shall be nominated for election and elected or appointed as provided in the Charter and Bylaws.

### (f) Removal and Vacancies.

- (i) Except as provided in Section 2.1(a)(ii) or as required by Applicable Law, the parties hereto agree that no Director designated pursuant to Section 2.1(a)(ii) above may be removed from office unless (A) such Director fails to meet the qualification criteria set forth in Section 2.1(c); (B) in the case of an Investor Designee, such removal is directed or approved by the OEP Majority Interest (on behalf of the OEP Stockholders); (C) in the case of the CEO Director, pursuant to Section 2.1(f)(iii); or (D) in the case of a Director designated pursuant to Section 2.1(a)(iii)(B), such removal is directed or approved by the Nominating and Corporate Governance Committee.
- (ii) If at any time any Director ceases to serve on the Board of Directors (whether due to death, disability, resignation, removal or otherwise), the Person or Persons that designated or nominated such Director pursuant to Section 2.1(a)(ii) or (iii) shall designate or nominate a successor to fill the vacancy created thereby on the terms and subject to the conditions of Section 2.1(a)(ii) or (iii). In the event that the

OEP Stockholders do not, pursuant to Section 2.1(a)(ii), have the right to designate an individual to fill such vacancy, then such vacancy may be filled as provided in the Charter and the Bylaws.

(iii) If for any reason the CEO Director shall cease to serve as the Chief Executive Officer of the Company, the Company shall seek to obtain the immediate resignation of the CEO Director as a Director of the Company contemporaneously with such CEO Director's termination of service to the Company as its Chief Executive Officer. In the event such resignation is not effective within ten (10) days of such termination of service, upon the written request of the OEP Majority Interest (on behalf of the OEP Stockholders), the Company shall call a special meeting of stockholders or seek the written consents of stockholders, in each case to

B-8

approve or consent to the removal of the CEO Director (to the extent permitted by Applicable Law, the Charter and the Bylaws). Any employment agreement between the Company and the Chief Executive Officer of the Company shall contain a requirement that the Chief Executive Officer of the Company resign as the CEO Director contemporaneously with termination of his service as the Chief Executive Officer of the Company. Notwithstanding anything to the contrary in the foregoing, an individual who formerly served as the CEO Director and/or Chief Executive Officer of the Company may be nominated, designated, and/or elected as a Director of the Company other than the CEO Director in accordance with Section 2.1(a) above.

- (iv) In the event that the OEP Stockholders cease to have the right to designate a person to serve as a Director pursuant to Section 2.1(a)(ii), if requested by a majority of the Directors then serving on the Board of Directors (other than any Investor Designees), that number of Directors for which the OEP Stockholders cease to have the right to designate to serve as a Director shall resign within one (1) month or, if earlier, such time as such Director's successor is appointed or elected (provided that, subject to the requirements set forth in Section 2.1(a)(ii), the OEP Majority Interest shall have the authority to select which such particular Director or Directors will resign).
- (g) <u>Expenses</u>. Each Director shall be entitled to reimbursement from the Company for his or her reasonable out-of-pocket expenses (including travel) incurred in attending any meeting of the Board of Directors or any committee thereof or governing body of any subsidiary of the Company or any committee thereof.
- Indemnification; Insurance. The Company shall not alter, in any manner adverse to the Investor Designees, any rights to indemnification and exculpation from liabilities currently afforded to members of the Board of Directors pursuant to the Charter, the Bylaws or any indemnification agreement, in each case, as in effect as of the Effective Time. The Company shall use commercially reasonable efforts to continue to maintain in effect directors' and officers' liability insurance and fiduciary liability insurance with benefits, terms, conditions, retentions and levels of coverage that are at least as favorable, in the aggregate, to the insureds as provided in the Company's existing policies as of the Effective Time. The Company hereby acknowledges that certain Investor Designees may have rights to indemnification, advancement of expenses and/or insurance provided by Persons other than the Company and its subsidiaries (collectively, the "Indemnitors"). The Company hereby agrees that, with respect to an action, suit or proceeding brought against an Investor Designee by reason of the fact that such Investor Designee is or was a director of the Company (A) the Company and its subsidiaries are the indemnitor of first resort (i.e., their obligations to the Investor Designees are primary and any obligation of the Indemnitors to advance expenses or to provide indemnification for the same expenses or liabilities incurred by any Investor Designee are secondary), (B) the Company and its subsidiaries shall be required to advance the full amount of expenses incurred by any Investor Designee and shall be liable for the full amount of all expenses, judgments, penalties, fines and amounts paid in settlement, in each case, to the extent legally permitted and as required by the terms of this Agreement, the Charter, the Bylaws, and certificate of incorporation, certificate of formation, bylaws, limited partnership agreement or limited liability company agreement or comparable organizational documents of any of the Company's subsidiaries (or any other agreement between the Company or any of its subsidiaries and any such Investor Designee related to indemnification), without regard to any rights such Investor Designee may have against the Indemnitors, and, (C) the Company and its subsidiaries irrevocably waive, relinquish and release the Indemnitors from any and all claims against the Indemnitors for contribution, subrogation or any other recovery of any kind in respect thereof. The Company further agrees that no advancement or payment by an Indemnitor on behalf of an Investor Designee with respect to any claim for which such Investor Designee has sought indemnification from the Company or its subsidiaries shall affect the foregoing and the applicable Indemnitor shall have a

B-9

right of contribution and/or be subrogated to the extent of such advancement or payment to all of the rights of recovery of such Investor Designee against the Company and its subsidiaries.

- (i) No <u>Liability for Election of Recommended Directors</u>. To the fullest extent permitted by Applicable Law, neither the OEP Stockholders nor any Affiliate of the OEP Stockholders, shall have any liability as a result of designating an individual for election as a Director for any act or omission by such designated individual in his or her capacity as a Director of the Company, nor shall the OEP Stockholders have any liability as a result of voting for any such designee in accordance with the provisions of this Agreement.
- (j) <u>Eligible OEP Shares</u>. For the purpose of determining the number of the Investor Designees that the OEP Stockholders shall be entitled to designate pursuant to Section 2.1(a)(ii), the calculation of Shares held by the OEP Stockholders shall exclude all Shares acquired by the OEP Stockholders after the Effective Date, except for the Shares acquired by the OEP Stockholders after the Effective Date pursuant to the Preemptive Rights under Section 5.1.

# 2.2 <u>Committees of the Board of Directors.</u>

(a) From and after the Effective Date, the Company shall, and each OEP Stockholder shall use its reasonable best efforts to, cause the Board of Directors to establish and maintain the following committees: (i) an Audit Committee, (ii) a Compensation Committee, (iii) a Nominating and Corporate Governance Committee and (iv) an Integration Committee (it being understood that the Integration Committee will only be maintained for such period of time as may be determined by the Board of Directors). The Board of Directors may also establish and maintain any other committee as the Board of Directors shall determine in its discretion.

- (b) For as long as an Investor Designee meets the criteria of Section 2.1(c)(ii) as an Independent Director, such Investor Designee will be eligible to be appointed and remain on all committees established by the Board of Directors, including the Audit Committee, the Compensation Committee, the Integration Committee (if then in existence) and the Nominating and Corporate Governance Committee; provided that, for the avoidance of doubt, any Investor Designee on the Integration Committee need not be an Independent Director.
- (c) For as long as the OEP Stockholders (i) have the right to designate at least one (1) Director under Section 2.1(a)(ii), and (ii) designate at least one (1) Director who is an Independent Director:
  - (i) The Nominating and Corporate Governance Committee shall be comprised of three (3) Independent Directors (A) who are selected by a majority of the Independent Directors then serving on the Board of Directors, (B) at least two (2) of whom are not Investor Designees, and (C) one (1) of whom who is an Investor Designee.
  - (ii) The Nominating and Corporate Governance Committee shall determine the size and membership of each of the Audit Committee, the Compensation Committee, the Nominating and Corporate Governance Committee and any committee of the Board of Directors established to consider any transaction between any OEP Stockholder or any of its Affiliates, on the one hand, and the Company, on the other hand, provided that such determination shall be subject in all cases to (A) the Company's obligation to comply with any applicable independence requirements under the Nasdaq Rules and SEC rules (and in the case of the Nominating and Corporate Governance Committee, with such Investor Designees otherwise being Independent Directors) and compliance with the requirements of Section 162(m) of the Internal Revenue Code

B-10

to have a compensation committee comprised solely of two (2) or more outside directors; (B) in the case of any committee of the Board of Directors established to consider any transaction between any OEP Stockholder or any of its Affiliates, on the one hand, and the Company, on the other hand, each member thereof being disinterested in the good faith determination of the Nominating and Corporate Governance Committee; and (C) to the extent consistent with the foregoing clause (A), for as long as the OEP Stockholders have the right to designate at least three (3) Directors under Section 2.1(a)(ii), at least one (1) member of each such committee shall be an Investor Designee.

- (d) No OEP Stockholder shall knowingly circumvent the director nominee process established by the Board of Directors' Nominating and Corporate Governance committee in accordance with the terms of this Agreement through proxy solicitations or contests.
- (e) For so long as the Investor Designees that the OEP Stockholders are entitled to designate under Section 2.1(a)(ii) represent at least a majority of the Board of Directors, an Investor Designee designated by the OEP Majority Interest shall be the Chairman of each of the Compensation Committee, the Nominating and Corporate Governance Committee and the Integration Committee. The Investor Designee designated and appointed as the Chairman of the Integration Committee shall not be required to be an Independent Director.
- (f) Each provision of this Section 2.2 shall (unless such provision otherwise expires earlier in accordance with its terms) expire on such date as when the OEP Stockholders no longer have a right to designate any Investor Designees under Section 2.1(a)(ii).

# 2.3 <u>Additional Management Provisions</u>.

- (a) Each OEP Stockholder and the Company agrees and acknowledges that, subject to Applicable Law (including the Investor Designees' fiduciary duties thereunder), the Investor Designees designated by the OEP Majority Interest may not share Confidential Information other than Company Information with the OEP Stockholders and their underlying direct or indirect members or controlling parent entities, or general or limited partners, each of whom have a need to know such information (each such party for purposes of this Section, a "Receiving Party.") and solely to be used in connection with the OEP Stockholders' management of their ownership of the Shares (and for no other purpose). As a condition to sharing such Company Information to a Receiving Party, each OEP Stockholder shall (i) require such Receiving Party to agree to be bound by confidentiality obligations substantially similar to (and no less restrictive than) those set forth in Section 2.3(b) as though it were a party hereto, and (ii) advise any such Receiving Party that such Company Information is being provided subject to limitations upon use and may include material non-public information and that applicable securities laws impose restrictions on trading securities when in possession of such information and on communicating such information to any other person under circumstances in which it is reasonably foreseeable that such person is likely to trade in such securities. For the purposes of this section, the application of internal policies and procedures of JPMorgan Chase & Co. regarding confidentiality shall satisfy the conditions of sharing such Confidential Information under this section 2.3(a).
- (b) Each Receiving Party shall keep all Confidential Information confidential and will not, except as permitted below, without the prior written consent of the Company, disclose any Confidential Information; provided, however, that such Receiving Party may disclose Company Information only to the extent (and in the manner): (i) requested or required by Applicable Law or pursuant to judicial process (by oral questions, interrogatories, requests for information or documents in legal proceedings, subpoena, civil investigative demand or other similar process including pursuant to regulations of any applicable stock exchange), which such Receiving Party shall reasonably promptly

B-11

notify the Company (if legally permitted) of the nature, scope and contents of such disclosure; (ii) required pursuant to a routine examination by any regulatory authority (including self-regulatory authority) not specifically targeted to the Company or the Company Information, which such Receiving Party shall, to the extent practicable and legally permissible, as applicable, (A) reasonably promptly notify the Company of the nature, scope and contents of such disclosure and (B) advise the applicable regulatory authority (including self-regulatory authority) of the confidential nature of such Company Information; (iii) used by such Receiving Party's attorneys, auditors or professional consultants on behalf of the Receiving Party; or (vi) such information is required to be disclosed in connection with any litigation or disputes involving that such Receiving Party. Notwithstanding any other provision hereof, with respect to each Receiving Party, the terms Confidential Information and Company Information shall not include information which (i) is or becomes generally available to the public other than as a result of a disclosure by such Receiving Party in violation of this Section 2.3(b), (ii) was within such Receiving Party's possession on a non-confidential basis prior to it being furnished or disclosed to such Receiving Party by or on behalf of the Company, provided that such Receiving Party did not know that the source of such information was bound by a confidentiality agreement with, or other contractual, legal or fiduciary obligation of confidentiality to, the Company with respect to such information, (iii) becomes available to such Receiving Party from a source other than the Company or

any of its Representatives, <u>provided</u> that such Receiving Party did not know at the time of receipt of such information that the source is bound by a confidentiality agreement with, or other contractual, legal or fiduciary obligation of confidentiality to, the Company with respect to such information, or (iv) is independently developed by or on behalf such Receiving Party without use of the Confidential Information of the Company. An OEP Stockholder shall be responsible for any breach of this Section 2.3(b) by any such Receiving Party to whom such OEP Stockholder provided Company Information to the same extent as if such breach had been committed by such OEP Stockholder.

- (c) The OEP Stockholders and the Company hereby agree, notwithstanding anything to the contrary in any other agreement or at law or in equity, that, to the maximum extent permitted by Applicable Law, when the OEP Stockholders take any action under this Agreement, in their capacity as stockholders of the Company, to give or withhold its consent, the OEP Stockholders shall have no duty (fiduciary or other) to consider the interests of the Company or the other stockholders of the Company and may act exclusively in its own interest; provided, however, that the foregoing shall in no way affect the obligations of the parties hereto to comply with the provisions of this Agreement.
- (d) Each of the parties covenants and agrees to take all Necessary Actions within its control to ensure that the Charter and the Bylaws do not, at any time, conflict with the provisions of this Agreement.
- 2.4 <u>Certain Transactions</u>. For as long as the OEP Stockholders have a right to designate any Investor Designees under Section 2.1(a)(ii), the Company shall not enter into any agreement or transaction (including any Change of Control Transaction) with any OEP Stockholder or any of its Affiliates without obtaining the prior approval of a majority of the Independent Directors.
- 2.5 <u>Removal of Chairman</u>. The initial Chairman of the Board of Directors shall serve during the period beginning on the Effective Date and ending on date of the Company's first annual meeting of stockholders after the Effective Date. The Board of Directors may elect a replacement Chairman of the Board in connection with or following the Company's first annual meeting of stockholders after the Effective Date, subject to the terms of this Agreement, the Charter and Bylaws.

B-12

# SECTION III. STANDSTILL PROVISIONS

# 3.1 <u>Standstill</u>.

(a) Except as expressly permitted herein, no Stockholder nor any of its Affiliates shall: (i) effect, agree, seek or make any proposal or offer with respect to, or announce any intention with respect to or cause or participate in or in any way assist, facilitate or encourage any other Person to effect or seek, directly or indirectly, (A) any acquisition of Beneficial Ownership of any Shares or any security that is convertible into Shares or any assets, indebtedness or businesses of the Company or any of its subsidiaries, (B) any financing of the acquisition of any Shares or any security convertible into Shares, (C) any tender or exchange offer, merger or other business combination involving the Company or any of its subsidiaries or assets of the Company or any of its subsidiaries constituting a significant portion of the consolidated assets of the Company and its subsidiaries, (D) any recapitalization, restructuring, liquidation, dissolution or Change of Control Transaction, or (E) any "solicitation" of "proxies" (as such terms are used in the proxy rules of the SEC) to vote any Shares or any consent solicitation or stockholder proposal, (ii) form, join or in any way participate in "a group" (as defined under the Exchange Act) with respect to the Company or enter into any voting agreement or otherwise act in concert with any Person or Group in respect of any voting Shares, (iii) except in accordance with this Agreement, otherwise act, alone or in concert with others, to seek representation on the Board of Directors (other than, in the case of a Stockholder that is not an OEP Stockholder, pursuant to non-public negotiations or discussions with the Company and the Board of Directors any of the types of matters set forth in clause (i) above; (iv) take any action which would or would reasonably be expected to cause the Company to make a public announcement under Applicable Law regarding any of the types of matters set forth in clause (i) above; (v) enter into any discussions or arrangements with any Person with res

# (b) Section 3.1(a) shall not prohibit or prevent:

- (i) any acquisition of Beneficial Ownership of any Shares, or any security that is convertible into Shares, by any Stockholder or its Affiliates if such acquisition would not result in all Stockholders and their respective Affiliates in the aggregate Beneficially Owning a number of voting Shares that is greater than (A) one hundred twenty percent (120%) of the Baseline Amount, or (B) sixty percent (60%) of the number of voting Shares issued and outstanding as of such date (it being understood that any such acquisition pursuant to this clause (A) of Shares in excess of the number of Shares issued to the Initial OEP Stockholders on the Effective Date pursuant to the Merger Agreement shall be permitted only if (x) prior to any such acquisition, the applicable Stockholder delivers to the Board of Directors written notice of the proposed acquisition and the material terms thereof, (y) such acquisition is made pursuant to open market purchases during Company-approved trading windows, and (z) a majority of the Independent Directors shall have determined not to cause the Company to repurchase a material percentage of Shares during such trading window);
- (ii) any Shares, or any security that is convertible into Shares, issued by the Company to Stockholders or their Affiliates pursuant to any stock split, stock dividend or the like effected by the Company;
- (iii) any acquisition of Beneficial Ownership of any Shares, or any security that is convertible into Shares, by any Stockholder or its Affiliates pursuant to Transfers effected on the Nasdaq Stock Market or other nationally recognized securities exchange following the issuance of any new voting Shares by the Company as consideration in the acquisition of another business or assets of another Person by the Company by merger or purchase of the assets or shares, reorganization or otherwise; <u>provided</u>, <u>that</u> immediately following such acquisition of Shares such Stockholder and its Affiliates, in the aggregate, do not Beneficially Own a percentage

B-13

- (iv) any Shares issued (including pursuant to exercise of stock options granted) to any Investor Designee or any officer or employee of the Company or its subsidiaries in respect of such Director's service on the Board of Directors or such officer's or employee's employment with the Company or its subsidiaries;
- (v) any acquisition of Beneficial Ownership of any Shares by an OEP Stockholder pursuant to the exercise of Preemptive Rights under Section 5.1;
  - (vi) Transfers of Shares permitted by and made in accordance with Section IV;
- (vii) any acquisition of Beneficial Ownership of any Shares, or any security that is convertible into Shares, by any Stockholder or its Affiliates or any other action that would otherwise be prohibited by Section 3.1(a) if approved in advance by a majority of the Independent Directors then serving on the Board of Directors (including pursuant to any merger, acquisition or other transaction that is approved in advance by a majority of the Independent Directors then serving on the Board of Directors);
  - (viii) any transaction, discussions, or arrangements solely between or among a Stockholder and its Affiliates; or
- (ix) any Director who is an Investor Designee from engaging, in his or her capacity as such, in confidential discussions with the Board of Directors regarding one or more transactions that would otherwise be prohibited by Section 3.1(a) so long as such discussions would not reasonably be expected to result in public disclosure by the OEP Stockholders or the Company under Applicable Law, including requirements of the SEC or any applicable stock exchange.
- (c) All of the restrictions set forth in this Section 3.1 shall terminate in respect of the Stockholders and their Affiliates upon the earlier to occur of:
  - (i) the entry by the Company into a definitive agreement with any Person providing for a Change of Control Transaction;
  - (ii) such date as when the OEP Stockholders no longer have a right to designate any Investor Designees under Section 2.1(a) (ii).
- (d) Notwithstanding anything to the contrary in Section 6.2 and Section 6.7, the provisions of this Section III may not be terminated, amended or modified unless such termination, amendment or modification is approved by (i) at least six (6) Directors, or at least two-thirds of the members of the Board of Directors at such time does not have nine (9) Directors, and (ii) a majority of the Independent Directors.
- 3.2 <u>Nonapplicability to Certain Affiliates</u>. Notwithstanding anything in this Agreement to the contrary, neither the standstill restrictions in Section 3.1(a) nor any other provision of this Agreement nor any other agreement between any of the OEP Stockholders or their Affiliates, on the one hand, and the Company or its subsidiaries, on the other, shall in any way restrict, prohibit or otherwise restrain

B-14

JPMorgan Chase & Co. and its Affiliates from operating in the ordinary course of business or engaging in their respective ordinary course business activities, whether through its corporate investment banking division, or asset and wealth management division, or otherwise.

3.3 <u>Nonintervention by Company</u>. The Company shall not, and shall not permit any of its subsidiaries to, take any action that would directly impair the ability of the Stockholders or their Affiliates to exercise their rights under Section 3.1(b).

# SECTION IV. TRANSFER RESTRICTIONS

and

# 4.1 <u>Transfer Restrictions</u>.

- (a) Except as otherwise approved by a majority of the Independent Directors then serving on the Board of Directors, during the period of one hundred and eighty (180) days after the Effective Date (the "<u>Initial Lock-Up Period</u>"), no Stockholder may Transfer any voting Shares that it Beneficially Owns to any Person, other than, in the case of OEP Stockholders, a Permitted Transferee or as may be required as a result of a Regulatory Requirement.
- (b) From and after the end of the Initial Lock-Up Period and until the third (3<sup>rd</sup>) anniversary of the Effective Date, except as otherwise approved by a majority of the Independent Directors then serving on the Board of Directors, no Stockholder may Transfer any voting Shares that it Beneficially Owns to any Person, other than a Permitted Transferee or, in the case of JPMorgan Chase & Co. and its controlled Affiliates, as may be required as a result of a Regulatory Requirement, if:
  - (i) such Transfer (or series of related Transfers) involves more than fifteen percent (15%) of the then outstanding voting Shares; or
  - (ii) such Transferee (together with its Affiliates), to the knowledge of such Stockholder, would Beneficially Own, after giving effect to such Transfer (or series of related Transfers), more than fifteen percent (15%) of the then outstanding voting Shares; provided, that the foregoing restriction shall not apply to a Transfer made by such Stockholder in a block trade (without knowledge by such Stockholder of the identity of the ultimate Transferee at the time of such Transfer) to a broker-dealer that is instructed by such Stockholder to comply with the Transfer restrictions of this Section IV with respect to any subsequent Transfer by such broker-dealer to the ultimate Transferee.
- (c) Until such time as when the OEP Stockholders no longer have a right to designate any Investor Designees under Section 2.1(a)(ii), and except as otherwise approved by a majority of the Independent Directors then serving on the Board of Directors, no OEP Stockholder may Transfer pursuant to Rule 144 under the Securities Act (other than in a privately negotiated sale, including block trades) any voting Shares that it Beneficially Owns if after giving effect to such Transfer (or series of related Transfers) such Stockholder (together with its Affiliates), in the aggregate, would Transfer more than one percent (1%) of the then outstanding voting Shares in any one (1) calendar quarter. For the avoidance of doubt, the restrictions in this Section 4.1(c) do

not apply to a bona fide public offering pursuant to an effective registration statement filed under the Securities Act, including pursuant to the Registration Rights Agreement.

(d) The restrictions in Section 4.1(b) shall not apply with respect to any Transferee that agrees in writing to (and to cause any subsequent Transferee to) be bound by, and comply with, the terms and conditions of Sections 3.1 of this Agreement by executing a Joinder Agreement with respect to such Section or, in the alternative, enters into a separate agreement with the Company (the "Standstill Agreement") that contains standstill restrictions that are at least as favorable to the Company as the terms of and conditions of Section 3.1; provided, however, that the terms and conditions of Section 3.1

B-15

or the corresponding requirements of the Standstill Agreement, as applicable, shall terminate and no longer apply with respect to any such Transferee (or subsequent Transferee, as the case may be) upon the earlier of (i) the third (3<sup>rd</sup>) anniversary of the Effective Date and (ii) such date as when such Transferee (or subsequent Transferee, as the case may be) and its Affiliates, in the aggregate, Beneficially Own not more than fifteen percent (15%) of the then outstanding voting Shares. Upon notice given by the OEP Stockholder to the Company that the OEP Stockholder is exploring a potential transfer of its Shares, which notice shall be given to the Company at least fifteen (15) days prior to the proposed Transfer, the Company shall negotiate in good faith and use its commercially reasonable efforts to enter into a customary confidentiality agreement with any potential Transferee identified by the OEP Stockholder to the Company and to enter a Standstill Agreement and such other related and customary transfer documentation with a Transferee (as applicable).

- (e) Notwithstanding any restrictions in Section 4.1 (but subject to Section 2.4 and Section 4.2), each Stockholder shall be permitted to tender any voting Shares it Beneficially Owns pursuant to a public tender offer made to all holders of Shares so long as a majority of the Independent Directors then serving on the Board of Directors has recommended to the holders of Shares that they accept such tender offer and tender their Shares in such tender offer.
- 4.2 <u>Change of Control Transactions</u>. The OEP Stockholders shall not enter into any definitive agreement with any Person providing for a Change of Control Transaction or participate in or in any way support, assist, facilitate or encourage any other Person to effect or seek, directly or indirectly, a Change of Control Transaction, including by Transferring any Shares in connection with a public tender or similar takeover offer made to all holders of Shares for all Shares, in each case, if as a result of such Change of Control Transaction the OEP Stockholders or their Affiliates would receive per Share consideration in excess of the per Share consideration to be received by the other holders of Shares (<u>provided</u>, <u>however</u>, that if the holders of Shares are granted the right to elect to receive one of two or more alternative forms of consideration, the foregoing provision shall be deemed satisfied if each holder of Shares is granted identical election rights), except as otherwise approved by a majority of the Independent Directors then serving on the Board of Directors.
  - 4.3 <u>Legend</u>. Each OEP Stockholder consents to the placement of the following legend on any Certificate representing Shares:

"THE SALE OR OTHER DISPOSITION OF ANY OF THE SHARES REPRESENTED BY THIS CERTIFICATE IS RESTRICTED BY THE PRINCIPAL STOCKHOLDERS AGREEMENT, DATED AS OF [ ], 2017, AS AMENDED FROM TIME TO TIME, AMONG CERTAIN OF THE STOCKHOLDERS OF THIS CORPORATION AND THIS CORPORATION (THE "AGREEMENT"). A COPY OF THE AGREEMENT IS AVAILABLE FOR INSPECTION DURING NORMAL BUSINESS HOURS AT THE PRINCIPAL EXECUTIVE OFFICE OF THIS CORPORATION."

The Company may also place stop-transfer instructions in respect of such Shares with respect to such legend.

# SECTION V. PREEMPTIVE RIGHTS

# 5.1 <u>Preemptive Rights</u>.

(a) For as long as the OEP Stockholders have a right to designate two (2) or more Investor Designees under Section 2.1(a)(ii), each OEP Stockholder shall have the right to purchase, in accordance with the procedures set forth herein, its pro rata portion, calculated based on the percentage

B-16

of the total issued and outstanding voting Shares owned by such OEP Stockholder immediately prior to issuance of the New Shares ("Pro Rata Portion") of any New Shares that the Company may, from time to time, propose to sell and issue (hereinafter referred to as the "Preemptive Right").

- (b) In the event that the Company proposes to issue and sell New Shares, the Company shall notify the OEP Stockholders in writing with respect to the proposed New Shares to be issued (the "New Shares Notice"). Each New Shares Notice shall set forth: (i) the number of New Shares proposed to be issued by the Company and the purchase price therefor; (ii) each OEP Stockholder's Pro Rata Portion of such New Shares; and (iii) any other material term (including, if known, the expected date of consummation of the purchase and sale of the New Shares).
- (c) The OEP Stockholders shall be entitled to exercise their right to purchase New Shares by delivering an irrevocable written notice to the Company within twenty (20) days from the date of receipt of any such New Shares Notice specifying the number of New Shares to be subscribed, which in any event can be no greater than each OEP Stockholder's Pro Rata Portion of such New Shares at the price and on the terms and conditions specified in the New Shares Notice.
- (d) If any OEP Stockholder does not elect within the applicable notice period described above to exercise its Preemptive Rights with respect to any of the New Shares proposed to be sold by the Company, the Company shall have one hundred twenty (120) days after expiration of such notice period to sell such unsubscribed New Shares proposed to be sold by the Company, at a price and on terms no more favorable to the purchaser than those set forth in the New Shares Notice. If the Company does not consummate the sale of the unsubscribed New Shares in accordance with the terms of the New Shares Notice within such one hundred twenty (120)-day period, then the Company may not issue or sell such New Shares unless it sends a new New Shares Notice and once again complies with the provisions of this Section 5.1 with respect to such New Shares. A failure by any OEP Stockholder to exercise its Preemptive Rights with respect to any of the New Shares shall not waive such OEP Stockholder's Preemptive Rights with respect to future issuances of the New Shares.

(e) Each OEP Stockholder shall take up and pay for any New Shares that such OEP Stockholder has elected to purchase pursuant to the Preemptive Right upon closing of the issuance of the New Shares, and shall have no right to acquire such New Shares if the issuance thereof is not consummated.

# SECTION VI. MISCELLANEOUS PROVISIONS

- Reliance. Each covenant and agreement made by a party in this Agreement or in any certificate, instrument or other document delivered pursuant to this Agreement is material, shall be deemed to have been relied upon by the other parties and shall remain operative and in full force and effect after the Effective Date regardless of any investigation. This Agreement shall not be construed so as to confer any right or benefit upon any Person other than the parties hereto and their respective successors and permitted assigns.
- Amendment and Waiver; Actions of the Board. Any party may waive in writing any provision hereof intended for its benefit. Except as provided in this Agreement, no action taken pursuant to this Agreement, including any investigation by or on behalf of any party, or delay or omission in the exercise of any right, power or remedy accruing to any party as a result of any breach or default hereunder by any other party shall be deemed to constitute a waiver by the party taking such action of compliance with any representations, warranties, covenants or agreements contained in this Agreement. The waiver by any party hereto of a breach of any provision hereunder shall not operate or be construed as a waiver of any prior or subsequent breach of the same or any other provision hereunder. The remedies provided

B-17

for herein are cumulative and are not exclusive of any remedies that may be available to any party at law or in equity or otherwise. This Agreement may be amended only with the prior written consent of the OEP Majority Interest and the Company, including the approval of a majority of the Independent Directors. Any consent given as provided in the preceding sentence shall be binding on all parties. Further, with the prior written consent of the OEP Majority Interest, at any time hereafter Permitted Transferees may be made parties hereto, with any such additional parties shall be treated as "OEP Stockholders" or "Stockholders", as applicable, for all purposes hereunder, by executing a counterpart signature page in the form attached as Exhibit A hereto, which signature page shall be attached to this Agreement and become a part hereof without any further action of any other party hereto. Notwithstanding anything to the contrary in the foregoing sentences of this Section 6.2, (a) no provision of this Agreement that requires approval by any specified number of Directors or Independent Directors or portion of the Board of Directors, as applicable, and (b) without limitation of the foregoing, no provision of this Agreement may be amended, modified or waived without the approval of a majority of the Independent Directors.

6.3 Notices. All notices and other communications hereunder shall be in writing and shall be deemed to have been duly delivered and received hereunder (a) four (4) Business Days after being sent by registered or certified mail, return receipt requested, postage prepaid, (b) one (1) Business Day after being sent for next Business Day delivery, fees prepaid, via a reputable nationwide overnight courier service, or (c) immediately upon delivery by hand or by facsimile or email (with a written or electronic confirmation of delivery), if sent during normal business hours of the recipient, then on the recipient's next Business Day, in each case to the intended recipient as set forth below:

If to the Company:

Solstice Sapphire Investments, Inc. 4 Technology Park Drive Westford, MA 01886
Attention: Jeffrey M. Snider Facsimile: 978 614 8101
Email: jsnider@sonusnet.com

With a copy (which shall not constitute notice):

Wilmer Cutler Pickering Hale and Dorr LLP 60 State Street Boston, MA 02109

Attention: Jay E. Bothwick Joseph B. Conahan

Facsimile: (617) 526-5000

Email: jay.bothwick@wilmerhale.com; joseph.conahan@wilmerhale.com

If to the OEP Stockholders:

c/o OEP Parent LLC 270 Park Avenue, 47<sup>th</sup> Floor New York, NY 10017 Attn: Richard W. Smith

B-18

Facsimile: [ · ]

Email: rick.w.smith@jpmorgan.com

With a copy (which shall not constitute notice):

885 Third Avenue New York, New York 10022 Attention: David S. Allinson Thomas J. Malone

Facsimile: (212) 751-4864 Email: [ · ]

If to any other Stockholder, at such Stockholder's address for notice as set forth in the books and records of the Company, or, as to each of the foregoing, at such other address as shall be designated by a party in a written notice to other parties complying as to delivery with the terms of this Section 6.3.

- 6.4 <u>Counterparts</u>. This Agreement may be executed in two or more counterparts, and delivered via facsimile, .pdf or other electronic transmission, each of which shall be deemed an original, but all of which together shall constitute one and the same agreement. Each counterpart may consist of a number of copies hereof each signed by less than all, but together signed by all of the parties hereto.
- Remedies; Severability. It is specifically understood and agreed that any breach of the provisions of this Agreement by any party will result in irreparable injury to the other parties, that the remedy at law alone will be an inadequate remedy for such breach, and that, in addition to any other legal or equitable remedies which they may have, such other parties shall be entitled to enforce their respective rights by bringing actions for specific performance or injunctive relief. If any provision (or part thereof) of this Agreement is invalid, illegal or unenforceable, that provision (or part thereof) will, to the extent possible, be modified in such a manner as to be valid, legal and enforceable but so as to retain most nearly the intent of the parties as expressed herein, and if such a modification is not possible, that provision (or part thereof) will be severed from this Agreement, and in either case the validity, legality and enforceability of the remaining provisions (or parts thereof) of this Agreement will not in any way be affected or impaired thereby. If any provision (or part thereof) of this Agreement is so broad as to be unenforceable, the provision (or part thereof) shall be interpreted to be only so broad as is enforceable.
- 6.6 Entire Agreement. This Agreement, the Exhibits and any documents delivered by the parties in connection herewith constitute the entire agreement among the parties with respect to the subject matter hereof and supersede all prior agreements and understandings among the parties with respect thereto. No addition to or modification of any provision of this Agreement shall be binding upon any party hereto unless made in writing and signed by all parties hereto.
- 6.7 <u>Termination</u>. This Agreement shall remain in effect until the earlier of (i) termination by written agreement of the OEP Majority Interest and the Company, including the approval of a majority of the Independent Directors, and (ii) such date as the OEP Stockholders, in the aggregate, Beneficially Own less than two percent (2%) of the issued and outstanding Shares.
- 6.8 <u>Governing Law</u>. This Agreement and the rights and obligations of the parties hereto shall be governed by and construed in accordance with the laws of the State of Delaware, without giving effect to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware.

B-19

- 6.9 <u>Successors and Assigns; Beneficiaries</u>. No party hereto may assign this Agreement, or any of its rights or obligations under this Agreement, to any Person without the prior written consent of the other parties. Subject to the foregoing sentence, this Agreement shall be binding upon and inure to the benefit of the parties and the respective successors and permitted assigns of the parties as contemplated herein.
  - 6.10 Consent to Jurisdiction; Specific Performance; WAIVER OF JURY TRIAL.
- (a) Each of the parties hereby irrevocably agrees that any legal action or proceeding with respect to this Agreement, the Transactions, and the rights and obligations arising hereunder, or for recognition and enforcement of any judgment in respect of this Agreement, the Transactions, and the rights and obligations arising hereunder brought by the other party hereto or its successors or assigns, shall be brought and determined exclusively in the Delaware Court of Chancery and any state appellate court therefrom within the State of Delaware (or, if the Delaware Court of Chancery declines to accept jurisdiction over a particular matter, any state or federal court within the State of Delaware) (collectively with Delaware Court of Chancery, the "Delaware Courts"). Each of the parties hereto further agrees not to commence any litigation relating to this Agreement or the Transactions except in the Delaware Courts, waives any objection to the laying of venue of any such litigation in the Delaware Courts and agrees not to plead or claim in any Delaware Court that such litigation brought therein has been brought in an inconvenient forum. The choice of forum set forth in this Section shall not be deemed to preclude the enforcement of any judgment of a Delaware federal or state court, or the taking of any action under this Agreement to enforce such a judgment, in any other appropriate jurisdiction.
- (b) EACH PARTY TO THIS AGREEMENT IRREVOCABLY WAIVES THE RIGHT TO A TRIAL BY JURY IN CONNECTION WITH ANY MATTER ARISING OUT OF THIS AGREEMENT AND, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY DEFENSE OR OBJECTION IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY PROCEEDING UNDER THIS AGREEMENT BROUGHT IN THE DELAWARE COURTS AND ANY CLAIM THAT ANY PROCEEDING UNDER THIS AGREEMENT BROUGHT IN ANY SUCH COURT HAS BEEN BROUGHT IN AN INCONVENIENT FORUM. EACH PARTY TO THIS AGREEMENT CERTIFIES THAT IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT OR INSTRUMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS SET FORTH ABOVE IN THIS SECTION 6.10. NO PARTY HAS IN ANY WAY AGREED WITH OR REPRESENTED TO ANY OTHER PARTY THAT THE PROVISIONS OF THIS SECTION WILL NOT BE FULLY ENFORCED IN ALL INSTANCES.
- 6.11 <u>Further Assurances; Company Logo</u>. At any time or from time to time after the Effective Date, the parties hereto agree to cooperate with each other, and at the request of any other party, to execute and deliver any further instruments or documents and to take all such further action as any other party may reasonably request in order to evidence or effectuate the provisions of this Agreement and to otherwise carry out the intent of the parties hereunder. The Company hereby grants the OEP Stockholders and their respective Affiliates permission to use the Company's and its subsidiaries' name and logo in marketing materials.
- 6.12 <u>Competitive Opportunity</u>. If any Stockholder or any of its Affiliates acquires knowledge of a potential transaction or matter which may be an investment or business opportunity or prospective economic or competitive advantage in which the Company could have an interest or expectancy (a "<u>Competitive Opportunity</u>") or otherwise is then exploiting any Competitive Opportunity, then, except with respect to any Competitive Opportunity described

n the following sentence of this Section 6.12, the Company shall have no interest in, and no only such interest or expectation being hereby renounced so that each Stockholder (other than	
B-20	
stockholder who is bound by any employment, consulting, non-competition or other agreement ommunicate or present such Competitive Opportunity to the Company and (ii) have the right stockholder's (and its agents', partners' or Affiliates') own account and benefit or to recomm to Persons other than the Company or any Affiliate of the Company. Notwithstanding the forein Investor Designee and to the extent that the Company identifies a Competitive Opportunity DEP Stockholders did not have knowledge of prior to receipt of such notice, (ii) the Board of Board of Directors determines the Company has or is reasonably capable of obtaining the requirement of such Investor Designee, and such Investor Designee shall not assist any OEP States as the Company ceases to pursue such Competitive Opportunity. Notwithstanding anything greement, none of the provisions of this Agreement or any other agreement shall in any way in their businesses unrelated to the Company and its Subsidiaries or in making investments.	to hold any such Competitive Opportunity for such end, assign or otherwise transfer such Competitive Opportunity going, as long as the OEP Stockholders have a right to designate to an Investor Designee that (i) the Investor Designee and the Directors resolves to cause the Company to pursue, and (iii) the usite funding to pursue, then no OEP Stockholder may seek the ockholder, in pursuing such Competitive Opportunity until such ing to the contrary contained in this Agreement or any other
6.13 <u>Recapitalization, Exchange, Etc. Affecting the Shares</u> . The provisions of the espect to any and all Shares or equity securities of any successor or assign of the Company (or or or otherwise) that may be issued in respect of, in exchange for, or in substitution or ividends, splits, reverse splits, combinations, recapitalizations, and the like occurring after the	whether by merger, consolidation, sale of assets, conversion to a of, the Shares and shall be appropriately adjusted for any
6.14 No Recourse. Notwithstanding anything that may be expressed or implied gree and acknowledge that no recourse under this Agreement or any document or instrument gainst any current or future director, officer, employee, agent, general or limited partner or not so such, whether by the enforcement of any assessment or by any legal or equitable proceeding aw, it being expressly acknowledged that no personal liability whatsoever shall attach to, be irrector, officer, employee, agent, general or limited partner or member of any Stockholder on the stockholder under this Agreement or any documents or instruments delivered in connecting reason of such obligations or their creation.	delivered in connection with this Agreement shall be had tember of any Stockholder or any Affiliate or assignee thereof, g, or by virtue of any statute, regulation or other Applicable imposed upon or otherwise be incurred by any current or future any Affiliate or assignee thereof, as such, for any obligation of
(SIGNATURE PAGE FOLLO	WS)
B-21	
IN WITNESS WHEREOF, the parties are signing this Principal Stockholders Agree	nent as of the date first set forth above.
SOLS	TICE SAPPHIRE INVESTMENTS, INC.
Ву:	Name: Susan Villare Title: President and Treasurer
Signature Page to Principal Stockholder	s Agreement
<u>INITIAL OE</u>	P STOCKHOLDERS:
HERI	TAGE PE (OEP) II, L.P.
Ву:	Name: Title:
HERI	TAGE PE (OEP) III, L.P.
Ву:	Name: Title:

# EXHIBIT A

Form of Joinder Agreement		
(OEP) II, L.P., a Cayman Isl	re page, [ ] hereby agrees to become a party to, and to be bound by the obligations of, and receive the benefits of, that rs Agreement, dated as of [ $\cdot$ ], 2017, by and among Solstice Sapphire Investments, Inc., a Delaware corporation, Heritage PE ands exempted limited partnership, Heritage PE (OEP) III, L.P., a Cayman Islands exempted limited partnership, and certain other ended from time to time thereafter.	
	[NAME]	
	Ву:	
	Name: Title:	
	Titte.	
	Notice Address:	
	EXHIBIT B	
	Registration Rights Agreement	
	See attached.	
	EXHIBIT C AGREED FORM	
	REGISTRATION RIGHTS AGREEMENT	
	among	
	SOLSTICE SAPPHIRE INVESTMENTS, INC.	
	AND	
	THE OEP STOCKHOLDERS	
	Dated as of [·], 2017	
	TABLE OF CONTENTS	
ARTICLE I GENER	AL PROVISIONS C-1	
SECTION 1.01	Defined Terms C-1	
SECTION 1.02	Interpretation C-4 Effectiveness C-4	
SECTION 1.03	Effectiveness C-4	
	RATION RIGHTS C-4	
SECTION 2.01	Demand Registration C-4	
SECTION 2.02 SECTION 2.03	Piggyback Rights C-7 Shelf Registration C-9	
SECTION 2.03	Withdrawal Rights C-10	
SECTION 2.05	Lock-up Agreements C-11	
SECTION 2.06	Registration Procedures C-12	
SECTION 2.07	Registration Expenses C-18	
SECTION 2.08	Miscellaneous C-18	
SECTION 2.09	Indemnification C-18	

C-22

C-22

ARTICLE III

SECTION 3.01

MISCELLANEOUS PROVISIONS

Termination

SECTION 3.02	Notices	C-22
SECTION 3.03	Severability	C-22
SECTION 3.04	Entire Agreement	C-23
SECTION 3.05	Amendments	C-23
SECTION 3.06	Waivers	C-23
SECTION 3.07	Assignment	C-23
SECTION 3.08	Benefit	C-23
SECTION 3.09	Governing Law; Consent to Jurisdiction	C-23
SECTION 3.10	Counterparts	C-24
SECTION 3.11	Enforcement of Agreement	C-24

This REGISTRATION RIGHTS AGREEMENT (this "<u>Agreement</u>"), dated as of [·] 2017, is by and among Solstice Sapphire Investments, Inc., a Delaware corporation (the "<u>Company</u>"), Heritage PE (OEP) II, L.P., a Cayman Islands exempted limited partnership ("<u>OEP II</u>"), and Heritage PE (OEP) III, L.P., a Cayman Islands exempted limited partnership (together with OEP II, the "<u>OEP Stockholders</u>"), and any other stockholder who from time to time becomes a party to this Agreement by execution of a joinder agreement in the form of <u>Exhibit A</u> hereto (a "<u>Joinder Agreement</u>") in accordance with Section 3.07 (collectively, the "<u>Stockholders</u>").

WHEREAS, on the Closing Date, the Company will issue Shares to the OEP Stockholders pursuant to the Merger Agreement (as defined below);

NOW, THEREFORE, the parties hereby agree as follows:

#### ARTICLE I

### **GENERAL PROVISIONS**

SECTION 1.01 <u>Defined Terms.</u> (a) Capitalized terms used herein and not otherwise defined herein shall have the respective meanings set forth in the Merger Agreement. In this Agreement, the following terms shall have the meanings set forth below:

"Affiliate" means, with respect to any specified Person, any other Person which, directly or indirectly, controls, is controlled by or is under common control with the specified Person, including if the specified Person is a private equity fund, (i) any general partner of the specified Person and (ii) any investment fund now or hereafter managed by, or which is controlled by or is under common control with, one or more general partners of the specified Person; provided, however, that, for purposes of this Agreement, (A) neither the Company nor any of its Subsidiaries shall be deemed to be an Affiliate of the OEP Stockholders, (B) no OEP Stockholder shall be deemed to be an Affiliate of the Company or any of its Subsidiaries, and (C) each OEP Stockholder shall be deemed to be an Affiliate of each other OEP Stockholder. For the purposes of this definition, "control" (including, with its correlative meanings, the terms "controlled by" and "under common control with"), as used with respect to any Person, shall mean the possession, directly or indirectly, of the power to direct, or cause the direction of the management and policies of such Person, whether through the ownership of securities, by contract or otherwise.

"Beneficial Ownership" by a Person of any securities means that such Person is a beneficial owner of such securities in accordance with Rule 13d-3 adopted by the Commission under the Exchange Act. The term "Beneficially Own" shall have a correlative meaning.

"Blackout Period" means (i) the period of any lock-up period that may apply to the Stockholders participating in the registration pursuant to which such Stockholders are not permitted to trade or (ii) in the event that the Board determines in good faith and in its reasonable judgment that the registration would reasonably be expected to materially and adversely affect or materially interfere with any bona fide material financing of the Company or any material transaction (including an acquisition, disposition or recapitalization or change in senior

C-1

management) involving the Company that is under consideration by the Company, a period of up to 100 days from the date such deferral commenced; <a href="mailto:provided">provided</a> such period shall end upon the earlier to occur of (1) the expiration of the 100-day period and (2) upon (x) the filing by the Company of a Form 8-K with respect to such financing or transaction or (y) the cessation of consideration of such financing or transaction by the Company, as reasonably determined by the Company.

"Board" means the board of directors of the Company.

"Commission" means the United States Securities and Exchange Commission or any successor agency.

"Company Equity Securities" means the Shares and any other equity securities of the Company.

"equity security" shall have the meaning given to such term in Rule 405 under the Securities Act.

"Merger Agreement" means the Agreement and Plan of Merger by and among GENBAND Holdings Company, Sonus Networks, Inc., the Company and the other parties thereto, dated as of May 23, 2017.

"<u>Permitted Transferee</u>" means any Transferee in any Transfer of Shares, where such Transfer of such Shares to such Transferee does not constitute a breach or violation of the Stockholders Agreement by the Transferor.

"Person" means an individual, corporation, partnership, limited liability company, joint venture, association, trust, unincorporated organization, government (or agency or political subdivision thereof) or any other entity or group (as defined in Section 13(d) of the Exchange Act).

"Public Offering" means an offering of Company Equity Securities pursuant to an effective registration statement under the Securities Act.

"Registrable Amount" means Registrable Securities representing 2.5% of the Shares outstanding.

"Registrable Securities" means any Shares held by the Stockholders and any other securities issued or issuable with respect to any Share held by a Stockholder, including by way of merger, exchange or similar event. As to any particular Registrable Securities, such securities shall cease to be Registrable Securities of a Stockholder when (i) a registration statement registering the offer and sale of such securities under the Securities Act has been declared effective and such securities have been sold or otherwise Transferred by the holder thereof pursuant to such effective registration statement or (ii) such securities have been sold, or are capable of being sold, by such Stockholder in accordance with Rule 144 (or any successor provision) promulgated under the Securities Act without the restriction as to the number of securities that can be sold during any time period.

C-2

"Securities Act" means the Securities Act of 1933 and the rules and regulations thereunder.

"Stockholders Agreement" means the Principal Stockholders Agreement, dated as of the date hereof, among the Company and the OEP Stockholders.

"Shares" means shares of common stock, par value \$0.0001 per share, of the Company.

"Transfer" means any direct or indirect transfer, donation, sale, assignment, pledge, hypothecation, grant of a security interest in or other disposal or attempted disposal (whether by merger, consolidation or otherwise by operation of law) of all or any portion of a security, any interest or rights in a security, or any rights under the Stockholders Agreement; provided, however, that any Transfer of equity securities of any Person, including as a result of a change of control of such Person, that Beneficially Owns any equity securities of any OEP Stockholder shall not, by itself, be deemed a Transfer of Shares for the purposes of this Agreement, unless the equity securities of any OEP Stockholder constitute such Person's primary asset or such Person was formed in contemplation of such Transfer.

"Transferee" means a Person acquiring Company Equity Securities through a Transfer.

"Transferor" means a Person Transferring any Company Equity Securities.

"<u>Underwritten Offering</u>" means a sale of securities of the Company to an underwriter or underwriters for reoffering to the public.

(b) Each of the following terms is defined in the Section listed opposite such term:

Term	Section
Agreement	Preamble
Company	Preamble
Demand	Section 2.01(a)
Demand Participating Stockholders	Section 2.01(b)
Demand Registration	Section 2.01(a)
Demand Right Holders	Section 2.01(a)
Final Prospectus Filing Date	Section 2.05
Form S-3	Section 2.03(a)
Free Writing Prospectus	Section 2.06(a)(iv)
Joinder Agreement	Preamble
Marketed Underwritten Shelf Offering	Section 2.03(e)
Maximum Amount	Section 2.01(g)
OEP II	Preamble
OEP Stockholders	Preamble
Other Demand Rights	Section 2.02(b)
Other Demanding Sellers	Section 2.02(b)
Piggyback Notice	Section 2.02(a)
Piggyback Registration	Section 2.02(a)
Piggyback Seller	Section 2.02(a)
	5-3

C-3

Section 2.07
Section 2.06(d)
Section 2.01(a)
Section 2.06(a)(i)
Section 2.03(a)
Section 2.03(e)
Section 2.03(a)
Preamble
Section 2.03(e)

SECTION 1.02 Interpretation. Except where expressly stated otherwise in this Agreement, the following rules of interpretation apply to this Agreement: (a) "either" and "or" are not exclusive and "include," "includes" and "including" are not limiting; (b) "hereof," "hereby," "herein" and "hereunder" and words of similar import when used in this Agreement refer to this Agreement as a whole and not to any particular provision of this Agreement; (c) "date of this Agreement" refers to the date set forth in the initial caption of this Agreement; (d) "extent" in the phrase "to the extent" means the degree to which a subject or other thing extends, and such phrase does not mean simply "if"; (e) the headings and table of contents included herein are included for convenience only and shall not affect in any way the meaning or interpretation of this Agreement or any provision hereof; (f) definitions

contained in this Agreement are applicable to the singular as well as the plural forms of such terms; (g) references to a contract or agreement mean such contract or agreement as amended or otherwise supplemented or modified from time to time; (h) references to a Person are also to its permitted successors and assigns; (i) references to an "Article," "Section," or "Exhibit" refer to an Article or Section of, or an Exhibit to, this Agreement; (j) references to "\$" or otherwise to dollar amounts refer to the lawful currency of the United States; and (k) references to a federal, state, local or foreign law include any rules, regulations and delegated legislation issued thereunder. The language used in this Agreement shall be deemed to be the language chosen by the parties hereto to express their mutual intent, and no rule of strict construction shall be applied against any party hereto. No summary of this Agreement prepared by any party shall affect the meaning or interpretation of this Agreement. If any date on which a party is required to make a payment or a delivery pursuant to the terms hereof is not a Business Day, then such party shall make such payment or delivery on the next succeeding Business Day. Time shall be of the essence in this Agreement.

SECTION 1.03 <u>Effectiveness</u>. This Agreement, and all rights and obligations hereunder, shall become effective upon the Closing Date. Upon any termination of the Merger Agreement prior to the Closing Date, this Agreement shall terminate automatically and shall be of no force or effect.

### ARTICLE II

# REGISTRATION RIGHTS

SECTION 2.01 <u>Demand Registration.</u> (a) <u>Registration.</u> Subject to the terms hereof and of the Stockholders Agreement, at any time after the 180th day following the Closing Date, any Stockholder or group of Stockholders holding at least 2.5% of the outstanding Shares

C-4

(collectively, the "<u>Demand Right Holders</u>") shall be entitled to make a written request of the Company (a "<u>Demand</u>" and any Demand Right Holders that makes such written request, the "<u>Requesting Stockholders</u>") for registration under the Securities Act of an amount equal to or greater than the Registrable Amount (a "<u>Demand Registration</u>") and thereupon the Company will, subject to the terms of this Agreement, use its reasonable best efforts to effect, as promptly as reasonably practicable, the registration under the Securities Act of:

- (i) the Registrable Securities which the Company has been so requested to register by the Requesting Stockholders for disposition in accordance with the intended method of disposition stated in such Demand;
- (ii) all other Registrable Securities which the Company has been requested to register pursuant to Section 2.01(b), but subject to Section 2.01(g); and
- (iii) all Shares which the Company may elect to register in connection with any offering of Registrable Securities pursuant to this Section 2.01, but subject to Section 2.01(g);

all to the extent necessary to permit the disposition (in accordance with the intended distribution methods in such request) of the Registrable Securities and the additional Shares, if any, to be so registered.

- (b) <u>Demands; Demand Participation.</u> A Demand shall specify: (i) the aggregate number of Registrable Securities requested to be registered in such Demand Registration, (ii) the intended method of disposition in connection with such Demand Registration, to the extent then known, and (iii) the identity of the Requesting Stockholder(s). Within five Business Days after receipt of a Demand, the Company shall give written notice of such Demand to each other Stockholder that holds any Registrable Securities. Subject to Section 2.01(g), the Company shall include in such registration all Registrable Securities with respect to which the Company has received a written request for inclusion therein within ten Business Days after the Company's notice required by this paragraph has been given (such participating Stockholders, the "<u>Demand Participating Stockholders</u>"). Such written notice shall include the same information included in the written request of the Requesting Stockholder(s) delivered pursuant to this Section 2.01(b).
- (c) <u>Number of Demands.</u> The Demand Right Holders (collectively) shall be entitled to unlimited Demand Registrations during the term of this Agreement.
- (d) Effective Registration Statement. A Demand Registration shall not be deemed to have been effected and shall not count as a Demand (i) unless a registration statement with respect thereto has become effective and has remained effective for a period of at least 120 days (or two years in the case of a Shelf Registration Statement) or such shorter period in which all Registrable Securities included in such registration statement have actually been sold thereunder (provided that such period shall be extended for a period of time equal to the period the holder of Registrable Securities refrains from selling any securities included in the effective registration statement at the request of the Company or the lead or co-managing underwriter(s) pursuant to the provisions of this Agreement), (ii) if, after it has become effective, but before any of the circumstances in clause (i) are satisfied, such registration statement becomes subject to any stop

C-5

order, injunction or other order or requirement of the Commission or other Governmental Authority for any reason, or (iii) if the conditions to closing specified in the purchase agreement or underwriting agreement entered into in connection with such registration statement are not satisfied, other than by reason of some act or omission by such Requesting Stockholders.

- (e) <u>Registration Statement Form.</u> Demand Registrations shall be on such appropriate registration form of the Commission as shall be selected by the Company and shall be reasonably acceptable to the Requesting Stockholder.
- (f) Restrictions on Demand Registrations. The Company shall not be obligated to (i) maintain the effectiveness of a registration statement under the Securities Act, filed pursuant to a Demand Registration, for a period longer than 120 days (or, in the case of a Shelf Registration Statement, two years), or (ii) effect any Demand Registration (A) within 90 days of a "firm commitment" underwritten registration in which all Stockholders holding a Registrable Amount were given piggyback rights pursuant to Section 2.02 (subject to Section 2.02(b)) and at least 80% of the number of Registrable Securities requested by such Stockholders to be included in such registration statement were included, (B) within three months of any other Demand

Registration, or (C) if, in the Company's reasonable judgment, it is not feasible for the Company to proceed with the Demand Registration because of the unavailability of audited financial statements. In addition, the Company shall be entitled to postpone the filing of a registration statement or the facilitation of a registered offering (upon written notice to all Stockholders) in the event of a Blackout Period until the expiration of the applicable Blackout Period. The Company may not postpone the filing of a registration statement or the facilitation of a registered offering more than twice in any period of 12 consecutive months, except if required by Applicable Law; provided that if the Company has previously postponed the filing of a registration statement or the facilitation of a registered offering, the Company may not again postpone the effectiveness of such registration statement until 30 days after the expiration of the previous postponement. If the Company postpones the filing or effectiveness of a registration statement for a Demand Registration, the holders of a majority of Registrable Securities held by the Requesting Stockholder(s) shall have the right to withdraw such Demand in accordance with Section 2.04.

Registration in Demand Registrations. The Company shall not include any securities other than Registrable Securities in a Demand Registration, except (i) for Shares that the Company proposes to sell for its own account and (ii) with the written consent (such consent not to be unreasonably withheld, delayed or conditioned) of Stockholders participating in such Demand Registration that hold a majority of the Registrable Securities in such Demand Registration. If, in connection with a Demand Registration, the lead managing or co-managing underwriter(s) advise(s) the Company, in writing, that, in its opinion, the inclusion of all of the securities, including securities of the Company that are not Registrable Securities, sought to be registered in connection with such Demand Registration would adversely affect the distribution of the Registrable Securities sought to be sold pursuant thereto, then the Company shall include in such registration statement only such securities as the Company is advised by such underwriter(s) can be sold without such adverse effect (the "Maximum Amount") as follows and in the following order of priority:

C-6

- (i) first, the number of Registrable Securities requested to be included in such registration by the Requesting Stockholders up to the Maximum Amount, allocated pro rata among such Requesting Stockholders requesting such registration on the basis of the number of such securities requested to be included by such Stockholders;
- (ii) second, Shares that the Company proposes to sell which, taken together with the Registrable Securities under clause (i) above, do not exceed the Maximum Amount; and
- (iii) third, all other securities of the Company duly requested to be included in such registration statement, pro rata on the basis of the amount of such other securities requested to be included or such other method determined by the Company, to the extent, when taken together with clause (i) and (ii) such number of securities does not exceed the Maximum Amount.
- (h) <u>Selection of Underwriters.</u> In connection with a Demand Registration, the Requesting Stockholder(s) may elect to have Registrable Securities sold in an Underwritten Offering. Anytime that a Demand Registration involves an Underwritten Offering, the Requesting Stockholder(s) may select the investment banker or investment bankers and managers that will serve as lead and co-managing underwriters with respect to the offering of such Registrable Securities, subject to the prior written consent of the Company, which consent shall not be unreasonably withheld, delayed or conditioned. In connection with any Underwritten Offering under this Section 2.01, each Demand Participating Stockholder shall be obligated to accept the terms of the underwriting as agreed upon between the Requesting Stockholder(s) and the lead or co-managing underwriters on terms no less favorable to such Demand Participating Stockholders than the Requesting Stockholders(s). In the event of a disagreement among the Requesting Stockholders, the decision of the Stockholder(s) holding a majority of the Registrable Securities shall govern for purposes of this Section 2.01(h).
- (i) <u>Demand Withdrawal.</u> The Requesting Stockholder or the Requesting Stockholders (with the consent of the Requesting Stockholder(s) holding a majority of the Registrable Securities), as the case may be, shall have the right to withdraw a Demand in accordance with Section 2.04.

SECTION 2.02 <u>Piggyback Rights.</u> (a) Subject to the terms and conditions hereof and the Stockholders Agreement, whenever the Company proposes to register any of its securities under the Securities Act (other than a registration by the Company (i) on a registration statement on Form S-4 or any successor form, a registration statement on Form S-8 or any successor form or (ii) pursuant to Section 2.01 or 2.03) (a "<u>Piggyback Registration</u>"), the Company shall give the Stockholders prompt written notice thereof (but not less than ten Business Days prior to the filing by the Company with the Commission of any registration statement with respect thereto). Such notice (a "<u>Piggyback Notice</u>") shall specify, at a minimum, the number of securities proposed to be registered, the proposed date of filing of such registration statement with the Commission, the proposed means of distribution, the proposed lead or co-managing underwriter(s) (if any and if known), and a good faith estimate by the Company of the proposed minimum offering price of such securities. Upon the written request of a Stockholder (a "<u>Piggyback Seller</u>") (which written request shall specify the number of Registrable Securities

C-7

then presently intended to be disposed of by such Stockholder) given within ten days after such Piggyback Notice is sent to such Stockholder, the Company, subject to the terms and conditions of this Agreement, shall use its reasonable best efforts to cause all such Registrable Securities held by Stockholders with respect to which the Company has received such written requests for inclusion to be included in such Piggyback Registration on the same terms and conditions as the Company's securities being sold in such Piggyback Registration.

- (b) <u>Priority on Piggyback Registrations</u>. If, in connection with a Piggyback Registration, the lead or co-managing underwriter(s) advise(s) the Company, in writing, that, in its opinion, the inclusion of all the securities sought to be included in such Piggyback Registration by the Company, by others who have sought to have Registrable Securities registered pursuant to any rights to demand registration (other than pursuant to so called "piggyback" or other incidental or participation registration rights described herein) (such demand rights being "<u>Other Demand Rights</u>" and such Persons being "<u>Other Demanding Sellers</u>"), by the Piggyback Sellers and by any other proposed sellers, as the case may be, would adversely affect the distribution of the securities sought to be sold pursuant thereto, then the Company shall include in the registration statement applicable to such Piggyback Registration only such securities as the Company is so advised by such lead or co-managing underwriter(s) can be sold without such an effect, as follows and in the following order of priority:
  - (i) if the Piggyback Registration is in connection with an offering for the Company's own account, then (A) first, such number of securities to be sold by the Company as the Company, in its reasonable judgment and acting in good faith and in accordance with sound financial practice, shall have determined, (B) second, Registrable Securities of Piggyback Sellers, pro rata on the basis of the amount of such Registrable

Securities sought to be registered by such Piggyback Sellers, (C) third, other Shares of the Company sought to be registered by the Other Demanding Sellers and (D) fourth, other shares held by any other proposed sellers; and

- (ii) if the Piggyback Registration relates to an offering other than for the Company's own account, then (A) first, such number of Registrable Securities sought to be registered by each Other Demanding Seller, pro rata in proportion to the number of securities sought to be registered by all such Other Demanding Sellers, (B) second, Registrable Securities of Piggyback Sellers pro rata on the basis of the amount of such Registrable Securities sought to be registered by such Piggyback Sellers, (C) third, Shares to be sold by the Company and (D) fourth, other shares of the Company held by any other proposed sellers.
- (c) <u>Terms of Underwriting.</u> In connection with any offering under this Section 2.02 involving an underwriting for the Company's account, the Company shall not be required to include a holder's Registrable Securities in the underwritten offering if, after the Company consults with such holder and considers such holder's positions in good faith, such holder refuses to agree to the terms of the underwriting as agreed upon between the Company and the lead or co-managing underwriter(s) whether secured by the Company or otherwise.
- (d) <u>Withdrawal by the Company.</u> If, at any time after giving written notice of its intention to register any of its securities as set forth in this Section 2.02 and prior to the time the

C-8

registration statement filed in connection with such registration is declared effective, the Company shall determine for any reason not to register such securities, the Company may, at its election, give written notice of such determination to each Stockholder and thereupon shall be relieved of its obligation to register any Registrable Securities in connection with such particular withdrawn or abandoned registration (but not from its obligation to pay the Registration Expenses in connection therewith as provided herein); provided that any participating Demand Right Holders may continue the registration as a Demand Registration pursuant to Section 2.01.

SECTION 2.03 Shelf Registration. (a) Subject to the terms hereof and of the Stockholders Agreement, in connection with a Demand Registration, subject to Section 2.03(d), and further subject to the availability of a registration statement on Form S-3 or any successor form ("Form S-3") to the Company, the Requesting Stockholder(s) making the Demand may by written notice delivered to the Company (the "Shelf Notice") require the Company to file as soon as practicable (but no later than 45 days after the date the Shelf Notice is delivered and no earlier than the 180th day following the Closing Date), and to use reasonable best efforts to cause to be declared effective by the Commission as soon as practicable, a Form S-3 providing for an offering to be made on a continuous or delayed basis pursuant to Rule 415 under the Securities Act relating to the offer and sale, from time to time, of the Registrable Securities Beneficially Owned by such Requesting Stockholder(s) and the other Stockholders holding any Registrable Securities who elect to participate therein as provided in Section 2.03(b) in accordance with the plan and method of distribution set forth in the prospectus included in such Form S-3 (the "Shelf Registration Statement"). In the event of a disagreement among the Requesting Stockholders, the decision of the Stockholder(s) holding a majority of the Registrable Securities shall govern for purposes of this Section 2.03(a).

- (b) Within five Business Days after receipt of a Shelf Notice pursuant to Section 2.03(a), the Company will deliver written notice thereof to each Stockholder holding any Registrable Securities. Each Stockholder may elect to participate in the Shelf Registration Statement in accordance with the plan and method of distribution set forth in such Shelf Registration Statement by delivering to the Company a written request to so participate within ten Business Days after the Shelf Notice is given to any such Stockholders.
- (c) Subject to Section 2.03(d), the Company will use reasonable best efforts to keep the Shelf Registration Statement continuously effective until the earlier of (i) two years after the Shelf Registration Statement has been declared effective and (ii) the date on which all Registrable Securities covered by the Shelf Registration Statement have been sold thereunder in accordance with the plan and method of distribution disclosed in the prospectus included in the Shelf Registration Statement, or otherwise.
- (d) The Company shall be entitled, from time to time, by providing written notice to the Stockholders who elected to participate in the Shelf Registration Statement, to require such Stockholders to suspend the use of the prospectus for sales of Registrable Securities under the Shelf Registration Statement for any Blackout Period. Immediately upon receipt of such notice, the Stockholders covered by the Shelf Registration Statement shall suspend the use of the prospectus until the requisite changes to the prospectus have been made as required below. After the expiration of any Blackout Period and without any further request from a Stockholder, the Company shall as promptly as reasonably practicable prepare a post-effective amendment or

C-9

supplement to the Shelf Registration Statement or the prospectus, or any document incorporated therein by reference, or file any other required document so that, as thereafter delivered to purchasers of the Registrable Securities included therein, the prospectus will not include any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to be stated in order to make the statements made, in light of the circumstances under which they were made, not misleading.

- (e) At any time that a Shelf Registration Statement is effective, if any Demand Right Holder delivers a notice to the Company (a "<u>Take-Down Notice</u>") stating that it intends to sell all or part of its Registrable Securities included by it on the Shelf Registration Statement in an Underwritten Offering (a "<u>Shelf Offering</u>"), then, the Company shall as promptly as reasonably practicable amend or supplement the Shelf Registration Statement as may be necessary in order to enable such Registrable Securities to be distributed pursuant to the Shelf Offering (taking into account, solely in connection with a Marketed Underwritten Shelf Offering (as defined below), the inclusion of Registrable Securities by any other holders pursuant to this Section 2.03(e)). In connection with any Shelf Offering that is an Underwritten Offering and where the plan of distribution set forth in the applicable Take-Down Notice includes a customary "road show" (including an "electronic road show") or other substantial marketing effort by the Company and the underwriters (a "<u>Marketed Underwritten Shelf Offering</u>"):
  - (i) the Company shall forward the Take-Down Notice to all other holders of Registrable Securities included on the Shelf Registration Statement and the Company and such proposing Demand Right Holder shall permit each such holder to include its Registrable Securities included on the Shelf Registration Statement in the Marketed Underwritten Shelf Offering if such holder notifies the proposing Demand Right Holder and the Company within five Business Days after delivery of the Take-Down Notice to such holder; and

(ii) if the lead or co-managing underwriter(s) advises the Company and the proposing Demand Right Holder that, in its opinion, the inclusion of all of the securities sought to be sold in connection with such Marketed Underwritten Shelf Offering would adversely affect the distribution thereof, then there shall be included in such Marketed Underwritten Shelf Offering only such securities as the proposing Demand Right Holder is advised by such lead or co-managing underwriter(s) can be sold without such adverse effect, and such number of Registrable Securities shall be allocated in the same manner as described in Section 2.01(g). Except as otherwise expressly specified in this Section 2.03, any Marketed Underwritten Shelf Offering shall be subject to the same requirements, limitations and other provisions of this Article II as would be applicable to a Demand Registration (i.e., as if such Marketed Underwritten Shelf Offering were a Demand Registration), including Section 2.01(f) and Section 2.01(g).

Notwithstanding anything in this Section 2.03 to the contrary, the Company shall not be required to participate in more than two Marketed Underwritten Shelf Offerings per fiscal year.

SECTION 2.04 <u>Withdrawal Rights.</u> Any Stockholder having notified or directed the Company to include any or all of its Registrable Securities in a registration statement under the Securities Act shall have the right to withdraw any such notice or direction with respect to

C-10

any or all of the Registrable Securities designated by it for registration by giving written notice to such effect to the Company prior to the effective date of such registration statement. In the event of any such withdrawal, the Company shall not include such Registrable Securities in the applicable registration and such Registrable Securities shall continue to be Registrable Securities for all purposes of this Agreement. No such withdrawal shall affect the obligations of the Company with respect to the Registrable Securities not so withdrawn; provided, however, that in the case of a Demand Registration, if such withdrawal shall reduce the number of Registrable Securities sought to be included in such registration below the Registrable Amount, then the Company shall as promptly as practicable give each holder of Registrable Securities sought to be registered notice to such effect and, within ten Business Days following the mailing of such notice, such holder of Registrable Securities still seeking registration shall, by written notice to the Company, elect to register additional Registrable Securities, when taken together with elections to register Registrable Securities by its Affiliates, to satisfy the Registrable Amount or elect that such registration statement not be filed or, if theretofore filed, be withdrawn. During such ten Business Day period, the Company shall not file such registration statement or, if such registration statement has already been filed, the Company shall not seek, and shall use reasonable best efforts to prevent, the effectiveness thereof. Any registration statement withdrawn or not filed (a) in accordance with an election by the Company, (b) in accordance with an election by the Company subsequent to the effectiveness of the applicable Demand registration statement because any post-effective amendment or supplement to the applicable Demand registration statement contains information regarding the Company which the Company deems adverse to the Company, shall not be counted as a Dem

SECTION 2.05 Lock-up Agreements. In connection with any Underwritten Offering, each Stockholder agrees to enter into customary agreements to not effect any public sale or distribution (including sales pursuant to Rule 144) of Company Equity Securities (a) for a Public Offering (other than a Demand Registration or Piggyback Registration), during the period between the date specified by the Company to such Stockholder in its notice of intention to commence a Public Offering (such date to be the Company's best estimate as to the date that is 10 days prior to the date of the filing of the "final" prospectus or "final" prospectus supplement if the Underwritten Offering is made pursuant to a Shelf Registration Statement, the "Final Prospectus Filing Date") and 120 days following the Final Prospectus Filing Date or (b) for a Demand Registration or Piggyback Registration, during the period between the date specified by the Company to such Stockholder in its notice of intention to commence an Underwritten Offering (such date to be the Company's best estimate as to the date that is 10 days prior to the Final Prospectus Filing Date) and 90 days following the Final Prospectus Filing Date. For the avoidance of doubt, the lock-up restrictions pursuant to any underwriting agreement to be entered into with the underwriters shall not exceed the time limits on the lock-up restrictions set forth herein without the written consent (such consent not to be unreasonably withheld, delayed or conditioned) of each Demand Right Holder. The Company also shall cause its executive officers and directors (and managers, if applicable) and shall use its reasonable best efforts to cause other holders of Shares who Beneficially Own any of the Shares participating in such offering (including the Company, if applicable), to enter into lockup agreements that contain restrictions that are no less restrictive than the restrictions contained herein. Notwithstanding anything in this Agreement to the contrary, no provision of this Agreement nor any othe

C-11

agreement between any of the OEP Stockholders or their Affiliates, on the one hand, and the Company or its subsidiaries, on the other, shall in any way restrict, prohibit or otherwise restrain JPMorgan Chase & Co. and its Affiliates from operating in the ordinary course business or engaging in their respective ordinary course business activities, whether through its investment banking division or otherwise.

SECTION 2.06 <u>Registration Procedures.</u> (a) <u>Registration.</u> If and whenever the Company is required to use reasonable best efforts to effect the registration of any Registrable Securities under the Securities Act as provided in Sections 2.01, 2.02, and 2.03, the Company shall as promptly as reasonably practicable:

(i) prepare and file with the Commission a registration statement to effect such registration and thereafter use reasonable best efforts to cause such registration statement to become and remain effective, pursuant to the terms of this Agreement; provided, however, that the Company may discontinue any registration of its securities which are not Registrable Securities at any time prior to the effective date of the registration statement relating thereto; provided, further, however, that at least five Business Days prior to filing any registration statement or any amendments thereto, the Company will furnish to the counsel selected by the holders of Registrable Securities which are to be included in such registration ("Selling Holders") copies of all such documents proposed to be filed, which documents will be subject to the review of such counsel (such review to be conducted with reasonable promptness) and other documents reasonably requested by such counsel, including any comment letter from the Commission, and if reasonably requested by such counsel reasonable opportunity to participate in the preparation of such registration statement and each prospectus included therein and such other opportunities to conduct a reasonable investigation within the meaning of the Securities Act, including reasonable access upon reasonable notice during normal business hours to the Company's books and records, officers, accountants and other advisors, so long as such access or request does not unreasonably disrupt the normal operations of the Company and its subsidiaries. The Company shall not file such registration statement or any amendments thereto if the Selling Holders shall in good faith reasonably object in writing to the filing of such documents, unless, in the good faith opinion of the Company, such filing is necessary to comply with

Applicable Law; <u>provided</u>, however, that the Selling Holders shall (and shall cause their representatives to) keep confidential any such information that is not generally publicly available at the time of delivery of such information;

(ii) prepare and file with the Commission such amendments and supplements to such registration statement and the prospectus used in connection therewith as may be necessary (A) to keep such registration statement effective, (B) to comply with the provisions of the Securities Act with respect to the disposition of all securities covered by such registration statement and (C) to not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to be stated in order to make the statements made, in light of the circumstances under which they were made, not misleading, until the earlier of such time as all of such securities have been disposed of in accordance with the intended methods of disposition by Selling Holders thereof set forth in such registration statement or the expiration of 120 days (or two years

C-12

in the case of a Shelf Registration Statement) after such registration statement becomes effective;

- (iii) if requested by the lead or co-managing underwriters, if any, or the holders of a majority of the then outstanding Registrable Securities being sold in connection with an underwritten offering, as promptly as reasonably practicable include in a prospectus supplement or post-effective amendment such information as the lead or co-managing underwriters, if any, and such holders may reasonably request in order to permit the intended method of distribution of such securities and make all required filings of such prospectus supplement or such post-effective amendment as soon as practicable after the Company has received such request; <u>provided</u>, <u>however</u>, that the Company shall not be required to take any actions under this Section 2.06(a)(iii) that are not, in the good faith opinion of the Company, in compliance with Applicable Law;
- (iv) furnish to each Selling Holder and each underwriter, if any, of the securities being sold by such Selling Holder such number of conformed copies of such registration statement and of each amendment and supplement thereto (in each case including all exhibits), such number of copies of the prospectus contained in such registration statement (including each preliminary prospectus and any summary prospectus) and each free writing prospectus (as defined in Rule 405 of the Securities Act) (a "Free Writing Prospectus") utilized in connection therewith and any other prospectus filed under Rule 424 under the Securities Act, in conformity with the requirements of the Securities Act, and such other documents as such Selling Holder and underwriter, if any, may reasonably request in order to facilitate the public sale or other disposition of the Registrable Securities owned by such Selling Holder;
- (v) use reasonable best efforts to register or qualify or cooperate with the Selling Holders, the underwriters, if any, and their respective counsel in connection with the registration or qualification (or exemption from such registration or qualification) of such Registrable Securities covered by such registration statement under such other securities laws or blue sky laws of such jurisdictions as any Selling Holder and any underwriter of the securities being sold by such Selling Holder shall reasonably request and to keep each such registration or qualification (or exemption therefrom) effective during the period such registration statement is required to be kept effective hereunder, and take any other action which may be reasonably necessary or advisable to enable such Selling Holder and underwriter to consummate the disposition in such jurisdictions of the Registrable Securities owned by such Selling Holder, except that the Company shall not for any such purpose be required to qualify generally to do business as a foreign corporation in any jurisdiction wherein it would not, but for the requirements of this subdivision (v), be obligated to be so qualified, to subject itself to taxation in any such jurisdiction or to file a general consent to service of process in any such jurisdiction;
- (vi) use reasonable best efforts to cause such Registrable Securities to be listed on each securities exchange on which the same securities issued by the Company are then listed;

C-13

- (vii) use reasonable best efforts to cause such Registrable Securities covered by such registration statement to be registered with or approved by such other governmental agencies or authorities as may be necessary to enable the Selling Holder(s) thereof to consummate the disposition of such Registrable Securities;
- (viii) use reasonable best efforts to provide and cause to be maintained a transfer agent and registrar for all Registrable Securities covered by such registration statement from and after a date not later than the effective date of such registration statement;
- (ix) make, in accordance with customary practice and upon reasonable notice during normal business hours, available for inspection by representatives of the Selling Holders, any underwriters and any counsel or accountant retained by the Selling Holders or underwriters all relevant financial and other records, pertinent corporate documents and properties of the Company and cause appropriate officers, managers, employees, outside counsel and accountants of the Company to supply all information reasonably requested by any such representative, underwriter, counsel or accountant in connection with their due diligence exercise, including through in-person meetings, but subject to the recipients of such information executing customary confidentiality agreements and to customary privilege constraints and so long as such access or request does not unreasonably disrupt the normal operations of the Company and its subsidiaries.
  - (x) in connection with an Underwritten Offering, obtain for each Selling Holder and underwriter:
  - (A) an opinion of counsel for the Company, covering the matters customarily covered in opinions requested in underwritten offerings and such other matters as may be reasonably requested by such Selling Holder and underwriters,
  - (B) a "comfort" letter (or, in the case of any such Person which does not satisfy the conditions for receipt of a "comfort" letter specified in Statement on Auditing Standards No. 72, an "agreed upon procedures" letter) signed by the independent public accountants who have certified the Company's financial statements and, to the extent required, any other financial statements included in such registration statement, covering the matters customarily covered in "comfort" or "agreed upon procedures" letters in connection with underwritten offerings; and
  - (C) to the extent requested and customary for the relevant transaction, enter into a securities sales agreement with the Selling Holders providing for, among other things, the appointment of a representative as agent for the Selling Holders for the purpose of soliciting

purchases of shares, which agreement shall be customary in form, substance and scope and shall contain customary representations, warranties and covenants;

(xi) as promptly as reasonably practicable notify, in writing, each Selling Holder and the underwriters, if any, of the following events:

C-14

- (A) the filing of the registration statement, the prospectus or any prospectus supplement related thereto or post-effective amendment to the registration statement or any Free Writing Prospectus utilized in connection therewith, and, with respect to the registration statement or any post-effective amendment thereto, when the same has become effective;
- (B) any request by the Commission or any other U.S. or state-governmental authority for amendments or supplements to the registration statement or the prospectus or for additional information;
- (C) the issuance by the Commission of any stop order suspending the effectiveness of the registration statement or the initiation of any proceedings by any Person for that purpose; and
- (D) the receipt by the Company of any notification with respect to the suspension of the qualification of any Registrable Securities for sale under the securities or blue sky laws of any jurisdiction or the initiation or threat of any proceeding for such purpose;
- (xii) notify each Selling Holder, at any time when a prospectus relating thereto is required to be delivered under the Securities Act, upon discovery that, or upon the happening of any event as a result of which, the registration statement, the prospectus included in such registration statement or any document incorporated or deemed to be incorporated therein by reference, as then in effect, includes an untrue statement of a material fact or omits to state a material fact required to be stated therein or necessary to be stated in order to make the statements therein, not misleading, and, at the request of any Selling Holder, as promptly as reasonably practicable prepare and furnish to such Selling Holder a reasonable number of copies of a supplement to or an amendment of such prospectus as may be necessary so that, as thereafter delivered to the purchasers of such Registrable Securities, such prospectus shall not include an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to be stated in order to make the statements made, in light of the circumstances under which they were made, not misleading;
- (xiii) make every reasonable effort to obtain the withdrawal of any order suspending the effectiveness of the registration statement and to prevent or obtain the lifting of any suspension of the qualification (or exemption from qualification) of any of the Registrable Securities for sale in any jurisdiction as any Selling Holder and any underwriter of the securities being sold by such Selling Holder shall reasonably request at the earliest date reasonably practicable;
- (xiv) otherwise use reasonable best efforts to comply with all applicable rules and regulations of the Commission, and make available to Selling Holders, as soon as reasonably practicable, an earnings statement covering the period of at least 12 months, but not more than 18 months, beginning with the first day of the Company's first full quarter after the effective date of such registration statement, which earnings statement

C-15

shall satisfy the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder;

- (xv) cooperate with each seller of Registrable Securities and each underwriter or agent participating in the disposition of such Registrable Securities and their respective counsel in connection with any filings required to be made with the Financial Industry Regulatory Authority, Inc.;
- (xvi) prior to the date on which the pricing of the relevant offering is expected to occur, provide a CUSIP number for the Registrable Securities;
- (xvii) use its reasonable best efforts to assist Stockholders who made a request of the Company to provide for a third party "market maker" for the Shares; <u>provided</u>, <u>however</u>, that the Company shall not be required to serve as such "market maker"; and
- (xviii) have appropriate officers of the Company prepare and make presentations at any "road shows" and before analysts and rating agencies, as the case may be, and otherwise use its reasonable best efforts to cooperate as reasonably requested by the underwriters in the offering, marketing or selling of the Registrable Securities.
- (b) Agreements. Without limiting any of the foregoing, the Company agrees to, in connection with registration of any Registrable Securities under this Article II, enter into an underwriting agreement in form, scope and substance as is customary in underwritten offerings including indemnification provisions and procedures substantially to the effect set forth in Section 2.09 hereof with respect to all parties to be indemnified pursuant thereto. In connection with any offering of Registrable Securities registered pursuant to this Agreement, the Company shall (i) furnish to the underwriter, if any (or, if no underwriter, the sellers of such Registrable Securities), unlegended (unless otherwise required by Applicable Law) certificates representing ownership of the Registrable Securities being sold under the registration statement, in such denominations and registered in such names as requested by the lead or comanaging underwriters or sellers, (ii) make available to the Company's transfer agent prior to the effectiveness of such registration statement a supply of such certificates and (iii) instruct any transfer agent and registrar of the Registrable Securities to release any stop transfer order with respect thereto.
- (c) Return of Prospectuses. Each Selling Holder agrees that upon receipt of any notice from the Company of the happening of any event of the kind described in clauses (B) through (D) of Section 2.06(a)(xi) or in Section 2.06(a)(xii), such Selling Holder shall forthwith discontinue such Selling Holder's disposition of Registrable Securities pursuant to the applicable registration statement and prospectus relating thereto until such Selling Holder's receipt of the copies of the supplemented or amended prospectus contemplated by Section 2.06(a)(xi) or until it is advised in writing by the Company that the use of the applicable prospectus may be resumed, and has received copies of any additional or supplemental filings that are incorporated or deemed to be incorporated by reference in such prospectus and, if so directed by the Company, deliver to the Company, at the Company's expense, all copies, other than

C-16

notice, any applicable 120 day or two year period during which such registration statement must remain effective pursuant to this Agreement shall be extended by the number of days during the period from the date of giving of a notice regarding the happening of an event of the kind described in Section 2.06(a)(xi) or Section 2.06(a)(xii) to the date when all such Selling Holders shall receive such a supplemented or amended prospectus and such prospectus shall have been filed with the Commission.

- (d) Requested Information. Not less than five Business Days before the expected filing date of each registration statement pursuant to this Agreement, the Company shall notify each Selling Holder of the information, documents and instruments from such Selling Holder that the Company or any underwriter reasonably requests in connection with such registration statement, including to a questionnaire, custody agreement, power of attorney, lock-up letter and underwriting agreement (the "Requested Information"). If the Company has not received, on or before the second day before the expected filing date, the Requested Information from such Selling Holder, the Company may file the registration statement without including Registrable Securities of such Selling Holder. The failure to so include in any registration statement the Registrable Securities of a Selling Holder (with regard to that registration statement) shall not in and of itself result in any liability on the part of the Company to such Selling Holder. In the event that, either immediately prior to or subsequent to the effectiveness of any registration statement, any Stockholder shall distribute Registrable Securities to its stockholders, partners or members, such Stockholder shall so advise the Company and provide such information as shall be necessary or advisable to permit an amendment to such registration statement or supplement to any prospectus to provide information with respect to such stockholders, partners or members, in their capacity as selling security holders (it being understood that no such distribution of any Shares may be effectuated following the pricing of an Underwritten Offering that includes such Shares). As soon as reasonably practicable following receipt of such information, the Company shall file an appropriate amendment to such registration statement or supplement to any prospectus reflecting the information so provided.
  - (e) <u>No Requirement to Participate.</u> Neither the Company nor any Stockholder shall be required to participate in any Public Offering.
- (f) Rule 144. The Company covenants that it will use its reasonable best efforts to (i) file in a timely fashion the reports required to be filed by it under the Securities Act and the Exchange Act and the rules and regulations adopted by the Commission thereunder (or, if it is not required to file such reports, it will, upon the request of any holder of Registrable Securities, make publicly available other information so long as necessary to permit sales in compliance with Rule 144 under the Securities Act), (ii) furnish to any holder of Registrable Securities, as promptly as reasonably practicable upon request, a written statement by the Company as to its compliance with the reporting requirements of Rule 144 under the Securities Act and of the Exchange Act, and (iii) take such further reasonable action, to the extent required from time to time to enable such holder to sell Registrable Securities without registration under the Securities Act within the limitation of the exemptions provided by Rule 144 under the Securities Act, as such Rule may be amended from time to time, or any similar rule or regulation hereafter adopted by the Commission.

C-17

SECTION 2.07 Registration Expenses. All expenses incident to the Company's performance of, or compliance with, its obligations under this Agreement including (a) all registration and filing fees, all fees and expenses of compliance with securities and blue sky laws (including the reasonable and documented fees and disbursements of counsel for the underwriters in connection with blue sky qualifications of the Registrable Securities pursuant to Section 2.06), (b) all printing and copying expenses (including expenses of printing certificates for the Registrable Securities in a form eligible for deposit with the Depository Trust Company and of printing prospectuses as requested by any holder of Registrable Securities), (c) all messenger and delivery expenses, (d) all fees and expenses of the Company's independent certified public accountants and counsel (including, with respect to "comfort" letters and opinions) and (e) all reasonable fees and disbursements of one single primary outside counsel and one outside local counsel for each jurisdiction that Registrable Securities shall be distributed for the holders thereof, which counsels shall be selected by the holders of a majority of the Registrable Securities being sold (collectively, the "Registration Expenses") shall be borne by the Company. The Registration Expenses shall be borne by the Company regardless of whether or not any registration statement is filed or becomes effective. the Company will pay its internal expenses (including all salaries and expenses of its officers and employees performing legal or accounting duties, the expense of any annual audit and the expense of any liability insurance), the expenses and fees for listing the securities to be registered on each securities exchange and included in each established over-the-counter market on which similar securities issued by the Company are then listed or traded and any expenses of the Company incurred in connection with any "road show". Each Selling Holder shall pay its pro rata portion (based on th

SECTION 2.08 <u>Miscellaneous</u>. The Company may grant demand, piggyback or shelf registration rights to third parties, provided that the terms of such rights are not senior to and do not conflict with the rights granted to the holders of Registrable Securities hereunder without the prior written consent (such consent not to be unreasonably withheld, delayed or conditioned) of the OEP Stockholders.

SECTION 2.09 <u>Indemnification.</u> (a) The Company shall indemnify and hold harmless each Selling Holder and their respective Affiliates, member, partners, directors, officers and employees and each Person, if any, who controls any Selling Holder within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act and each of their respective representatives as follows:

(i) against any and all losses, liabilities, claims, damages, judgments and reasonable expenses whatsoever, as incurred, arising out of any untrue statement or alleged untrue statement of a material fact contained in any Registration Statement (or any amendment thereto) pursuant to which Registrable Securities were registered under the Securities Act, including all documents incorporated therein by reference, or the omission or alleged omission therefrom of a material fact required to be stated therein or necessary to make the statements therein not misleading or arising out of any untrue statement or alleged untrue statement of a material fact contained in any prospectus (or any amendment or supplement thereto) including all documents incorporated therein by

reference, or the omission or alleged omission therefrom of a material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading;

- (ii) against any and all losses, liabilities, claims, damages, judgments and reasonable expenses whatsoever, as incurred, to the extent of the aggregate amount paid in settlement of any litigation, investigation or proceeding by any governmental agency or body, commenced or threatened, or of any other claim whatsoever based upon any such untrue statement or omission, or any such alleged untrue statement or omission, if such settlement is effected with the written consent of the Company; and
- (iii) against any and all reasonable expense whatsoever, as incurred (including, subject to Section 2.09(c), fees and disbursements of counsel) incurred in investigating, preparing or defending against any litigation, investigation or proceeding by any governmental agency or body, commenced or threatened, in each case whether or not such Person is a party, or any claim whatsoever based upon any such untrue statement or omission, or any such alleged untrue statement or omission, to the extent that any such expense is not paid under subparagraph (i) or (ii) above;

provided, however, that this indemnity agreement does not apply to any Selling Holder with respect to any loss, liability, claim, damage, judgment or expense to the extent arising out of any untrue statement or omission or alleged untrue statement or omission (A) made in reliance upon and in conformity with written information furnished to the Company by any Selling Holder expressly for use in a Registration Statement (or any amendment thereto) or any related prospectus (or any amendment or supplement thereto) or (B) if such untrue statement or omission or alleged untrue statement or omission was corrected in an amended or supplemented Registration Statement or prospectus and the Company had furnished copies thereof to the Person asserting such loss, liability, claim, damage, judgment or expense purchased the securities that are the subject thereof prior to the date of sale by such Selling Holder to such Person.

(b) Indemnification by Selling Holders. Each Selling Holder shall severally (but not jointly) indemnify and hold harmless the Company, and each other Selling Holder, and each of their respective Affiliates, members, partners, directors, officers and employees (including each officer of the Company who signed the Registration Statement) and each Person, if any, who controls the Company, or any other Selling Holder within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act and each of their respective representatives, against any and all losses, liabilities, claims, damages, judgments and expenses described in the indemnity contained in Section 2.09(a) (provided that any settlement of the type described therein is effected with the written consent of such Selling Holder) as incurred, but only with respect to untrue statements or omissions, or alleged untrue statements or omissions, made in a Registration Statement (or any amendment thereto) in reliance upon and in conformity with written information furnished to the Company by such Selling Holder expressly for use in such Registration Statement (or any amendment thereto) or such prospectus (or any amendment or supplement thereto); provided, however, that an indemnifying Selling Holder shall not be required to provide indemnification in any amount in excess of the amount by which (x) the total price at which the Registrable Securities sold by such indemnifying Selling Holder and its Affiliates and distributed to the public were offered to the public exceeds (y) the amount

C-19

of any damages which such indemnifying Selling Holder has otherwise been required to pay by reason of such untrue or alleged untrue statement or omission or alleged omission. The Company shall be entitled, to the extent customary, to receive indemnification and contribution from underwriters, selling brokers, dealer managers and similar securities industry professionals participating in the distribution, to the same extent as provided above with respect to information so furnished in writing by such Persons specifically for inclusion in any prospectus or Registration Statement.

Conduct of Indemnification Proceedings. Each indemnified party or parties shall give reasonably prompt notice to each indemnifying party or parties of any action or proceeding commenced against it in respect of which indemnity may be sought hereunder, but failure to so notify an indemnifying party or parties shall not relieve it or them from any liability which it or they may have under this indemnity agreement, except to the extent that the indemnifying party is materially prejudiced by such failure to give notice. If the indemnifying party or parties so elects within a reasonable time after receipt of such notice, the indemnifying party or parties may assume the defense of such action or proceeding at such indemnifying party's or parties' expense with counsel chosen by the indemnifying party or parties and approved by the indemnified party defendant in such action or proceeding, which approval shall not be unreasonably withheld; provided, however, that, if such indemnified party or parties reasonably determine that a conflict of interest exists and that therefore it is advisable for such indemnified party or parties to be represented by separate counsel or that, upon advice of counsel, there may be legal defenses available to it or them which are different from or in addition to those available to the indemnifying party, then the indemnifying party or parties shall not be entitled to assume such defense and the indemnified party or parties shall be entitled to separate counsel (limited in each jurisdiction to one counsel for all indemnified parties under this Agreement) at the indemnifying party's or parties' expense. If any indemnifying party or parties are not so entitled to assume the defense of such action or do not assume such defense, after having received the notice referred to in the first sentence of this paragraph, the indemnifying party or parties will pay the reasonable fees and expenses of counsel for the indemnified party or parties (limited in each jurisdiction to one counsel for all indemnified parties under this Agreement). In such event, however, no indemnifying party or parties will be liable for any settlement effected without the written consent of such indemnifying party or parties (which consent shall not be unreasonably withheld or delayed); provided, however, that if at any time an indemnified party or parties shall have requested an indemnifying party or parties to reimburse the indemnified party or parties for fees and expenses of counsel as contemplated by this paragraph, the indemnifying party or parties shall be liable for any settlement of any proceeding effected without the written consent of such indemnifying party or parties if (x) such settlement is entered into more than 15 business days after receipt by such indemnifying party or parties of the aforesaid request accompanied by supporting documents reasonably satisfactory to the indemnifying party or parties and (y) such indemnifying party or parties shall not have reimbursed the indemnified party or parties in accordance with such request prior to the date of such settlement. If an indemnifying party is entitled to assume, and assumes, the defense of such action or proceeding in accordance with this paragraph, such indemnifying party or parties shall not, except as otherwise provided in this Section 2.09(c), be liable for any fees and expenses of counsel for the indemnified parties incurred thereafter in connection with such action or proceeding.

C-20

(d) <u>Contribution.</u> (i)In order to provide for just and equitable contribution in circumstances in which the indemnity agreement provided for in this Section 2.09 is for any reason held to be unenforceable by the indemnified parties although applicable in accordance with its terms in respect of any losses, liabilities, claims, damages, judgments and expenses suffered by an indemnified party referred to therein, each applicable indemnifying party, in lieu of indemnifying such indemnified party, shall contribute to the amount paid or payable by such indemnified party as a result of such losses, liabilities, claims, damages, judgments and expenses in such proportion as is appropriate to reflect the relative fault of the Company, on the one hand, and of the liable Selling Holders (including, in each case, that of their respective officers, directors, employees and agents), on the other, in connection with the statements or

omissions which resulted in such losses, liabilities, claims, damages, judgments or expenses, as well as any other relevant equitable considerations. The relative fault of the Company, on the one hand, and of the liable Selling Holders (including, in each case, that of their respective officers, directors, employees and agents), on the other, shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by the Company, on the one hand, or by or on behalf of the Selling Holders, on the other, and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission. The amount paid or payable by a party as a result of the losses, liabilities, claims, damages, judgments and expenses referred to above shall be deemed to include, subject to the limitations set forth in Section 2.09(c), any legal or other fees or expenses reasonably incurred by such party in connection with investigating or defending any action or claim.

- (ii) The Company and each Selling Holder agree that it would not be just and equitable if contribution pursuant to this Section 2.09(d) were determined by pro rata allocation or by any other method of allocation which does not take account of the equitable considerations referred to in sub-paragraph (i) above. Notwithstanding this Section 2.09(d), in the case of distributions to the public, an indemnifying Selling Holder shall not be required to contribute any amount in excess of the amount by which (A) the total price at which the Registrable Securities sold by such indemnifying Selling Holder and its Affiliates and distributed to the public were offered to the public exceeds (B) the amount of any damages which such indemnifying Selling Holder has otherwise been required to pay by reason of such untrue or alleged untrue statement or omission or alleged omission. No Person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation.
- (iii) For purposes of this Section, each Person, if any, who controls a Selling Holder within the meaning of Section 15 of the Securities Act shall have the same rights to contribution as such Selling Holder; and each director of the Company, each officer of the Company who signed the Registration Statement, and each Person, if any, who controls the Company within the meaning of Section 15 of the Securities Act, shall have the same rights to contribution as the Company.

C-21

#### ARTICLE III

### MISCELLANEOUS PROVISIONS

SECTION 3.01 <u>Termination.</u> This Agreement (other than Section 2.07 and Section 2.09) will terminate on the date on which all Shares cease to be Registrable Securities.

SECTION 3.02 Notices. All notices and other communications hereunder shall be in writing and shall be deemed to have been duly delivered and received hereunder (a) four Business Days after being sent by registered or certified mail, return receipt requested, postage prepaid, (b) one Business Day after being sent for next Business Day delivery, fees prepaid, via a reputable nationwide overnight courier service, or (c) immediately upon delivery by hand or by facsimile or email (with a written or electronic confirmation of delivery), if sent during normal business hours of the recipient, or if not sent during normal business hours of the recipient, then on the recipient's next Business Day, in each case to the intended recipient as set forth below:

If to the Company, addressed to it at:

Solstice Sapphire Investments, Inc. 4 Technology Park Drive Westford, MA 01886 Tel: (978) 614-8100 Fax: 978 614-8101

Attention: Jeffrey M. Snider Email: jsnider@sonusnet.com

with a copy to (for information purposes only):

Wilmer Cutler Pickering Hale and Dorr LLP 60 State Street

Boston, MA 02109 Tel: (617) 526-6000 Fax: (617) 526-5000

Attention: Jay E. Bothwick

Joseph B. Conahan

Email: Jay.Bothwick@wilmerhale.com

Joseph.Conahan@wilmerhale.com

If to any Stockholder, addressed to it at:

The address for such Stockholder reflected in the stock record books of the Company

SECTION 3.03 Severability. If any provision (or part thereof) of this Agreement is invalid, illegal or unenforceable, that provision (or part thereof) will, to the extent possible, be modified in such a manner as to be valid, legal and enforceable but so as to retain most nearly the intent of the parties as expressed herein, and if such a modification is not possible, that provision (or part thereof) will be severed from this Agreement, and in either case the validity, legality and

enforceability of the remaining provisions (or parts thereof) of this Agreement will not in any way be affected or impaired thereby. If any provision (or part thereof) of this Agreement is so broad as to be unenforceable, the provision (or part thereof) shall be interpreted to be only so broad as is enforceable.

SECTION 3.04 Entire Agreement. This Agreement, and any documents delivered by the parties in connection herewith constitute the entire agreement among the parties with respect to the subject matter hereof and supersede all prior agreements and understandings among the parties with respect thereto. No addition to or modification of any provision of this Agreement shall be binding upon any party hereto unless made in writing and signed by all parties hereto.

SECTION 3.05 Amendments. Any provision of this Agreement may be amended if, and only if, such amendment is in writing and signed by the Company and the OEP Stockholders; provided that (a) any amendment that would have a material adverse effect on a Stockholder relative to the OEP Stockholders shall require the written consent of that Stockholder and (b) this Section 3.05 may not be amended without the prior written consent of the Stockholders (other than the OEP Stockholders) holding a majority of the outstanding Registrable Securities of such Stockholders.

SECTION 3.06 <u>Waivers</u>. Except as provided in this Agreement, no action taken pursuant to this Agreement, including any investigation by or on behalf of any party, or delay or omission in the exercise of any right, power or remedy accruing to any party as a result of any breach or default hereunder by any other party shall be deemed to constitute a waiver by the party taking such action of compliance with any representations, warranties, covenants or agreements contained in this Agreement. The waiver by any party hereto of a breach of any provision hereunder shall not operate or be construed as a waiver of any prior or subsequent breach of the same or any other provision hereunder.

SECTION 3.07 <u>Assignment.</u> No Stockholder shall assign any of its rights under this Agreement, in whole or in part, to any Person, without first obtaining the prior written consent of the Company; provided, that, without the consent of the Company, a Stockholder may assign its rights under this Agreement with respect to any Registrable Securities to any Permitted Transferee of such Registrable Securities who executes a Joinder Agreement prior to or concurrently with the Transfer of such Registrable Securities to such Permitted Transferee, and any assignment in contravention hereof shall be null and void. This Agreement will be binding upon, inure to the benefit of and be enforceable by the parties and their respective permitted successors and assigns.

SECTION 3.08 Benefit. Notwithstanding anything contained in this Agreement to the contrary, nothing in this Agreement, expressed or implied, is intended to confer on any person other than the parties hereto or their respective successors and permitted assigns any rights, remedies, obligations or liabilities under or by reason of this Agreement.

SECTION 3.09 Governing Law; Consent to Jurisdiction. This Agreement and the rights and obligations of the parties hereto shall be governed by and construed in accordance with the laws of the State of Delaware, without giving effect to any choice or conflict of law

C-23

provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware. Each of the parties hereby irrevocably agrees that any legal action or proceeding with respect to this Agreement and the rights and obligations arising hereunder, or for recognition and enforcement of any judgment in respect of this Agreement and the rights and obligations arising hereunder brought by the other party hereto or its successors or assigns, shall be brought and determined exclusively in the Delaware Court of Chancery and any state appellate court therefrom within the State of Delaware (or, if the Delaware Court of Chancery declines to accept jurisdiction over a particular matter, any state or federal court within the State of Delaware) (collectively with Delaware Court of Chancery, the "Delaware Courts"). Each of the parties hereto further agrees not to commence any litigation relating to this Agreement except in the Delaware Courts, waives any objection to the laying of venue of any such litigation in the Delaware Courts and agrees not to plead or claim in any Delaware Court that such litigation brought therein has been brought in an inconvenient forum. EACH PARTY TO THIS AGREEMENT IRREVOCABLY WAIVES THE RIGHT TO A TRIAL BY JURY IN CONNECTION WITH ANY MATTER ARISING OUT OF THIS AGREEMENT AND, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY DEFENSE OR OBJECTION IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY PROCEEDING UNDER THIS AGREEMENT BROUGHT IN ANY SUCH COURT HAS BEEN BROUGHT IN AN INCONVENIENT FORUM.

SECTION 3.10 <u>Counterparts</u>. This Agreement may be executed by the parties hereto in separate counterparts, each of which when so executed and delivered shall be an original, but all such counterparts shall together constitute one and the same instrument. Each counterpart may consist of a number of copies hereof each signed by less than all, but together signed by all of the parties hereto.

SECTION 3.11 <u>Enforcement of Agreement</u>. The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with its specific terms or if this Agreement was otherwise breached and that monetary damages, even if available, would not be an adequate remedy hereunder. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any Delaware Court without proof of actual damages and each party hereto waives any requirement for the securing or posting of any bond in connection with such remedy, this being in addition to any other remedy to which they are entitled at law or in equity. The parties further agree not to assert that a remedy of specific enforcement is unenforceable, invalid, contrary to Applicable Law or in equity for any reason, nor to assert that a remedy of monetary damages would provide an adequate remedy for such breach.

[Signature Pages Follow]

C-24

IN WITNESS WHEREOF, the parties have caused this Agreement to be executed and delivered by their authorized representatives as of the date first above written.

Solstice Sapphire Investments, Inc.

By:

Name: Susan Villare

[Signature Page to Registr	ation Rights Agreement]
F	HERITAGE PE (OEP) II, L.P.
E	Зу:
	Name: Citle:
F	HERITAGE PE (OEP) III, L.P.
_	
	By: Name:
	Title:
[Signature Page to Registre	ation Rights Agreement]
EXHIE	BIT A
Joinder Ag	<u>greement</u>
By execution of this signature page, [ ] hereby agrees to become the benefits of a Stockholder, under that certain Registration Rights Agreement, Delaware corporation, Heritage PE (OEP) II, L.P., a Cayman Islands exempted I limited partnership, and the other Stockholders who become parties thereto from	imited partnership, Heritage PE (OEP) III, L.P., a Cayman Islands exempted
]	NAME]
F	By:
	Name:
7	Title:
ľ	Notice Address:
	EXHIBIT I AGREED FORM
PROMISSO	RY NOTE
THIS NOTE HAS NOT BEEN REGISTERED UNDER THE SECURITIES ACTRANSFERRED IN THE ABSENCE OF REGISTRATION OR AN EXEMPT	
Solstice Sapphire l	investments, Inc.
\$22,500,000	[ , 2017] (the " <u>Issue Date</u> "
Solstice Sapphire Investments, Inc., a Delaware corporation ("Payor"),	for value received, promises to pay to each person identified on Schedule I

Title:

President and Treasurer

This Note is made pursuant to that certain Agreement and Plan of Merger (the "Merger Agreement"), dated May 23, 2017 by and among Sonus Networks Inc., Genband Holdings Company, Genband Inc., Genband II, Inc., the Payor and the other parties thereto, and is the "Promissory Note" referred to therein. Payee is receiving this Note pursuant to the Merger Agreement. All capitalized terms used herein and not defined herein shall have the meanings assigned to such terms in the Merger Agreement.

hereto (each, a "<u>Payee</u>") the principal amount set forth opposite its name on <u>Schedule I</u> together with accrued interest thereon, calculated and payable as set forth below (together with any PIK Notes issued pursuant to Section 1.1 below, the "<u>Note</u>"). The principal and interest on this Note is payable in lawful money of the United States of America in immediately available funds at such place in the United States as Payee may from time to time designate in writing

to Payor.

# 1. Payment of Principal and Interest

## 1.1 <u>Calculation and Payment of Interest.</u>

Interest on the principal balance of this Note outstanding from time to time until paid in full in cash shall accrue at the rate equal to the Applicable Rate *per annum*, computed on the basis of a 360-day year, for the actual number of days elapsed, commencing on the date hereof. "Applicable Rate" means, 7.5% on or before the date that is exactly six months after the Issue Date, and after the date that is exactly six months after the Issue Date means 10.0%. Interest shall be payable quarterly in arrears, beginning on the last day of the first calendar quarter following the Issue Date and on the last date of each calendar quarter thereafter (each, an "Interest Payment Date") until the Maturity Date; provided, however, that any amount of cash interest which is not paid on an Interest Payment Date as a result of a Payment Restriction or pursuant to Section 3 hereof shall be made either (i) by the issuance of a promissory note in a principal amount equal to interest accrued but not otherwise paid (by the issuance of a PIK Note or otherwise) on the principal amount hereof through and including such Interest Payment Date and otherwise having such terms and provisions that are the same as the terms and provisions of this Note (each such promissory note a "PIK Note") and Payor shall be deemed to have issued a PIK Note for any such interest regardless of whether Payor shall have actually delivered any such PIK Note or (ii) by increasing the principal amount of this Note by the amount of such cash interest. Payee, by its acceptance hereof, acknowledges (i) that Payor is contractually bound hereunder to pay cash interest only to the extent not prohibited by a Payment Restriction or by Section 3 hereof, (ii) that any cash interest not so paid shall be paid in the form of a PIK Note or by increasing the principal amount of this Note by the amount of such cash interest, and (iii) the failure to pay cash interest as a result of a Payment Restriction or pursuant to Section 3 hereof shall not constitute a default or Event of D

1.2 <u>Payment on Maturity Date</u>. The principal balance of, and any accrued and unpaid interest on, this Note (including, for the avoidance of doubt, the principal balance of and any accrued and unpaid interest on any PIK Note) shall be payable in cash on the third anniversary of the Issue Date the ("<u>Maturity Date</u>").

# 1.3 <u>Prepayment</u>.

(a) Payor may, at its option at any time, without premium or penalty, prepay all or any portion of this Note.

D-2

- (b) Any prepayment of this Note shall be applied as follows: <u>first</u>, to payment of accrued interest; and <u>second</u>, to payment of principal. Upon any partial prepayment, at the request either of Majority Payees or Payor, this Note shall be surrendered to Payor in exchange for a substitute note, which shall set forth the revised principal amount but otherwise be identical to this Note. In the event that this Note is prepaid in its entirety, this Note shall be surrendered to Payor for cancellation as a condition to any such prepayment.
- 1.4 <u>Payment Only on Business Days; Payments Free and Clear.</u> Any payment hereunder which, but for this Section 1.4, would be payable on a day which is not a Business Day, shall instead be due and payable on the Business Day next following such date for payment. All payments hereunder shall be made free and clear of any deduction, withholding or offset and in immediately available funds, except to the extent otherwise required by applicable law.

## 2. Events of Default

- (a) The following shall constitute "Events of Default" under this Note:
- (i) Failure by Payor to make any payment required under this Note when the same becomes due and payable (whether at maturity, by acceleration or otherwise); and with respect to the payment of any interest the continuation of such failure for a period of thirty (30) days thereafter;
  - (ii) Payor voluntarily liquidates;
  - (iii) the Payor pursuant to or within the meaning of any Bankruptcy Law:
    - (A) commences a voluntary case or proceeding;
    - (B) consents to the entry of an order for relief against it in an involuntary case or proceeding;
    - (C) consents to the appointment of a Custodian of it or for all or substantially all of its property;

D-3

- (D) makes a general assignment for the benefit of its creditors;
- (E) admits in writing that it is generally unable to pay its debts as they become due;
- (iv) a court of competent jurisdiction enters an order or decree (that remains unstayed and in effect for thirty (30) days) under any Bankruptcy Law that:
  - (A) is for relief against the Payor in an involuntary case or proceeding;

- (B) appoints a Custodian of the Payor or for all or substantially all of Payor's property; or
- (C) orders the liquidation of Payor; or
- (v) a Change of Control.
- (b) Acceleration. If an Event of Default specified in Section 2(a)(i) shall have occurred and be continuing and any Senior Indebtedness shall then be outstanding, subject to the provisions of Section 3 hereof, the Majority Payees may, at their option, by notice in writing to Payor and to the agents under the Senior Indebtedness Documents (the "Acceleration Notice"), declare the entire principal amount of this Note and the interest accrued thereon to be due and payable upon the earlier of (i) one hundred eighty (180) days after the receipt of the Acceleration Notice by Payor and the agents under the Senior Indebtedness Documents or (ii) an acceleration under any of the Senior Indebtedness Documents, and upon any such declaration the same shall become due and payable at such time. If an Event of Default specified in Section 2(a)(i) shall have occurred and be continuing and no Senior Indebtedness shall then be outstanding, the Majority Payees may, at their option, declare the entire principal balance of this Note and the accrued and unpaid interest thereon to be due and payable upon the date which is five Business Days after the date of delivery by Payee to Payor of a written notice of acceleration, and upon any such declaration the same shall become due and payable at such time. If any other Event of Default occurs, the principal balance of this Note and the accrued and unpaid interest thereon

D-4

shall become due and payable immediately without any declaration or other act on the part of the Majority Payees and without presentment, demand, protest or other notice or action of any kind, all of which are hereby expressly waived.

If any Event of Default shall have occurred and be continuing, subject to the provisions of Section 3 hereof, the Majority Payees may proceed to protect and enforce their rights either by suit in equity or by action at law, or both, whether for specific performance of any provision of this Note or in aid of the exercise of any power granted to any Payee under this Note.

### 3. Subordination

Note Subordinated to Senior Indebtedness. To the extent and in the manner hereinafter set forth in this Section 3, the indebtedness represented by this Note and the payment of the principal of and the interest on this Note and any claim for rescission of the purchase of this Note, and any claim which is the equivalent of or substitute for principal of or interest on this Note, for damages arising from the purchase of this Note or for reimbursement or contribution on account of such a claim, and all other payments with respect to or on account of this Note (collectively, the "Subordinated Debt") are hereby expressly made subordinate and subject in right of payment to the prior payment in full in cash of all Senior Indebtedness then outstanding. This Section 3 constitutes a continuing offer to all Persons who become holders of, or continue to hold, Senior Indebtedness, each of whom is an obligee hereunder and is entitled to enforce such holder's rights hereunder, subject to the provisions hereof, without any act or notice of acceptance hereof or reliance hereon. For purposes of this Section 3, Senior Indebtedness shall not be deemed to have been paid and shall be deemed to be outstanding in full until the termination of all commitments or other obligations by any holder thereof and unless all such holders shall have received payment in full in cash of all obligations under or in respect of Senior Indebtedness (including, without limitation, post-petition interest, if any).

# 3.2 <u>No Payment on Note in Certain Circumstances.</u>

(a) To the extent any payment hereunder is blocked by a Payment Restriction (or there occurs and is continuing a payment default or payment event of default under any Senior Indebtedness), no direct or indirect payment of any kind shall be made, asked for,

D-5

demanded, accepted, received or retained with respect to principal, interest or other amounts due under the Note nor shall any holder thereof exercise any remedies with respect thereto.

- (b) Payee agrees that, so long as payments or distributions for or on account of the Subordinated Debt are not permitted pursuant to this Section 3, Payee will not take, sue for, ask or demand from Payor payment of all or any amounts under or in respect of this Note, or commence, or join with any creditor other than the holders of Senior Indebtedness and their agents in commencing, directly or indirectly cause Payor to commence, or assist Payor in commencing, any proceeding referred to in Section 3.3, and Payee shall not take or receive from Payor, directly or indirectly or on its behalf, in cash or other property or by set-off or in any other manner, including, without limitation, from or by way of collateral, payment of all or any amounts under or in respect of the Subordinated Debt. In the event that notwithstanding the foregoing provisions of this Section 3.2, any payment or distribution of any kind or character, whether in cash, property or securities, shall be received by Payee by or on account of or in respect of the Subordinated Debt while a Payment Restriction exists or while payments or distributions for or on account of the Subordinated Debt are otherwise not permitted pursuant to this Section 3, such payment or distribution shall be received and held in trust for, and shall be paid over (in the same form as so received, to the extent practicable, and with any necessary endorsement) to the holders of the Senior Indebtedness remaining unpaid or their representative or representatives, or to the trustee or trustees under any such indenture or agreement under which any Senior Indebtedness may have been issued, for application (in the case of cash) to, or as collateral (in the case of non-cash property or securities) for the payment or prepayment of Senior Indebtedness, until all Senior Indebtedness shall have been paid in full in cash, after giving effect to any concurrent payment or distribution to the holders of such Senior Indebtedness.
- 3.3 <u>Dissolution; Liquidation; Bankruptcy; Acceleration</u>. In the event of (i) any insolvency or bankruptcy case or proceeding, or any receivership, liquidation, reorganization or other similar proceeding in connection therewith, relative to the Payor or any of its assets, or (ii) any liquidation, dissolution or other winding up of the Payor, whether voluntary or involuntary or whether or not involving insolvency or bankruptcy, or (iii) any assignment for the benefit of creditors or any other marshalling of assets or liabilities of the Payor, or (iv) the acceleration of

the Senior Indebtedness by reason of the occurrence of a default or an event of default thereunder:

- (a) The holders of all Senior Indebtedness shall first be entitled to receive payment in full in cash of all Senior Indebtedness before any direct or indirect payment may be made for or on account of payments under or in respect of the Subordinated Debt, whether in cash, property or securities of any kind;
- (b) Any payment or distribution of any kind or character, whether in cash, property or securities (including any payment or distribution that may be payable by reason of any other indebtedness of Payor being subordinated to payment of the Subordinated Debt), to which Payee would be entitled except for the provisions of this Section 3, shall be paid by the liquidating trustee or agent or other person making such payment or distribution, whether a trustee in bankruptcy, a receiver or liquidating trustee or other trustee or agent, directly to the holders of Senior Indebtedness or their representative or representatives, or to the trustee or trustees under any indenture or other agreement under which any of such Senior Indebtedness may have been issued for application (in the case of cash) to, or as collateral (in the case of non-cash property or securities) for the payment or prepayment of Senior Indebtedness, to the extent necessary to make payment in full of all Senior Indebtedness remaining unpaid, after giving effect to any concurrent payment or distribution to the holders of such Senior Indebtedness;
- (c) The holders of Senior Indebtedness are hereby irrevocably authorized and empowered (in their own names or in the name of Payee or otherwise), but shall have no obligation, to demand, sue for, collect and receive every payment or distribution referred to in paragraph (b) above to file (but not vote) claims and proofs of claim as they may deem necessary or advisable for the exercise or enforcement of any of the rights or interests of the holders of Senior Indebtedness hereunder.
- (d) Payee shall duly and promptly take such action as the holders of Senior Indebtedness may reasonably request to execute and deliver to the holders of Senior Indebtedness such powers of attorney, assignments, or other instruments as the holders of Senior Indebtedness may request in order to enable the holders of Senior Indebtedness to enforce any

D-7

and all claims with respect to, and any security interests and other liens securing payment of, the amounts owing under the Subordinated Debt.

- (e) In the event that, any payment or distribution of any kind or character, whether in cash, property or securities (including any payment or distribution that may be payable by reason of any other indebtedness of Payor being subordinated to payment of the Subordinated Debt), shall be received by Payee for or on account of or in respect of the Subordinated Debt in contravention of this Section 3.3 before all Senior Indebtedness is paid in full in cash, such payment or distribution shall be received and held in trust for, and shall be paid over (in the same form as so received, to the extent practicable, and with any necessary endorsement) to the holders of the Senior Indebtedness remaining unpaid or their representative or representatives, or to the trustee or trustees under any such indenture or agreement under which any Senior Indebtedness may have been issued, for application (in the case of cash) to, or as collateral (in the case of non-cash property or securities) for the payment or prepayment of Senior Indebtedness, until all Senior Indebtedness shall have been paid in full in cash, after giving effect to any concurrent payment or distribution to the holders of such Senior Indebtedness.
- 3.4 <u>Subrogation</u>. Upon the final payment in full in cash of all Senior Indebtedness, Payee shall be subrogated to the rights of the holders of Senior Indebtedness to receive payments or distributions of cash, property or securities of Payor applicable to the Senior Indebtedness until the principal of and interest on and all other amounts payable under the Subordinated Debt shall be paid in full in cash, and for the purposes of such subrogation, no payments or distributions to the holders of the Senior Indebtedness of any cash, property or securities to which Payee would be entitled except for the provisions of this Section 3 and no payment pursuant to the provisions of this Section 3 to the holders of Senior Indebtedness by Payee shall, as between Payor, its creditors other than holders of Senior Indebtedness, and Payee, be deemed to be a payment by Payor to or on account of the Senior Indebtedness. It is understood that the provisions of this Section 3 are and are intended solely for the purpose of defining the relative rights of Payee, on the one hand, and the holders of the Senior Indebtedness, on the other hand.

D-8

- 3.5 <u>Obligations of Payor Unconditional</u>. Nothing contained in this Section 3 or elsewhere in this Note is intended to or shall impair, as among Payor, its creditors (other than the holders of Senior Indebtedness) and Payee, the obligation of Payor, which is absolute and unconditional, to pay to Payee the principal of and interest on and all other amounts due under this Note in accordance with its terms, or is intended to or shall affect the relative rights of Payee and creditors of Payor (other than the holders of the Senior Indebtedness), nor shall anything herein prevent Payee from exercising all remedies otherwise permitted by applicable law upon default under this Note, subject to the provisions of this Section 3 and to the rights of holders of Senior Indebtedness to receive distributions and payments otherwise payable to Payee.
- 3.6 Reliance on Judicial Order or Certificate of Liquidating Agent. Upon any payment or distribution of assets of Payor referred to in this Section 3, Payee shall be entitled to rely upon any order or decree made by any court of competent jurisdiction in which bankruptcy, dissolution, winding-up, liquidation or reorganization proceedings are pending, or a certificate of the receiver, trustee in bankruptcy, liquidating trustee, agent or other person making such payment or distribution, delivered to Payee, for the purpose of ascertaining the persons entitled to participate in such distribution, the holders of the Senior Indebtedness and other indebtedness of Payor, the amount thereof or payable thereon, the amount or amounts paid or distributed thereon and all other facts pertinent thereto or to Section 3 of this Note.
- 3.7 <u>Subordination Rights Not Impaired by Acts or Omissions of Payor or Holders of Senior Indebtedness</u>. No right of any present or future holders of any Senior Indebtedness to enforce subordination as provided herein will at any time in any way be prejudiced or impaired by any act or failure to act on the part of Payor or by any act or failure to act by any such holder, or by any act, failure to act or noncompliance by Payor, the holders of Senior Indebtedness or their respective agents with the terms of this Note, regardless of any knowledge thereof which any such holder or Payor may have or otherwise be charged with. No amendment, waiver or other modification of this Note shall in any way adversely affect the rights of the holders of any Senior Indebtedness under this Section 3 unless such holders of Senior Indebtedness consent in writing to such amendment, waiver or modification. The provisions of this Section 3 are intended for the benefit of and shall be enforceable directly by the holders of the Senior Indebtedness.

3.8 <u>Further Assurances</u>. Payee and Payor each will, at Payor's expense and at any time and from time to time, promptly execute and deliver all further instruments and documents, and take all further action, that may be necessary or desirable, or that the holders of Senior Indebtedness may request, in order to protect any right or interest granted or purported to be granted hereby or to enable the holders of Senior Indebtedness to exercise and enforce their rights and remedies hereunder.

# 3.9 Agreements in Respect of Subordinated Debt.

Payor agrees that it will not make any payment for or on account of or in respect of this Note, or take any other action, in contravention of the provisions of this Section 3.

- 3.10 <u>Obligations Hereunder Not Affected</u>. All rights and interests of the holders of Senior Indebtedness hereunder, and all agreements and obligations of Payee and Payor under this Section 3, shall remain in full force and effect irrespective of:
  - (i) any lack of validity or enforceability of any present or future guaranty of the Credit Agreement or any other Senior Indebtedness Document;
  - (ii) any change in the time, manner or place of payment of, or in any other term of, all or any of the Senior Indebtedness, or any other amendment or waiver of or any consent to any departure from the Credit Agreement or any successor agreement or any other Senior Indebtedness Document, including, without limitation, any increase in the Senior Indebtedness resulting from the extension of additional credit to Payor or any of its Subsidiaries or otherwise;
  - (iii) any taking, exchange, release or non-perfection of any other collateral, or any taking, release, amendment or waiver of or consent to departure from any guaranty, for all or any of the Senior Indebtedness;

D-10

- (iv) any manner of application of collateral, or proceeds thereof, to all or any of the Senior Indebtedness, or any manner of sale or other disposition of any collateral for all or any of the Senior Indebtedness or any other assets of Payor or any of its Subsidiaries;
  - (v) any change, restructuring or termination of the corporate structure or existence of Payor or any of its Subsidiaries; or
- (vi) any other circumstance which might otherwise constitute a defense available to, or a discharge of, Payor or a subordinated creditor.

The provisions of this Section 3 shall continue to be effective or be reinstated, as the case may be, if at any time any payment of any of the Senior Indebtedness is rescinded or must otherwise be returned by the holders of Senior Indebtedness upon the insolvency, bankruptcy or reorganization of Payor or otherwise, all as though such payment had not been made.

- 3.11 <u>Waiver</u>. Payee hereby waives promptness, diligence and notice of acceptance with respect to any of the Senior Indebtedness and this Section 3 and any requirement that the holders of Senior Indebtedness protect, secure, perfect or insure any security interest or lien on any property subject thereto or exhaust any right or take any action against Payor or any other person or entity or any collateral.
- 3.12 <u>No Waiver; Remedies</u>. No failure on the part of the holders of Senior Indebtedness to exercise, and no delay in exercising, any right hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right hereunder preclude any other or further exercise thereof or the exercise of any other right. The remedies herein provided are cumulative and not exclusive of any remedies provided by law.
- 3.13 <u>Continuing Agreement; Assignments Under Senior Indebtedness Agreements</u>. The provisions of this Section 3 constitute a continuing agreement and shall (i) remain in full force and effect until the earlier of (x) the date the obligations under this Note are satisfied in full in accordance with this Section 3 and (y) the date that is six months or such shorter period as the holders of a majority of the Senior Indebtedness may agree after the indefeasible payment in full

D-11

in cash of the Senior Indebtedness, (ii) be binding upon Payee, Payor and their respective successors and assigns, and (iii) inure to the benefit of, and be enforceable by, the holders of Senior Indebtedness and their successors, transferees and assigns. Without limiting the generality of the foregoing clause (iii), the holders of Senior Indebtedness may assign or otherwise transfer all or any portion of their rights and obligations under the Senior Credit Agreement or any other Senior Indebtedness Document, as applicable, to any other Person, and such other Person shall thereupon become vested with all the rights in respect thereof granted to the holders of Senior Indebtedness herein or otherwise.

# 4. <u>Certain Definitions</u>

"Bankruptcy Law" means Title 11, U.S. Code or any similar federal or state law for the relief of debtors.

"Business Day" means each day other than Saturdays, Sundays and days when commercial banks are authorized or required by law to be closed for business in New York. New York.

"Change of Control" means the occurrence of any of the following: (i) Payor ceases to beneficially and of record own directly and control 100% of the equity interests of each of its subsidiaries; (ii) all or substantially all of the Payor's assets are, directly or indirectly, sold to any Person or related group of Persons; (iii) the consummation of one or more transactions (excluding any transaction involving only the sale of equity interests by a Payee and its

Affiliates (and no other Person), but including, for the avoidance of doubt, mergers and tenders offers) the result of which is that the Payee and their Affiliates
cease to directly or indirectly own and control, beneficially and of record, at least 45.0% of the equity interests of the Payor; (iv) any "person" or "group" (as
such terms are used in Sections 13(d) and 14(d) of the Exchange Act of 1934, as amended), other than one or more of the Payees and their Affiliates, is or
becomes the beneficial owner (as defined in Rules 13d-3 and 13d-5 under the Exchange Act of 1934, as amended, except that for purposes of this clause such
person or group shall be deemed to have "beneficial ownership" of all securities that such person or group has the right to acquire, whether such right is
exercisable immediately or only after the passage of time), directly or

D-12

indirectly, of more than 35% of the equity interests of Payor; or (v) during any period of two consecutive years, individuals who at the beginning of such period constituted the Board of Directors of Payor (together with any new directors whose election to such Board of Directors or whose nomination for election was approved by a vote of a majority of the members of the Board of Directors of Payor) cease for any reason to constitute a majority of the Board of Directors of Payor.

"Credit Agreement" means the Credit Agreement first dated as of June 27, 2014 by and among Sonus Networks, Inc., as borrower, Bank of America, N.A., as Administrative Agent, Swing Line Lender and L/C Issuer, and the other lenders from time to time party thereto, as amended.

"Custodian" means any receiver, trustee, assignee, liquidator, sequestrator or similar office under any Bankruptcy Law.

"Event of Default" means any of the occurrences specified under Section 2 of this Note.

"Majority Payees" means one or more Payees entitled to a majority of the aggregate principal amount then owing under this Note.

"Payment Restriction" means the Payor is prohibited by the terms of the Senior Indebtedness from making any applicable payment on this

Note.

"<u>Person</u>" means any individual, corporation, partnership, limited liability company, joint venture, association, joint-stock company, trust, unincorporated organization or government or any agency or political subdivision thereof.

"Senior Indebtedness" means (i) indebtedness incurred pursuant to the Credit Agreement, and (ii) any indebtedness incurred to refinance, replace or otherwise restructure all or any part of any indebtedness described in clause (i) above or this clause (ii) whether by the same or any other agent, lender, debtholder or group of lenders or debtholders, including any

D-13

new facility entered into after the termination of any debt facility, whether or not contemporaneous.

"Senior Indebtedness Documents" means the Credit Agreement, any other note, agreement, indenture, mortgage, guaranty, pledge, security agreement or instrument evidencing or securing Senior Indebtedness or pursuant to which Senior Indebtedness is incurred, in each case as such agreement or document may be amended, modified or supplemented from time to time, including without limitation any agreement or document extending the maturity of, increasing the aggregate commitments under, or refinancing, replacing or otherwise restructuring all or any part of indebtedness under such agreement or document or any replacement or successor agreement or document and whether by the same or any other agent, lender or group of lenders.

# 5. Miscellaneous

- 5.1 <u>Section Headings</u>. The section headings contained in this Note are for reference purposes only and shall not affect the meaning or interpretation of this Note.
- 5.2 <u>Amendment and Waiver</u>. Subject to Section 5.10 hereof, no provision of this Note may be amended or waived unless Payor shall have obtained the written agreement of the Majority Payees and (unless there is no Senior Indebtedness outstanding and no commitments outstanding under the Credit Agreement and/or any other Senior Indebtedness Document) the holders of all of Senior Indebtedness. No failure or delay in exercising any right, power or privilege hereunder shall imply or otherwise operate as a waiver of any rights of Payee, nor shall any single or partial exercise thereof preclude any other or future exercise thereof or the exercise of any other right, power or privilege.
- 5.3 <u>Successors, Assigns and Transferors</u>. This Note may not be assigned or transferred by Payee to any competitor, customer or supplier of Payor or any of its subsidiaries. Subject to the foregoing, this note may be assigned or transferred by Payee provided that any such transfer complies with all applicable federal and state securities laws. Subject to the foregoing, the obligations of Payor and Payee under this Note shall be binding upon, and inure to the benefit of, and be enforceable by, Payor and Payee, and their respective successors and

D-14

permitted assigns, whether or not so expressed.

5.4 <u>Governing Law.</u> This Note shall be governed by, and construed in accordance with, the laws of the State of New York without giving effect to any conflicts of laws principles thereof that would otherwise require the application of the law of any other jurisdiction. Any proceeding to enforce, interpret, challenge the validity of, or recover for the breach of any provision of, this Note shall be filed exclusively in the United States District Court for the Southern District of New York or the state courts located in the State of New York, and the parties hereto expressly consent to the exclusive jurisdiction of such courts and expressly waive any and all objections to personal jurisdiction, service of process or venue in connection therewith. Final judgment in any action, suit or proceeding shall be conclusive and may be enforced in any other jurisdiction by suit on the judgment. Payor hereby acknowledges that this

Note constitutes an instrument for the payment of money, and consents and agrees that the Majority Payees, at their sole option, in the event of a dispute in the payment of any moneys due hereunder, shall have the right to bring a motion-action under New York CPLR Section 3213. Nothing in this Section 5.4 shall affect the right of the Majority Payees to (i) commence legal proceedings or otherwise sue Payor in any other court having jurisdiction over Payor or (ii) serve process upon Payor in any manner authorized by the laws of any such jurisdiction.

- 5.5 Lost, Stolen, Destroyed or Mutilated Note. Upon receipt of evidence reasonably satisfactory to Payor of the loss, theft, destruction or mutilation of this Note and of indemnity arrangements reasonably satisfactory to Payor from or on behalf of the holder of this Note, and upon surrender or cancellation of this Note if mutilated, Payor shall make and deliver a new note of like tenor in lieu of such lost, stolen, destroyed or mutilated Note, at Payee's expense.
- 5.6 <u>Waiver of Presentment, Etc.</u> Except as otherwise provided herein, presentment, demand, protest, notice of dishonor and all other demands and notices are hereby expressly waived by Payor.
- 5.7 <u>Usury.</u> Nothing contained in this Note shall be deemed to establish or require the payment of a rate of interest in excess of the maximum rate legally enforceable. If the rate of interest called for under this Note at any time exceeds the maximum rate legally enforceable, the

D-15

rate of interest required to be paid hereunder shall be automatically reduced to the maximum rate legally enforceable. If such interest rate is so reduced and thereafter the maximum rate legally enforceable is increased, the rate of interest required to be paid hereunder shall be automatically increased to the lesser of the maximum rate legally enforceable and the rate otherwise provided for in this Note.

5.8 <u>Notices</u>. Any notice, request, instruction or other document to be given hereunder by either party to the other shall be in writing and shall be deemed given when received and shall be (i) delivered personally or (ii) mailed by certified mail, postage prepaid, return receipt requested or (iii) delivered by FedEx or a similar overnight courier or (iv) sent via facsimile transmission to the fax number given below, as follows:

# If to a Payor, addressed to:

Sonus Networks, Inc. 4 Technology Park Dr. Westford, Massachusetts 01886 Attention: General Counsel Email: jsnider@sonusnet.com

Facsimile: (978) 614-8913

If to Payee, addressed to the address set forth on Schedule I:

or to such other place and with such other copies as either party may designate as to itself by written notice to the other party.

In the event that any notice under this Note is required to be made on or as of a day which is not a Business Day, then such notice shall not be required to be made until the first day thereafter which is a Business Day.

5.9 Representations and Warranties of Payor. Payor hereby represents and warrants to Payee that: (a) Payor is duly organized, validly existing and in good standing under the laws of the State of Delaware; (b) Payor has duly authorized, executed and delivered this Note; and (c) this Note constitutes a legally valid and binding obligation of Payor, enforceable against Payor in accordance with its terms, subject to the effects of bankruptcy, insolvency, reorganization,

D-16

moratorium or other similar laws now or hereafter in effect relating to or affecting the rights or remedies of creditors and the effect of general principles of equity, whether enforcement is considered in a proceeding in equity or at law, and the discretion of the court before which any proceeding therefor may be brought.

- 5.10 Actions by Majority Payees. Subject to the provisions of this Section 5.10, Majority Payees and Payor may enter into agreements for the purpose of adding or modifying provisions of the Note or changing in any manner the rights of the Payees or Payor hereunder or waiving any default or Event of Default hereunder and no amendment or modification of this Note shall be effective without the consent of Majority Payees and Payor; *provided*, *however*, that no change may be made to this Note which would either modify the subordination provisions hereof or would otherwise adversely affect the rights of the holders of Senior Indebtedness without the written consent, prior to the indefeasible repayment thereof in full in cash, of the holders of a majority of Senior Indebtedness outstanding at such time.
- 5.11 <u>Fees.</u> Payor agrees to pay all costs (including attorney's and paralegal fees and expenses) incurred or paid by Payee in enforcing collection of the Note.

[signature pages follow]

D-17

		By: Name: Title:
Accepted and Agreed to:		
[Payee]		
By: Name: Title:		
Tite.	[Signature Page to	Promissory Note]
	-	

Solstice Sapphire Investments, Inc.

# Schedule I Payee Schedule

Payee	Principal Amount	t	
	\$	[	]
[ADDRESS]			
	\$	[	]
[ADDRESS]			

EXHIBIT E AGREED FORM

# SONUS NETWORKS, INC.

# RESTATED CERTIFICATE OF INCORPORATION

ARTICLE I NAME

The name of the corporation (the "Corporation") is Sonus Networks, Inc.

# ARTICLE II REGISTERED AGENT

The address of the Corporation's registered office in the State of Delaware is 2711 Centerville Road, Suite 400, in the City of Wilmington, County of New Castle, 19808; and the name of its registered agent is Corporation Service Company.

ARTICLE III PURPOSE

The nature of the business or purposes to be conducted or promoted by the Corporation is to engage in any lawful act or activity for which corporations may be organized under the General Corporation Law of the State of Delaware (the "<u>DGCL</u>").

# ARTICLE IV CAPITAL STOCK

The total number of shares of all classes of stock that the Corporation shall have authority to issue is 1,100 shares, consisting solely of:

- 1,000 shares of common stock, par value \$0.001 per share (the "Common Stock"); and
- shares of preferred stock, par value \$0.01 per share (the "Preferred Stock").

The following is a statement of the powers, designations, preferences, privileges, and relative rights in respect of each class of capital stock of the Corporation.

A. COMMON STOCK.

- 1. *General*. The voting, dividend and liquidation rights of the holders of Common Stock are subject to and qualified by the rights of the holders of Preferred Stock.
- 2. *Voting.* The holders of Common Stock are entitled to one vote for each share held at all meetings of stockholders. There shall be no cumulative voting.
- 3. *Dividends*. Dividends may be declared and paid on the Common Stock from funds lawfully available therefor if, as and when determined by the Board of Directors and subject to any preferential dividend rights of any then outstanding shares of Preferred Stock.

E-1

4. *Liquidation*. Upon the dissolution or liquidation of the Corporation, whether voluntary or involuntary, holders of Common Stock will be entitled to receive all assets of the Corporation available for distribution to its stockholders, subject to any preferential rights of any then outstanding shares of Preferred Stock.

# B. PREFERRED STOCK.

#### 1. General.

Shares of Preferred Stock may be issued from time to time in one or more series, each of such series to have such powers, designations, preferences, and relative, participating, optional, or other special rights, if any, and such qualifications and restrictions, if any, of such preferences and rights, as are stated or expressed in the resolution or resolutions of the Board of Directors providing for such series of Preferred Stock. Different series of Preferred Stock shall not be construed to constitute different classes of shares for the purposes of voting by classes unless expressly so provided in such resolution or resolutions.

Authority is hereby granted to the Board of Directors from time to time to issue the Preferred Stock in one or more series, and in connection with the creation of any such series, by resolution or resolutions to determine and fix the powers, designations, preferences, and relative, participating, optional, or other special rights, if any, and the qualifications and restrictions, if any, of such preferences and rights, including without limitation dividend rights, conversion rights, voting rights (if any), redemption privileges, and liquidation preferences, of such series of Preferred Stock (which need not be uniform among series), all to the fullest extent now or hereafter permitted by the DGCL. Without limiting the generality of the foregoing, the resolution or resolutions providing for the creation or issuance of any series of Preferred Stock may provide that such series shall be superior to, rank equally with, or be junior to the Preferred Stock of any other series, all to the fullest extent permitted by law. No resolution, vote, or consent of the holders of the capital stock of the Corporation shall be required in connection with the creation or issuance of any shares of any series of Preferred Stock authorized by and complying with the conditions of this Restated Certificate of Incorporation, the right to any such resolution, vote, or consent being expressly waived by all present and future holders of the capital stock of the Corporation.

Any resolution or resolutions adopted by the Board of Directors pursuant to the authority vested in them by this Article IV shall be set forth in a certificate of designation along with the number of shares of stock of such series as to which the resolution or resolutions shall apply and such certificate shall be executed, acknowledged, filed, recorded, and shall become effective, in accordance with Section 103 of the DGCL. Unless otherwise provided in any such resolution or resolutions, the number of shares of stock of any such series to which such resolution or resolutions apply may be increased (but not above the total number of authorized shares of the class) or decreased (but not below the number of shares thereof then outstanding) by a certificate likewise executed, acknowledged, filed and recorded, setting forth a statement that a specified increase or decrease therein has been authorized and directed by a resolution or resolutions likewise adopted by the Board of Directors. In case the number of such shares shall be decreased, the number of shares so specified in the certificate shall resume the status which they had prior to the adoption of the first resolution or resolutions. When no shares of any such class

E-2

or series are outstanding, either because none were issued or because none remain outstanding, a certificate setting forth a resolution or resolutions adopted by the Board of Directors that none of the authorized shares of such class or series are outstanding, and that none will be issued subject to the certificate of designations previously filed with respect to such class or series, may be executed, acknowledged, filed and recorded in the same manner as previously described and it shall have the effect of eliminating from this Restated Certificate of Incorporation all matters set forth in the certificate of designations with respect to such class or series of stock. If no shares of any such class or series established by a resolution or resolutions adopted by the Board of Directors have been issued, the voting powers, designations, preferences and relative, participating, optional or other rights, if any, with the qualifications, limitations or restrictions thereof, may be amended by a resolution or resolutions adopted by the Board of Directors. In the event of any such amendment, a certificate which (i) states that no shares of such class or series have been issued, (ii) sets forth the copy of the amending resolution or resolutions and (iii) if the designation of such class or series is being changed, indicates the original designation and the new designation, shall be executed, acknowledged, filed, recorded, and shall become effective, in accordance with Section 103 of the DGCL.

# ARTICLE V BOARD OF DIRECTORS

The following provisions are inserted for the management of the business and for the conduct of the affairs of the Corporation and for defining and regulating the powers of the Corporation and its directors and stockholders and are in furtherance and not in limitation of the powers conferred upon the Corporation by statute:

(a) Each director shall hold office until the next annual meeting of stockholders and until his or her respective successor shall have been duly elected and qualified, subject, however, to prior death, resignation, retirement, disqualification or removal from office. Except as may otherwise be provided by the DGCL or the By-Laws of the Corporation, any director or the entire Board of Directors may be removed from office at any time, with or without cause but only by the affirmative vote of the holders of at least 66 2/3% of the voting power of the shares of the Corporation's stock entitled to vote for the election of directors. The number of directors constituting the full Board of Directors shall be such number as the Board of Directors from time to time may determine.

(b) The Board of Directors shall have the power and authority: (i) to adopt, amend or repeal By-Laws of the Corporation, subject only to such limitations, if any, as may be from time to time imposed by other provisions of this Certificate, by law, or by the By-Laws; and (ii) to the full extent permitted or not prohibited by law, and without the consent of or other action by the stockholders, to authorize or create mortgage, pledges or other liens or encumbrances upon any or all of the assets, real, personal or mixed, and franchises of the Corporation, including after-acquired property, and to exercise all of the powers of the Corporation in connection therewith.

E-3

# ARTICLE VI LIMITATION OF LIABILITY

No director of the Corporation shall be personally liable to the Corporation or to any of its stockholders for monetary damages for breach of fiduciary duty as a director, notwithstanding any provision of law imposing such liability; PROVIDED, HOWEVER, that to the extent required from time to time by applicable law, this Article VI shall not eliminate or limit the liability of a director, to the extent such liability is provided by applicable law, (i) for any breach of the director's duty of loyalty to the Corporation or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) under Section 174 of Title 8 of the DGCL, or (iv) for any transactions from which the director derived an improper personal benefit. No amendment to or repeal of this Article VI shall apply to or have any effect on the liability or alleged liability of any director for or with respect to any acts or omissions of such director occurring prior to the effective date of such amendment or repeal.

# ARTICLE VII INDEMNIFICATION

The Corporation shall, to the fullest extent permitted by Section 145 of the DGCL, as amended from time to time, indemnify each person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that he is or was, or has agreed to become, a director or officer of the Corporation, or is or was serving, or has agreed to serve, at the request of the Corporation, as a director, officer or trustee of, or in a similar capacity with, another corporation, partnership, joint venture, trust or other enterprise (including any employee benefit plan), or by reason of any action alleged to have been taken or omitted in such capacity, against all expenses (including attorneys' fees), judgements, fines and amounts paid in settlement actually and reasonably incurred by him or on his behalf in connection with such action, suit or proceeding and any appeal therefrom.

Indemnification may include payment by the Corporation of expenses in defending an action or proceeding in advance of the final disposition of such action or proceeding upon receipt of an undertaking by the person indemnified to repay such payment if it is ultimately determined that such person is not entitled to indemnification under this Article VII, which undertaking may be accepted without reference to the financial ability of such person to make such repayment.

The Corporation shall not indemnify any such person seeking indemnification in connection with a proceeding (or part thereof) initiated by such person unless the initiation thereof was approved by the Board of Directors.

The indemnification rights provided in this Article VII (i) shall not be deemed exclusive of any other rights to which those indemnified may be entitled under any law, agreement or vote of stockholders or disinterested directors or otherwise, and (ii) shall inure to the benefit of the heirs, executors and administrators of such persons. The Corporation may, to the extent authorized from time to time by its Board of Directors, grant indemnification rights to other employees or agents of the Corporation or other persons serving the Corporation and such rights may be equivalent to, or greater or less than, those set forth in this Article VII.

E-4

# ARTICLE VIII COMPROMISES AND ARRANGEMENTS

Whenever a compromise or arrangement is proposed between the Corporation and its creditors or any class of them and/or between the Corporation and its stockholders or any class of them, any court of equitable jurisdiction within the State of Delaware may, on the application in a summary way of the Corporation or of any creditor or stockholder thereof or on the application of any receiver or receivers appointed for the Corporation under Section 291 of Title 8 of the Delaware Code or on the application of trustees in dissolution or of any receiver or receivers appointed for the Corporation under Section 279 of Title 8 of the Delaware Code order a meeting of the creditors or class of creditors, and/or of the stockholders or class of stockholders of the Corporation, as the case may be, to be summoned in such manner as the said court directs. If a majority in number representing three-fourths (3/4ths) in value of the creditors or class of creditors, and/or of the stockholders or class of stockholders of the Corporation, as the case may be, agree to any compromise or arrangement and to any reorganization of the Corporation as consequence of such compromise or arrangement, the said compromise or arrangement and the said reorganization shall, if sanctioned by the court to which the said application has been made, be binding on all the creditors or class of creditors, and/or on all the stockholders or class of stockholders, of the Corporation, as the case may be, and also on the Corporation.

# ARTICLE IX CERTAIN TRANSACTIONS

The Board of Directors, when considering a tender offer or merger or acquisition proposal, may take into account factors in addition to potential economic benefits to stockholders, including without limitation (i) comparison of the proposed consideration to be received by stockholders in relation to the then current market price of the Corporation's capital stock, the estimated current value of the Corporation in a freely negotiated transaction, and the estimated future value of the Corporation as an independent entity, (ii) the impact of such a transaction on the employees, suppliers, and customers of the Corporation and its effect on the communities in which the Corporation operates, and (iii) the impact of such a transaction on the unique corporate culture and atmosphere of the Corporation.

ARTICLE X
STOCKHOLDER ACTION

Any action required or permitted to be taken by the stockholders of the Corporation may be taken either at a duly called annual or special meeting of the stockholders or by written consent in lieu of such a meeting, and special meetings of stockholders may be called only by the Chairman of the Board of Directors, the President, or a majority of the Board of Directors.

# ARTICLE XI SEVERABILITY

To the extent that any provision of this Restated Certificate of Incorporation is found to be invalid or unenforceable, such invalidity or unenforceability shall not affect the validity or enforceability of any other provision of this Restated Certificate of Incorporation, and following

E-5

any determination by a court of competent jurisdiction that any provision of this Restated Certificate of incorporation is invalid or unenforceable, this Restated Certificate of Incorporation shall contain only such provisions (i) as were in effect immediately prior to such determination and (ii) were not so determined to be invalid or unenforceable.

# ARTICLE XII AMENDMENTS

The affirmative vote of the holders of at least 66 2/3% of the voting power of the outstanding stock of the Corporation entitled to vote thereon (in addition to any separate class vote required by law or that may in the future be required pursuant to the terms of any outstanding Preferred Stock), voting together as a single class, shall be required to amend or repeal the provisions of Articles IV (to the extent it relates to the authority of the Board of Directors to issue shares of Preferred Stock in one or more series, the terms of which may be determined by the Board of Directors), V, VII, IX, X, or XII of this Restated Certificate of Incorporation or to reduce the numbers of authorized shares of Common Stock or Preferred Stock.

\* \* \*

E-6

EXHIBIT F

AGREED FORM

THE COMPANIES LAW (2016 REVISION)
OF THE CAYMAN ISLANDS
COMPANY LIMITED BY SHARES

# AMENDED AND RESTATED MEMORANDUM AND ARTICLES OF ASSOCIATION

OF

GENBAND Holdings Company (adopted by special resolution dated [\*] 2017)

THE COMPANIES LAW (2016 REVISION)
OF THE CAYMAN ISLANDS
COMPANY LIMITED BY SHARES

AMENDED AND RESTATED
MEMORANDUM OF ASSOCIATION
OF
GENBAND Holdings Company

GENBAND Holdings Company (adopted by special resolution dated [\*] 2017)

- 1 The name of the Company is **GENBAND Holdings Company**.
- The Registered Office of the Company shall be at the offices of Maples Corporate Services Limited, PO Box 309, Ugland House, Grand Cayman, KY1-1104, Cayman Islands, or at such other place within the Cayman Islands as the Directors may decide.
- The objects for which the Company is established are unrestricted and the Company shall have full power and authority to carry out any object not prohibited by the laws of the Cayman Islands.
- 4 The liability of each Member is limited to the amount unpaid on such Member's shares.
- The share capital of the Company is US\$100,000.00 divided into 10,000,000,000,000 shares of a par value of US\$0.00001 each.

- 6 The Company has power to register by way of continuation as a body corporate limited by shares under the laws of any jurisdiction outside the Cayman Islands and to be deregistered in the Cayman Islands.
- Capitalised terms that are not defined in this Memorandum of Association bear the respective meanings given to them in the Articles of Association of the Company.

F-1

# THE COMPANIES LAW (2016 REVISION) OF THE CAYMAN ISLANDS **COMPANY LIMITED BY SHARES**

# AMENDED AND RESTATED ARTICLES OF ASSOCIATION

**GENBAND Holdings Company** (adopted by special resolution dated [\*] 2017)

#### Interpretation 1

1.1 In the Articles Table A in the First Schedule to the Statute does not apply and, unless there is something in the subject or context inconsistent therewith:

"Articles" means these articles of association of the Company.

"Auditor" means the person for the time being performing the duties of auditor of the Company (if any).

"Company" means the above named company.

"Directors" means the directors for the time being of the Company.

"Dividend" means any dividend (whether interim or final) resolved to be paid on Shares pursuant to the Articles.

"Electronic Record" has the same meaning as in the Electronic Transactions Law.

"Electronic Transactions Law" means the Electronic Transactions Law (2003 Revision) of the Cayman Islands.

"Member" has the same meaning as in the Statute.

"Memorandum" means the memorandum of association of the Company.

"Ordinary Resolution" means a resolution passed by a simple majority of the Members as, being entitled to do so, vote in person or, where proxies are allowed, by proxy at a general meeting, and includes a unanimous

written resolution. In computing the majority when a poll is demanded regard shall be had to the

number of votes to which each Member is entitled by the Articles.

F-2

"Register of Members" means the register of Members maintained in accordance with the Statute and includes (except

where otherwise stated) any branch or duplicate register of Members.

"Registered Office" means the registered office for the time being of the Company.

"Seal" means the common seal of the Company and includes every duplicate seal.

"Share" means a share in the Company and includes a fraction of a share in the Company.

"Special Resolution" has the same meaning as in the Statute, and includes a unanimous written resolution.

"Statute" means the Companies Law (2016 Revision) of the Cayman Islands.

"Subscriber" means the subscriber to the Memorandum.

"Treasury Share" means a Share held in the name of the Company as a treasury share in accordance with the Statute.

#### 1.2 In the Articles:

- (a) words importing the singular number include the plural number and vice versa;
- (b) words importing the masculine gender include the feminine gender;
- words importing persons include corporations as well as any other legal or natural person; (c)

- (d) "written" and "in writing" include all modes of representing or reproducing words in visible form, including in the form of an Electronic Record;
- (e) "shall" shall be construed as imperative and "may" shall be construed as permissive;
- (f) references to provisions of any law or regulation shall be construed as references to those provisions as amended, modified, re-enacted or replaced;
- (g) any phrase introduced by the terms "including", "include", "in particular" or any similar expression shall be construed as illustrative and shall not limit the sense of the words preceding those terms;
- (h) the term "and/or" is used herein to mean both "and" as well as "or." The use of "and/or" in certain contexts in no respects qualifies or modifies the use of the terms "and" or "or" in others. The term "or" shall not be interpreted to be exclusive and the term "and" shall not be interpreted to require the conjunctive (in each case, unless the context otherwise requires);
- (i) headings are inserted for reference only and shall be ignored in construing the Articles;

F-3

- (j) any requirements as to delivery under the Articles include delivery in the form of an Electronic Record;
- (k) any requirements as to execution or signature under the Articles including the execution of the Articles themselves can be satisfied in the form of an electronic signature as defined in the Electronic Transactions Law;
- (l) sections 8 and 19(3) of the Electronic Transactions Law shall not apply;
- (m) the term "clear days" in relation to the period of a notice means that period excluding the day when the notice is received or deemed to be received and the day for which it is given or on which it is to take effect; and
- (n) the term "holder" in relation to a Share means a person whose name is entered in the Register of Members as the holder of such Share.

### 2 Commencement of Business

- 2.1 The business of the Company may be commenced as soon after incorporation of the Company as the Directors shall see fit.
- 2.2 The Directors may pay, out of the capital or any other monies of the Company, all expenses incurred in or about the formation and establishment of the Company, including the expenses of registration.

# 3 Issue of Shares

- 3.1 Subject to the provisions, if any, in the Memorandum (and to any direction that may be given by the Company in general meeting) and without prejudice to any rights attached to any existing Shares, the Directors may allot, issue, grant options over or otherwise dispose of Shares (including fractions of a Share) with or without preferred, deferred or other rights or restrictions, whether in regard to Dividend or other distribution, voting, return of capital or otherwise and to such persons, at such times and on such other terms as they think proper, and may also (subject to the Statute and the Articles) vary such rights.
- 3.2 The Company shall not issue Shares to bearer.

# 4 Register of Members

- 4.1 The Company shall maintain or cause to be maintained the Register of Members in accordance with the Statute.
- 4.2 The Directors may determine that the Company shall maintain one or more branch registers of Members in accordance with the Statute. The Directors may also determine which register of Members shall constitute the principal register and which shall constitute the branch register or registers, and to vary such determination from time to time.

# 5 Closing Register of Members or Fixing Record Date

5.1 For the purpose of determining Members entitled to notice of, or to vote at any meeting of Members or any adjournment thereof, or Members entitled to receive payment of any Dividend or

F-4

other distribution, or in order to make a determination of Members for any other purpose, the Directors may provide that the Register of Members shall be closed for transfers for a stated period which shall not in any case exceed forty days.

5.2 In lieu of, or apart from, closing the Register of Members, the Directors may fix in advance or arrears a date as the record date for any such determination of Members entitled to notice of, or to vote at any meeting of the Members or any adjournment thereof, or for the purpose of determining the Members entitled to receive payment of any Dividend or other distribution, or in order to make a determination of Members for any other purpose.

5.3 If the Register of Members is not so closed and no record date is fixed for the determination of Members entitled to notice of, or to vote at, a meeting of Members or Members entitled to receive payment of a Dividend or other distribution, the date on which notice of the meeting is sent or the date on which the resolution of the Directors resolving to pay such Dividend or other distribution is passed, as the case may be, shall be the record date for such determination of Members. When a determination of Members entitled to vote at any meeting of Members has been made as provided in this Article, such determination shall apply to any adjournment thereof.

# **6** Certificates for Shares

- A Member shall only be entitled to a share certificate if the Directors resolve that share certificates shall be issued. Share certificates representing Shares, if any, shall be in such form as the Directors may determine. Share certificates shall be signed by one or more Directors or other person authorised by the Directors. The Directors may authorise certificates to be issued with the authorised signature(s) affixed by mechanical process. All certificates for Shares shall be consecutively numbered or otherwise identified and shall specify the Shares to which they relate. All certificates surrendered to the Company for transfer shall be cancelled and subject to the Articles no new certificate shall be issued until the former certificate representing a like number of relevant Shares shall have been surrendered and cancelled.
- 6.2 The Company shall not be bound to issue more than one certificate for Shares held jointly by more than one person and delivery of a certificate to one joint holder shall be a sufficient delivery to all of them.
- 6.3 If a share certificate is defaced, worn out, lost or destroyed, it may be renewed on such terms (if any) as to evidence and indemnity and on the payment of such expenses reasonably incurred by the Company in investigating evidence, as the Directors may prescribe, and (in the case of defacement or wearing out) upon delivery of the old certificate.
- 6.4 Every share certificate sent in accordance with the Articles will be sent at the risk of the Member or other person entitled to the certificate. The Company will not be responsible for any share certificate lost or delayed in the course of delivery.

### 7 Transfer of Shares

7.1 Shares are transferable subject to the approval of the Directors by resolution who may, in their absolute discretion, decline to register any transfer of Shares without giving any reason. If the Directors refuse to register a transfer they shall notify the transferee within two months of such refusal.

F-5

7.2 The instrument of transfer of any Share shall be in writing and shall be executed by or on behalf of the transferor (and if the Directors so require, signed by or on behalf of the transferee). The transferor shall be deemed to remain the holder of a Share until the name of the transferee is entered in the Register of Members.

# 8 Redemption, Repurchase and Surrender of Shares

- 8.1 Subject to the provisions of the Statute the Company may issue Shares that are to be redeemed or are liable to be redeemed at the option of the Member or the Company. The redemption of such Shares shall be effected in such manner and upon such other terms as the Company may, by Special Resolution, determine before the issue of the Shares.
- 8.2 Subject to the provisions of the Statute, the Company may purchase its own Shares (including any redeemable Shares) in such manner and on such other terms as the Directors may agree with the relevant Member.
- 8.3 The Company may make a payment in respect of the redemption or purchase of its own Shares in any manner permitted by the Statute, including out of capital.
- 8.4 The Directors may accept the surrender for no consideration of any fully paid Share.

# 9 Treasury Shares

- 9.1 The Directors may, prior to the purchase, redemption or surrender of any Share, determine that such Share shall be held as a Treasury Share.
- 9.2 The Directors may determine to cancel a Treasury Share or transfer a Treasury Share on such terms as they think proper (including, without limitation, for nil consideration).

# 10 Variation of Rights of Shares

- 10.1 If at any time the share capital of the Company is divided into different classes of Shares, all or any of the rights attached to any class (unless otherwise provided by the terms of issue of the Shares of that class) may, whether or not the Company is being wound up, be varied without the consent of the holders of the issued Shares of that class where such variation is considered by the Directors not to have a material adverse effect upon such rights; otherwise, any such variation shall be made only with the consent in writing of the holders of not less than two thirds of the issued Shares of that class, or with the approval of a resolution passed by a majority of not less than two thirds of the votes cast at a separate meeting of the holders of the Shares of that class. For the avoidance of doubt, the Directors reserve the right, notwithstanding that any such variation may not have a material adverse effect, to obtain consent from the holders of Shares of the relevant class. To any such meeting all the provisions of the Articles relating to general meetings shall apply *mutatis mutandis*, except that the necessary quorum shall be one person holding or representing by proxy at least one third of the issued Shares of the class and that any holder of Shares of the class present in person or by proxy may demand a poll.
- 10.2 For the purposes of a separate class meeting, the Directors may treat two or more or all the classes of Shares as forming one class of Shares if the Directors consider that such class of Shares would be affected in the same way by the proposals under consideration, but in any other case shall treat them as separate classes of Shares.

10.3 The rights conferred upon the holders of the Shares of any class issued with preferred or other rights shall not, unless otherwise expressly provided by the terms of issue of the Shares of that class, be deemed to be varied by the creation or issue of further Shares ranking pari passu therewith.

# 11 Commission on Sale of Shares

The Company may, in so far as the Statute permits, pay a commission to any person in consideration of his subscribing or agreeing to subscribe (whether absolutely or conditionally) or procuring or agreeing to procure subscriptions (whether absolutely or conditionally) for any Shares. Such commissions may be satisfied by the payment of cash and/or the issue of fully or partly paid-up Shares. The Company may also on any issue of Shares pay such brokerage as may be lawful.

# 12 Non Recognition of Trusts

The Company shall not be bound by or compelled to recognise in any way (even when notified) any equitable, contingent, future or partial interest in any Share, or (except only as is otherwise provided by the Articles or the Statute) any other rights in respect of any Share other than an absolute right to the entirety thereof in the holder.

### 13 Lien on Shares

- 13.1 The Company shall have a first and paramount lien on all Shares (whether fully paid-up or not) registered in the name of a Member (whether solely or jointly with others) for all debts, liabilities or engagements to or with the Company (whether presently payable or not) by such Member or his estate, either alone or jointly with any other person, whether a Member or not, but the Directors may at any time declare any Share to be wholly or in part exempt from the provisions of this Article. The registration of a transfer of any such Share shall operate as a waiver of the Company's lien thereon. The Company's lien on a Share shall also extend to any amount payable in respect of that Share.
- 13.2 The Company may sell, in such manner as the Directors think fit, any Shares on which the Company has a lien, if a sum in respect of which the lien exists is presently payable, and is not paid within fourteen clear days after notice has been received or deemed to have been received by the holder of the Shares, or to the person entitled to it in consequence of the death or bankruptcy of the holder, demanding payment and stating that if the notice is not complied with the Shares may be sold.
- To give effect to any such sale the Directors may authorise any person to execute an instrument of transfer of the Shares sold to, or in accordance with the directions of, the purchaser. The purchaser or his nominee shall be registered as the holder of the Shares comprised in any such transfer, and he shall not be bound to see to the application of the purchase money, nor shall his title to the Shares be affected by any irregularity or invalidity in the sale or the exercise of the Company's power of sale under the Articles.
- 13.4 The net proceeds of such sale after payment of costs, shall be applied in payment of such part of the amount in respect of which the lien exists as is presently payable and any balance shall (subject to a like lien for sums not presently payable as existed upon the Shares before the sale) be paid to the person entitled to the Shares at the date of the sale.

F-7

# 14 Call on Shares

- Subject to the terms of the allotment and issue of any Shares, the Directors may make calls upon the Members in respect of any monies unpaid on their Shares (whether in respect of par value or premium), and each Member shall (subject to receiving at least fourteen clear days' notice specifying the time or times of payment) pay to the Company at the time or times so specified the amount called on the Shares. A call may be revoked or postponed, in whole or in part, as the Directors may determine. A call may be required to be paid by instalments. A person upon whom a call is made shall remain liable for calls made upon him notwithstanding the subsequent transfer of the Shares in respect of which the call was made.
- 14.2 A call shall be deemed to have been made at the time when the resolution of the Directors authorising such call was passed.
- 14.3 The joint holders of a Share shall be jointly and severally liable to pay all calls in respect thereof.
- If a call remains unpaid after it has become due and payable, the person from whom it is due shall pay interest on the amount unpaid from the day it became due and payable until it is paid at such rate as the Directors may determine (and in addition all expenses that have been incurred by the Company by reason of such non-payment), but the Directors may waive payment of the interest or expenses wholly or in part.
- An amount payable in respect of a Share on issue or allotment or at any fixed date, whether on account of the par value of the Share or premium or otherwise, shall be deemed to be a call and if it is not paid all the provisions of the Articles shall apply as if that amount had become due and payable by virtue of a call.
- 14.6 The Directors may issue Shares with different terms as to the amount and times of payment of calls, or the interest to be paid.
- 14.7 The Directors may, if they think fit, receive an amount from any Member willing to advance all or any part of the monies uncalled and unpaid upon any Shares held by him, and may (until the amount would otherwise become payable) pay interest at such rate as may be agreed upon between the Directors and the Member paying such amount in advance.
- 14.8 No such amount paid in advance of calls shall entitle the Member paying such amount to any portion of a Dividend or other distribution payable in respect of any period prior to the date upon which such amount would, but for such payment, become payable.

# 15 Forfeiture of Shares

- 15.1 If a call or instalment of a call remains unpaid after it has become due and payable the Directors may give to the person from whom it is due not less than fourteen clear days' notice requiring payment of the amount unpaid together with any interest which may have accrued and any expenses incurred by the Company by reason of such non-payment. The notice shall specify where payment is to be made and shall state that if the notice is not complied with the Shares in respect of which the call was made will be liable to be forfeited.
- 15.2 If the notice is not complied with, any Share in respect of which it was given may, before the payment required by the notice has been made, be forfeited by a resolution of the Directors.

F-8

Such forfeiture shall include all Dividends, other distributions or other monies payable in respect of the forfeited Share and not paid before the forfeiture.

- 15.3 A forfeited Share may be sold, re-allotted or otherwise disposed of on such terms and in such manner as the Directors think fit and at any time before a sale, re-allotment or disposition the forfeiture may be cancelled on such terms as the Directors think fit. Where for the purposes of its disposal a forfeited Share is to be transferred to any person the Directors may authorise some person to execute an instrument of transfer of the Share in favour of that person.
- A person any of whose Shares have been forfeited shall cease to be a Member in respect of them and shall surrender to the Company for cancellation the certificate for the Shares forfeited and shall remain liable to pay to the Company all monies which at the date of forfeiture were payable by him to the Company in respect of those Shares together with interest at such rate as the Directors may determine, but his liability shall cease if and when the Company shall have received payment in full of all monies due and payable by him in respect of those Shares.
- A certificate in writing under the hand of one Director or officer of the Company that a Share has been forfeited on a specified date shall be conclusive evidence of the facts stated in it as against all persons claiming to be entitled to the Share. The certificate shall (subject to the execution of an instrument of transfer) constitute a good title to the Share and the person to whom the Share is sold or otherwise disposed of shall not be bound to see to the application of the purchase money, if any, nor shall his title to the Share be affected by any irregularity or invalidity in the proceedings in reference to the forfeiture, sale or disposal of the Share.
- 15.6 The provisions of the Articles as to forfeiture shall apply in the case of non payment of any sum which, by the terms of issue of a Share, becomes payable at a fixed time, whether on account of the par value of the Share or by way of premium as if it had been payable by virtue of a call duly made and notified.

#### 16 Transmission of Shares

- 16.1 If a Member dies the survivor or survivors (where he was a joint holder) or his legal personal representatives (where he was a sole holder), shall be the only persons recognised by the Company as having any title to his Shares. The estate of a deceased Member is not thereby released from any liability in respect of any Share, for which he was a joint or sole holder.
- Any person becoming entitled to a Share in consequence of the death or bankruptcy or liquidation or dissolution of a Member (or in any other way than by transfer) may, upon such evidence being produced as may be required by the Directors, elect, by a notice in writing sent by him to the Company, either to become the holder of such Share or to have some person nominated by him registered as the holder of such Share. If he elects to have another person registered as the holder of such Share he shall sign an instrument of transfer of that Share to that person. The Directors shall, in either case, have the same right to decline or suspend registration as they would have had in the case of a transfer of the Share by the relevant Member before his death or bankruptcy or liquidation or dissolution, as the case may be.
- 16.3 A person becoming entitled to a Share by reason of the death or bankruptcy or liquidation or dissolution of a Member (or in any other case than by transfer) shall be entitled to the same Dividends, other distributions and other advantages to which he would be entitled if he were the holder of such Share. However, he shall not, before becoming a Member in respect of a Share, be entitled in respect of it to exercise any right conferred by membership in relation to general

F-9

meetings of the Company and the Directors may at any time give notice requiring any such person to elect either to be registered himself or to have some person nominated by him be registered as the holder of the Share (but the Directors shall, in either case, have the same right to decline or suspend registration as they would have had in the case of a transfer of the Share by the relevant Member before his death or bankruptcy or liquidation or dissolution or any other case than by transfer, as the case may be). If the notice is not complied with within ninety days of being received or deemed to be received (as determined pursuant to the Articles) the Directors may thereafter withhold payment of all Dividends, other distributions, bonuses or other monies payable in respect of the Share until the requirements of the notice have been complied with.

# 17 Amendments of Memorandum and Articles of Association and Alteration of Capital

- 17.1 The Company may by Ordinary Resolution:
  - (a) increase its share capital by such sum as the Ordinary Resolution shall prescribe and with such rights, priorities and privileges annexed thereto, as the Company in general meeting may determine;
  - (b) consolidate and divide all or any of its share capital into Shares of larger amount than its existing Shares;
  - (c) convert all or any of its paid-up Shares into stock, and reconvert that stock into paid-up Shares of any denomination;

- (d) by subdivision of its existing Shares or any of them divide the whole or any part of its share capital into Shares of smaller amount than is fixed by the Memorandum or into Shares without par value; and
- (e) cancel any Shares that at the date of the passing of the Ordinary Resolution have not been taken or agreed to be taken by any person and diminish the amount of its share capital by the amount of the Shares so cancelled.
- All new Shares created in accordance with the provisions of the preceding Article shall be subject to the same provisions of the Articles with reference to the payment of calls, liens, transfer, transmission, forfeiture and otherwise as the Shares in the original share capital.
- 17.3 Subject to the provisions of the Statute and the provisions of the Articles as regards the matters to be dealt with by Ordinary Resolution, the Company may by Special Resolution:
  - (a) change its name;
  - (b) alter or add to the Articles;
  - (c) alter or add to the Memorandum with respect to any objects, powers or other matters specified therein; and
  - (d) reduce its share capital or any capital redemption reserve fund.

F-10

### 18 Offices and Places of Business

Subject to the provisions of the Statute, the Company may by resolution of the Directors change the location of its Registered Office. The Company may, in addition to its Registered Office, maintain such other offices or places of business as the Directors determine.

### 19 General Meetings

- 19.1 All general meetings other than annual general meetings shall be called extraordinary general meetings.
- 19.2 The Company may, but shall not (unless required by the Statute) be obliged to, in each year hold a general meeting as its annual general meeting, and shall specify the meeting as such in the notices calling it. Any annual general meeting shall be held at such time and place as the Directors shall appoint and if no other time and place is prescribed by them, it shall be held at the Registered Office on the second Wednesday in December of each year at ten o'clock in the morning. At these meetings the report of the Directors (if any) shall be presented.
- 19.3 The Directors may call general meetings, and they shall on a Members' requisition forthwith proceed to convene an extraordinary general meeting of the Company.
- 19.4 A Members' requisition is a requisition of Members holding at the date of deposit of the requisition not less than ten per cent. in par value of the issued Shares which as at that date carry the right to vote at general meetings of the Company.
- 19.5 The Members' requisition must state the objects of the meeting and must be signed by the requisitionists and deposited at the Registered Office, and may consist of several documents in like form each signed by one or more requisitionists.
- 19.6 If there are no Directors as at the date of the deposit of the Members' requisition or if the Directors do not within twenty-one days from the date of the deposit of the Members' requisition duly proceed to convene a general meeting to be held within a further twenty-one days, the requisitionists, or any of them representing more than one-half of the total voting rights of all of the requisitionists, may themselves convene a general meeting, but any meeting so convened shall be held no later than the day which falls three months after the expiration of the said twenty-one day period.
- 19.7 A general meeting convened as aforesaid by requisitionists shall be convened in the same manner as nearly as possible as that in which general meetings are to be convened by Directors.

# 20 Notice of General Meetings

At least five clear days' notice shall be given of any general meeting. Every notice shall specify the place, the day and the hour of the meeting and the general nature of the business to be conducted at the general meeting and shall be given in the manner hereinafter mentioned or in such other manner if any as may be prescribed by the Company, provided that a general meeting of the Company shall, whether or not the notice specified in this Article has been given and whether or not the provisions of the Articles regarding general meetings have been complied with, be deemed to have been duly convened if it is so agreed:

F-11

- (a) in the case of an annual general meeting, by all of the Members entitled to attend and vote thereat; and
- (b) in the case of an extraordinary general meeting, by a majority in number of the Members having a right to attend and vote at the meeting, together holding not less than ninety five per cent. in par value of the Shares giving that right.
- 20.2 The accidental omission to give notice of a general meeting to, or the non receipt of notice of a general meeting by, any person entitled to receive such notice shall not invalidate the proceedings of that general meeting.

### 21 Proceedings at General Meetings

- 21.1 No business shall be transacted at any general meeting unless a quorum is present. Two Members being individuals present in person or by proxy or if a corporation or other non-natural person by its duly authorised representative or proxy shall be a quorum unless the Company has only one Member entitled to vote at such general meeting in which case the quorum shall be that one Member present in person or by proxy or (in the case of a corporation or other non-natural person) by its duly authorised representative or proxy.
- A person may participate at a general meeting by conference telephone or other communications equipment by means of which all the persons participating in the meeting can communicate with each other. Participation by a person in a general meeting in this manner is treated as presence in person at that meeting.
- 21.3 A resolution (including a Special Resolution) in writing (in one or more counterparts) signed by or on behalf of all of the Members for the time being entitled to receive notice of and to attend and vote at general meetings (or, being corporations or other non-natural persons, signed by their duly authorised representatives) shall be as valid and effective as if the resolution had been passed at a general meeting of the Company duly convened and held.
- 21.4 If a quorum is not present within half an hour from the time appointed for the meeting to commence or if during such a meeting a quorum ceases to be present, the meeting, if convened upon a Members' requisition, shall be dissolved and in any other case it shall stand adjourned to the same day in the next week at the same time and/or place or to such other day, time and/or place as the Directors may determine, and if at the adjourned meeting a quorum is not present within half an hour from the time appointed for the meeting to commence, the Members present shall be a quorum.
- 21.5 The Directors may, at any time prior to the time appointed for the meeting to commence, appoint any person to act as chairman of a general meeting of the Company or, if the Directors do not make any such appointment, the chairman, if any, of the board of Directors shall preside as chairman at such general meeting. If there is no such chairman, or if he shall not be present within fifteen minutes after the time appointed for the meeting to commence, or is unwilling to act, the Directors present shall elect one of their number to be chairman of the meeting.
- 21.6 If no Director is willing to act as chairman or if no Director is present within fifteen minutes after the time appointed for the meeting to commence, the Members present shall choose one of their number to be chairman of the meeting.

F-12

- 21.7 The chairman may, with the consent of a meeting at which a quorum is present (and shall if so directed by the meeting) adjourn the meeting from time to time and from place to place, but no business shall be transacted at any adjourned meeting other than the business left unfinished at the meeting from which the adjournment took place.
- When a general meeting is adjourned for thirty days or more, notice of the adjourned meeting shall be given as in the case of an original meeting.

  Otherwise it shall not be necessary to give any such notice of an adjourned meeting.
- A resolution put to the vote of the meeting shall be decided on a show of hands unless before, or on the declaration of the result of, the show of hands, the chairman demands a poll, or any other Member or Members collectively present in person or by proxy (or in the case of a corporation or other non-natural person, by its duly authorised representative or proxy) and holding at least ten per cent. in par value of the Shares giving a right to attend and vote at the meeting demand a poll.
- 21.10 Unless a poll is duly demanded and the demand is not withdrawn a declaration by the chairman that a resolution has been carried or carried unanimously, or by a particular majority, or lost or not carried by a particular majority, an entry to that effect in the minutes of the proceedings of the meeting shall be conclusive evidence of that fact without proof of the number or proportion of the votes recorded in favour of or against such resolution.
- 21.11 The demand for a poll may be withdrawn.
- 21.12 Except on a poll demanded on the election of a chairman or on a question of adjournment, a poll shall be taken as the chairman directs, and the result of the poll shall be deemed to be the resolution of the general meeting at which the poll was demanded.
- A poll demanded on the election of a chairman or on a question of adjournment shall be taken forthwith. A poll demanded on any other question shall be taken at such date, time and place as the chairman of the general meeting directs, and any business other than that upon which a poll has been demanded or is contingent thereon may proceed pending the taking of the poll.
- 21.14 In the case of an equality of votes, whether on a show of hands or on a poll, the chairman shall be entitled to a second or casting vote.

# 22 Votes of Members

- Subject to any rights or restrictions attached to any Shares, on a show of hands every Member who (being an individual) is present in person or by proxy or, if a corporation or other non-natural person is present by its duly authorised representative or by proxy, shall have one vote and on a poll every Member present in any such manner shall have one vote for every Share of which he is the holder.
- In the case of joint holders the vote of the senior holder who tenders a vote, whether in person or by proxy (or, in the case of a corporation or other non-natural person, by its duly authorised representative or proxy), shall be accepted to the exclusion of the votes of the other joint holders, and seniority shall be determined by the order in which the names of the holders stand in the Register of Members.

F-13

committee, receiver, curator bonis or other person may vote by proxy.

- 22.4 No person shall be entitled to vote at any general meeting unless he is registered as a Member on the record date for such meeting nor unless all calls or other monies then payable by him in respect of Shares have been paid.
- 22.5 No objection shall be raised as to the qualification of any voter except at the general meeting or adjourned general meeting at which the vote objected to is given or tendered and every vote not disallowed at the meeting shall be valid. Any objection made in due time in accordance with this Article shall be referred to the chairman whose decision shall be final and conclusive.
- On a poll or on a show of hands votes may be cast either personally or by proxy (or in the case of a corporation or other non-natural person by its duly authorised representative or proxy). A Member may appoint more than one proxy or the same proxy under one or more instruments to attend and vote at a meeting. Where a Member appoints more than one proxy the instrument of proxy shall state which proxy is entitled to vote on a show of hands and shall specify the number of Shares in respect of which each proxy is entitled to exercise the related votes.
- On a poll, a Member holding more than one Share need not cast the votes in respect of his Shares in the same way on any resolution and therefore may vote a Share or some or all such Shares either for or against a resolution and/or abstain from voting a Share or some or all of the Shares and, subject to the terms of the instrument appointing him, a proxy appointed under one or more instruments may vote a Share or some or all of the Shares in respect of which he is appointed either for or against a resolution and/or abstain from voting a Share or some or all of the Shares in respect of which he is appointed.

### 23 Proxies

- 23.1 The instrument appointing a proxy shall be in writing and shall be executed under the hand of the appointor or of his attorney duly authorised in writing, or, if the appointor is a corporation or other non natural person, under the hand of its duly authorised representative. A proxy need not be a Member.
- 23.2 The Directors may, in the notice convening any meeting or adjourned meeting, or in an instrument of proxy sent out by the Company, specify the manner by which the instrument appointing a proxy shall be deposited and the place and the time (being not later than the time appointed for the commencement of the meeting or adjourned meeting to which the proxy relates) at which the instrument appointing a proxy shall be deposited. In the absence of any such direction from the Directors in the notice convening any meeting or adjourned meeting or in an instrument of proxy sent out by the Company, the instrument appointing a proxy shall be deposited physically at the Registered Office not less than 48 hours before the time appointed for the meeting or adjourned meeting to commence at which the person named in the instrument proposes to vote.
- 23.3 The chairman may in any event at his discretion declare that an instrument of proxy shall be deemed to have been duly deposited. An instrument of proxy that is not deposited in the manner permitted, or which has not been declared to have been duly deposited by the chairman, shall be invalid.

F-14

- 23.4 The instrument appointing a proxy may be in any usual or common form (or such other form as the Directors may approve) and may be expressed to be for a particular meeting or any adjournment thereof or generally until revoked. An instrument appointing a proxy shall be deemed to include the power to demand or join or concur in demanding a poll.
- 23.5 Votes given in accordance with the terms of an instrument of proxy shall be valid notwithstanding the previous death or insanity of the principal or revocation of the proxy or of the authority under which the proxy was executed, or the transfer of the Share in respect of which the proxy is given unless notice in writing of such death, insanity, revocation or transfer was received by the Company at the Registered Office before the commencement of the general meeting, or adjourned meeting at which it is sought to use the proxy.

# 24 Corporate Members

Any corporation or other non-natural person which is a Member may in accordance with its constitutional documents, or in the absence of such provision by resolution of its directors or other governing body, authorise such person as it thinks fit to act as its representative at any meeting of the Company or of any class of Members, and the person so authorised shall be entitled to exercise the same powers on behalf of the corporation which he represents as the corporation could exercise if it were an individual Member.

# 25 Shares that May Not be Voted

Shares in the Company that are beneficially owned by the Company shall not be voted, directly or indirectly, at any meeting and shall not be counted in determining the total number of outstanding Shares at any given time.

### 26 Directors

There shall be a board of Directors consisting of not less than one person (exclusive of alternate Directors) provided however that the Company may by Ordinary Resolution increase or reduce the limits in the number of Directors.

## 27 Powers of Directors

- 27.1 Subject to the provisions of the Statute, the Memorandum and the Articles and to any directions given by Special Resolution, the business of the Company shall be managed by the Directors who may exercise all the powers of the Company. No alteration of the Memorandum or Articles and no such direction shall invalidate any prior act of the Directors which would have been valid if that alteration had not been made or that direction had not been given. A duly convened meeting of Directors at which a quorum is present may exercise all powers exercisable by the Directors.
- All cheques, promissory notes, drafts, bills of exchange and other negotiable or transferable instruments and all receipts for monies paid to the Company shall be signed, drawn, accepted, endorsed or otherwise executed as the case may be in such manner as the Directors shall determine by resolution.

27.3 The Directors on behalf of the Company may pay a gratuity or pension or allowance on retirement to any Director who has held any other salaried office or place of profit with the Company or to his widow or dependants and may make contributions to any fund and pay premiums for the purchase or provision of any such gratuity, pension or allowance.

F-15

27.4 The Directors may exercise all the powers of the Company to borrow money and to mortgage or charge its undertaking, property and assets (present and future) and uncalled capital or any part thereof and to issue debentures, debenture stock, mortgages, bonds and other such securities whether outright or as security for any debt, liability or obligation of the Company or of any third party.

### 28 Appointment and Removal of Directors

- 28.1 The Company may by Ordinary Resolution appoint any person to be a Director or may by Ordinary Resolution remove any Director.
- 28.2 The Directors may appoint any person to be a Director, either to fill a vacancy or as an additional Director provided that the appointment does not cause the number of Directors to exceed any number fixed by or in accordance with the Articles as the maximum number of Directors.

### 29 Vacation of Office of Director

The office of a Director shall be vacated if:

- (a) the Director gives notice in writing to the Company that he resigns the office of Director; or
- (b) the Director absents himself (for the avoidance of doubt, without being represented by proxy or an alternate Director appointed by him) from three consecutive meetings of the board of Directors without special leave of absence from the Directors, and the Directors pass a resolution that he has by reason of such absence vacated office; or
- (c) the Director dies, becomes bankrupt or makes any arrangement or composition with his creditors generally; or
- (d) the Director is found to be or becomes of unsound mind; or
- (e) all of the other Directors (being not less than two in number) determine that he should be removed as a Director, either by a resolution passed by all of the other Directors at a meeting of the Directors duly convened and held in accordance with the Articles or by a resolution in writing signed by all of the other Directors.

# 30 Proceedings of Directors

- 30.1 The quorum for the transaction of the business of the Directors may be fixed by the Directors, and unless so fixed shall be two if there are two or more Directors, and shall be one if there is only one Director. A person who holds office as an alternate Director shall, if his appointor is not present, be counted in the quorum. A Director who also acts as an alternate Director shall, if his appointor is not present, count twice towards the quorum.
- 30.2 Subject to the provisions of the Articles, the Directors may regulate their proceedings as they think fit. Questions arising at any meeting shall be decided by a majority of votes. In the case of an equality of votes, the chairman shall have a second or casting vote. A Director who is also an alternate Director shall be entitled in the absence of his appointor to a separate vote on behalf of his appointor in addition to his own vote.

F-16

- A person may participate in a meeting of the Directors or any committee of Directors by conference telephone or other communications equipment by means of which all the persons participating in the meeting can communicate with each other at the same time. Participation by a person in a meeting in this manner is treated as presence in person at that meeting. Unless otherwise determined by the Directors the meeting shall be deemed to be held at the place where the chairman is located at the start of the meeting.
- A resolution in writing (in one or more counterparts) signed by all the Directors or all the members of a committee of the Directors or, in the case of a resolution in writing relating to the removal of any Director or the vacation of office by any Director, all of the Directors other than the Director who is the subject of such resolution (an alternate Director being entitled to sign such a resolution on behalf of his appointor and if such alternate Director is also a Director, being entitled to sign such resolution both on behalf of his appointer and in his capacity as a Director) shall be as valid and effectual as if it had been passed at a meeting of the Directors, or committee of Directors as the case may be, duly convened and held.
- A Director or alternate Director may, or other officer of the Company on the direction of a Director or alternate Director shall, call a meeting of the Directors by at least two days' notice in writing to every Director and alternate Director which notice shall set forth the general nature of the business to be considered unless notice is waived by all the Directors (or their alternates) either at, before or after the meeting is held. To any such notice of a meeting of the Directors all the provisions of the Articles relating to the giving of notices by the Company to the Members shall apply *mutatis mutandis*.
- 30.6 The continuing Directors (or a sole continuing Director, as the case may be) may act notwithstanding any vacancy in their body, but if and so long as their number is reduced below the number fixed by or pursuant to the Articles as the necessary quorum of Directors the continuing Directors or Director may act for the purpose of increasing the number of Directors to be equal to such fixed number, or of summoning a general meeting of the Company, but for no other purpose.
- 30.7 The Directors may elect a chairman of their board and determine the period for which he is to hold office; but if no such chairman is elected, or if at any meeting the chairman is not present within five minutes after the time appointed for the meeting to commence, the Directors present may choose

one of their number to be chairman of the meeting.

- 30.8 All acts done by any meeting of the Directors or of a committee of the Directors (including any person acting as an alternate Director) shall, notwithstanding that it is afterwards discovered that there was some defect in the appointment of any Director or alternate Director, and/or that they or any of them were disqualified, and/or had vacated their office and/or were not entitled to vote, be as valid as if every such person had been duly appointed and/or not disqualified to be a Director or alternate Director and/or had not vacated their office and/or had been entitled to vote, as the case may be.
- 30.9 A Director but not an alternate Director may be represented at any meetings of the board of Directors by a proxy appointed in writing by him. The proxy shall count towards the quorum and the vote of the proxy shall for all purposes be deemed to be that of the appointing Director.

F-17

# 31 Presumption of Assent

A Director or alternate Director who is present at a meeting of the board of Directors at which action on any Company matter is taken shall be presumed to have assented to the action taken unless his dissent shall be entered in the minutes of the meeting or unless he shall file his written dissent from such action with the person acting as the chairman or secretary of the meeting before the adjournment thereof or shall forward such dissent by registered post to such person immediately after the adjournment of the meeting. Such right to dissent shall not apply to a Director or alternate Director who voted in favour of such action.

#### 32 Directors' Interests

- 32.1 A Director or alternate Director may hold any other office or place of profit under the Company (other than the office of Auditor) in conjunction with his office of Director for such period and on such terms as to remuneration and otherwise as the Directors may determine.
- 32.2 A Director or alternate Director may act by himself or by, through or on behalf of his firm in a professional capacity for the Company and he or his firm shall be entitled to remuneration for professional services as if he were not a Director or alternate Director.
- 32.3 A Director or alternate Director may be or become a director or other officer of or otherwise interested in any company promoted by the Company or in which the Company may be interested as a shareholder, a contracting party or otherwise, and no such Director or alternate Director shall be accountable to the Company for any remuneration or other benefits received by him as a director or officer of, or from his interest in, such other company.
- 32.4 No person shall be disqualified from the office of Director or alternate Director or prevented by such office from contracting with the Company, either as vendor, purchaser or otherwise, nor shall any such contract or any contract or transaction entered into by or on behalf of the Company in which any Director or alternate Director shall be in any way interested be or be liable to be avoided, nor shall any Director or alternate Director so contracting or being so interested be liable to account to the Company for any profit realised by or arising in connection with any such contract or transaction by reason of such Director or alternate Director holding office or of the fiduciary relationship thereby established. A Director (or his alternate Director in his absence) shall be at liberty to vote in respect of any contract or transaction in which he is interested provided that the nature of the interest of any Director or alternate Director in any such contract or transaction shall be disclosed by him at or prior to its consideration and any vote thereon.
- 32.5 A general notice that a Director or alternate Director is a shareholder, director, officer or employee of any specified firm or company and is to be regarded as interested in any transaction with such firm or company shall be sufficient disclosure for the purposes of voting on a resolution in respect of a contract or transaction in which he has an interest, and after such general notice it shall not be necessary to give special notice relating to any particular transaction.

## 33 Minutes

The Directors shall cause minutes to be made in books kept for the purpose of recording all appointments of officers made by the Directors, all proceedings at meetings of the Company or the holders of any class of Shares and of the Directors, and of committees of the Directors, including the names of the Directors or alternate Directors present at each meeting.

F-18

### 34 Delegation of Directors' Powers

- 34.1 The Directors may delegate any of their powers, authorities and discretions, including the power to sub-delegate, to any committee consisting of one or more Directors. They may also delegate to any managing director or any Director holding any other executive office such of their powers, authorities and discretions as they consider desirable to be exercised by him provided that an alternate Director may not act as managing director and the appointment of a managing director shall be revoked forthwith if he ceases to be a Director. Any such delegation may be made subject to any conditions the Directors may impose and either collaterally with or to the exclusion of their own powers and any such delegation may be revoked or altered by the Directors. Subject to any such conditions, the proceedings of a committee of Directors shall be governed by the Articles regulating the proceedings of Directors, so far as they are capable of applying.
- 34.2 The Directors may establish any committees, local boards or agencies or appoint any person to be a manager or agent for managing the affairs of the Company and may appoint any person to be a member of such committees, local boards or agencies. Any such appointment may be made subject to any conditions the Directors may impose, and either collaterally with or to the exclusion of their own powers and any such appointment may be revoked or altered by the Directors. Subject to any such conditions, the proceedings of any such committee, local board or agency shall be governed by the Articles regulating the proceedings of Directors, so far as they are capable of applying.

- 34.3 The Directors may by power of attorney or otherwise appoint any person to be the agent of the Company on such conditions as the Directors may determine, provided that the delegation is not to the exclusion of their own powers and may be revoked by the Directors at any time.
- 34.4 The Directors may by power of attorney or otherwise appoint any company, firm, person or body of persons, whether nominated directly or indirectly by the Directors, to be the attorney or authorised signatory of the Company for such purpose and with such powers, authorities and discretions (not exceeding those vested in or exercisable by the Directors under the Articles) and for such period and subject to such conditions as they may think fit, and any such powers of attorney or other appointment may contain such provisions for the protection and convenience of persons dealing with any such attorneys or authorised signatories as the Directors may think fit and may also authorise any such attorney or authorised signatory to delegate all or any of the powers, authorities and discretions vested in him.
- 34.5 The Directors may appoint such officers of the Company (including, for the avoidance of doubt and without limitation, any secretary) as they consider necessary on such terms, at such remuneration and to perform such duties, and subject to such provisions as to disqualification and removal as the Directors may think fit. Unless otherwise specified in the terms of his appointment an officer of the Company may be removed by resolution of the Directors or Members. An officer of the Company may vacate his office at any time if he gives notice in writing to the Company that he resigns his office.

### 35 Alternate Directors

Any Director (but not an alternate Director) may by writing appoint any other Director, or any other person willing to act, to be an alternate Director and by writing may remove from office an alternate Director so appointed by him.

F-19

- An alternate Director shall be entitled to receive notice of all meetings of Directors and of all meetings of committees of Directors of which his appointor is a member, to attend and vote at every such meeting at which the Director appointing him is not personally present, to sign any written resolution of the Directors, and generally to perform all the functions of his appointor as a Director in his absence.
- 35.3 An alternate Director shall cease to be an alternate Director if his appointor ceases to be a Director.
- 35.4 Any appointment or removal of an alternate Director shall be by notice to the Company signed by the Director making or revoking the appointment or in any other manner approved by the Directors.
- 35.5 Subject to the provisions of the Articles, an alternate Director shall be deemed for all purposes to be a Director and shall alone be responsible for his own acts and defaults and shall not be deemed to be the agent of the Director appointing him.

# 36 No Minimum Shareholding

The Company in general meeting may fix a minimum shareholding required to be held by a Director, but unless and until such a shareholding qualification is fixed a Director is not required to hold Shares.

### 37 Remuneration of Directors

- 37.1 The remuneration to be paid to the Directors, if any, shall be such remuneration as the Directors shall determine. The Directors shall also be entitled to be paid all travelling, hotel and other expenses properly incurred by them in connection with their attendance at meetings of Directors or committees of Directors, or general meetings of the Company, or separate meetings of the holders of any class of Shares or debentures of the Company, or otherwise in connection with the business of the Company or the discharge of their duties as a Director, or to receive a fixed allowance in respect thereof as may be determined by the Directors, or a combination partly of one such method and partly the other.
- 37.2 The Directors may by resolution approve additional remuneration to any Director for any services which in the opinion of the Directors go beyond his ordinary routine work as a Director. Any fees paid to a Director who is also counsel, attorney or solicitor to the Company, or otherwise serves it in a professional capacity shall be in addition to his remuneration as a Director.

### 38 Seal

- 38.1 The Company may, if the Directors so determine, have a Seal. The Seal shall only be used by the authority of the Directors or of a committee of the Directors authorised by the Directors. Every instrument to which the Seal has been affixed shall be signed by at least one person who shall be either a Director or some officer of the Company or other person appointed by the Directors for the purpose.
- 38.2 The Company may have for use in any place or places outside the Cayman Islands a duplicate Seal or Seals each of which shall be a facsimile of the common Seal of the Company and, if the Directors so determine, with the addition on its face of the name of every place where it is to be used.

F-20

A Director or officer, representative or attorney of the Company may without further authority of the Directors affix the Seal over his signature alone to any document of the Company required to be authenticated by him under seal or to be filed with the Registrar of Companies in the Cayman Islands or elsewhere wheresoever.

# 39 Dividends, Distributions and Reserve

39.1 Subject to the Statute and this Article and except as otherwise provided by the rights attached to any Shares, the Directors may resolve to pay Dividends and other distributions on Shares in issue and authorise payment of the Dividends or other distributions out of the funds of the Company lawfully available therefor. A Dividend shall be deemed to be an interim Dividend unless the terms of the resolution pursuant to which the Directors

- resolve to pay such Dividend specifically state that such Dividend shall be a final Dividend. No Dividend or other distribution shall be paid except out of the realised or unrealised profits of the Company, out of the share premium account or as otherwise permitted by law.
- 39.2 Except as otherwise provided by the rights attached to any Shares, all Dividends and other distributions shall be paid according to the par value of the Shares that a Member holds. If any Share is issued on terms providing that it shall rank for Dividend as from a particular date, that Share shall rank for Dividend accordingly.
- 39.3 The Directors may deduct from any Dividend or other distribution payable to any Member all sums of money (if any) then payable by him to the Company on account of calls or otherwise.
- 39.4 The Directors may resolve that any Dividend or other distribution be paid wholly or partly by the distribution of specific assets and in particular (but without limitation) by the distribution of shares, debentures, or securities of any other company or in any one or more of such ways and where any difficulty arises in regard to such distribution, the Directors may settle the same as they think expedient and in particular may issue fractional Shares and may fix the value for distribution of such specific assets or any part thereof and may determine that cash payments shall be made to any Members upon the basis of the value so fixed in order to adjust the rights of all Members and may vest any such specific assets in trustees in such manner as may seem expedient to the Directors.
- Except as otherwise provided by the rights attached to any Shares, Dividends and other distributions may be paid in any currency. The Directors may determine the basis of conversion for any currency conversions that may be required and how any costs involved are to be met.
- 39.6 The Directors may, before resolving to pay any Dividend or other distribution, set aside such sums as they think proper as a reserve or reserves which shall, at the discretion of the Directors, be applicable for any purpose of the Company and pending such application may, at the discretion of the Directors, be employed in the business of the Company.
- Any Dividend, other distribution, interest or other monies payable in cash in respect of Shares may be paid by wire transfer to the holder or by cheque or warrant sent through the post directed to the registered address of the holder or, in the case of joint holders, to the registered address of the holder who is first named on the Register of Members or to such person and to such address as such holder or joint holders may in writing direct. Every such cheque or warrant shall be made payable to the order of the person to whom it is sent. Any one of two or more joint holders may give effectual receipts for any Dividends, other distributions, bonuses, or other monies payable in respect of the Share held by them as joint holders.

F-21

- 39.8 No Dividend or other distribution shall bear interest against the Company.
- Any Dividend or other distribution which cannot be paid to a Member and/or which remains unclaimed after six months from the date on which such Dividend or other distribution becomes payable may, in the discretion of the Directors, be paid into a separate account in the Company's name, provided that the Company shall not be constituted as a trustee in respect of that account and the Dividend or other distribution shall remain as a debt due to the Member. Any Dividend or other distribution which remains unclaimed after a period of six years from the date on which such Dividend or other distribution becomes payable shall be forfeited and shall revert to the Company.

# 40 Capitalisation

The Directors may at any time capitalise any sum standing to the credit of any of the Company's reserve accounts or funds (including the share premium account and capital redemption reserve fund) or any sum standing to the credit of the profit and loss account or otherwise available for distribution; appropriate such sum to Members in the proportions in which such sum would have been divisible amongst such Members had the same been a distribution of profits by way of Dividend or other distribution; and apply such sum on their behalf in paying up in full unissued Shares for allotment and distribution credited as fully paid-up to and amongst them in the proportion aforesaid. In such event the Directors shall do all acts and things required to give effect to such capitalisation, with full power given to the Directors to make such provisions as they think fit in the case of Shares becoming distributable in fractions (including provisions whereby the benefit of fractional entitlements accrue to the Company rather than to the Members concerned). The Directors may authorise any person to enter on behalf of all of the Members interested into an agreement with the Company providing for such capitalisation and matters incidental or relating thereto and any agreement made under such authority shall be effective and binding on all such Members and the Company.

### 41 Books of Account

- 41.1 The Directors shall cause proper books of account (including, where applicable, material underlying documentation including contracts and invoices) to be kept with respect to all sums of money received and expended by the Company and the matters in respect of which the receipt or expenditure takes place, all sales and purchases of goods by the Company and the assets and liabilities of the Company. Such books of account must be retained for a minimum period of five years from the date on which they are prepared. Proper books shall not be deemed to be kept if there are not kept such books of account as are necessary to give a true and fair view of the state of the Company's affairs and to explain its transactions.
- 41.2 The Directors shall determine whether and to what extent and at what times and places and under what conditions or regulations the accounts and books of the Company or any of them shall be open to the inspection of Members not being Directors and no Member (not being a Director) shall have any right of inspecting any account or book or document of the Company except as conferred by Statute or authorised by the Directors or by the Company in general meeting.
- 41.3 The Directors may cause to be prepared and to be laid before the Company in general meeting profit and loss accounts, balance sheets, group accounts (if any) and such other reports and accounts as may be required by law.

- 42.1 The Directors may appoint an Auditor of the Company who shall hold office on such terms as the Directors determine.
- 42.2 Every Auditor of the Company shall have a right of access at all times to the books and accounts and vouchers of the Company and shall be entitled to require from the Directors and officers of the Company such information and explanation as may be necessary for the performance of the duties of the Auditor.
- 42.3 Auditors shall, if so required by the Directors, make a report on the accounts of the Company during their tenure of office at the next annual general meeting following their appointment in the case of a company which is registered with the Registrar of Companies as an ordinary company, and at the next extraordinary general meeting following their appointment in the case of a company which is registered with the Registrar of Companies as an exempted company, and at any other time during their term of office, upon request of the Directors or any general meeting of the Members.

#### 43 Notices

- 43.1 Notices shall be in writing and may be given by the Company to any Member either personally or by sending it by courier, post, cable, telex, fax or e-mail to him or to his address as shown in the Register of Members (or where the notice is given by e-mail by sending it to the e-mail address provided by such Member). Any notice, if posted from one country to another, is to be sent by airmail.
- Where a notice is sent by courier, service of the notice shall be deemed to be effected by delivery of the notice to a courier company, and shall be deemed to have been received on the third day (not including Saturdays or Sundays or public holidays) following the day on which the notice was delivered to the courier. Where a notice is sent by post, service of the notice shall be deemed to be effected by properly addressing, pre paying and posting a letter containing the notice, and shall be deemed to have been received on the fifth day (not including Saturdays or Sundays or public holidays in the Cayman Islands) following the day on which the notice was posted. Where a notice is sent by cable, telex or fax, service of the notice shall be deemed to be effected by properly addressing and sending such notice and shall be deemed to have been received on the same day that it was transmitted. Where a notice is given by e-mail service shall be deemed to be effected by transmitting the e-mail to the e-mail address provided by the intended recipient and shall be deemed to have been received on the same day that it was sent, and it shall not be necessary for the receipt of the e-mail to be acknowledged by the recipient.
- 43.3 A notice may be given by the Company to the person or persons which the Company has been advised are entitled to a Share or Shares in consequence of the death or bankruptcy of a Member in the same manner as other notices which are required to be given under the Articles and shall be addressed to them by name, or by the title of representatives of the deceased, or trustee of the bankrupt, or by any like description at the address supplied for that purpose by the persons claiming to be so entitled, or at the option of the Company by giving the notice in any manner in which the same might have been given if the death or bankruptcy had not occurred.
- 43.4 Notice of every general meeting shall be given in any manner authorised by the Articles to every holder of Shares carrying an entitlement to receive such notice on the record date for such meeting except that in the case of joint holders the notice shall be sufficient if given to the joint

F-23

holder first named in the Register of Members and every person upon whom the ownership of a Share devolves by reason of his being a legal personal representative or a trustee in bankruptcy of a Member where the Member but for his death or bankruptcy would be entitled to receive notice of the meeting, and no other person shall be entitled to receive notices of general meetings.

### 44 Winding Up

- 44.1 If the Company shall be wound up the liquidator shall apply the assets of the Company in satisfaction of creditors' claims in such manner and order as such liquidator thinks fit. Subject to the rights attaching to any Shares, in a winding up:
  - (a) if the assets available for distribution amongst the Members shall be insufficient to repay the whole of the Company's issued share capital, such assets shall be distributed so that, as nearly as may be, the losses shall be borne by the Members in proportion to the par value of the Shares held by them; or
  - (b) if the assets available for distribution amongst the Members shall be more than sufficient to repay the whole of the Company's issued share capital at the commencement of the winding up, the surplus shall be distributed amongst the Members in proportion to the par value of the Shares held by them at the commencement of the winding up subject to a deduction from those Shares in respect of which there are monies due, of all monies payable to the Company for unpaid calls or otherwise.
- 44.2 If the Company shall be wound up the liquidator may, subject to the rights attaching to any Shares and with the approval of a Special Resolution of the Company and any other approval required by the Statute, divide amongst the Members in kind the whole or any part of the assets of the Company (whether such assets shall consist of property of the same kind or not) and may for that purpose value any assets and determine how the division shall be carried out as between the Members or different classes of Members. The liquidator may, with the like approval, vest the whole or any part of such assets in trustees upon such trusts for the benefit of the Members as the liquidator, with the like approval, shall think fit, but so that no Member shall be compelled to accept any asset upon which there is a liability.

## 45 Indemnity and Insurance

Every Director and officer of the Company (which for the avoidance of doubt, shall not include auditors of the Company), together with every former Director and former officer of the Company (each an "Indemnified Person") shall be indemnified out of the assets of the Company against any liability, action, proceeding, claim, demand, costs, damages or expenses, including legal expenses, whatsoever which they or any of them may incur as a result of any act or failure to act in carrying out their functions other than such liability (if any) that they may incur by reason of their own actual fraud or wilful default. No Indemnified Person shall be liable to the Company for any loss or damage incurred by the Company as a result (whether direct or indirect) of the carrying out of their functions unless that liability arises through the actual fraud or wilful default of such Indemnified Person. No person shall be found to have committed actual fraud or wilful default under this Article unless or until a court of competent jurisdiction shall have made a finding to that effect.

investigation involving such Indemnified Person for which indemnity will or could be sought. In connection with any advance of any expenses hereunder, the Indemnified Person shall execute an undertaking to repay the advanced amount to the Company if it shall be determined by final judgment or other final adjudication that such Indemnified Person was not entitled to indemnification pursuant to this Article. If it shall be determined by a final judgment or other final adjudication that such Indemnified Person was not entitled to indemnification with respect to such judgment, costs or expenses, then such party shall not be indemnified with respect to such judgment, costs or expenses and any advancement shall be returned to the Company (without interest) by the Indemnified Person.

45.3 The Directors, on behalf of the Company, may purchase and maintain insurance for the benefit of any Director or other officer of the Company against any liability which, by virtue of any rule of law, would otherwise attach to such person in respect of any negligence, default, breach of duty or breach of trust of which such person may be guilty in relation to the Company.

### 46 Financial Year

Unless the Directors otherwise prescribe, the financial year of the Company shall end on 31st December in each year and, following the year of incorporation, shall begin on 1st January in each year.

# 47 Transfer by Way of Continuation

If the Company is exempted as defined in the Statute, it shall, subject to the provisions of the Statute and with the approval of a Special Resolution, have the power to register by way of continuation as a body corporate under the laws of any jurisdiction outside the Cayman Islands and to be deregistered in the Cayman Islands.

### 48 Mergers and Consolidations

The Company shall have the power to merge or consolidate with one or more other constituent companies (as defined in the Statute) upon such terms as the Directors may determine and (to the extent required by the Statute) with the approval of a Special Resolution.

F-25

EXHIBIT G AGREED FORM

# SONUS NETWORKS, INC.

### RESTATED CERTIFICATE OF INCORPORATION

The present name of the corporation is Sonus Networks, Inc. The corporation was incorporated under the name Solstice Sapphire Investments, Inc. by the filing of its original Certificate of Incorporation with the Secretary of State of the State of Delaware on May 19, 2017. This Restated Certificate of Incorporation of the corporation, which restates and integrates and also further amends the provisions of the corporation's Certificate of Incorporation, was duly adopted in accordance with the provisions of Sections 242 and 245 of the General Corporation Law of the State of Delaware and by the written consent of its stockholders in accordance with Section 228 of the General Corporation Law of the State of Delaware. The Certificate of Incorporation of the corporation is hereby amended, integrated and restated to read in its entirety as follows:

ARTICLE I NAME

The name of the corporation (the "Corporation") is Sonus Networks, Inc.

## ARTICLE II REGISTERED AGENT

The address of the Corporation's registered office in the State of Delaware is 2711 Centerville Road, Suite 400, in the City of Wilmington, County of New Castle, 19808; and the name of its registered agent is Corporation Service Company.

ARTICLE III PURPOSE

The nature of the business or purposes to be conducted or promoted by the Corporation is to engage in any lawful act or activity for which corporations may be organized under the General Corporation Law of the State of Delaware (the "<u>DGCL</u>").

ARTICLE IV CAPITAL STOCK

The total number of shares of all classes of stock that the Corporation shall have authority to issue is 250,000,000 shares, consisting solely of:

240,000,000 shares of common stock, par value \$0.001 per share (the "Common Stock"); and

10,000,000 shares of preferred stock, par value \$0.01 per share (the "Preferred Stock").

The following is a statement of the powers, designations, preferences, privileges, and relative rights in respect of each class of capital stock of the Corporation.

G-1

#### A. COMMON STOCK.

- 1. *General*. The voting, dividend and liquidation rights of the holders of Common Stock are subject to and qualified by the rights of the holders of Preferred Stock.
- 2. *Voting.* The holders of Common Stock are entitled to one vote for each share held at all meetings of stockholders. There shall be no cumulative voting.
- 3. *Dividends*. Dividends may be declared and paid on the Common Stock from funds lawfully available therefor if, as and when determined by the Board of Directors and subject to any preferential dividend rights of any then outstanding shares of Preferred Stock.
- 4. *Liquidation*. Upon the dissolution, liquidation or winding up of the Corporation, whether voluntary or involuntary, holders of Common Stock will be entitled to receive all assets of the Corporation available for distribution to its stockholders, subject to any preferential rights of any then outstanding shares of Preferred Stock.
- 5. Subject to any rights of any then outstanding shares of Preferred Stock, the number of authorized shares of Common Stock or Preferred Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority in voting power of the stock of the Corporation entitled to vote thereon irrespective of the provisions of Section 242(b)(2) of the DGCL (or any successor provision thereto), and no vote of the holders of either the Common Stock or the Preferred Stock voting separately as a class shall be required therefor.

### B. PREFERRED STOCK.

Shares of Preferred Stock may be issued from time to time in one or more series, each of such series to have such powers, designations, preferences, and relative, participating, optional, or other special rights, if any, and such qualifications, limitations and restrictions, if any, of such preferences and rights, as are stated or expressed in the resolution or resolutions of the Board of Directors providing for such series of Preferred Stock.

Authority is hereby granted to the Board of Directors from time to time to issue the Preferred Stock in one or more series, and in connection with the creation of any such series, by resolution or resolutions to determine and fix the powers, designations, preferences, and relative, participating, optional, or other special rights, if any, and the qualifications and restrictions, if any, of such preferences and rights, including without limitation dividend rights, conversion rights, voting rights (if any), redemption privileges, and liquidation preferences, of such series of Preferred Stock (which need not be uniform among series), all to the fullest extent now or hereafter permitted by the DGCL. The powers, preferences and relative, participating, optional and other special rights of each series of Preferred Stock, and the qualifications, limitations or restrictions thereof, if any, may differ from those of any and all other series at any time outstanding. Without limiting the generality of the foregoing, the resolution or resolutions providing for the creation or issuance of any series of Preferred Stock may provide that such series shall be superior to, rank equally with, or be junior to the Preferred Stock of any other series, all to the fullest extent permitted by law

G-2

Any resolution or resolutions adopted by the Board of Directors pursuant to the authority vested in them by this Article IV shall be set forth in a certificate of designation along with the number of shares of stock of such series as to which the resolution or resolutions shall apply and such certificate shall be executed, acknowledged, filed, recorded, and shall become effective, in accordance with Section 103 of the DGCL. Unless otherwise provided in any such resolution or resolutions, the number of shares of stock of any such series to which such resolution or resolutions apply may be increased (but not above the total number of authorized shares of the class) or decreased (but not below the number of shares thereof then outstanding) by a certificate likewise executed, acknowledged, filed and recorded, setting forth a statement that a specified increase or decrease therein has been authorized and directed by a resolution or resolutions likewise adopted by the Board of Directors. In case the number of such shares shall be decreased, the number of shares so specified in the certificate shall resume the status which they had prior to the adoption of the first resolution or resolutions. When no shares of any such class or series are outstanding, either because none were issued or because none remain outstanding, a certificate setting forth a resolution or resolutions adopted by the Board of Directors that none of the authorized shares of such class or series are outstanding, and that none will be issued subject to the certificate of designations previously filed with respect to such class or series, may be executed, acknowledged, filed and recorded in the same manner as previously described and it shall have the effect of eliminating from this Restated Certificate of Incorporation all matters set forth in the certificate of designations with respect to such class or series of stock. If no shares of any such class or series established by a resolution or resolutions adopted by the Board of Directors have been issued, the voting powers, designations, preferences and relative, participating, optional or other rights, if any, with the qualifications, limitations or restrictions thereof, may be amended by a resolution or resolutions adopted by the Board of Directors. In the event of any such amendment, a certificate which (i) states that no shares of such class or series have been issued, (ii) sets forth the copy of the amending resolution or resolutions and (iii) if the designation of such class or series is being changed, indicates the original designation and the new designation, shall be executed, acknowledged, filed, recorded, and shall become effective, in accordance with Section 103 of the DGCL.

Notwithstanding the foregoing, except as otherwise required by law, holders of Common Stock, as such, shall not be entitled to vote on any amendment to this Restated Certificate of Incorporation (including any certificate of designation relating to any series of Preferred Stock) that relates solely to the terms of one or more outstanding series of Preferred Stock if the holders of such affected series are entitled, either separately or together with the holders of one or more other such series, to vote thereon pursuant to this Restated Certificate of Incorporation (including any certificate of designation relating to any series of Preferred Stock) or pursuant to the DGCL.

### ARTICLE V BOARD OF DIRECTORS

The following provisions are inserted for the management of the business and for the conduct of the affairs of the Corporation and for defining and regulating the powers of the Corporation and its directors and stockholders and are in furtherance and not in limitation of the powers conferred upon the Corporation by statute:

G-3

- (a) The number of directors constituting the full Board of Directors shall be as determined from time to time by resolution adopted by a majority of the total number of authorized directors, whether or not there exist any vacancies in previously authorized directorships (the "Whole Board"), in a manner consistent (for so long as it remains in effect) with the Principal Stockholders Agreement, dated [·], 2017, by and among the Corporation, Heritage PE (OEP) II, L.P., Heritage PE (OEP) III, L.P. and certain other stockholder parties thereto as such agreement may be amended from time to time (the "Stockholders Agreement"). Each director shall hold office until the next annual meeting of stockholders and until his or her respective successor shall have been duly elected and qualified, subject, however, to prior death, resignation, retirement, disqualification or removal from office. Except as may otherwise be provided by the DGCL, the By-Laws of the Corporation or the Stockholders Agreement (for so long as the Stockholders Agreement is in effect), any director or the entire Board of Directors may be removed from office at any time, (a) for cause by the affirmative vote of the holders of a majority of voting power of the shares of the Corporation's stock entitled to vote for the election of directors, voting together as a single class, or (b) without cause by (i) subject to clause (ii), the affirmative vote of the holders of at least 66 2/3% of the voting power of the shares of the Corporation's stock entitled to vote for the election of directors, voting together as a single class or (ii) in the event recommended by at least two-thirds of the Whole Board, including the approval of a majority of the Independent Directors (as such term is defined in the Stockholders Agreement, the "Independent Directors, voting as a single class.
- (b) The Board of Directors shall have the power and authority: (i) to adopt, amend or repeal the By-Laws of the Corporation, by resolution of the Board of Directors duly adopted by a majority of the directors then constituting the full Board of Directors, including (for so long as the Stockholders Agreement remains in effect) the approval of a majority of the Independent Directors; and (ii) to the full extent permitted or not prohibited by law, and without the consent of or other action by the stockholders, to authorize or create mortgage, pledges or other liens or encumbrances upon any or all of the assets, real, personal or mixed, and franchises of the Corporation, including after-acquired property, and to exercise all of the powers of the Corporation in connection therewith. With respect to the power of holders of capital stock to adopt, amend and repeal By-Laws of the Corporation, notwithstanding any provision of the By-Laws of the Corporation or any other provision that might otherwise permit a lesser vote or no vote, in addition to any vote of the holders of any class or series of capital stock of the Corporation required in the By-Laws of the Corporation or by law, the affirmative vote of the holders of at least 66 2/3% of the voting power of the shares of the Corporation's stock entitled to vote thereon, voting together as a single class, shall be required for any such alteration, amendment, repeal, or adoption by the vote of the holders of any class or series of capital stock of the Corporation.
- (c) Unless and except to the extent that the By-Laws of the Corporation shall so require, the election of directors of the corporation need not be by written ballot.

G-4

### ARTICLE VI LIMITATION OF LIABILITY

No director of the Corporation shall be personally liable to the Corporation or to any of its stockholders for monetary damages for breach of fiduciary duty as a director, notwithstanding any provision of law imposing such liability; provided, however, that to the extent required from time to time by applicable law, this Article VI shall not eliminate or limit the liability of a director, to the extent such liability is provided by applicable law, (i) for any breach of the director's duty of loyalty to the Corporation or its stockholders, (ii) for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) under Section 174 of Title 8 of the DGCL, or (iv) for any transactions from which the director derived an improper personal benefit. No amendment to or repeal of this Article VI shall apply to or have any effect on the liability or alleged liability of any director for or with respect to any acts or omissions of such director occurring prior to the effective date of such amendment or repeal.

# ARTICLE VII INDEMNIFICATION

The Corporation shall, to the fullest extent permitted by Section 145 of the DGCL, as amended from time to time, indemnify each person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative, by reason of being or having been a director or officer of the Corporation or, while a director or officer of the Corporation, serving or having served at the request of the Corporation as a director, trustee, officer, employee or agent of another corporation or of a partnership, joint venture, trust or other enterprise, including service with respect to an employee benefit plan, or by reason of any action alleged to have been taken or omitted in such capacity, against all expense, liability and loss (including attorneys' fees, judgements, fines, ERISA excise taxes or penalties and amounts paid in settlement) reasonably incurred or suffered by him in connection with such action, suit or proceeding and any appeal therefrom, and such indemnification shall continue as to an Indemnitee who has ceased to be a director, trustee, officer, employee, or agent.

Indemnification may include payment by the Corporation of expenses in defending an action or proceeding in advance of the final disposition of such action or proceeding upon receipt of an undertaking by the person indemnified to repay such payment if it is ultimately determined that such person is not entitled to indemnification under this Article VII, which undertaking may be accepted without reference to the financial ability of such person to make such repayment.

The Corporation shall not indemnify any such person seeking indemnification in connection with a proceeding (or part thereof) initiated by such person unless the initiation thereof was approved by the Board of Directors.

The indemnification rights provided in this Article VII (i) shall not be deemed exclusive of any other rights to which those indemnified may be entitled under any law, bylaw, agreement or vote of stockholders or disinterested directors or otherwise, (ii) may, to the extent authorized from time to time by

G-5

indemnification rights to other employees or agents of the Corporation or other persons serving the Corporation and such rights may be equivalent to, or greater or less than, those set forth in this Article VII.

# ARTICLE VIII COMPROMISES AND ARRANGEMENTS

Whenever a compromise or arrangement is proposed between the Corporation and its creditors or any class of them and/or between the Corporation and its stockholders or any class of them, any court of equitable jurisdiction within the State of Delaware may, on the application in a summary way of the Corporation or of any creditor or stockholder thereof or on the application of any receiver or receivers appointed for the Corporation under Section 291 of Title 8 of the Delaware Code or on the application of trustees in dissolution or of any receiver or receivers appointed for the Corporation under Section 279 of Title 8 of the Delaware Code order a meeting of the creditors or class of creditors, and/or of the stockholders or class of stockholders of the Corporation, as the case may be, to be summoned in such manner as the said court directs. If a majority in number representing three fourths (3/4ths) in value of the creditors or class of creditors, and/or of the stockholders or class of stockholders or the Corporation, as the case may be, agree to any compromise or arrangement and to any reorganization of the Corporation as consequence of such compromise or arrangement, the said compromise or arrangement and the said reorganization shall, if sanctioned by the court to which the said application has been made, be binding on all the creditors or class of creditors, and/or on all the stockholders or class of stockholders, of the Corporation, as the case may be, and also on the Corporation.

# ARTICLE IX BUSINESS COMBINATIONS WITH INTERESTED STOCKHOLDERS

- (a) The Corporation expressly elects not to be governed by the provisions of Section 203 of the DGCL.
- (b) Notwithstanding any other provisions of the DGCL or this Restated Certificate of Incorporation (including paragraph (a) of this Article IX), the Corporation shall not engage in any business combination with any interested stockholder for a period of three years following the time that such stockholder became an interested stockholder unless:
- (1) Prior to such time the Board of Directors of the Corporation approved either the business combination or the transaction which resulted in the stockholder becoming an interested stockholder;
- Upon consummation of the transaction which resulted in the stockholder becoming an interested stockholder, the interested stockholder owned at least 85% of the voting stock of the Corporation outstanding at the time the transaction commenced, excluding for purposes of determining the voting stock outstanding (but not the outstanding voting stock owned by the interested stockholder) those shares owned (i) by persons who are directors and also officers and (ii) employee stock plans in which employee participants do not have the right to determine confidentially whether shares held subject to the plan will be tendered in a tender or exchange offer; or

G-6

- (3) At or subsequent to such time the business combination is approved by (i) two-thirds of the Whole Board, and (ii) a majority of the Independent Directors.
  - (c) The restrictions contained in this Article IX shall not apply if:
- (1) A stockholder becomes an interested stockholder inadvertently and (i) as soon as practicable divests itself of ownership of sufficient shares so that the stockholder ceases to be an interested stockholder; and (ii) would not, at any time within the three-year period immediately prior to a business combination between the Corporation and such stockholder, have been an interested stockholder but for the inadvertent acquisition of ownership; or
- The business combination is proposed prior to the consummation or abandonment of and subsequent to the earlier of the public announcement or the notice required hereunder of a proposed transaction which (i) constitutes one of the transactions described in the second sentence of this paragraph (2); (ii) is with or by a person who either was not an interested stockholder during the previous three years or who became an interested stockholder with the approval of the Corporation's Board of Directors; and (iii) is approved or not opposed by a majority of the members of the Board of Directors then in office (but not less than one) who were directors prior to any person becoming an interested stockholder during the previous three years or were recommended for election or elected to succeed such directors by a majority of such directors. The proposed transactions referred to in the preceding sentence are limited to (x) a merger or consolidation of the Corporation (except for a merger in respect of which, pursuant to Section 251(f) of the DGCL, no vote of the stockholders of the Corporation is required); (y) a sale, lease, exchange, mortgage, pledge, transfer or other disposition (in one transaction or a series of transactions), whether as part of a dissolution or otherwise, of assets of the Corporation or of any direct or indirect majority-owned subsidiary of the Corporation (other than to any direct or indirect wholly-owned subsidiary or to the Corporation) having an aggregate market value equal to 50% or more of either that aggregate market value of all of the assets of the Corporation determined on a consolidated basis or the aggregate market value of all the outstanding stock of the Corporation; or (z) a proposed tender or exchange offer for 50% or more of the outstanding voting stock of the Corporation. The Corporation shall give not less than 20 days' notice to all interested stockholders prior to the consummation of any of the transactions described in clause (x) or (y) of the second sentence of this paragraph
  - (d) As used in this Article IX only, the term:
- (1) "Affiliate" means a person that directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, another person.

(2) "Associate," when used to indicate a relationship with any person, means: (i) any corporation, partnership, unincorporated association or other entity of which such person is a director, officer or partner or is, directly or indirectly, the owner of 20% or more of any class of voting stock; (ii) any trust or other estate in which such person has at least a 20% beneficial interest or as to which such person serves as trustee or in a similar
G-7
fiduciary capacity; and (iii) any relative or spouse of such person, or any relative of such spouse, who has the same residence as such person.
(3) "Business combination," when used in reference to the Corporation and any interested stockholder of the Corporation, means:
(i) Any merger or consolidation of the Corporation or any direct or indirect majority-owned subsidiary of the Corporation with (A) the interested stockholder, or (B) with any other corporation, partnership, unincorporated association or other entity if the merger or consolidation is caused by the interested stockholder and as a result of such merger or consolidation subsection (b) of this Article IX is not applicable to the surviving entity;
(ii) Any sale, lease, exchange, mortgage, pledge, transfer or other disposition (in one transaction or a series of transactions), except proportionately as a stockholder of the Corporation, to or with the interested stockholder, whether as part of a dissolution or otherwise, of assets of the Corporation or of any direct or indirect majority-owned subsidiary of the Corporation which assets have an aggregate market value equal to 10% or more of either the aggregate market value of all the assets of the Corporation determined on a consolidated basis or the aggregate market value of all the outstanding stock of the Corporation;
(iii) Any transaction which results in the issuance or transfer by the Corporation or by any direct or indirect majority-owned subsidiary of the Corporation of any stock of the Corporation or of such subsidiary to the interested stockholder, except: (A) pursuant to the exercise, exchange or conversion of securities exercisable for, exchangeable for or convertible into stock of the Corporation or any such subsidiary which securities were outstanding prior to the time that the interested stockholder became such; (B) pursuant to a merger under Section 251(g) of the DGCL; (C) pursuant to a dividend or distribution paid or made, or the exercise, exchange or conversion of securities exercisable for, exchangeable for or convertible into stock of the Corporation or any such subsidiary which security is distributed, pro rata to all holders of a class or series of stock of the Corporation subsequent to the time the interested stockholder became such; (D) pursuant to an exchange offer by the Corporation to purchase stock made on the same terms to all holders of said stock; or (E) any issuance or transfer of stock by the Corporation; provided however, that in no case under items (C)-(E) of this subparagraph (iii) shall there be an increase in the interested stockholder's proportionate share of the stock of any class or series of the Corporation or of the voting stock of the Corporation;
(iv) Any transaction involving the Corporation or any direct or indirect majority-owned subsidiary of the Corporation which has the effect, directly or indirectly, of increasing the proportionate share of the stock of any class or series, or securities convertible into the stock of any class or series, of the Corporation or of any such subsidiary which is owned by the interested stockholder, except as a result of immaterial changes due to fractional share adjustments or as a result of any purchase or redemption of any shares of stock not caused, directly or indirectly, by the interested stockholder; or
G-8
(v) Any receipt by the interested stockholder of the benefit, directly or indirectly (except proportionately a stockholder of the Corporation), of any loans, advances, guarantees, pledges or other financial benefits (other than those expressly permitted in paragraphs (d)(3)(i)-(iv) of this Article IX) provided by or through the Corporation or any direct or indirect majority-owned subsidiary.  (4) "Control," including the terms "controlling," "controlled by" and "under common control with," means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a person, whether through the ownership of voting stock, by contract or otherwise. A person who is the owner of 20% or more of the outstanding voting stock of any corporation, partnership, unincorporated association or other entity shall be presumed to have control of such entity, in the absence of proof by a preponderance of the evidence to the contrary. Notwithstanding the foregoing, a presumption of control shall not apply where such person holds voting stock, in good faith and not for the purpos of circumventing this Article IX, as an agent, bank, broker, nominee, custodian or trustee for one or more owners who do not individually or as a group have control of such entity.
(5) "Interested stockholder" means any person (other than the Corporation and any direct or indirect majority-owner subsidiary of the Corporation) that (i) is the owner of 15% or more of the outstanding voting stock of the Corporation, or (ii) is an affiliate or associate of the Corporation and was the owner of 15% or more of the outstanding voting stock of the Corporation at any time within the three-year period immediately priod to the date on which it is sought to be determined whether such person is an interested stockholder, and the affiliates and associates of such person; provided however, that the term "interested stockholder" shall not include any person whose ownership of shares in excess of the 15% limitation set forth herein is the result of action taken solely by the Corporation; provided that such person shall be an interested stockholder if thereafter such person acquires additional shares of voting stock of the Corporation, except as a result of further corporate action by the Corporation not caused, directly or indirectly, by such person. For the purpose of determining whether a person is an interested stockholder, the voting stock of the Corporation deemed to be outstanding shall include stock deemed to be owned by the person through application of paragraph (6) of this subsection (d) but shall not include any other unissued stock of the Corporation which may be issuable pursuant to any agreement, arrangement or understanding, or upon exercise of conversion rights, warrants or options, or otherwise.
(6) "Owner," including the terms "own" and "owned," when used with respect to any stock, means a person that individually or with or through any of its affiliates or associates:
(i) Beneficially owns such stock, directly or indirectly; or

	(ii)	Has (A) the right to acquire such stock (whether such right is exercisable immediately or only after the
passage of time) pursuant to any agreement, a	arrangem	ent or understanding, or upon the exercise of conversion rights, exchange rights, warrants or options, o
otherwise; provided, however, that a person s	hall not	be deemed the owner of stock tendered pursuant to a tender or exchange offer made by such person or
any		

G-9

of such person's affiliates or associates until such tendered stock is accepted for purchase or exchange; or (B) the right to vote such stock pursuant to any agreement, arrangement or understanding; provided, however, that a person shall not be deemed the owner of any stock because of such person's right to vote such stock if the agreement, arrangement or understanding to vote such stock arises solely from a revocable proxy or consent given in response to a proxy or consent solicitation made to ten or more persons; or

- (iii) Has any agreement, arrangement or understanding for the purpose of acquiring, holding, voting (except voting pursuant to a revocable proxy or consent as described in item (B) of subparagraph (ii) of this paragraph (6)), or disposing of such stock with any other person that beneficially owns, or whose affiliates or associates beneficially own, directly or indirectly, such stock.
  - (7) "Person" means any individual, corporation, partnership, unincorporated association or other entity.
  - (8) "Stock" means, with respect to any corporation, capital stock and, with respect to any other entity, any equity

interest.

(9) "<u>Voting stock</u>" means, with respect to any corporation, stock of any class or series entitled to vote generally in the election of directors and, with respect to any entity that is not a corporation, any equity interest entitled to vote generally in the election of the governing body of such entity. Every reference to a percentage of voting stock shall refer to such percentage of the votes of such voting stock.

# ARTICLE X CERTAIN TRANSACTIONS

The Board of Directors, when considering a tender offer or merger or acquisition proposal, may, to the fullest extent permitted by applicable law, take into account factors in addition to potential economic benefits to stockholders, including without limitation (i) comparison of the proposed consideration to be received by stockholders in relation to the then current market price of the Corporation's capital stock, the estimated current value of the Corporation in a freely negotiated transaction, and the estimated future value of the Corporation as an independent entity, (ii) the impact of such a transaction on the employees, suppliers, and customers of the Corporation and its effect on the communities in which the Corporation operates, and (iii) the impact of such a transaction on the unique corporate culture and atmosphere of the Corporation.

For so long as the Stockholders Agreement remains in effect, the Corporation shall not, either directly or indirectly by amendment, merger, consolidation, waiver or otherwise, take any action or enter into any transaction that would violate or conflict with any provision of the Stockholders Agreement requiring the written consent, waiver or affirmative vote of a majority of the Independent Directors (or of a specified number of directors) without (in addition to any other vote required by law, this Restated Certificate of Incorporation or the By-Laws of the Corporation) obtaining the written consent, waiver or affirmative vote of a majority of the Independent Directors (or such specified number of directors), given in writing or by vote at a

G-10

meeting, and any such act or transaction entered into without such consent, waiver or vote shall be null and void *ab initio*, and of no force or effect. Notwithstanding anything to the contrary herein, for so long as the Stockholders Agreement remains in effect, in the event that any provision herein has a corresponding provision in, or otherwise relates or is applicable to, the Stockholders Agreement, such provision herein shall be interpreted in a manner consistent with, and giving full effect to, the provisions of the Stockholders Agreement. A copy of the Stockholders Agreement will be provided to any stockholder of the Corporation upon request.

# ARTICLE XI STOCKHOLDER ACTION

Any action required or permitted to be taken by the stockholders of the Corporation may be taken only at a duly called annual or special meeting of the stockholders, and not by written consent in lieu of such a meeting, and special meetings of stockholders may be called only by a majority of the Whole Board or (for so long as the Stockholders Agreement remains in effect) a majority of the Independent Directors.

# ARTICLE XII EXCLUSIVE FORUM

Unless the Corporation consents in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware shall, to the fullest extent permitted by law, be the sole and exclusive forum for: (i) any derivative action or proceeding brought on behalf of the Corporation, (ii) any action asserting a claim of breach of a fiduciary duty owed by any director, officer, other employee or stockholder of the Corporation to the Corporation or the Corporation's stockholders, (iii) any action asserting a claim arising pursuant to any provision of the DGCL or as to which the DGCL confers jurisdiction on the Court of Chancery of the State of Delaware, or (iv) any action asserting a claim arising pursuant to any provision of the Restated Certificate of Incorporation or the By-Laws of the Corporation (in each case, as they may be amended from time to time) or governed by the internal affairs doctrine. Any person or entity purchasing or otherwise acquiring or holding any interest in shares of capital stock of the Corporation shall be deemed to have notice of and consented to the provisions of this Article XII.

ARTICLE XIII SEVERABILITY

To the extent that any provision of this Restated Certificate of Incorporation is found to be invalid or unenforceable, such invalidity or unenforceability shall not affect the validity or enforceability of any other provision of this Restated Certificate of Incorporation, and following any determination by a court of competent jurisdiction that any provision of this Restated Certificate of incorporation is invalid or unenforceable, this Restated Certificate of Incorporation shall contain only such provisions (i) as were in effect immediately prior to such determination and (ii) were not so determined to be invalid or unenforceable.
G-11
ARTICLE XIV AMENDMENTS
The affirmative vote of the holders of at least 66 2/3% of the voting power of the outstanding voting stock of the Corporation entitled to vote thereo (in addition to any separate class vote required by law or that may in the future be required pursuant to the terms of any outstanding Preferred Stock), voting together as a single class, shall be required to amend or repeal the provisions of Articles IV (to the extent it relates to the authority of the Board of Directors issue shares of Preferred Stock in one or more series, the terms of which may be determined by the Board of Directors), V, VII, IX, X, XI, XII, XIII or XIV of this Restated Certificate of Incorporation or to reduce the numbers of authorized shares of Common Stock or Preferred Stock.
***
G-12
IN WITNESS WHEREOF, this Restated Certificate of Incorporation has been executed by its duly authorized officer this day of , 2017.  SONUS NETWORKS, INC.
By:

: Name:
Title:

### SIGNATURE PAGE TO RESTATED CERTIFICATE OF INCORPORATION

EXHIBIT H AGREED FORM

### SONUS NETWORKS, INC.

# AMENDED AND RESTATED BY-LAWS

# Article I - General.

- 1.1 Offices. The registered office of Sonus Networks, Inc. (the "Company") shall be in the City of Wilmington, County of New Castle, State of Delaware. The Company may also have offices at such other places both within and without the State of Delaware as the Board of Directors may from time to time determine or the business of the Company may require.
- 1.2 <u>Seal</u>. The seal, if any, of the Company shall be in the form of a circle and shall have inscribed thereon the name of the Company, the year of its organization and the words "Corporate Seal, Delaware."
  - 1.3 **Fiscal Year**. The fiscal year of the Company shall be the period from January 1 through December 31.

# Article II - Stockholders.

- 2.1 Place of Meetings. Each meeting of the stockholders shall be held upon notice as hereinafter provided, at such place, if any, as the Board of Directors shall have determined and as shall be stated in such notice.
- 2.2 Annual Meeting. The annual meeting of the stockholders shall be held each year on such date and at such time as the Board of Directors may determine. At each annual meeting the stockholders entitled to vote shall elect such members of the Board of Directors as are standing for election, by ballot, and they may transact such other corporate business as may properly be brought before the meeting.
- 2.3 **Quorum**. At all meetings of the stockholders the holders of a majority in voting power of the shares of stock issued and outstanding and entitled to vote thereat, present in person or represented by proxy, shall constitute a quorum requisite for the transaction of business except as otherwise provided by law, the Company's Certificate of Incorporation (the "Certificate of Incorporation"), or these by-laws. Whether or not there is such a quorum at any meeting, the chairman of the meeting or the stockholders entitled to vote thereat, present in person or by proxy, by a majority vote, may adjourn the meeting from time to time without notice of such adjourned meeting if the time and place, if any, thereof, are announced at the meeting at which the adjournment is taken. If the adjournment is for more than thirty (30) days, or if after the adjournment a new record date is fixed for the adjourned meeting, a notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting. At such adjourned meeting, at which a quorum shall be present, any business may be transacted that might have been transacted if the meeting had been held as originally called. The stockholders present

in person or by proxy at a duly called meeting at which a quorum is present may continue to transact business until adjournment, notwithstanding the withdrawal of enough stockholders to leave less than a quorum.

2.4 **Right to Vote; Proxies**. Subject to the provisions of the Company's Certificate of Incorporation, each holder of a share or shares of capital stock of the Company having the right to vote at any meeting shall be entitled to one vote for each such share of stock held by him. Any stockholder entitled to vote at any meeting of stockholders may vote either in person or by proxy, but no proxy that is dated more than three years prior to the meeting at which it is offered shall confer the right to vote thereat unless the proxy provides that it shall be effective for a longer period. A proxy may be granted by a

H-1

writing executed by the stockholder or his authorized agent or by transmission or authorization of transmission of a telegram, cablegram, or other means of electronic transmission to the person who will be the holder of the proxy or to a proxy solicitation firm, proxy support service organization, or like agent duly authorized by the person who will be the holder of the proxy to receive such transmission, subject to the conditions set forth in Section 212 of the General Corporation Law of the State of Delaware, as it may be amended from time to time (the "General Corporation Law"). A proxy shall be irrevocable if it states that it is irrevocable and if, and only as long as, it is coupled with an interest sufficient in law to support an irrevocable power.

### 2.5 **Voting**

- (a) At all meetings of stockholders, when a quorum is present, in all matters other than the election of directors, the affirmative vote of the holders of a majority in voting power of the shares of stock present in person or represented by proxy at the meeting and entitled to vote on such matter shall be the act of the stockholders unless a different or minimum vote is required by the Certificate of Incorporation, these by-laws, the rules or regulations of any stock exchange applicable to the Company, or any law or regulation applicable to the Company or its securities, in which case such different or minimum vote shall be the applicable vote on the matter.
- (b) At all meetings of stockholders at which directors are to be elected, other than in a Contested Election Meeting (as defined below), when a quorum is present, a nominee for election as a director at such meeting shall be elected to the Board of Directors if the votes cast "for" such nominee's election exceed the votes cast "against" such nominee's election (with "abstentions" and "broker non-votes" not counted as a vote "for" or "against" such nominee's election). In a Contested Election Meeting, when a quorum is present, directors shall be elected by a plurality of the votes of the shares present in person or represented by proxy at such Contested Election Meeting and entitled to vote on the election of directors. If directors are to be elected by a plurality vote, stockholders shall not be permitted to vote against a nominee. A meeting of stockholders shall be a "Contested Election Meeting" if the number of nominees for election as directors exceeds the number of directors to be elected at such meeting, as of the tenth (10th) day preceding the date of the Company's first notice to stockholders of such meeting sent pursuant to Section 2.9, as applicable, of these by-laws (the "Determination Date"); provided, however, that if, in accordance with Section 2.12 of these by-laws, stockholders are entitled to nominate persons for election as a director in accordance with Section 2.12.
- Notice of Annual Meetings. Unless otherwise provided by law, the Certificate of Incorporation or these by-laws, notice of the date, time, place (if any), and the means of remote communications (if any), by which stockholders and proxyholders may be deemed to be present in person and vote, of each annual meeting of the stockholders shall be given, at least ten (10) days but not more than sixty (60) days prior to the meeting, to each stockholder of record entitled to vote. Such notice shall be given personally or by mail or, to the extent and in the manner permitted by applicable law, by a form of electronic transmission consented to by the stockholder to whom the notice is given. Notices are deemed given as provided in Section 6.2(a) of these by-laws. The Board of Directors may postpone, reschedule or cancel any annual meeting of the stockholders at its discretion, even after notice thereof has been given. Notice need not be given to any stockholder who submits a written waiver of notice signed by him or waives notice by electronic transmission, whether before or after the time of such meeting. Attendance of a stockholder at a meeting of stockholders shall constitute a waiver of notice of such meeting, except when the stockholder attends the meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or

H-2

convened. Neither the business to be transacted at, nor the purpose of, any annual meeting of the stockholders need be specified in any waiver of notice.

- Stockholders' List. A complete list of the stockholders entitled to vote at any meeting of stockholders, arranged in alphabetical order and showing the address of each stockholder, and the number of shares registered in the name of each stockholder, shall be prepared by the Company at least ten (10) days before every meeting of stockholders. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting, at least ten (10) days prior to the meeting: (i) on a reasonably accessible electronic network, provided that the information required to gain access to such list is provided with the notice of meeting, or (ii) during ordinary business hours, at the principal place of business of the Company. If the meeting is to be held at a place, then the list shall be produced and kept at the time and place of the meeting during the whole time thereof, and may be inspected by any stockholder who is present. If the meeting is to be held solely by means of remote communication, then the list shall also be open to the examination of any stockholder during the whole time of the meeting on a reasonable accessible electronic network.
- Special Meetings. Special meetings of the stockholders for any purpose or purposes, unless otherwise provided by law, may be called only by a majority of the Whole Board (as defined below) or (for so long as the Principal Stockholders Agreement, dated [·], 2017, by and among the Company, Heritage PE (OEP) II, L.P., Heritage PE (OEP) III, L.P. and certain other stockholder parties thereto as such agreement may be amended from time to time (the "Stockholders Agreement") remains in effect) a majority of the Independent Directors (as such term is defined in the Stockholders Agreement, the "Independent Directors"). The Board of Directors may postpone, reschedule or cancel any special meeting of the stockholders at its discretion, even after notice thereof has been given.
- 2.9 Notice of Special Meetings. Unless otherwise provided by law, the Certificate of Incorporation or these by-laws, notice of the date, time, place (if any), the means of remote communications (if any), by which stockholders and proxyholders may be deemed to be present in person and vote, and the purpose or purposes of each special meeting of the stockholders shall be given, at least ten (10) days but not more than sixty (60) days prior to the meeting, to each stockholder of record entitled to vote. Such notice shall be given personally or by mail or, to the extent and in the manner permitted by applicable law, by a form of electronic transmission consented to by the stockholder to whom the notice is given. Notices are deemed given as provided in

Section 6.2(a) of these by-laws. No business may be transacted at such meeting except that referred to in said notice, or in a supplemental notice given also in compliance with the provisions hereof, or such other business as may be germane or supplementary to that stated in said notice or notices as determined by the Board of Directors. Notice need not be given to any stockholder who submits a written waiver of notice signed by him or waives notice by electronic transmission, whether before or after the time of such meeting. Attendance of a stockholder at a meeting of stockholders shall constitute a waiver of notice of such meeting, except when the stockholder attends the meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened. Neither the business to be transacted at, nor the purpose of, any special meeting of the stockholders need be specified in any waiver of notice.

Inspectors. The Company shall, in advance of any meeting of stockholders, appoint one or more inspectors to act at the meeting and make a written report thereof. Such inspectors shall have the powers and duties set forth in Section 231 of the General Corporation Law as currently in effect or as the same may hereafter be amended. Each inspector, before discharging his or her duties, shall take and sign an oath faithfully to execute the duties of inspector with strict impartiality and according to the best of such inspector's ability. At the meeting for which the inspector or inspectors are appointed, he or they shall, in addition to the duties set forth in Section 231 of the General Corporation Law, receive and take charge of the proxies and ballots, and decide all questions touching on the qualifications of voters, the

H-3

validity of proxies, and the acceptance and rejection of votes. If any inspector previously appointed shall fail to attend or refuse or be unable to act at a meeting of stockholders, the presiding officer shall appoint an inspector to act at the meeting.

2.11 <u>Stockholders' Consent in Lieu of Meeting</u>. Unless otherwise provided in the Company's Certificate of Incorporation, any action required to be taken at any annual or special meeting of stockholders of the Company, or any action that may be taken at any annual or special meeting of such stockholders, may be taken only at such a meeting, and not by written consent of stockholders.

### 2.12 Notice of Stockholder Business and Nominations.

- (a) Annual Meetings of Stockholders. (1) Nominations of persons for election to the Board of Directors of the Company and the proposal of other business to be considered by the stockholders may be made at an annual meeting of stockholders only (A) pursuant to the Company's notice of meeting (or any supplement thereto), (B) subject to the Stockholders Agreement (for so long as the Stockholders Agreement remains in effect), by or at the direction of the Board of Directors or any committee thereof, or (C) subject to the Stockholders Agreement (for so long as the Stockholders Agreement remains in effect), by any stockholder of the Company who was a stockholder of record of the Company at the time the notice provided for in this Section 2.12 is delivered to the Secretary of the Company, who is entitled to vote at the meeting and who complies with the notice procedures set forth in this Section 2.12.
- For any nominations or other business to be properly brought before an annual meeting by a stockholder pursuant to (2)clause (C) of paragraph (a)(1) of this Section 2.12, the stockholder must have given timely notice thereof in writing to the Secretary of the Company and any such proposed business (other than the nominations of persons for election to the Board of Directors) must constitute a proper matter for stockholder action. To be timely, a stockholder's notice shall be delivered to the Secretary at the principal executive offices of the Company not later than the close of business on the ninetieth (90th) day, nor earlier than the close of business on the one hundred twentieth (120th) day, prior to the first anniversary of the preceding year's annual meeting (provided, however, that in the event that no annual meeting was held in the prior year or the date of the annual meeting is more than thirty (30) days before or more than seventy (70) days after such anniversary date, notice by the stockholder must be so delivered not earlier than the close of business on the one hundred twentieth (120th) day prior to such annual meeting and not later than the close of business on the later of the ninetieth (90th) day prior to such annual meeting or the tenth (10th) day following the day on which public announcement of the date of such meeting is first made by the Company). For purposes of the first annual meeting following the adoption of these By-Laws, the date of the first anniversary of the preceding year's annual meeting shall be deemed to be June 9, 2018. In no event shall the public announcement of an adjournment or postponement of an annual meeting commence a new time period (or extend any time period) for the giving of a stockholder's notice as described above. Such stockholder's notice shall set forth: (A) as to each person whom the stockholder proposes to nominate for election as a director (i) all information relating to such person that is required to be disclosed in solicitations of proxies for election of directors in an election contest, or is otherwise required, in each case pursuant to and in accordance with Regulation 14A under the Securities Exchange Act of 1934, as amended (the "Exchange Act") and (ii) such person's written consent to being named in the Company's proxy statement as a nominee and to serving as a director if elected; (B) as to any other business that the stockholder proposes to bring before the meeting, a brief description of the business desired to be brought before the meeting, the text of the proposal or business (including the text of any resolutions proposed for consideration and in the event that such business includes a proposal to amend the by-laws of the Company, the language of the proposed amendment), the reasons for conducting such business at the meeting and any material interest in such business of such stockholder and the beneficial owner, if any, on whose behalf the proposal is made; and

H-4

(C) as to the stockholder giving the notice and the beneficial owner, if any, on whose behalf the nomination or proposal is made (i) the name and address of such stockholder, as they appear on the Company's books, and of such beneficial owner, (ii) the class or series and number of shares of capital stock of the Company which are owned beneficially and of record by such stockholder and such beneficial owner, (iii) a description of any agreement, arrangement or understanding with respect to the nomination or proposal between or among such stockholder and such beneficial owner, any of their respective affiliates or associates, and any others acting in concert with any of the foregoing, (iv) a description of any agreement, arrangement or understanding (including any derivative or short positions, profit interests, options, warrants, stock appreciation or similar rights, hedging transactions, and borrowed or loaned shares) that has been entered into as of the date of the stockholder's notice by, or on behalf of, such stockholder and such beneficial owners, the effect or intent of which is to mitigate loss to, manage risk or benefit of share price changes for, or increase or decrease the voting power of, such stockholder or such beneficial owner, with respect to shares of stock of the Company, (v) a representation that the stockholder is a holder of record of stock of the Company entitled to vote at such meeting and intends to appear in person or by proxy at the meeting to propose such business or nomination, and (vi) a representation whether the stockholder or the beneficial owner, if any, intends or is part of a group which intends (a) to deliver a proxy statement and/or form of proxy to holders of at least the percentage of the Company's outstanding capital stock required to approve or adopt the proposal or elect the nominee and/or (b) otherwise to solicit proxies or votes from stockholders in support of such proposal or nomination. The foregoing notice requirements of this Section 2.12 shall be deemed satisfied

in a proxy statement that has been prepared by the Company to solicit proxies for such annual meeting. The Company may require any proposed nominee to furnish such other information as it may reasonably require to determine the eligibility of such proposed nominee to serve as a director of the Company.

- (3) Notwithstanding anything in the second sentence of paragraph (a)(2) of this Section 2.12 to the contrary, in the event that the number of directors to be elected to the Board of Directors of the Company is increased effective at the annual meeting and there is no public announcement by the Company naming the nominees for the additional directorships at least one hundred (100) days prior to the first anniversary of the preceding year's annual meeting, a stockholder's notice required by this Section 2.12 shall also be considered timely, but only with respect to nominees for the additional directorships, if it shall be delivered to the Secretary at the principal executive offices of the Company not later than the close of business on the tenth (10th) day following the day on which such public announcement is first made by the Company.
- (b) Special Meetings of Stockholders. Only such business shall be conducted at a special meeting of stockholders as shall have been brought before the meeting pursuant to the Company's notice of meeting. Nominations of persons for election to the Board of Directors may be made at a special meeting of stockholders at which directors are to be elected pursuant to the Company's notice of meeting (1) subject to the Stockholders Agreement (for so long as the Stockholders Agreement remains in effect), by or at the direction of the Board of Directors or any committee thereof, or (2) provided that the Board of Directors has determined that directors shall be elected at such meeting and subject to the Stockholders Agreement (for so long as the Stockholders Agreement remains in effect), by any stockholder of the Company who is a stockholder of record at the time the notice provided for in this Section 2.12 is delivered to the Secretary of the Company, who is entitled to vote at the meeting and upon such election and who complies with the notice procedures set forth in this Section 2.12. In the event the Company calls a special meeting of stockholders for the purpose of electing one or more directors to the Board of Directors, any such stockholder entitled to vote in such election of directors may nominate a person or persons (as the case may be) for election to such position(s) as specified in the Company's

H-5

notice of meeting, if the stockholder's notice required by paragraph (a)(2) of this Section 2.12 shall be delivered to the Secretary at the principal executive offices of the Company not earlier than the close of business on the one hundred twentieth (120th) day prior to such special meeting and not later than the close of business on the later of the ninetieth (90th) day prior to such special meeting or the tenth (10th) day following the day on which public announcement is first made by the Company of the date of the special meeting and of the nominees proposed by the Board of Directors to be elected at such meeting. In no event shall the public announcement of an adjournment or postponement of a special meeting commence a new time period (or extend any time period) for the giving of a stockholder's notice as described above.

- General. (1) Only such persons who are nominated in accordance with the procedures set forth in this Section 2.12 shall be eligible to be elected at an annual or special meeting of stockholders of the Company to serve as directors and only such business shall be conducted at a meeting of stockholders as shall have been brought before the meeting in accordance with the procedures set forth in this Section 2.12. Except as otherwise provided by law, the chairman of the meeting shall have the power and duty (A) to determine whether a nomination or any business proposed to be brought before the meeting was made or proposed, as the case may be, in accordance with the procedures set forth in this Section 2.12 (including whether the stockholder or beneficial owner, if any, on whose behalf the nomination or proposal is made solicited (or is part of a group which solicited) or did not so solicit, as the case may be, proxies or votes in support of such stockholder's nominee or proposal in compliance with such stockholder's representation as required by clause (a)(2)(C)(vi) of this Section 2.12) and (B) if any proposed nomination or business was not made or proposed in compliance with this Section 2.12, to declare that such nomination shall be disregarded or that such proposed business shall not be transacted. Notwithstanding the foregoing provisions of this Section 2.12, unless otherwise required by law, if the stockholder (or a qualified representative of the stockholder) does not appear at the annual or special meeting of stockholders of the Company to present a nomination or proposed business, such nomination shall be disregarded and such proposed business shall not be transacted, notwithstanding that proxies in respect of such vote may have been received by the Company. For purposes of this Section 2.12, to be considered a qualified representative of the stockholder, a person must be a duly authorized officer, manager or partner of such stockholder or must be authorized by a writing executed by such stockholder or an electronic transmission delivered by such stockholder to act for such stockholder as proxy at the meeting of stockholders and such person must produce such writing or electronic transmission, or a reliable reproduction of the writing or electronic transmission, at the meeting of stockholders.
- (2) For purposes of this Section 2.12, "public announcement" shall include disclosure in a press release reported by the Dow Jones News Service, Associated Press or other national news service or in a document publicly filed by the Company with the Securities and Exchange Commission pursuant to Section 13, 14 or 15(d) of the Exchange Act.
- (3) Notwithstanding the foregoing provisions of this Section 2.12, a stockholder shall also comply with all applicable requirements of the Exchange Act and the rules and regulations thereunder with respect to the matters set forth in this Section 2.12; provided however, that any references in these by-laws to the Exchange Act or the rules promulgated thereunder are not intended to and shall not limit any requirements applicable to nominations or proposals as to any other business to be considered pursuant to this Section 2.12 (including paragraphs (a)(1)(C) and (b) hereof), and compliance with paragraphs (a)(1)(C) and (b) of this Section 2.12 shall be the exclusive means for a stockholder to make nominations or submit other business (other than, as provided in the penultimate sentence of (a)(2), matters brought properly under and in compliance with Rule 14a-8 of the Exchange Act, as may be amended from time to time). Nothing in this Section 2.12 shall be deemed to affect any rights (A) of stockholders to request inclusion of proposals in the Company's proxy statement pursuant to applicable rules and regulations promulgated under the Exchange Act or (B) of the holders of any series

H-6

of Preferred Stock to elect directors pursuant to any applicable provisions of the Company's Certificate of Incorporation.

Conduct of Meetings. The date and time of the opening and the closing of the polls for each matter upon which the stockholders will vote at a meeting shall be announced at the meeting by the person presiding over the meeting. The Board of Directors may adopt by resolution such rules and regulations for the conduct of the meeting of stockholders as it shall deem appropriate. Except to the extent inconsistent with such rules and regulations as adopted by the Board of Directors, the person presiding over any meeting of stockholders shall have the right and authority to convene, to recess and to adjourn the meeting, to prescribe such rules, regulations and procedures and to do all such acts as, in the judgment of such presiding person, are appropriate for the proper conduct of the meeting. Such rules, regulations or procedures, whether adopted by the Board of Directors or prescribed by the presiding person of the meeting, may include, without limitation, the following: (i) the establishment of an agenda or order of business for the meeting; (ii) rules and

procedures for maintaining order at the meeting and the safety of those present; (iii) limitations on attendance at or participation in the meeting to stockholders of record of the Company, their duly authorized and constituted proxies or such other persons as the presiding person of the meeting shall determine; (iv) restrictions on entry to the meeting after the time fixed for the commencement thereof; and (v) limitations on the time allotted to questions or comments by participants. The presiding person at any meeting of stockholders, in addition to making any other determinations that may be appropriate to the conduct of the meeting, shall, if the facts warrant, determine and declare to the meeting that a matter or business was not properly brought before the meeting and if such presiding person should so determine, such presiding person shall so declare to the meeting and any such matter or business not properly brought before the meeting shall not be transacted or considered. Unless and to the extent determined by the Board of Directors or the person presiding over the meeting, meetings of stockholders shall not be required to be held in accordance with the rules of parliamentary procedure.

### **Article III - Directors.**

### 3.1 **Number of Directors**.

- (a) Except as otherwise provided by law, the Company's Certificate of Incorporation, or these by-laws, the property and business of the Company shall be managed by or under the direction of a board of directors. Directors need not be stockholders, residents of Delaware, or citizens of the United States. The use of the phrase "Whole Board" herein refers to the total number of directors which the Company would have if there were no vacancies or unfilled newly-created directorships.
- (b) The number of directors constituting the full Board of Directors shall be as determined from time to time by resolution adopted by a majority of the Whole Board in a manner consistent (for so long as it remains in effect) with the Stockholders Agreement. Each director shall be elected annually at each annual meeting of the Company's stockholders. Members of the Board of Directors shall hold office until the annual meeting of stockholders at which their respective successors are elected and qualified or until their earlier death, incapacity, resignation, or removal.
- (c) Except as the General Corporation Law, the Company's Certificate of Incorporation or the Stockholders Agreement (for so long as the Stockholders Agreement is in effect) may otherwise require, newly created directorships resulting from any increase in the authorized number of directors or any vacancies in the Board of Directors resulting from death, incapacity, disqualification, resignation, or removal from office or other cause may only be filled by the affirmative vote of a majority of the remaining directors then in office, although less than a quorum, or by the sole remaining director.

H-7

Any director so chosen shall hold office until the next annual election of the directors and until such director's successor shall be elected and qualified. No decrease in the number of directors shall shorten the term of any incumbent director.

- 3.2 **Resignation**. Any director of the Company may resign at any time by giving notice in writing or by electronic transmission to the Chairman of the Board, the President, or the Secretary of the Company. Such resignation shall take effect at the time specified therein, at the time of receipt if no time is specified therein and at the time of acceptance if the effectiveness of such resignation is conditioned upon its acceptance. Unless otherwise specified therein, the acceptance of such resignation shall not be necessary to make it effective.
- 3.3 **Removal.** Except as may otherwise be provided by the General Corporation Law, the Company's Certificate of Incorporation or the Stockholders Agreement (for so long as the Stockholders Agreement is in effect), any director or the entire Board of Directors may be removed from office at any time, (a) for cause by the affirmative vote of the holders of a majority of the voting power of the shares of the Company's stock entitled to vote for the election of directors, voting together as a single class, or (b) without cause by (i) subject to clause (ii), the affirmative vote of the holders of at least 66 2/3% of the voting power of the shares of the Company's stock entitled to vote for the election of directors, voting together as a single class or (ii) in the event recommended by at least two-thirds of the Whole Board, including the approval of a majority of the Independent Directors, the affirmative vote of the holders of a majority of the voting power of the shares of the Company's stock entitled to vote for the election of directors, voting together as a single class.
- 3.4 <u>Place of Meetings and Books</u>. The Board of Directors may hold their meetings and keep the books of the Company outside the State of Delaware, at such places as they may from time to time determine.
- 3.5 <u>General Powers</u>. In addition to the powers and authority expressly conferred upon them by these by-laws, the board may exercise all such powers of the Company and do all such lawful acts and things as are not by law or by the Company's Certificate of Incorporation or by these by-laws directed or required to be exercised or done by the stockholders.
- 3.6 <u>Committees</u>. The Board of Directors may designate one or more committees, each committee to consist of one or more directors of the Company. Any such committee, to the extent provided in the resolution of the Board of Directors or in these by-laws, shall have and may exercise all the powers and authority of the Board of Directors in the management of the business and affairs of the Company to the extent permitted by law, and may authorize the seal of the Company to be affixed to all papers which may require it. Such committee or committees shall have such name or names as may be determined from time to time by resolution adopted by the Board of Directors.
- 3.7 <u>Powers Denied to Committees</u>. Committees of the Board of Directors shall not, in any event, have the power or authority to: (i) approve or adopt, or recommend to the stockholders, any action or matter (other than the election or removal of directors) expressly required by the General Corporation Law to be submitted to stockholders for approval, or (ii) adopt, amend or repeal any by-law of the Company.
- 3.8 <u>Substitute Committee Member</u>. Subject to the restrictions set forth in the Stockholders Agreement (for so long as the Stockholders Agreement is in effect), the Board of Directors may designate one or more directors as alternate members of any committee, who may replace any absent or disqualified member at any meeting of the committee. Subject to the restrictions set forth in the Stockholders Agreement (for so long as the Stockholders Agreement is in effect), in the absence or disqualification of a

H-8

member. Any committee shall keep regular minutes of its proceedings and report the same to the Board of Directors as may be required by the Board of Directors.

- 3.9 **Compensation of Directors.** The Board of Directors shall have the power to fix the compensation of directors and members of committees of the Board. The directors may be paid their expenses, if any, of attendance at each meeting of the Board of Directors, may be paid a fixed sum for attendance at each meeting of the Board of Directors or a stated salary as director and may receive stock options, grants and issuances of restricted stock under the Company's equity incentive plan(s). No such payment shall preclude any director from serving the Company in any other capacity and receiving compensation therefor. Members of special or standing committees may be allowed like compensation for attending committee meetings.
- 3.10 **Regular Meetings**. No notice shall be required for regular meetings of the Board of Directors for which the time and place (within or without the State of Delaware) have been fixed by resolution of the Board of Directors.
- 3.11 Special Meetings. Special meetings of the Board of Directors may be held at any time or place, within or without the State of Delaware, whenever called by a majority of the directors then constituting the full Board of Directors. Notice of a special meeting of the Board of Directors shall be given by the person or persons calling the meeting at least forty-eight hours before the special meeting to each director, personally or by telephone, facsimile, electronic mail or other electronic transmission to the extent and in the manner permitted by applicable law. Notice need not be given to any director who submits a written waiver of notice signed by him or waives notice by electronic transmission, whether before or after the time of such meeting. Attendance of any director at a meeting shall constitute a waiver of notice of such meeting, except when such director attends a meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened. Neither the business to be transacted at, nor the purpose of, any special meeting of the directors need be specified in any waiver of notice.
- 3.12 **Quorum**. At all meetings of the Board of Directors, a majority of the Whole Board shall be necessary and sufficient to constitute a quorum for the transaction of business, and the act of a majority of the directors present at any meeting at which there is a quorum shall be the act of the Board of Directors, except as may be otherwise specifically permitted or provided by law, or by the Company's Certificate of Incorporation, or by these by-laws. If at any meeting of the board there shall be less than a quorum present, a majority of those directors present may adjourn the meeting from time to time until a quorum is obtained, and no further notice of the adjourned meeting need be given if the time and place, if any, thereof, are announced at the meeting at which the adjournment is taken.
- 3.13 <u>Telephonic Participation in Meetings</u>. Members of the Board of Directors, or any committee designated by the Board of Directors, may participate in a meeting thereof by means of conference telephone or other communications equipment by means of which all persons participating in the meeting can hear each other, and participation in a meeting pursuant to this by-law shall constitute presence in person at such meeting.
- 3.14 Action by Consent. Unless otherwise restricted by the Company's Certificate of Incorporation or these by-laws, any action required or permitted to be taken at any meeting of the Board of Directors, or of any committee thereof, may be taken without a meeting if all members of the Board of Directors or such committee, as the case may be, consent thereto in writing or by electronic transmission

H-9

and the writing or writings or electronic transmissions are filed with the minutes of proceedings of the Board of Directors or committee in accordance with applicable law.

## Article IV - Officers.

- 4.1 <u>Selection; Statutory Officers</u>. The officers of the Company shall be chosen by the Board of Directors. There shall be a President, a Secretary, and a Treasurer, and there may be a Chairman of the Board of Directors, one or more Vice Presidents, one or more Assistant Secretaries, and one or more Assistant Treasurers, as the Board of Directors may elect. Any number of offices may be held by the same person, except that the offices of President and Secretary shall not be held by the same person simultaneously.
- 4.2 **Time of Election**. The officers designated in Section 4.1 shall be chosen by the Board of Directors at its first meeting after each annual meeting of stockholders. None of said officers need be a director.
- 4.3 <u>Additional Officers</u>. The Board of Directors may appoint such other officers and agents as it shall deem necessary, who shall hold their offices for such terms and shall exercise such powers and perform such duties as shall be determined from time to time by the Board of Directors.
- 4.4 <u>Terms of Office</u>. Each officer of the Company shall hold office until his successor is chosen and qualified, or until his earlier resignation or removal. Any officer elected or appointed by the Board of Directors may be removed at any time by the Board of Directors, but such removal shall be without prejudice to the contractual rights of such officer. Any officer may resign at any time upon written notice or notice by electronic transmission to the Company.
- 4.5 <u>Compensation of Officers</u>. The Board of Directors shall have power to fix the compensation of all officers of the Company. It may authorize any officer, upon whom the power of appointing subordinate officers may have been conferred, to fix the compensation of such subordinate officers.
- 4.6 **Chairman of the Board**. The Chairman of the Board of Directors shall preside at all meetings of the stockholders and directors, and shall have such other duties as may be assigned to him from time to time by the Board of Directors.
- 4.7 President. Unless the Board of Directors otherwise determines, the President shall be the chief executive officer and head of the Company. Unless there is a Chairman of the Board, the President shall preside at all meetings of directors (if he is also a director) and stockholders. Under the supervision of the Board of Directors, the President shall have the general control and management of its business and affairs, subject, however, to the right of the Board of Directors to confer any specific power, except such as may be by law exclusively conferred on the President, upon any other officer or officers of the Company. The President shall perform and do all acts and things incident to the position of President and such other duties as may be assigned to him from time to time by the Board of Directors.

H-10

- Treasurer. The Treasurer shall have the care and custody of all the funds and securities of the Company that may come into his hands as Treasurer, and the power and authority to endorse checks, drafts and other instruments for the payment of money for deposit or collection when necessary or proper and to deposit the same to the credit of the Company in such bank or banks or depository as the Board of Directors, or the officers or agents to whom the Board of Directors may delegate such authority, may designate, and he may endorse all commercial documents requiring endorsements for or on behalf of the Company. He may sign all receipts and vouchers for the payments made to the Company. He shall render an account of his transactions to the Board of Directors as often as the board or the committee shall require the same. He shall enter regularly in the books to be kept by him for that purpose full and adequate account of all moneys received and paid by him on account of the Company. He shall perform all acts incident to the position of Treasurer, subject to the control of the Board of Directors. He shall when requested, pursuant to vote of the Board of Directors, give a bond to the Company conditioned for the faithful performance of his duties, the expense of which bond shall be borne by the Company.
- 4.10 <u>Secretary</u>. The Secretary shall keep the minutes of all meetings of the Board of Directors and of the stockholders; he shall attend to the giving and serving of all notices of the Company. Except as otherwise ordered by the Board of Directors, he shall attest the seal of the Company upon all contracts and instruments executed under such seal and shall affix the seal of the Company thereto and to all certificates of shares of capital stock of the Company. He shall have charge of the stock certificate book, transfer book and stock ledger, and such other books and papers as the Board of Directors may direct. He shall, in general, perform all the duties of Secretary, subject to the control of the Board of Directors.
- 4.11 <u>Assistant Secretary</u>. The Board of Directors or any two of the officers of the Company acting jointly may appoint or remove one or more Assistant Secretaries of the Company. Any Assistant Secretary upon his appointment shall perform such duties of the Secretary, and also any and all such other duties as the Board of Directors or the President or the Executive Vice-President or the Treasurer or the Secretary may designate.
- 4.12 <u>Assistant Treasurer</u>. The Board of Directors or any two of the officers of the Company acting jointly may appoint or remove one or more Assistant Treasurers of the Company. Any Assistant Treasurer upon his appointment shall perform such of the duties of the Treasurer, and also any and all such other duties as the Board of Directors or the President or the Executive Vice-President or the Treasurer or the Secretary may designate.
- 4.13 <u>Subordinate Officers</u>. The Board of Directors may select such subordinate officers as it may deem desirable. Each such officer shall hold office for such period, have such authority, and perform such duties as the Board of Directors may prescribe. The Board of Directors may, from time to time, authorize any officer to appoint and remove subordinate officers and to prescribe the powers and duties thereof.

### Article V - Stock.

5.1 Stock. The shares of the Company shall be represented by certificates, provided that the Board of Directors may provide by resolution or resolutions that some or all of any or all classes or series of its stock shall be uncertificated shares. Any such resolution shall not apply to shares represented by a certificate until such certificate is surrendered to the Company. Every holder of stock represented by certificates shall be entitled to have a certificate signed by, or in the name of the Company by any two authorized officers of the Company, including, but not limited to, the Chairman of the Board of Directors, a Vice-Chairman of the Board of Directors, the President, a Vice-President, the Treasurer, an Assistant Treasurer, the Secretary or an Assistant Secretary representing the number or shares registered in

H-11

certificate form and shall be sealed with the corporate seal of the Company. The certificates of stock of the Company shall be numbered and shall be entered in the books of the Company as they are issued. Any or all the signatures on the certificate may be a facsimile. In case any officer or officers, transfer agent or registrar who shall have signed, or whose facsimile signature or signatures shall have been used on, any such certificate or certificates shall cease to be such officer or officers, transfer agent or registrar of the Company, whether because of death, resignation or otherwise, before such certificate or certificates shall have been issued, such certificate or certificates may nevertheless be issued by the Company with the same effect as though the person or persons who signed such certificate or certificates or whose facsimile signature shall have been used thereon had not ceased to be such officer or officers, transfer agent or registrar of the Company.

- 5.2 **Fractional Share Interests**. The Company may, but shall not be required to, issue fractions of a share. If the Company does not issue fractions of a share, it shall (i) arrange for the disposition of fractional interests by those entitled thereto, (ii) pay in cash the fair value of fractions of a share as of the time when those entitled to receive such fractions are determined, or (iii) issue scrip or warrants in registered form (either represented by a certificate or uncertificated) or in bearer form (represented by a certificate) that shall entitle the holder to receive a full share upon the surrender of such scrip or warrants aggregating a full share. A certificate for a fractional share or an uncertificated fractional share shall, but scrip or warrants shall not unless otherwise provided therein, entitle the holder to exercise voting rights, to receive dividends thereon, and to participate in any of the assets of the Company in the event of liquidation. The Board of Directors may cause scrip or warrants to be issued subject to the conditions that they shall become void if not exchanged for certificates representing full shares or uncertificated full shares before a specified date, or subject to the conditions that the shares for which scrip or warrants are exchangeable may be sold by the Company and the proceeds thereof distributed to the holders of scrip or warrants, or subject to any other conditions that the Board of Directors may impose.
- 5.3 <u>Transfers of Stock</u>. Subject to any transfer restrictions then in force, the shares of stock of the Company shall be transferable only upon its books by the holders thereof in person or by their duly authorized attorneys or legal representatives, and upon the surrender of the certificate or certificates for such shares properly endorsed (or, with respect to uncertificated shares, by delivery of duly executed instructions or in any other manner permitted by applicable law). The Company shall be entitled to treat the holder of record of any share or shares of stock as the holder in fact thereof and accordingly shall not be bound to recognize any equitable or other claim to or interest in such share on the part of any other person whether or not it shall have express or other notice thereof save as expressly provided by the laws of Delaware.

5.4 <b>Record Date</b> . For the purpose of determining the stockholders entitled to notice of or to vote at any meeting of stockholders or any
adjournment thereof, or entitled to receive payment of any dividend or other distribution or the allotment of any rights, or entitled to exercise any rights in
respect of any change, conversion, or exchange of stock or for the purpose of any other lawful action, the Board of Directors may fix, in advance, a record
date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board of Directors, and which record date
shall not be more than sixty (60) days nor less than ten (10) days before the date of such meeting, nor more than sixty (60) days prior to any other action. If
no such record date is fixed by the Board of Directors, the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders
shall be at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next
preceding the day on which the meeting is held; and the record date for determining stockholders for any other purpose shall be at the close of business on the
day on which the Board of Directors adopts the resolution relating thereto. A determination of stockholders of record entitled to notice of or to vote at any
meeting of stockholders shall apply to any adjournment of the

H-12

meeting; provided, however, that the Board of Directors may fix a new record date for the adjourned meeting.

5.5 **Transfer Agent and Registrar**. The Board of Directors may appoint one or more transfer agents or transfer clerks and one or more registrars and may require all certificates of stock to bear the signature or signatures of any of them.

### 5.6 **Dividends**.

- (a) **Power to Declare**. Dividends upon the capital stock of the Company, subject to the provisions of the Company's Certificate of Incorporation, if any, may be declared by the Board of Directors at any regular or special meeting, pursuant to law. Dividends may be paid in cash, in property, or in shares of the capital stock, subject to the provisions of the Company's Certificate of Incorporation and the General Corporation Law.
- (b) Reserves. Before payment of any dividend, there may be set aside out of any funds of the Company lawfully available for dividends such sum or sums as the directors from time to time, in their absolute discretion, think proper as a reserve or reserves to meet contingencies, or for equalizing dividends, or for repairing or maintaining any property of the Company, or for such other purpose as the directors shall think conducive to the interest of the Company, and the directors may modify or abolish any such reserve in the manner in which it was created.
- 5.7 **Lost, Stolen, or Destroyed Certificates.** No certificates for shares of stock of the Company (or uncertificated shares, as the case may be) shall be issued in place of any certificate alleged to have been lost, stolen, or destroyed, except upon production of such evidence of the loss, theft, or destruction and upon indemnification of the Company and its agents to such extent and in such manner as the Board of Directors may from time to time prescribe.
- 5.8 <u>Inspection of Books</u>. Any stockholder, in person or by attorney or other agent, shall, upon written demand under oath stating the purpose thereof, have the right during the usual hours for business to inspect for any proper purpose the Company's stock ledger, a list of its stockholders, and its other books and records; and no stockholder shall have any right to inspect any account or book or document of the Company except as conferred by law.

### Article VI - Miscellaneous Management Provisions.

6.1 <u>Checks, Drafts, and Notes</u>. All checks, drafts, or orders for the payment of money, and all notes and acceptances of the Company shall be signed by such officer or officers, or such agent or agents, as the Board of Directors may designate.

### 6.2 Notices.

(a) Notices to directors may be given personally or by telephone, mail, facsimile, electronic mail or other electronic transmission to the extent and in the manner permitted by applicable law. Notices to stockholders may be given personally or by mail or, to the extent and in the manner permitted by applicable law, by a form of electronic transmission consented to by the stockholder to whom the notice is given. In accordance with Section 232 of the General Corporation Law, notices are deemed given (i) if by mail, when deposited in the United States mail, postage prepaid, directed to the stockholder at such stockholder's address as it appears on the records of the Company, or, if a stockholder shall have filed with the Secretary of the Company a written request that notices to such stockholder be mailed to some other address, then directed to such stockholder at such other address; (ii) if by facsimile,

H-13

when directed to a number at which the stockholder has consented to receive notice; (iii) if by electronic mail, when directed to an electronic mail address at which the stockholder has consented to receive such notice; (iv) if by posting on an electronic network together with a separate notice to the stockholder of such specific posting, upon the later to occur of (A) such posting and (B) the giving of such separate notice of such posting; and (v) if by any other form of electronic transmission, when directed to the stockholder in the manner consented to by the stockholder. Notice shall be deemed to have been given to all stockholders of record who share an address if notice is given in accordance with Section 233 of the General Corporation Law. An affidavit of the Secretary, Assistant Secretary or any transfer agent of the Company that the notice has been given shall, in the absence of fraud, be prima facie evidence of the facts stated therein.

- (b) Whenever any notice is required to be given under the provisions of any applicable law or of the Company's Certificate of Incorporation or of these by-laws, a written waiver of notice, signed by the person or persons entitled to said notice, or a waiver of notice by electronic transmission, whether before or after the time stated therein or the meeting or action to which such notice relates, shall be deemed equivalent to notice. Attendance of a person at a meeting shall constitute a waiver of notice of such meeting except when the person attends a meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened.
- 6.3 <u>Conflict of Interest</u>. No contract or transaction between the Company and one or more of its directors or officers, or between the Company and any other corporation, partnership, association, or other organization in which one or more of its directors or officers are directors or officers, or have a financial interest, shall be void or voidable solely for this reason, or solely because the director or officer is present at or participates in the meeting of the

board or committee thereof that authorized the contract or transaction, or solely because any such director's or officer's votes are counted for such purpose, if:
(i) the material facts as to the director's or officer's relationship or interest and as to the contract or transaction are disclosed or are known to the Board of Directors or the committee, and the Board of Directors or committee in good faith authorizes the contract or transaction by the affirmative vote of a majority of the disinterested directors, even though the disinterested directors be less than a quorum; or (ii) the material facts as to the director's or officer's relationship or interest and as to the contract or transaction are disclosed or are known to the stockholders of the Company entitled to vote thereon, and the contract or transaction as specifically approved in good faith by vote of such stockholders; or (iii) the contract or transaction is fair as to the Company as of the time it is authorized, approved, or ratified, by the Board of Directors, a committee or the stockholders. Common or interested directors may be counted in determining the presence of a quorum at a meeting of the Board of Directors or of a committee that authorizes the contract or transaction.

6.4 **Voting of Securities Owned by the Company**. Subject always to the specific directions of the Board of Directors, (i) any shares or other securities issued by any other entity and owned or controlled by the Company may be voted in person at any meeting of security holders of such other entity by the President of the Company if he is present at such meeting, or in his absence by the Treasurer of the Company if he is present at such meeting, and (ii) whenever, in the judgment of the President, it is desirable for the Company to execute a proxy or written consent in respect to any shares or other securities issued by any other entity and owned by the Company, such proxy or consent shall be executed in the name of the Company by the President, without the necessity of any authorization by the Board of Directors, affixation of corporate seal or countersignature or attestation by another officer, provided that if the President is unable to execute such proxy or consent by reason of sickness, absence from the United States or other similar cause, the Treasurer may execute such proxy or consent. Any person or persons designated in the manner above stated as the proxy or proxies of the Company shall have full right, power

H-14

and authority to vote the shares or other securities issued by such other entity and owned by the Company the same as such shares or other securities might be voted by the Company.

#### Article VII - Indemnification.

- **Right to Indemnification**. Each person who was or is made a party or is threatened to be made a party to or is otherwise involved in any action, suit or proceeding, whether civil, criminal, administrative or investigative (a "Proceeding"), by reason of being or having been a director or officer of the Company or, while a director or officer of the Company, serving or having served at the request of the Company as a director, trustee, officer, employee or agent of another corporation or of a partnership, joint venture, trust or other enterprise, including service with respect to an employee benefit plan (an "Indemnitee"), shall be indemnified and held harmless by the Company to the fullest extent permitted by applicable law, as the same exists or may hereafter be amended, against all expense, liability and loss (including attorneys' fees, judgments, fines, ERISA excise taxes or penalties and amounts paid in settlement) reasonably incurred or suffered by such Indemnitee in connection therewith and such indemnification shall continue as to an Indemnitee who has ceased to be a director, trustee, officer, employee, or agent and shall inure to the benefit of the Indemnitee's heirs, executors, and administrators; provided, however, that, except as provided in Section 7.2 hereof with respect to Proceedings to enforce rights to indemnification or Advancement of Expenses, the Company shall indemnify any such Indemnitee in connection with a Proceeding (or part thereof) initiated by such Indemnitee only if such Proceeding (or part thereof) was approved by the Board of Directors of the Company. The right to indemnification conferred in this Article 7 shall be a contract right and shall include the right to be paid by the Company the expenses (including attorneys' fees) incurred in defending any such Proceeding in advance of its final disposition (an "Advancement of Expenses"); provided, however, that, to the extent required by law, such Advancement of Expenses shall be made only upon the Company's receipt of an undertaking (an "Undertaking"), by or on behalf of such Indemnitee, to repay all amounts so advanced if it shall ultimately be determined by final judicial decision from which there is no further right to appeal (a "Final Adjudication") that such Indemnitee is not entitled to be indemnified for such expenses under this Article 7 or otherwise.
- Right of Indemnitee to Bring Suit. If a claim under Section 7.1 hereof is not paid in full by the Company within sixty days after a written claim has been received by the Company, except in the case of a claim for an Advancement of Expenses, in which case the applicable period shall be twenty days, the Indemnitee may at any time thereafter bring suit against the Company to recover the unpaid amount of the claim. If successful in whole or in part in any such suit, or in a suit brought by the Company to recover an Advancement of Expenses pursuant to the terms of an Undertaking, the Indemnitee shall be entitled to be paid also the expense of prosecuting or defending such suit to the fullest extent permitted by applicable law. In (i) any suit brought by the Indemnitee to enforce a right to indemnification hereunder (but not in a suit brought by the Indemnitee to enforce a right to an Advancement of Expenses) it shall be a defense that, and (ii) in any suit by the Company to recover an Advancement of Expenses pursuant to the terms of an Undertaking the Company shall be entitled to recover such expenses upon a Final Adjudication that, the Indemnitee has not met the applicable standard of conduct set forth in Section 145 of the General Corporation Law. Neither the failure of the Company (including its Board of Directors, independent legal counsel, or its stockholders) to have made a determination prior to the commencement of such suit that indemnification of the Indemnitee is proper in the circumstances because the Indemnitee has met the applicable standard of conduct set forth in Section 145 of the General Corporation Law, nor an actual determination by the Company (including its Board of Directors, independent legal counsel, or its stockholders) that the Indemnitee has not met such applicable standard of conduct, shall create a presumption that the Indemnitee has not met the applicable standard of conduct or, in the case of such a suit brought by the Indemnitee, be a defense to such suit. In any suit brought by the

H-15

hereunder, or by the Company to recover an Advancement of Expenses pursuant to the terms of an Undertaking, the burden of proving that the Indemnitee is not entitled to be indemnified, or to such Advancement of Expenses, under this Article 7 or otherwise shall be on the Company.

- 7.3 Non-Exclusivity of Rights. The rights to indemnification and to the Advancement of Expenses conferred in this Article 7 shall not be exclusive of any other right that any person may have or hereafter acquire under any law, the Company's Certificate of Incorporation, these by-laws, agreement, vote of stockholders or disinterested directors or otherwise.
- 7.4 **Insurance**. The Company may maintain insurance, at its expense, to protect itself and any director, officer, employee or agent of the Company or another corporation, partnership, joint venture, trust or other enterprise against any expense, liability or loss, whether or not the Company would have the power to indemnify such person against such expense, liability or loss under this Article 7 or under the General Corporation Law.

- 7.5 <u>Indemnification of Employees and Agents of the Company</u>. The Company may, to the extent authorized from time to time by the Board of Directors, grant rights to indemnification, and to the Advancement of Expenses, to any employee or agent of the Company to the fullest extent of the provisions of this Article 7 with respect to the indemnification and Advancement of Expenses of directors and officers of the Company.
- 7.6 **Amendment or Repeal**. Any repeal or modification of the foregoing provisions of this Article VII shall not adversely affect any right or protection hereunder of any Indemnitee in respect of any act or omission occurring prior to the time of such repeal or modification.

# Article VIII - Amendments.

Amendments. Subject always to any limitations imposed by the Company's Certificate of Incorporation, these by-laws may be altered, amended, or repealed, or new by-laws may be adopted, by resolution of the Board of Directors duly adopted by a majority of the Whole Board, including (for so long as the Stockholders Agreement remains in effect) the approval of a majority of the Independent Directors. With respect to the power of holders of capital stock to adopt, amend and repeal by-laws of the Company, notwithstanding any other provision of these by-laws or any provision that might otherwise permit a lesser vote or no vote, in addition to any vote of the holders of any class or series of capital stock of the Company required herein or by law, the affirmative vote of the holders of the voting power of at least 66 2/3% of the shares of the Company's stock entitled to vote thereon, voting together as a single class, shall be required for any such alteration, amendment, repeal, or adoption by the vote of the holders of any class or series of the capital stock of the Company.

H-16

EXHIBIT I AGREED FORM

### CERTIFICATE OF SONUS NETWORKS, INC.

, 2017

Re: Agreement and Plan of Merger dated May 23, 2017 (the "Agreement") by and among Sonus Networks, Inc., Solstice Sapphire Investments, Inc., Solstice Sapphire Investments LLC, Green Sapphire LLC, Genband Holdings Company, Genband Inc., and Genband II, Inc.

This certificate is supplied to you in connection with your rendering of an opinion regarding certain U.S. federal income tax consequences of certain of the Mergers to be undertaken pursuant to the Agreement. Unless otherwise indicated, capitalized terms not defined herein have the meanings set forth in the Agreement.

- A. <u>Representations</u>. The undersigned hereby certifies and represents that the following facts are true, correct and complete in all respects and will be true, correct and complete in all respects at the Effective Time:
- 1. Solstice and, to the best knowledge of the undersigned, the stockholders of Solstice ("Equityholders") are participating in the Solstice Merger for good and valid business reasons and not for tax purposes. The terms of the Agreement and all other agreements entered into in connection with the Agreement are the products of arm's length negotiation;
- 2. The Mergers will occur on the same date at approximately the same time as part of a single plan pursuant to the terms of the Agreement, and the Agreement defines the rights of each of the parties to the transactions contemplated by the Agreement;
- 3. Each Equityholder will transfer its stock in Solstice pursuant to the Solstice Merger solely in exchange for Company stock (and/or cash received in lieu of fractional shares of Company stock is the only consideration other than Company stock that will be received by Equityholders in connection with the transfer of their stock in Solstice in the Solstice Merger. Any Company stock to be received by the Equityholders will be issued by the Company;
- 4. Other than with regard to Company Stock issued in exchange for Compensatory Solstice Stock, which Company stock shall not exceed 1.5% of the value of all Company stock that is issued and outstanding after the Mergers, no Company stock or securities will be issued in the Solstice Merger for services rendered to or for the benefit of the Company or its Subsidiaries in connection with the Solstice Merger. "Compensatory Solstice Stock" shall

I-1

mean any Solstice Restricted Shares and any Solstice stock received in settlement of Solstice RSUs that vest by reason of the Solstice Merger;

- 5. No Company stock or securities will be issued in the Solstice Merger for indebtedness of the Company or for interest on indebtedness of the Company;
- 6. There is no intercorporate indebtedness existing between Solstice, on the one hand, and the Company, on the other hand, that was issued, acquired, or will be settled at a discount;
  - 7. The Solstice Merger is not the result of the solicitation by a promoter, broker, or investment house;
- 8. The Equityholders will not retain any rights in the Solstice stock to be transferred to the Company pursuant to the Solstice Merger at or after the Effective Time;
- 9. No Company stock received in the Solstice Merger will be received in exchange for any licenses, leases or similar rights between the Equityholders and the Company;

- 10. To the best knowledge of the undersigned, no property transferred to the Company in connection with the Mergers will be leased back to an Equityholder, another shareholder of the Company or a related party;
- 11. Solstice and persons related to Solstice, as defined in Treas. Reg. Section 1.368-1(e)(3), have not redeemed, purchased, exchanged or otherwise acquired any Solstice stock and Solstice has not made any distributions with respect to its stock, in connection with the Mergers;
- 12. The Solstice stock to be exchanged for Company stock in the Solstice Merger was not acquired by the Equityholders in connection with the liquidation of another corporation;
  - 13. No liabilities of any Equityholders will be assumed by the Company in the Solstice Merger;
- 14. To the best knowledge of the undersigned, the Solstice stock to be transferred to the Company in the Solstice Merger is not subject to any liabilities;
- 15. To the best knowledge of the undersigned, at the time of the Mergers, there will be no indebtedness between the Company and the Equityholders, and there will be no indebtedness created in favor of the Equityholders as a result of the Solstice Merger;
- 16. To the best knowledge of the undersigned, following the Solstice Merger, Solstice will continue its "historic business" within the meaning of Treasury Regulation Section 1.368-1(d) or use a significant portion of its historic business assets in a business;
- 17. To the best knowledge of the undersigned, there is no plan or intention on the part of the Company to redeem or otherwise reacquire any stock or indebtedness to be issued by the Company in the Mergers. To the best knowledge of the undersigned, no person related to the

I-2

Company, as defined in Treas. Reg. Section 1.368-1(e)(3), and no person acting as intermediary for the Company or such related person (i) has acquired any stock of Solstice or (ii) has a plan or intention to acquire any of the Company stock to be issued in the Mergers;

- 18. To the best knowledge of the undersigned, the Equityholders do not have any plan or intention to sell, exchange or otherwise dispose of any of the Company stock to be issued in the Solstice Merger;
- 19. To the best knowledge of the undersigned, taking into account any issuance of additional shares of Company stock, any issuance of Company stock for services (including the Company stock that will be issued to holders of Class E Shares (as defined in the Amended and Restated Memorandum and Articles of Association of Glass) of Glass and the Company stock that will be issued to holders of Compensatory Solstice Stock in respect of such Compensatory Solstice Stock, which Company stock, in the aggregate, shall not exceed 1.5% of the value of all Company stock that is issued and outstanding after the Mergers (such Company common stock issued to such holders, the "Compensatory Company Stock")), the exercise of any Company stock rights, warrants or subscriptions, a public offering of Company stock, and the sale, exchange, transfer by gift, or other disposition of any of the stock of Company to be received in the Mergers, the direct shareholders of Glass immediately prior to the Mergers (excluding for this purpose (i) any shareholders of Glass who receive Compensatory Company Stock and (ii) the Glass Corporate Members and their stockholders), together with the Equityholders (excluding for this purpose any Equityholders who receive Compensatory Company Stock in their capacity as holders of Compensatory Solstice Stock) (together, such holders, the "Control Transferors"), will be in "control" (within the meaning of Sections 351(a) and 368(c) of the Code) of the Company immediately after the Mergers;
- 20. To the best knowledge of the undersigned, there exist no rights to acquire Company stock or to vote (or restrict or otherwise control the vote of) shares of Company stock which, if exercised, would affect the Control Transferors' acquisition and retention of "control" (within the meaning of Sections 351(a) and 368(c) of the Code) of the Company. To the best knowledge of the undersigned, there is no plan or intention for the Company to issue additional shares of Company stock after the Mergers, or take any other action, that would result in the Control Transferors not being in "control" of the Company after the Mergers;
- 21. The Equityholders will receive Company stock with a fair market value approximately equal to the fair market value of the Solstice stock disposed of in the Solstice Merger;
- 22. No Company stock will be placed in escrow or will be issued after the Closing Date pursuant to a contingent stock arrangement, in each case, in connection with the Mergers;
- 23. The principal purpose of the formation and use of the Company is not the avoidance of U.S. federal income tax or to obtain a U.S. federal income tax benefit that would have otherwise been unavailable;

I-3

- 24. To the best knowledge of the undersigned, the Company will remain in existence and retain and use the property transferred to it in a trade or business (or such property will reflect the ownership of subsidiaries engaged in a trade or business);
- 25. To the best knowledge of the undersigned, the Company has no plan or intention (i) to liquidate Solstice, (ii) to merge Glass or Solstice with or into another entity, (iii) to sell or otherwise dispose of the shares of Glass or the stock of Solstice acquired in the Mergers, (iv) to sell or otherwise dispose of the assets of Glass or Solstice or (v) to cause Glass or Solstice to sell or otherwise dispose of any of its assets, except, in the case of (iv) and (v), dispositions of assets made in the ordinary course of business and, in the case of (iii), (iv) and (v), transfers or successive transfers of stock or assets to one or more corporations controlled (within the meaning of Section 368(c) of the Code) in each case by the transferor corporation;
  - 26. Each of the parties to the Mergers will pay its own expenses, if any, incurred in connection with the proposed transactions;

27. To the best knowledge of the undersigned, at the Effective Time and immediately thereafter, the Company will not be an investment company within the meaning of Section 351(e)(1) of the Code and Treasury Regulations Section 1.351-1(c)(1)(ii);

28. Neither Solstice nor, to the best knowledge of the undersigned, the Company or any Equityholder is under the jurisdiction of a court in a title 11 or similar case (within the meaning of Section 368(a)(3)(A) of the Code);

29. At the Effective Time and immediately thereafter, to the best knowledge of the undersigned, the Company will not be a "personal service corporation" within the meaning of Section 269A of the Code;

30. The Solstice Merger will be consummated in compliance with the terms of the Agreement and none of the terms and conditions therein have been or will have been waived or modified. The Agreement and the other documents described in Section 9.4 of the Agreement constitute the entire agreement among the parties thereto with respect to the matters governed thereby;

31. To the best knowledge of the undersigned, the Equityholders and the Company will comply with the reporting and record-keeping requirements set forth in Treasury Regulations Section 1.351-3;

32. Prior to the Solstice Merger, the Company will be in "control" of Solstice Merger Sub within the meaning of Section 368(c) of the Code;

33. Solstice has no plan or intention to issue additional shares of its stock that would result in the Company losing control of Solstice within the meaning of Section 368(c) of the Code following the Solstice Merger;

**I-4** 

who is (or was) a service provider of Solstice will be separate consideration for, or allocable to, such Equityholder's shares of Solstice stock surrendered in the Solstice Merger; none of the shares of Company stock received in the Solstice Merger by any Equityholder who is (or was) a service provider of Solstice will be separate consideration for any past or future services; and the compensation paid to any such Equityholder will be for services actually rendered and

"Opinion") will be based on the representations set forth herein and will assume that all of the representations and statements set forth herein are true without regard to any qualification as to knowledge or belief and will be based on the statements contained in the Agreement and the documents related thereto, (ii) each Opinion will be subject to certain limitations and qualifications including that it may not be relied upon if any such representations are not accurate in all material respects without regard to any qualification as to knowledge or belief, and (iii) notwithstanding any provisions of the Agreement to the

I-5

By:

Name:

Title:

In witness whereof, the undersigned executed this certificate as of the date and year first above-written.

35. The facts which relate to the Mergers which are described in the Agreement and the Joint Proxy Statement/Prospectus are true, correct

36. None of the compensation received from the Company or its subsidiaries in connection with or after the Mergers by any Equityholder

37. The undersigned officer of Solstice is authorized to make all of the representations set forth herein on behalf of itself and Solstice.

1. The undersigned recognizes and agrees that (i) the opinions of WilmerHale LLP and Latham & Watkins LLP (the "Law Firms") (each, an

The undersigned recognizes and agrees that no Opinion will address any tax consequences of the Mergers or any action taken in

The undersigned hereby undertakes to inform the Law Firms immediately in writing should any of the statements or representations set

Sonus Networks, Inc.

34. Prior to the Solstice Merger, the Company has never owned any shares of Solstice.

will be commensurate with amounts paid to third parties bargaining at arm's-length for similar services; and

Reliance by Law Firms in Rendering Opinions: Limitation on the Opinions.

forth in this letter become untrue, incorrect or incomplete in any respect on or prior to the Effective Time.

contrary, the representations set forth in this letter will survive without limitation.

connection therewith except as expressly set forth in such Opinion.

and complete in all material respects;

В.

EXHIBIT J AGREED FORM

### CERTIFICATE OF GENBAND HOLDINGS COMPANY

, 2017

Re: Agreement and Plan of Merger dated May 23, 2017 (the "Agreement") by and among Sonus Networks, Inc., Solstice Sapphire Investments, Inc., Solstice Sapphire Investments LLC, Green Sapphire LLC, Genband Holdings company, Genband Inc., and Genband II, Inc.

This certificate is supplied to you in connection with your rendering of an opinion regarding certain U.S. federal income tax consequences of certain of the Mergers to be undertaken pursuant to the Agreement. Unless otherwise indicated, capitalized terms not defined herein have the meanings set forth in the Agreement.

- A. <u>Representations</u>. The undersigned hereby certifies and represents that the following facts are true, correct and complete in all respects and will be true, correct and complete in all respects at the Effective Time:
- 1. Glass and, to the best knowledge of the undersigned, the direct shareholders in Glass (such shareholders, the "Equityholders") are participating in the Cayman Merger for good and valid business reasons and not for tax purposes. The terms of the Agreement and all other agreements entered into in connection with the Agreement are the products of arm's length negotiation;
- 2. The Mergers will occur on the same date at approximately the same time as part of a single plan pursuant to the terms of the Agreement, and the Agreement defines the rights of each of the parties to the transactions contemplated by the Agreement;
- 3. Each Equityholder (other than the Glass Corporate Members and the Equityholders entitled to receive the Promissory Note pursuant to Section 3.2(d)(ii) of the Agreement) who is entitled to receive consideration in the Cayman Merger will dispose of its shares in Glass pursuant to the Cayman Merger solely in exchange for Company stock (and/or cash received in lieu of fractional shares of Company stock). The Promissory Note and any cash received in lieu of fractional shares of Company stock that will be received by Equityholders in the Mergers. The Company stock to be received by the Equityholders will be issued by the Company;
- 4. Other than with regard to Company stock issued to holders of Class E Shares (as defined in the Amended and Restated Memorandum and Articles of Association of Glass) of Glass, which Company stock shall not exceed 1.5% of the value of all Company stock that is

**J-**1

issued and outstanding after the Mergers, no Company stock or securities will be issued in the Cayman Merger for services rendered to or for the benefit of the Company or its Subsidiaries in connection with the Cayman Merger;

- 5. No Company stock or securities will be issued in the Cayman Merger for indebtedness of the Company or for interest on indebtedness of the Company;
  - 6. The Cayman Merger is not the result of the solicitation by a promoter, broker, or investment house;
- 7. The Equityholders will not retain any rights in the Glass shares to be disposed of pursuant to the Cayman Merger at or after the Effective Time:
- 8. No Company stock received in the Cayman Merger will be received in exchange for any licenses, leases or similar rights between the Equityholders and the Company;
- 9. To the best knowledge of the undersigned, no property transferred to the Company in connection with the Mergers will be leased back to an Equityholder, another shareholder of the Company, or a related party;
- 10. The Glass shares to be disposed of in the Cayman Merger were not acquired by the Equityholders in connection with the liquidation of another corporation;
  - 11. No liabilities of any Equityholders will be assumed by the Company in the Cayman Merger;
- 12. To the best knowledge of the undersigned, the Glass shares to be transferred to the Company in the Cayman Merger are not subject to any liabilities;
- 13. To the best knowledge of the undersigned, at the time of the Mergers, there will be no indebtedness between the Company and the Equityholders, and other than the Promissory Note, there will be no indebtedness created in favor of the Equityholders as a result of the Cayman Merger;
- 14. To the best knowledge of the undersigned, following the Mergers, Solstice will continue its "historic business" within the meaning of Treasury Regulation Section 1.368-1(d) or use a significant portion of its historic business assets in a business;
- 15. To the best knowledge of the undersigned, there is no plan or intention on the part of the Company to redeem or otherwise reacquire any stock or indebtedness to be issued by the Company in the Mergers. To the best knowledge of the undersigned, no person related to the Company, as defined in Treas. Reg. Section 1.368-1(e)(3), and no person acting as intermediary for the Company or such related person (i) has acquired any shares of Glass or (ii) has a plan or intention to acquire any of the Company stock to be issued in the Mergers;
- 16. To the best knowledge of the undersigned, the Equityholders do not have any plan or intention to sell, exchange or otherwise dispose of any of the Company stock to be issued in the Cayman Merger;

- 17. To the best knowledge of the undersigned, taking into account any issuance of additional shares of Company stock, any issuance of Company stock for services (including the Company stock that will be issued to holders of Class E Shares (as defined in the Amended and Restated Memorandum and Articles of Association of Glass) of Glass and the Company stock that will be issued in exchange for any Solstice Restricted Shares and any Solstice stock received in settlement of Solstice RSUs that vest by reason of the Solstice Merger, which Company stock, in the aggregate, shall not exceed 1.5% of the value of all Company stock that is issued and outstanding after the Mergers (such Company common stock issued to such holders, the "Compensatory Company Stock")), the exercise of any Company stock rights, warrants or subscriptions, a public offering of Company stock, and the sale, exchange, transfer by gift, or other disposition of any of the stock of Company to be received in the Mergers, the Equityholders (excluding for this purpose (i) the Equityholders who receive Compensatory Company Stock and (ii) the Glass Corporate Members and their stockholders), together with the direct stockholders of Solstice immediately prior to the Mergers (excluding for this purpose any stockholders of Solstice who receive Compensatory Company Stock in their capacity as holders of the Solstice stock exchanged for such Compensatory Company Stock) (together, such holders, the "Control Transferors") will be in "control" (within the meaning of Sections 351(a) and 368(c) of the Code) of the Company immediately after the Mergers;
- 18. To the best knowledge of the undersigned, there exist no rights to acquire Company stock or to vote (or restrict or otherwise control the vote of) shares of Company stock which, if exercised, would affect the Control Transferors' acquisition and retention of "control" (within the meaning of Sections 351(a) and 368(c) of the Code) of the Company. To the best knowledge of the undersigned, there is no plan or intention for the Company to issue additional shares of Company stock after the Mergers, or take any other action, that would result in the Control Transferors not being in "control" of the Company after the Mergers;
- 19. In exchange for the Glass shares disposed of in the Cayman Merger, the Equityholders will receive Company stock or other property with a fair market value approximately equal to the fair market value of the Glass shares disposed of in the Cayman Merger;
- 20. No Company stock will be placed in escrow or will be issued after the Closing Date pursuant to a contingent stock arrangement, in each case, in connection with the Mergers;
- 21. The principal purpose of the formation and use of the Company is not the avoidance of U.S. federal income tax or to obtain a U.S. federal income tax benefit that would have otherwise been unavailable;
- 22. To the best knowledge of the undersigned, the Company will remain in existence and retain and use the property transferred to it in a trade or business (or such property will reflect the ownership of subsidiaries engaged in a trade or business);
- 23. To the best knowledge of the undersigned, the Company has no plan or intention (i) to liquidate Solstice, (ii) to merge Glass or Solstice with or into another entity, (iii) to sell or otherwise dispose of the shares of Glass or the stock of Solstice acquired in the Mergers, (iv) to sell or otherwise dispose of the assets of Glass or Solstice, or (v) to cause Glass

J**-**3

or Solstice to sell or otherwise dispose of any of its assets, except, in the case of (iv) and (v), dispositions of assets made in the ordinary course of business and, in the case of (iii), (iv) and (v), transfers or successive transfers of stock or assets to one or more corporations controlled (within the meaning of Section 368(c) of the Code) in each case by the transferor corporation;

- 24. Each of the parties to the Mergers will pay its own expenses, if any, incurred in connection with the proposed transactions;
- 25. To the best knowledge of the undersigned, at the Effective Time and immediately thereafter, the Company will not be an investment company within the meaning of Section 351(e)(1) of the Code and Treasury Regulations Section 1.351-1(c)(1)(ii);
- 26. Neither Glass nor, to the best knowledge of the undersigned, any Equityholder is under the jurisdiction of a court in a title 11 or similar case (within the meaning of Section 368(a)(3)(A) of the Code);
- 27. At the Effective Time and immediately thereafter, to the best knowledge of the undersigned, the Company will not be a "personal service corporation" within the meaning of Section 269A of the Code;
- 28. The Cayman Merger will be consummated in compliance with the terms of the Agreement and none of the terms and conditions therein have been or will have been waived or modified. The Agreement and the other documents described in Section 9.4 of the Agreement constitute the entire agreement among the parties thereto with respect to the matters governed thereby;
- 29. To the best knowledge of the undersigned, the Equityholders and the Company will comply with the reporting and record-keeping requirements set forth in Treasury Regulations Section 1.351-3;
- 30. To the best knowledge of the undersigned, the Company has no plan or intention to cause Solstice to issue additional shares of its stock that would result in the Company losing control of Solstice within the meaning of Section 368(c) of the Code following the Solstice Merger
- 31. The facts which relate to the Mergers which are described in the Agreement and the Joint Proxy Statement/Prospectus are true, correct and complete in all material respects;
- 32. None of the compensation received from the Company or its subsidiaries in connection with or after the Mergers by any Equityholder who is (or was) a service provider of Glass will be separate consideration for, or allocable to, such Equityholder's shares of Glass surrendered in the Cayman Merger; none of the shares of Company stock received in the Cayman Merger by any Equityholder who is (or was) a service provider of Glass (other than the Compensatory Company Stock) will be separate consideration for any past or future services; and the compensation paid to any such Equityholder will be for services actually rendered and will be commensurate with amounts paid to third parties bargaining at arm's-length for similar services; and

- 33. The undersigned officer of Glass is authorized to make all of the representations set forth herein on behalf of itself and Glass.
- B. Reliance by Law Firms in Rendering Opinions: Limitation on the Opinions.
- 1. The undersigned recognizes and agrees that (i) the opinions of WilmerHale LLP and Latham & Watkins LLP (the "Law Firms") (each, an "Opinion") will be based on the representations set forth herein and will assume that all of the representations and statements set forth herein are true without regard to any qualification as to knowledge or belief and will be based on the statements contained in the Agreement and the documents related thereto, (ii) each Opinion will be subject to certain limitations and qualifications including that it may not be relied upon if any such representations are not accurate in all material respects without regard to any qualification as to knowledge or belief, and (iii) notwithstanding any provisions of the Agreement to the contrary, the representations set forth in this letter will survive without limitation.
- 2. The undersigned recognizes and agrees that no Opinion will address any tax consequences of the Mergers or any action taken in connection therewith except as expressly set forth in such Opinion.
- 3. The undersigned hereby undertakes to inform the Law Firms immediately in writing should any of the statements or representations set forth in this letter become untrue, incorrect or incomplete in any respect on or prior to the Effective Time.

TE

In witness whereof, the undersigned executed this certificate as of the date and year first above-written.
Genband Holdings Company
Ву:

EXHIBIT K AGREED FORM

## CERTIFICATE OF GENBAND INC.

Name:

Title:

, 2017

Re: Agreement and Plan of Merger dated May 23, 2017 (the "Agreement") by and among Sonus Networks, Inc., Solstice Sapphire Investments, Inc., Solstice Sapphire Investments LLC, Green Sapphire LLC, Genband Holdings Company, Genband Inc., and Genband II, Inc.

This certificate is supplied to you in connection with your rendering of an opinion regarding certain U.S. federal income tax consequences of certain of the Mergers to be undertaken pursuant to the Agreement. Unless otherwise indicated, capitalized terms not defined herein have the meanings set forth in the Agreement.

- A. <u>Representations</u>. The undersigned hereby certifies and represents that the following facts are true, correct and complete in all respects and will be true, correct and complete in all respects at the Effective Time:
- 1. GB and, to the best knowledge of the undersigned, the direct stockholders in GB (such stockholders, the "Equityholders") are participating in the GB Merger for good and valid business reasons and not for tax purposes. The terms of the Agreement and all other agreements entered into in connection with the Agreement are the products of arm's length negotiation;
- 2. The Mergers will occur on the same date at approximately the same time as part of a single plan pursuant to the terms of the Agreement, and the Agreement defines the rights of each of the parties to the transactions contemplated by the Agreement;
- 3. Each Equityholder will dispose of its stock in GB pursuant to the GB Merger solely in exchange either (i) for no consideration or (ii) for Company stock (and/or cash received in lieu of fractional shares of Company stock). Any cash received in lieu of fractional shares of Company stock is the only consideration other than Company stock that will be received by Equityholders in connection with the disposition of their equity in GB via the GB Merger. Any Company stock to be received by the Equityholders will be issued by the Company;
- 4. No Company stock or securities will be issued in the GB Merger for services rendered to or for the benefit of the Company or its Subsidiaries in connection with the GB Merger;
- 5. No Company stock or securities will be issued in the GB Merger for indebtedness of the Company or for interest on indebtedness of the Company;

- 6. The GB Merger is not the result of the solicitation by a promoter, broker, or investment house;
- 7. The Equityholders will not retain any rights in the GB stock to be disposed of pursuant to the GB Merger at or after the Effective Time;
- 8. No Company stock received in the GB Merger will be received in exchange for any licenses, leases or similar rights between the Equityholders and the Company;
- 9. To the best knowledge of the undersigned, no property transferred to the Company in connection with the Mergers will be leased back to an Equityholder, another shareholder of the Company or a related party;
- 10. The GB stock disposed of in the GB Merger was not acquired by the Equityholders in connection with the liquidation of another corporation, and the Glass shares held by GB were not acquired by GB in connection with the liquidation of another corporation;
  - 11. No liabilities of any Equityholders will be assumed by the Company in the GB Merger;
  - 12. To the best knowledge of the undersigned, the GB stock to be disposed of in the GB Merger is not subject to any liabilities;
- 13. To the best knowledge of the undersigned, at the time of the Mergers, there will be no indebtedness between the Company and the Equityholders, and there will be no indebtedness created in favor of the Equityholders as a result of the GB Merger;
- 14. To the best knowledge of the undersigned, following the Mergers, Solstice will continue its "historic business" within the meaning of Treasury Regulation Section 1.368-1(d) or use a significant portion of its historic business assets in a business;
- 15. To the best knowledge of the undersigned, there is no plan or intention on the part of the Company to redeem or otherwise reacquire any stock or indebtedness to be issued by the Company in the Mergers. To the best knowledge of the undersigned, no person related to the Company, as defined in Treas. Reg. Section 1.368-1(e)(3), and no person acting as intermediary for the Company or such related person (i) has acquired any stock of GB or (ii) has a plan or intention to acquire any of the Company stock to be issued in the Mergers;
- 16. To the best knowledge of the undersigned, the Equityholders do not have any plan or intention to sell, exchange or otherwise dispose of any of the Company stock issued in the GB Merger;
- 17. To the best knowledge of the undersigned, taking into account any issuance of additional shares of Company stock, any issuance of Company stock for services (including the Company stock that will be issued to holders of Class E Shares (as defined in the Amended and Restated Memorandum and Articles of Association of Glass) of Glass and the Company stock that will be issued in exchange for any Solstice Restricted Shares and any Solstice stock received in settlement of Solstice RSUs that vest by reason of the Solstice Merger, which Company stock,

K-2

in the aggregate, shall not exceed 1.5% of the value of all Company stock that is issued and outstanding after the Mergers (such Company common stock issued to such holders, the "Compensatory Company Stock")), the exercise of any Company stock rights, warrants or subscriptions, a public offering of Company stock, and the sale, exchange, transfer by gift, or other disposition of any of the stock of Company to be received in the Mergers, the direct shareholders of Glass immediately prior to the Mergers (excluding for this purpose (i) any shareholders of Glass who receive Compensatory Company Stock and (ii) the Glass Corporate Members and their stockholders), together with the direct stockholders of Solstice immediately prior to the Mergers (excluding for this purpose any stockholders of Solstice who receive Compensatory Company Stock in their capacity as holders of the Solstice stock exchanged for such Compensatory Company Stock) (together, such holders, the "Control Transferors"), will be in "control" (within the meaning of Sections 351(a) and 368(c) of the Code) of the Company immediately after the Mergers;

- 18. To the best knowledge of the undersigned, there exist no rights to acquire Company stock or to vote (or restrict or otherwise control the vote of) shares of Company stock which, if exercised, would affect the Control Transferors' acquisition and retention of "control" (within the meaning of Sections 351(a) and 368(c) of the Code) of the Company. To the best knowledge of the undersigned, there is no plan or intention for the Company to issue additional shares of Company stock after the Mergers, or take any other action, that would result in the Control Transferors not being in "control" of the Company after the Mergers;
- 19. The Equityholders will receive Company stock or other property approximately equal to the fair market value of the GB stock disposed of in the GB Merger;
- 20. No Company stock will be placed in escrow or will be issued after the Closing Date pursuant to a contingent stock arrangement, in each case, in connection with the Mergers;
- 21. The principal purpose of the formation and use of the Company is not the avoidance of U.S. federal income tax or to obtain a U.S. federal income tax benefit that would have otherwise been unavailable;
- 22. To the best knowledge of the undersigned, the Company will remain in existence and retain and use the property transferred to it in a trade or business (or such property will reflect the ownership of subsidiaries engaged in a trade or business);
- 23. To the best knowledge of the undersigned, , the Company has no plan or intention (i) to liquidate Solstice, (ii) to merge Glass or Solstice with or into another entity, (iii) to sell or otherwise dispose of the shares of Glass or the stock of Solstice acquired in the Mergers, (iv) to sell or otherwise dispose of the assets of Glass or Solstice, or (v) to cause Glass or Solstice to sell or otherwise dispose of any of its assets, except, in the case of (iv) and (v), dispositions of assets made in the ordinary course of business and, in the case of (iii), (iv) and (v), transfers or successive transfers of stock or assets to one or more corporations controlled (within the meaning of Section 368(c) of the Code) in each case by the transferor corporation;
  - 24. Each of the parties to the Mergers will pay its own expenses, if any, incurred in connection with the proposed transactions;

- 25. To the best knowledge of the undersigned, at the Effective Time and immediately thereafter, the Company will not be an investment company within the meaning of Section 351(e)(1) of the Code and Treasury Regulations Section 1.351-1(c)(1)(ii);
- 26. Neither GB nor, to the best knowledge of the undersigned, any Equityholder is under the jurisdiction of a court in a title 11 or similar case (within the meaning of Section 368(a)(3)(A) of the Code);
- 27. At the Effective Time and immediately thereafter, to the best knowledge of the undersigned, the Company will not be a "personal service corporation" within the meaning of Section 269A of the Code;
- 28. The GB Merger will be consummated in compliance with the terms of the Agreement and none of the terms and conditions therein have been or will have been waived or modified. The Agreement and the other documents described in Section 9.4 of the Agreement constitute the entire agreement among the parties thereto with respect to the matters governed thereby;
- 29. To the best knowledge of the undersigned, the Equityholders and the Company will comply with the reporting and record-keeping requirements set forth in Treasury Regulations Section 1.351-3;
- 30. To the best knowledge of the undersigned, the Company has no plan or intention to cause Solstice to issue additional shares of its stock that would result in the Company losing control of Solstice within the meaning of Section 368(c) of the Code following the Solstice Merger;
- 31. The facts which relate to the Mergers which are described in the Agreement and the Joint Proxy Statement/Prospectus are true, correct and complete in all material respects;
- 32. None of the compensation received from the Company or its subsidiaries in connection with or after the Mergers by any Equityholder who is (or was) a service provider of GB will be separate consideration for, or allocable to, such Equityholder's stock of GB surrendered in the GB Merger; none of the shares of Company stock received in the GB Merger by any Equityholder who is (or was) a service provider of GB will be separate consideration for any past or future services; and the compensation paid to any such Equityholder will be for services actually rendered and will be commensurate with amounts paid to third parties bargaining at arm's-length for similar services; and
  - 33. The undersigned officer of GB is authorized to make all of the representations set forth herein on behalf of itself and GB.
  - B. Reliance by Law Firms in Rendering Opinions: Limitation on the Opinions.
- 1. The undersigned recognizes and agrees that (i) the opinions of WilmerHale LLP and Latham & Watkins LLP (the "Law Firms") (each, an "Opinion") will be based on the representations set forth herein and will assume that all of the representations and statements set forth herein are true without regard to any qualification as to knowledge or belief and will be

K-4

based on the statements contained in the Agreement and the documents related thereto, (ii) each Opinion will be subject to certain limitations and qualifications, including that it may not be relied upon if any such representations are not accurate in all material respects without regard to any qualification as to knowledge or belief, and (iii) notwithstanding any provisions of the Agreement to the contrary, the representations set forth in this letter will survive without limitation.

- 2. The undersigned recognizes and agrees that no Opinion will address any tax consequences of the Mergers or any action taken in connection therewith except as expressly set forth in such Opinion.
- 3. The undersigned hereby undertakes to inform the Law Firms immediately in writing should any of the statements or representations set forth in this letter become untrue, incorrect or incomplete in any respect on or prior to the Effective Time.

K-5

In witness whereof, the undersigned executed this certificate as of the date and year first above-written.

Genband Inc.

By:

Name:

Title:

### CERTIFICATE OF GENBAND II, INC.

, 2017

Re: Agreement and Plan of Merger dated May 23, 2017 (the "Agreement") by and among Sonus Networks, Inc., Solstice Sapphire Investments, Inc., Solstice Sapphire Investments LLC, Green Sapphire LLC, Genband Holdings Company, Genband Inc., and Genband II, Inc.

This certificate is supplied to you in connection with your rendering of an opinion regarding certain U.S. federal income tax consequences of certain of the Mergers to be undertaken pursuant to the Agreement. Unless otherwise indicated, capitalized terms not defined herein have the meanings set forth in the Agreement.

- A. <u>Representations</u>. The undersigned hereby certifies and represents that the following facts are true, correct and complete in all respects and will be true, correct and complete in all respects at the Effective Time:
- 1. GB II and, to the best knowledge of the undersigned, the direct stockholders in GB II (such stockholders, the "Equityholders") are participating in the GB II Merger for good and valid business reasons and not for tax purposes. The terms of the Agreement and all other agreements entered into in connection with the Agreement are the products of arm's length negotiation;
- 2. The Mergers will occur on the same date at approximately the same time as part of a single plan pursuant to the terms of the Agreement, and the Agreement defines the rights of each of the parties to the transactions contemplated by the Agreement;
- 3. Each Equityholder will dispose of its stock in GB II pursuant to the GB II Merger solely in exchange either (i) for no consideration or (ii) for Company stock (and/or cash received in lieu of fractional shares of Company stock). Any cash received in lieu of fractional shares of Company stock is the only consideration other than Company stock that will be received by Equityholders in connection with the disposition of their stock in GB II via the GB II Merger. Any Company stock to be received by the Equityholders will be issued by the Company;
- 4. No Company stock or securities will be issued in the GB II Merger for services rendered to or for the benefit of the Company or its Subsidiaries in connection with the GB II Merger;

L-1

- 5. No Company stock or securities will be issued in the GB II Merger for indebtedness of the Company or for interest on indebtedness of the Company;
  - 6. The GB II Merger is not the result of the solicitation by a promoter, broker, or investment house;
- 7. The Equityholders will not retain any rights in the GB II stock to be disposed of pursuant to the GB II Merger at or after the Effective Time;
- 8. No Company stock received in the GB II Merger will be received in exchange for any licenses, leases or similar rights between the Equityholders and the Company;
- 9. To the best knowledge of the undersigned, no property transferred to the Company in connection with the Mergers will be leased back to an Equityholder, another shareholder of the Company or a related party;
- 10. The GB II stock disposed of in the GB II Merger was not acquired by the Equityholders in connection with the liquidation of another corporation, and the Glass shares held by GB II were not acquired by GB II in connection with the liquidation of another corporation;
  - 11. No liabilities of any Equityholders will be assumed by the Company in the GB II Merger;
  - 12. To the best knowledge of the undersigned, the GB II stock to be disposed of in the GB II Merger is not subject to any liabilities;
- 13. To the best knowledge of the undersigned, at the time of the Mergers, there will be no indebtedness between the Company and the Equityholders, and there will be no indebtedness created in favor of the Equityholders as a result of the GB II Merger;
- 14. To the best knowledge of the undersigned, following the Mergers, Solstice will continue its "historic business" within the meaning of Treasury Regulation Section 1.368-1(d) or use a significant portion of its historic business assets in a business;
- 15. To the best knowledge of the undersigned, there is no plan or intention on the part of the Company to redeem or otherwise reacquire any stock or indebtedness to be issued by the Company in the Mergers. To the best knowledge of the undersigned, no person related to the Company, as defined in Treas. Reg. Section 1.368-1(e)(3), and no person acting as intermediary for the Company or such related person (i) has acquired any stock of GB II or (ii) has a plan or intention to acquire any of the Company stock to be issued in the Mergers;
- 16. To the best knowledge of the undersigned, the Equityholders do not have any plan or intention to sell, exchange or otherwise dispose of any of the Company stock issued in the GB II Merger;
- 17. To the best knowledge of the undersigned, taking into account any issuance of additional shares of Company stock, any issuance of Company stock for services (including the

Company stock that will be issued to holders of Class E Shares (as defined in the Amended and Restated Memorandum and Articles of Association of Glass) of Glass and the Company stock that will be issued in exchange for any Solstice Restricted Shares and any Solstice stock received in settlement of Solstice RSUs that vest by reason of the Solstice Merger, which Company stock, in the aggregate, shall not exceed 1.5% of the value of all Company stock that is issued and outstanding after the Mergers (such Company common stock issued to such holders, the "Compensatory Company Stock")), the exercise of any Company stock rights, warrants or subscriptions, a public offering of Company stock, and the sale, exchange, transfer by gift, or other disposition of any of the stock of Company to be received in the Mergers, the direct shareholders of Glass immediately prior to the Mergers (excluding for this purpose (i) any shareholders of Glass who receive Compensatory Company Stock and (ii) the Glass Corporate Members and their stockholders), together with the direct stockholders of Solstice immediately prior to the Mergers (excluding for this purpose any stockholders of Solstice who receive Compensatory Company Stock in their capacity as holders of the Solstice stock exchanged for such Compensatory Company Stock) (together, such holders, the "Control Transferors"), will be in "control" (within the meaning of Sections 351(a) and 368(c) of the Code) of the Company immediately after the Mergers;

- 18. To the best knowledge of the undersigned, there exist no rights to acquire Company stock or to vote (or restrict or otherwise control the vote of) shares of Company stock which, if exercised, would affect the Control Transferors' acquisition and retention of "control" (within the meaning of Sections 351(a) and 368(c) of the Code) of the Company. To the best knowledge of the undersigned, there is no plan or intention for the Company to issue additional shares of Company stock after the Mergers, or take any other action, that would result in the Control Transferors not being in "control" of the Company after the Mergers;
- 19. The Equityholders will receive Company stock or other property approximately equal to the fair market value of the GB II stock disposed of in the GB II Merger;
- 20. No Company stock will be placed in escrow or will be issued after the Closing Date pursuant to a contingent stock arrangement, in each case, in connection with the Mergers;
- 21. The principal purpose of the formation and use of the Company is not the avoidance of U.S. federal income tax or to obtain a U.S. federal income tax benefit that would have otherwise been unavailable;
- 22. To the best knowledge of the undersigned, the Company will remain in existence and retain and use the property transferred to it in a trade or business (or such property will reflect the ownership of subsidiaries engaged in a trade or business);
- 23. To the best knowledge of the undersigned, , the Company has no plan or intention (i) to liquidate Solstice, (ii) to merge Glass or Solstice with or into another entity, (iii) to sell or otherwise dispose of the shares of Glass or the stock of Solstice acquired in the Mergers, (iv) to sell or otherwise dispose of the assets of Glass or Solstice, or (v) to cause Glass or Solstice to sell or otherwise dispose of any of its assets, except, in the case of (iv) and (v), dispositions of assets made in the ordinary course of business and, in the case of (iii), (iv) and

L-3

- (v), transfers or successive transfers of stock or assets to one or more corporations controlled (within the meaning of Section 368(c) of the Code) in each case by the transferor corporation;
  - 24. Each of the parties to the Mergers will pay its own expenses, if any, incurred in connection with the proposed transactions;
- 25. To the best knowledge of the undersigned, at the Effective Time and immediately thereafter, the Company will not be an investment company within the meaning of Section 351(e)(1) of the Code and Treasury Regulations Section 1.351-1(c)(1)(ii);
- 26. Neither GB II nor, to the best knowledge of the undersigned, any Equityholder is under the jurisdiction of a court in a title 11 or similar case (within the meaning of Section 368(a)(3)(A) of the Code);
- 27. At the Effective Time and immediately thereafter, to the best knowledge of the undersigned, the Company will not be a "personal service corporation" within the meaning of Section 269A of the Code;
- 28. The GB II Merger will be consummated in compliance with the terms of the Agreement and none of the terms and conditions therein have been or will have been waived or modified. The Agreement and the other documents described in Section 9.4 of the Agreement constitute the entire agreement among the parties thereto with respect to the matters governed thereby;
- 29. To the best knowledge of the undersigned, the Equityholders and the Company will comply with the reporting and record-keeping requirements set forth in Treasury Regulations Section 1.351-3;
- 30. To the best knowledge of the undersigned, the Company has no plan or intention to cause Solstice to issue additional shares of its stock that would result in the Company losing control of Solstice within the meaning of Section 368(c) of the Code following the Solstice Merger;
- 31. The facts which relate to the Mergers which are described in the Agreement and the Joint Proxy Statement/Prospectus are true, correct and complete in all material respects;
- 32. None of the compensation received from the Company or its subsidiaries in connection with or after the Mergers by any Equityholder who is (or was) a service provider of GB II will be separate consideration for, or allocable to, such Equityholder's stock of GB II surrendered in the GB II Merger; none of the shares of Company stock received in the GB II Merger by any Equityholder who is (or was) a service provider of GB II will be separate consideration for any past or future services; and the compensation paid to any such Equityholder will be for services actually rendered and will be commensurate with amounts paid to third parties bargaining at arm's-length for similar services; and
  - 33. The undersigned officer of GB II is authorized to make all of the representations set forth herein on behalf of itself and GB II.

<ol> <li>Reliance by Law Firms in Renderi</li> </ol>	g Opinions: Limitation on the Opinions
--	--

1. The undersigned recognizes and agrees that (i) the opinions of WilmerHale LLP and Latham & Watkins LLP (the "Law Firms") (each, an "Opinion") will be based on the representations set forth herein and will assume that all of the representations and statements set forth herein are true without regard to any qualification as to knowledge or belief and will be based on the statements contained in the Agreement and the documents related thereto, (ii) each Opinion will be subject to certain limitations and qualifications, including that it may not be relied upon if any such representations are not accurate in all material respects without regard to any qualification as to knowledge or belief, and (iii) notwithstanding any provisions of the Agreement to the contrary, the representations set forth in this letter will survive without limitation.

- 2. The undersigned recognizes and agrees that no Opinion will address any tax consequences of the Mergers or any action taken in connection therewith except as expressly set forth in such Opinion.
- 3. The undersigned hereby undertakes to inform the Law Firms immediately in writing should any of the statements or representations set forth in this letter become untrue, incorrect or incomplete in any respect on or prior to the Effective Time.

	L-5
In witness whereof, the undersigned executed this certificat	re as of the date and year first above-written.
	Genband II, Inc.
	Ву:
	Name:
	Title:

### Amendment to Second Amended and Restated By-Laws

The Second Amended and Restated By-Laws of Sonus Networks, Inc. are hereby amended by adding thereto the following provision as a new Section 6.5 thereof:

"6.5 Exclusive Forum. Unless the corporation consents in writing to the selection of an alternative forum, the Court of Chancery of the State of Delaware shall, to the fullest extent permitted by law, be the sole and exclusive forum for: (a) any derivative action or proceeding brought on behalf of the corporation, (b) any action asserting a claim of breach of a fiduciary duty owed by any current or former director, officer, other employee, agent or stockholder of the corporation to the corporation or the corporation's stockholders, including, without limitation, a claim alleging the aiding and abetting of such a breach of fiduciary duty, (c) any action asserting a claim arising pursuant to any provision of the General Corporation Law of the State of Delaware, the Company's Certificate of or these by-laws (as each may be amended from time to time) or as to which the General Corporation Law of the State of Delaware confers jurisdiction on the Court of Chancery of the State of Delaware, or (d) any action asserting a claim governed by the internal affairs doctrine or other "internal corporate claim" as that term is defined in Section 115 of the General Corporation Law of the State of Delaware. Any person or entity purchasing or otherwise acquiring or holding any interest in shares of capital stock of the corporation shall be deemed to have notice of and consented to the provisions of this Section 6.5."

Adopted by the Board of Directors of Sonus Networks, Inc. on May 22, 2017.

	Sonus Networks, Inc and		Sonus Networks, Inc.
May 23, 2017	GENBAND Merger Call	SONS	
Date ▲	Event Type 🛦	Ticker 🛦	Company 🛦

### PARTICIPANTS

### **Corporate Participants**

Susan M. Villare — Chief Financial Officer (Interim), Sonus Networks, Inc.

Raymond P. Dolan — President, Chief Executive Officer & Director, Sonus Networks, Inc.

David A. Walsh — Chairman, President & Chief Executive Officer, GENBAND, Inc.

### **Other Participants**

Dmitry G. Netis — Analyst, William Blair & Co. LLC
Greg Mesniaeff — Analyst, Drexel Hamilton LLC
Michael A. Kerlan — Analyst, Wells Fargo Securities LLC
Fahad Najam — Analyst, Cowen & Co. LLC

### **MANAGEMENT DISCUSSION SECTION**

Operator: Ladies and gentlemen, thank you for standing by. Welcome to Sonus and GENBAND Conference Call. During the presentation, all participants will be in a listen-only mode. Afterwards, we will conduct a question-and-answer session. [Operator Instructions] As a reminder, this conference is being recorded, Tuesday, May 23, 2017.

I would now like to turn the conference over to Susan Villare, Interim Chief Financial Officer of Sonus. Please go ahead.

### Susan M. Villare, Chief Financial Officer (Interim), Sonus Networks, Inc.

Good morning, everyone. Thank you for joining us on such short notice to discuss today's announcement that Sonus and GENBAND have signed a definitive merger agreement.

On the call today, you'll hear from Ray Dolan, Sonus' President and Chief Executive Officer; and David Walsh, GENBAND Chief Executive Officer and Chairman. A press release and investor presentation have been posted to our IR website and filed with the SEC.

During our prepared remarks, we will be referring to the investor presentation with supporting information. Please take a moment to locate these documents on our IR website. A recording of this call and the transcript will be available on our IR website shortly after this call.

As shown on slide two, please note that during this call, we will be making a number of forward-looking statements regarding the proposed combination. Such forward-looking statements are based on the current beliefs and expectations of Sonus as well as of GENBAND and are subject to business, economic, and competitive uncertainties and contingencies, many of which are difficult to predict.

Actual events or financial results may differ materially from the forward-looking statements described in this call due to a number of risks and uncertainties detailed in the documents filed or furnished by Sonus with the SEC. A discussion of these and other factors that may affect our future results is contained in our most recent Form 10-Q filed with the SEC on April 27, 2017 and in our press release, both of which are available on our IR website. While we may elect to update or revise forward-looking statements at some point, we specifically disclaim any obligation to do so.

1

Sonus Networks, Inc.		Sonus Networks, Inc and	
	SONS	GENBAND Merger Call	May 23, 2017
Company ▲	Ticker 🛦	Event Type 🛦	Date 🛦

During our call, we will be referring to certain GAAP and non-GAAP financial measures. A reconciliation of the non-GAAP to comparable GAAP financial measures is included in our press release issued today.

With that, let me turn it over to the President and Chief Executive Officer of Sonus, Mr. Ray Dolan.

# Raymond P. Dolan, President, Chief Executive Officer & Director, Sonus Networks, Inc.

Thank you, Susan, and good morning to everyone on today's call. I appreciate that you've joined us this morning on such short notice to hear more about the exciting combination of Sonus and GENBAND. I'm also very pleased to have David Walsh, CEO and Chairman of GENBAND here with me on the call today.

Our companies have known each other for a long time, and we've always had mutual respect for each other's businesses and teams. David and I have kept in touch because we've recognized how powerful our growth platform could be as a combined company.

Today, the timing is just right. At Sonus, we successfully transformed to a software-based company and we're at an important inflection point in our business where we believe that a combination with GENBAND will accelerate our ability to drive growth and continue our investment in the new cloud architecture.

Today, the world's largest service providers and enterprises are in the midst of accelerating the modernization and transformation of their networks, and we believe the combined company will be better positioned to capture more of these opportunities. By coming together, we are creating a strong global leader in

real-time communication software and cloud solutions, and ultimately, creating meaningful value for our shareholders.

For those listeners who may not be familiar, GENBAND is a leading provider of carrier and enterprise network transformation in real-time communication solutions, which is currently privately-owned. David will tell you more about their business in a few moments. We're both incredibly enthusiastic about the potential of our company's combination and the compelling strategic and financial rational that it presents.

When you look at our businesses, you'll see that our product portfolios fit together very well. Together, we'll also have greater scale, a broader geographic reach and a deeper sales footprint, creating a strong combination to serve our diversified global customer base.

Importantly, we believe we'll have significantly stronger financial profile, which will be supported by the meaningful cost synergies from this transaction. I'll go into a bit more detail on this later, but we expect to achieve between \$40 million and \$50 million of anticipated annual cost synergies by the end of 2018. So the key takeaway is that we expect this combination to bring benefits to our customers, employees and investors and will provide us with greater scale and resources to invest in and achieve our growth initiatives at a faster pace than either company could do on its own.

Now, let's move to slide four. All of these details are outlined in our press release, so I'll just provide a quick overview. Under the terms of the merger agreement, Sonus and GENBAND shareholders are expected to own approximately 50% of the combined company on a fully diluted basis after close. GENBAND's equity-holders will be issued approximately 50 million shares of the combined company along with a \$22.5 million unsecured note. Based on the closing price of Sonus common stock yesterday and estimated net cash of the time of closing, the transaction values the combined company at an enterprise value of approximately \$745 million.

2

Sonus Networks, Inc.		Sonus Networks, Inc and	
	SONS	GENBAND Merger Call	May 23, 2017
Company ▲	Ticker 🛦	Event Type 🛦	Date 🛦

Following the completion of the transaction, which we expect will occur in the second half of 2017 subject to its customary closing conditions, I will become the CEO of the combined company. David will oversee the Kandy business, which will be a division of the combined company and is currently GENBAND's cloud communication platform as a service, or CPaaS, for global service providers and enterprises. It's a very exciting growth initiative that we'll discuss later in the call.

I'm very pleased that David will also play a key role in assisting in the integration of our two companies. The board of directors of the combined company will have five representatives designated by GENBAND and four representatives designated by Sonus.

I will be joined on the board by Dick Lynch, the current Chairman of Sonus and former CTO of Verizon, who will become the Chairman of the combined company board. I think we'll have an exceptional leadership team and board who are well-positioned to oversee the combined company's strategic plans and drive long-term shareholder value.

Turning to slide five, we want to provide a side-by-side comparison of our companies in the few key areas; and to keep this simple, we've based this comparison on 2016 numbers. At a high level, you can see that while GENBAND is a larger company than Sonus, our businesses have similarities in terms of size of our customer and employee basis, geographies, and sales forces as well as revenue breakdown.

On our side, as a customer-focused organization, I'm proud to say we have nearly 700 customers around the world who rely on our products and services to enable and secure their real-time communications. It's a testament to approximately 1,400 employees and contractors who work in 22 different countries and territories that we're able to ensure that each of these customers have the tools they need to ensure their business communications function seamlessly.

Given today's market dynamics and the continuing shifts that service providers and enterprises are making, we believe the combined company will have a strong customer value prop and we'll be well-positioned to deliver superior comprehensive solution to service providers and enterprises in the increasingly important areas of security, mobility, cloud, network transformation, and evolution to CPaaS.

Before I introduce David Walsh, Chairman and CEO of GENBAND, I'd just like to express the excitement that I have to be working with him and the entire GENBAND team. David is a proven executive with a distinguished career in technology and telecom as well as the strong financial background and a great track record of overseeing important transactions and combinations.

So now I'd like to turn it over to David.

#### David A. Walsh, Chairman, President & Chief Executive Officer, GENBAND, Inc.

Thank you, Ray. It's great being here with you and everyone on this call today. I do share your enthusiasm for this merger. The combination benefits are very compelling.

As you can see, GENBAND has over 850 customers worldwide and nearly 2,000 employees in 24 countries. We have a good mix of revenues, customers around the world, as well as a good split between products and services.

About GENBAND. GENBAND focuses in two product areas: network transformation products and CPaaS products that we brand as Kandy. The network transformation products have more than 850 customers, many of the largest carriers, enterprises and governmental agencies in the world.

3

GENBAND specializes in migrating large-scale customer networks from legacy to IP. GENBAND has migrated 42 million lines to IP and has transformed over 2,200 central offices. Our broad product offering allows us to be a turnkey solution provider, while our large installed base or Nortel switches provide a solid opportunity for network transformation projects.

Three years ago, we created our second focused area called Kandy, which is our CPaaS platform focused on unified communications, embedded communications and over-the-top. These products are aimed at helping our carrier customers protect and grow their incumbent position real-time communications.

The Kandy platform is built on top of GENBAND's portfolio of science and is accessible via our cloud or can be incorporated into a carrier's network. The market is validating our strategy. And while it's early days, we're excited about the traction we're getting in the marketplace.

Now, I'll turn it back over to Ray to talk about how our offerings fit together.

#### Raymond P. Dolan, President, Chief Executive Officer & Director, Sonus Networks, Inc.

Thanks, David. Turning to slide seven. As I said earlier, a critical aspect of this transaction is about being able to offer customers a more comprehensive portfolio of products and services as they transition to the cloud in an all-IP world. As you can see here, the core strengths that each company is bringing to the table will enable us to do just that.

Both of our companies have pursued strategies to enable network transformation for our customers. At Sonus, we've been concentrating on providing next-generation cloud-based SIP and 4G/VoLTE solutions by providing network layer control, security, interworking, policy, and session management for the delivery of real-time communications.

Our complete portfolio of hardware-based and virtualized session border controllers, diameter signaling controllers, policy/routing servers, network intelligence applications, media and signaling gateways, and network analytics tools have helped us become one of the industry's market-leading supplier.

GENBAND strategy is centered on seamlessly supporting its customer base of carriers and enterprises to modernize their network to IP with a broad product portfolio that includes soft switches, media gateways, application servers, and leveraging that IP infrastructure with a rapid service creation capability.

In addition, I briefly mentioned Kandy before, and David will talk about it in greater detail in a moment, but this is a really exciting CPaaS product and one we think has promising growth opportunities. Together, we also believe we can drive benefits from our mutual investment in virtualization.

The complementary maintenance streams from across both businesses and our common strength in professional services. We're confident that our complementary strengths will be a true differentiator over our competitors and will prove to be a great value to carriers and enterprises who are transitioning to the cloud and an all-IP world.

Turning to slide eight. I want to touch on the three primary reasons that this is a financially compelling combination for Sonus shareholders. First, we expect the transaction to be substantially accretive to our non-GAAP EPS in 2018. Second, while Sonus' net cash position will be temporarily lower in closing than it is today, we expect to drive strong cash flow from operations from the first year on. And third, the combined company will be a financially stronger company right out of the

4

Sonus Networks, Inc.	SONS	Sonus Networks, Inc and GENBAND Merger Call	
Company ▲	Ticker 🛦	Event Type 🛦	Date ▲

gate, generating at least \$100 million of expected EBITDA once we realize the full impact of the annualized synergies by the end of 2018.

We're projecting fiscal 2020 EBITDA to be approximately \$140 million, so it's a very nice ramp-up there. On the left-hand side of the page, we've detailed combined 2016 revenues by both geography and revenue category. So you can get a sense for the diversified revenue streams that the combined company will have.

Slide nine gives you more detailed snapshot of the combined company; again, using 2016 numbers and not including synergies, or any impact of purchase accounting. As you can see, we're creating a significantly bigger and more diversified company with higher revenues, gross profit, earnings and cash flow from operations. Additionally, we expect Sonus and GENBAND's complementary growth strategies in cloud and virtualized network architectures will also have a positive impact on gross margin over time.

Moving to slide 10, we expect to achieve \$40 million to \$50 million of annualized synergies from four key areas: COGS, sales and marketing, R&D, and G&A. While we aren't providing a specific breakdown by area today, it's important to understand that we expect that cost synergies will be driven from all four buckets.

As I've mentioned, the cost synergies achieved by this transaction are compelling and will allow us to provide even stronger financial performance, while at the same time, increasing our technical and marketing investment in our key growth initiatives.

If you turn to slide 11, we'll quickly touch on what those are from both companies and I'll ask David to tell you more about Kandy first.

#### David A. Walsh, Chairman, President & Chief Executive Officer, GENBAND, Inc.

All right. Thanks, Ray. GENBAND's Kandy initiative enables service providers and other market participants to invent voice, video, messaging, chat presence, and more into any business process or application, enhancing outcomes for business, consumers, or Internet of Thing deployments. And Kandy CPaaS allows carriers to use simple APIs and SDKs and micro-apps, which we call Kandy Wrappers, to rapidly create new revenue streams for customers and partners.

The Kandy platform was built on top of GENBAND's core carrier-grade technology, which was built for scale. Our strategy is perfectly aligned with our carrier customers who need access to a platform, but want to use their network to better service their customers' needs for real-time communications.

Back to you, Ray.

#### Raymond P. Dolan, President, Chief Executive Officer & Director, Sonus Networks, Inc.

On the Sonus side, our cloud security initiative unifies security across all IP flows and we're offering new security analytics and control platforms to real-time flows and digital services. It allows the unification of real-time and IP data security and provides end-to-end visibility and analytics enabling network-wide flow control, threat detection and mitigation. They can be used as a direct sell or a managed security offering. We think that taken together, these growth initiatives are really compelling and aimed to create a path for customers and partners to quickly and securely launch new real-time communications offerings.

5

Sonus Networks, Inc.		Sonus Networks, Inc and	
	SONS	GENBAND Merger Call	May 23, 2017
Company ▲	Ticker 🛦	Event Type 🛦	Date 🛦

So, in conclusion, we believe this transaction is a win-win for all our stakeholders. Together, Sonus and GENBAND are set to generate increased cash flow driven by increased scale and market reach across products, customers and geographies. The combined company will be better positioned to create value for shareholders as a market leader in software-based cloud communications in the future.

That concludes our prepared remarks. I'd now like to open the line for questions. Benjamin?

6

Sonus Networks, Inc.		Sonus Networks, Inc and	
,	SONS	GENBAND Merger Call	
Company ▲	Ticker 🛦	Event Type 🛦	Date 🛦

#### **— QUESTION AND ANSWER SECTION**

Operator: Thank you. [Operator Instructions] Our first question comes from the line of Dmitry Netis with William Blair. Please proceed.

- **Q Dmitry Netis William Blair & Co. LLC>:** Thank you very much. Congratulations to Sonus-GENBAND team for making this transaction. It sure sounds quite synergistic. So a couple of questions. What are your revenue growth expectations from David, on the GENBAND side of things, we know the Sonus side, but give us a little bit of a preview what the growth may look like. And if you could then on a combined basis talk about some revenue, the synergies potential here given there are couple of areas where you obviously have similar products. So if you could comment on what you expect to do there maybe early days, but that would be helpful, just to get an idea.
- < A Dave Walsh GENBAND, Inc.>: Sure, Dmitry. Thank you. On the revenue growth side, this industry, as we all know, isn't growing, or in some segments, there's modest growth. So our business has been flat over the last couple of years. And we decided a couple of years ago that the way to get around that is we have to find some growth areas.

So we started making some moves in the CPaaS area, which is a very exciting and attractive growth market. We looked at wireless, which is also growing. That's why, we invested heavily in media and transcoding. And we also have a large installed base of Nortel products in the government sector. So we went out and got JITC certification, and that is proving to be an attractive and fertile ground.

So we're starting to position ourselves nicely because, as you know, in the fixed network transformation space, that's not growing. But these are some areas that we feel, heading out to 2018, we have the potential to grow.

On the second question about revenue synergies, Ray touched on it briefly about the synergy of the portfolio, which is really interesting. It's something we've known for a long time. If you look at the SBC marketplace, if you look at the strength that GENBAND brings in media handling and things like transcoding, coupled with the feature capability and the signaling capability that Sonus has, you can picture a very attractive product moving forward. So we think that's an area of synergy.

Another one is looking at the strength of our Class 5 capability in the end office with Sonus' Class 4 capability. We think that pairs up really nicely. So this will take time, as you know. But we think that these new areas that we've invested in have the potential to position us versus some growth in the future. And, Ray, I don't know if you have anything to add to that?

- <a --- Sonus Networks, Inc.>: [indiscernible] (20:05)
- <**Q Dmitry Netis William Blair & Co. LLC>:** If I could just follow up. I think, Dave, I've heard you say there's 42 million lines that you've already migrated to IPs. This presents us a very interesting sort of transformational opportunity. So as far as this opportunity goes, how many lines do you think are there globally that are up for grabs? And what, as a combined entity, think you or you can expect to get either as a number of lines or market share, however you want to slice it or dice it, but how are you thinking about this transformational opportunity?
- < A Dave Walsh GENBAND, Inc.>: I would say here's how we'd frame it out. If you I would say, on the Class 4 on the trunking side of transformation, I'd say, we're probably 65% through that. And you got to look back, I would say Sonus started that in like 1997, so we're here 20 years later,

	Sonus Networks, Inc and		Sonus Networks, Inc.
May 23, 2017	GENBAND Merger Call	SONS	
Date 🛦	Event Type 🛦	Ticker 🛦	Company ▲

So we're still like right in the need of this opportunity. And it's going to be around for a long time. But we can probably in a one-on-one, Dmitry, give you a more market information to better help you with just like sort of where we are there.

- <**Q Dmitry Netis William Blair & Co. LLC>:** All right. That's fine. And then, maybe if I could ask Dave on Kandy side of things that sounds like an exciting kind of new emerging area. What are the plans for Kandy as a combined are you planning to sort of is that becomes kind of a growth driver, is that a standalone-type operation, what are the plan on that business unit and how are you thinking about that one as well?
- <a Dave Walsh GENBAND, Inc.>: Well, what we've been trying to do is find growth in a market that is struggling to grow. And we wanted to come up with a strategy that leverage the assets that GENBAND had. And if you look at the science that we own, in real-time communications, and the sales footprint that we have, creating a CPaaS platform was a natural. And we all know the market very active for these types of services today, and the carriers are desperate to find growth.

So here's an opportunity to go back to our existing customers, not just GENBAND customers but Sonus customers, and show them how they can bolt-on the Kandy platform and immediately enter a new growing market for things like OTT, embedded communications, and UC. So we're still early days. But we do have deployments happening around the world. We've got significant traction with real serious carriers. So this is a business that we are really hopeful is going to be a driving force in value going forward.

- < Q Dmitry Netis William Blair & Co. LLC>: Can you disclose the revenue in that category? Or is that too early to talk about it?
- < A Ray Dolan Sonus Networks, Inc.>: Yeah. It's too early. It's too early, Dmitry. We're not going to break that out until we get a little more traction.
- < Q Dmitry Netis William Blair & Co. LLC>: All right. Very good. Thank you for that and congrats on the deal.
- < A Ray Dolan Sonus Networks, Inc.>: Thanks.
- < A Dave Walsh GENBAND, Inc.>: Thanks.

Operator: [Operator Instructions] Our next question comes from the line of Greg Mesniaeff with Drexel. Please proceed.

- **Q Greg Mesniaeff Drexel Hamilton LLC>:** Yes. Thanks. Good morning. I have a question on the accretion targets that you laid out for 2018. Can you give us a little bit more granularity how you plan to get there given the significant disparity in the margins of the two companies?
- < A Ray Dolan Sonus Networks, Inc.>: Yeah. Greg, this is Ray. Based on the difference in the we'll end up with a blended margin between the two companies, and then, we'll just continue to drive margin improvement together as a combined company across a much larger revenue base.

I don't think that's going to in any way burden our opportunity to deliver an accretive outcome in 2018. Our accretive outcome in 2018 is simply this: first, we'll have a substantially larger EBITDA combined. As we said, we're going to target on an annualized synergy run rate about \$100 million of EBITDA, and that will be substantially larger than our current EBITDA at Sonus, right?

8

Sonus Networks, Inc.		Sonus Networks, Inc and	
	SONS	GENBAND Merger Call	May 23, 2017
Company ▲	Ticker 🛦	Event Type 🛦	Date 🛦

So our 50% of that \$100 million will be by itself accretive. And what we need to do is really at that point just put the two companies together and drive \$40 million to \$50 million of synergies, which we think is totally achievable in 2018. Okay?

- < Q Greg Mesniaeff Drexel Hamilton LLC>: So basically, you're saying scale will base will lift you guys into a more accretive plane?
- <A Ray Dolan Sonus Networks, Inc.>: That's correct. Right out of the box, the scale will get us accretive. Some level of balance sheet was used to get that, and then, we'll restore that balance sheet some time by the end of 2018 to 2019 as we generate positive cash flow. But the key message is that the synergies of \$40 million to \$50 million, which is actually a relatively small portion of our combined cost structure, drives substantially greater EBITDA in aggregate as well as as a percent of revenue.
- < Q Greg Mesniaeff Drexel Hamilton LLC>: Okay. Thank you.
- < A Ray Dolan Sonus Networks, Inc.>: Thank you, Greg.

Operator: Our next question comes from the line of Jess Lubert with Wells Fargo. Please proceed.

< — **Mike Kerlan** — **Wells Fargo Securities LLC>:** Hey, guys. This is Mike Kerlan on for Jess. Congratulations on the deal. Can you just talk a little bit about where the two of you play on the session border controller market, where there might be overlap and where there might be opportunity to leverage your

combined strength to capture additional share?

< A — Ray Dolan — Sonus Networks, Inc.>: Sure. I'll talk a little bit about it. And then, David, I'd welcome you to talk about it from your perspective as well with GENBAND. Thanks, Mike. Yeah. We play in all spaces, search, provider and enterprise, core and edge and large scale and scale down. And we have fully virtualized. And we're able to announce the selection in Verizon's Virtual Network Service as their SBC as a service, which we were able to do in the first quarter, and we're in the process of being selected by a number of other service providers around the world in their edge client.

So it's a very pervasive portfolio. I'll let David talk about GENBAND's portfolio, but I think the opportunity to come together, cross-selling against our existing customer bases and to leverage a lot of the effort that they put into the development there and the team, frankly, that's quite talented that developed the service — the SBC on both sides will give us both the best talent and the best product portfolio across a very diverse customer base.

But, David, I'd welcome your comments about your SBC.

< A — Dave Walsh — GENBAND, Inc.>: Yeah. I think the fit here works really well, and it's going to be well-received from our customers. We at GENBAND didn't have the R&D dollars to take the interim step, which is virtualization. We were forced to go to the end state called NFE.

And so Sonus has been able to invest in both of those steps. So together, we are going to have continuity, which I'm really excited about rather than what we've been doing today is selling a future end state, which is we have a great future end state, but the interim step to get there, we had to move past. We just didn't have the R&D.

The second piece that's attractive on the enterprise side, we went about getting JITC certified for large government projects, which we now are winning for network transformation. And we would have to use third-party providers as related to enterprise SBCs. What a perfect fit for our combined company going forward to provide the total solution. So I think those are two areas that are going to be really helpful to us going forward. [indiscernible] (29:01).

9

Sonus Networks, Inc.		Sonus Networks, Inc and	
	SONS	GENBAND Merger Call	May 23, 2017
Company ▲	Ticker 🛦	Event Type 🛦	Date 🛦

< Q — Mike Kerlan — Wells Fargo Securities LLC>: Any rough idea about the — yeah, it's great. Just any idea what the combined share in the specific area of the SBC market where you play might stand out together?

<A — Ray Dolan — Sonus Networks, Inc.>: Where we play together? I think David did a good job explaining just one opportunity that's now clearly in front of us in the federal space and the public sector space where they've done some work to create JITC certification. That's the standard government certification for compliance with their needs. We've had JITC certification across our entire suite of SBCs, hardware, and software. And now, we'll be able to market together obviously once we bring the two companies together and close. We won't be able to do it prior to that. But giving those customers that choice for a more deeply integrated solution will give them a much more powerful product opportunity and give us a broader reach. That same opportunity exists in the enterprise. That same opportunity exists in the service provider.

In the areas where GENBAND is a very deeply trusted supplier to global service providers, we'll be able to layer our technology on top in areas like signaling and SBCs and SD-WAN. And in areas where we've been a trusted supplier, we'll be able to bolt-on Kandy to both the GENBAND product line as well as the Sonus product line and leverage that growth vertical as well.

< Q — Mike Kerlan — Wells Fargo Securities LLC>: Okay. Thanks.

Operator: Our next question comes from the line of Fahad Najam with Cowen. Please proceed with your question.

<**Q** — **Fahad Najam** — **Cowen & Co. LLC>:** Thanks for taking my question. If I did my math right, if I look at the combined revenue of the two entities and to get back to a similar operating margin — pro forma operating margin profile that Sonus has, you need to cut your combined OpEx by approximately \$20 million. So the synergies that you're talking about really incremental synergies is about \$20 million to \$25 million, is that a correct way of looking at this?

And then, on Kandy, with the CPaaS platform, would this essentially make the combined company in a way competitor to some of your customers like Vonage and BroadSoft?

< A — Ray Dolan — Sonus Networks, Inc.>: Okay. Fahad, I'll take the first question. I'll address the second, and then, hand it over to David to talk a little bit more deeply about CPaaS, because he's got much more domain experience than I do on that.

First, I'm not sure how you do your math to get to \$20 million and gets us back to the operating margin, but I will give you that, yes, we are at a \$40 million to \$50 total synergy in 2018. And we think the synergies will continue to build beyond that, but that's what we're guiding to now. So I do believe that incremental, if you're doing your math your way, which I can't put on the slide here, it does give us incremental \$20 million of synergies above that breakeven, which is why this is accretive right out of the box.

We will not compete with our customers. We will, however, offer the industry a comprehensive solution, and we'll offer that to all of our existing customers as well, but I don't see this as necessarily a competing product with anybody that you said.

David, do you have any comments on CPaaS?

< A — Dave Walsh — GENBAND, Inc.>: Yeah. I think the way to look at how our CPaaS offering is positioned in the market is we're the technology that allows the carriers once they connect or bolt-on the Kandy platform, they can get into this new attractive space. Essentially, what we're doing is

Sonus Networks, Inc.		Sonus Networks, Inc and	
	SONS	GENBAND Merger Call	May 23, 2017
Company 🛦	Ticker 🛦	Event Type 🛦	Date 🛦

we're creating Twilios. So every carrier has the ability once they're connected to Kandy to become Twilio.

And the difference being is we don't compete with the carrier, we allow them to use our technology, which is no different than what we do today in our core business, it's just that we enable it in a different form. So this is something that the carriers are very interested because they're watching new entrants come into the market and take their customers away. This is a way for them to protect their installed base and then grow.

As it relates to BroadSoft, BroadSoft is a phenomenal company, by the way. Mike Tessler has done an amazing job with that business. And the sort of business he's built with our carriers, I really admire.

Now, having said that, the carriers are looking to go beyond just pure unified communications. And what a platform allows you to do is not just do unified communications, it allows you do to embedded communications, OTT, and anything else. It's a developer framework. So we think what we're going to do is try to go beyond where BroadSoft is, but recognizing that BroadSoft has done an unbelievable job in the marketplace.

<Q — Fahad Najam — Cowen & Co. LLC>: Got it. Thanks.

< A — Ray Dolan — Sonus Networks, Inc.>: Any follow-ups, Fahad? If not, Benjamin, I believe we have no further questions. Is that correct?

Operator: We have no further questions for the phone lines. At this time, I will now turn the call back over to you.

#### Raymond P. Dolan, President, Chief Executive Officer & Director, Sonus Networks, Inc.

Okay. This is Ray Dolan. Again, thank you, everyone for participating at short notice. We'll continue to get our information out to you through all vehicles possible. Thanks for your support of Sonus in the past, and as we move through a disclosure and hopefully closing later this year, thanks for your support of the potential combined company. Have a great day.

Operator: Ladies and gentlemen, that does conclude today's conference call. We thank you for your participation, and ask you please disconnect your lines.

#### IMPORTANT INFORMATION ABOUT THE TRANSACTION WILL BE FILED WITH THE SEC

In connection with the proposed transaction, Sonus Networks, Inc., (the "Company") will cause Solstice Sapphire Investments, Inc., a wholly-owned subsidiary of the Company formed to act as a holding company in connection with the transaction ("NewCo"), to file with the SEC a Registration Statement on Form S-4 that will include a joint proxy statement of the Company and GENBAND Holdings Company ("GENBAND") and certain of its affiliates (the "GENBAND Parties") and a prospectus of NewCo and the Company and the parties may file with the SEC other relevant documents concerning the proposed transaction. The Company will mail the definitive joint proxy statement/prospectus to the Company stockholders and the GENBAND Party equity holders. THE COMPANY STOCKHOLDERS AND GENBAND PARTY EQUITY HOLDERS ARE URGED TO READ THE REGISTRATION STATEMENT AND JOINT PROXY STATEMENT/PROSPECTUS REGARDING THE PROPOSED TRANSACTION WHEN IT BECOMES AVAILABLE AND ANY OTHER RELEVANT DOCUMENTS FILED WITH THE SEC, AS WELL AS ANY AMENDMENTS

11

0 Y. I.	1	G N. 1 T 1	
Sonus Networks, Inc.	SONS	Sonus Networks, Inc and GENBAND Merger Call	May 23, 2017
Company ▲	Ticker 🛦	Event Type 🛦	Date 🛦

OR SUPPLEMENTS TO THOSE DOCUMENTS, BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION. Investors and security holders may obtain a free copy of the joint proxy statement/prospectus (when available) and other filings containing information about the Company at the SEC's website at www.sec.gov. The joint proxy statement/prospectus (when available) and the other filings may also be obtained free of charge from the Company's Investor Relations website (http://investors.sonusnet.com/) or by requesting them from the Company corporate secretary at Sonus Networks, Inc., 4 Technology Park Drive, Westford, Massachusetts 01886, Attention: Corporate Secretary.

The Company, NewCo, the GENBAND Parties and certain of their respective directors and executive officers, under the SEC's rules, may be deemed to be participants in the solicitation of proxies of the Company's stockholders in connection with the proposed transaction. Information about the directors and executive officers of the Company and their ownership of Company common stock is set forth in the proxy statement for the Company's 2017 annual meeting of stockholders, as filed with the SEC on Schedule 14A on April 28, 2017.

Additional information regarding the interests of those participants and other persons who may be deemed participants in the transaction may be obtained by reading the joint proxy statement/prospectus regarding the proposed transaction when it becomes available. Free copies of this document may be obtained as described in the preceding paragraphs.

#### NO OFFERS OR SOLICITATIONS

This communication shall not constitute an offer to sell or the solicitation of an offer to sell or the solicitation of an offer to buy any securities, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offer of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the Securities Act of 1933, as amended.

#### CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS

This communication contains "forward-looking statements"—that is, statements related to future, not past, events. In this context, forward-looking statements often address expected future business and financial performance and financial condition, and often contain words such as "expect," "anticipate," "intend," "plan," "believe," "see," "see," "will," "would," or "target." Forward-looking statements by their nature address matters that are, to different degrees, uncertain, such as statements about the consummation of the proposed transaction and the expected performance of the combined companies. Uncertainties that could cause actual results to be materially different than those expressed in the Company's and the GENBAND Parties' forward-looking statements include the failure to consummate the proposed transaction or to make or take any filing or other action required to consummate such transaction in a timely matter or at all. These or other uncertainties may cause actual future results to be materially different from those expressed in the Company's and the GENBAND Parties' forward-looking statements. These risks and uncertainties include, but are not limited to, economic, competitive, legal, governmental and technological factors. Accordingly, there is no assurance that the expectations of the Company or any GENBAND Party will be realized. Many factors could cause actual results to differ materially from these forward-looking statements with respect to the proposed transaction, including risks relating to the completion of the proposed transaction on anticipated terms and timing, including obtaining equity holder and regulatory approvals, anticipated ax treatment, unforeseen liabilities, future capital expenditures, revenues, expenses, earnings, synergies, economic performance, indebtedness, financial condition, losses, future prospects, business and management strategies for the management, expansion and growth of the new combined company's operations, the ability of the Company and the GENBAND Parties to

12

Sonus Networks, Inc.		Sonus Networks, Inc and	
	SONS	GENBAND Merger Call	May 23, 2017
Company ▲	Ticker 🛦	Event Type 🛦	Date 🛦

disruptions from the proposed transaction will harm the Company's or the GENBAND Parties' business. While the list of factors presented here is considered representative, no such list should be considered to be a complete statement of all potential risks and uncertainties. Unlisted factors may present significant additional obstacles to the realization of forward looking statements. Consequences of material differences in results as compared with those anticipated in the forward-looking statements could include, among other things, business disruption, operational problems, financial loss, legal liability to third parties and similar risks, any of which could have a material adverse effect on the Company's or the GENBAND Parties' consolidated financial condition, results of operations or liquidity. Neither the Company nor any GENBAND Party assumes any obligation to provide revisions to any forward-looking statements should circumstances change, except as otherwise required by securities and other applicable laws.

#### Sonus Employee note from R. Dolan to be sent with PDF attachment of release

Dear Sonus Teammates,

I am pleased to share the exciting news that this morning we announced that Sonus will combine with GENBAND to create what we believe will be a new leader in real-time communications (RTC), with increased size and market reach across products, customers and geographies. The press release announcing the news is attached.

This combination is about the growth and customer benefits that can be achieved by bringing two complementary companies together. For those who may be unfamiliar with GENBAND, the company is a leading provider of carrier and enterprise network transformation and real-time communications solutions. Together with GENBAND, we will be a financially stronger company with a greater ability to pursue new market opportunities in IP and further the technology leadership of the combined company. The combined company will also have greater resources to support our customers worldwide, with a broader and deeper sales footprint, an increased ability to invest in growth, more efficient and effective R&D, and a comprehensive RTC product offering.

Importantly, we also anticipate there will be enhanced opportunities for employees to advance and grow within the newly formed organization. We truly believe this combination is a win for our customers, partners, shareholders and employees.

The close of the transaction is expected in the second half of this year. Sonus and GENBAND both wholeheartedly embrace cultures that are customer-focused and value excellence, collaboration and innovation. We expect these shared values will facilitate a smooth integration process for everyone.

Today's news is just the first step in the process of the combination. It is still early days and premature to discuss specifics. Both companies will continue to operate independently until the close, so it is business as usual for us. It is critical that we remain focused on providing outstanding customer service and delivering excellent products and services like we always do. If you have any questions, don't hesitate to reach out to your manager.

I will be holding an All Hands meeting from Westford today at 10:30am ET to share more details of the combination and answer your questions. David Walsh, CEO and Chairman of GENBAND, will be joining me so you can learn more about them. I encourage you to tune in, either in person or via videocast, details of which will be sent shortly.

To that end, I would like to thank you all for your continued hard work. We truly would not be in the position we are today if it wasn't for your ongoing dedication to Sonus. I am very excited for the future and the many opportunities that lie ahead for us.

Sincerely,

Rav

#### IMPORTANT INFORMATION ABOUT THE TRANSACTION WILL BE FILED WITH THE SEC

In connection with the proposed transaction, Sonus will cause Solstice Sapphire Investments, Inc., a wholly-owned subsidiary of Sonus formed to act as a holding company in connection with the transaction ("NewCo"), to file with the SEC a Registration Statement on Form S-4 that will include a joint proxy statement of Sonus and GENBAND and certain of its affiliates (the "GENBAND Parties") and a prospectus of NewCo and Sonus and the parties may file with the SEC other relevant documents concerning the proposed transaction. Sonus will mail the joint proxy statement/prospectus to the Sonus stockholders and the GENBAND Party equity holders. SONUS STOCKHOLDERS AND GENBAND PARTY EQUITY HOLDERS ARE URGED TO READ THE REGISTRATION STATEMENT AND JOINT

1

PROXY STATEMENT/PROSPECTUS REGARDING THE PROPOSED TRANSACTION WHEN IT BECOMES AVAILABLE AND ANY OTHER RELEVANT DOCUMENTS FILED WITH THE SEC, AS WELL AS ANY AMENDMENTS OR SUPPLEMENTS TO THOSE DOCUMENTS, BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION. Investors and security holders may obtain a free copy of the joint proxy statement/prospectus (when available) and other filings containing information about Sonus at the SEC's website at www.sec.gov. The joint proxy statement/prospectus (when available) and the other filings may also be obtained free of charge from Sonus' Investor Relations website (http://investors.sonusnet.com/) or by requesting them from Sonus' corporate secretary at Sonus Networks, Inc., 4 Technology Park Drive, Westford, Massachusetts 01886, Attention: Corporate Secretary.

Sonus, NewCo, the GENBAND Parties and certain of their respective directors and executive officers, under the SEC's rules, may be deemed to be participants in the solicitation of proxies of Sonus stockholders in connection with the proposed transaction. Information about the directors and executive officers of Sonus and their ownership of Sonus common stock is set forth in the proxy statement for Sonus' 2017 annual meeting of stockholders, as filed with the SEC on Schedule 14A on April 28, 2017.

Additional information regarding the interests of those participants and other persons who may be deemed participants in the transaction may be obtained by reading the joint proxy statement/prospectus regarding the proposed transaction when it becomes available. Free copies of this document may be obtained as described in the preceding paragraphs.

# NO OFFERS OR SOLICITATIONS

This communication shall not constitute an offer to sell or the solicitation of an offer to sell or the solicitation of an offer to buy any securities, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offer of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the Securities Act of 1933, as amended.

# CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS

This communication contains "forward-looking statements"—that is, statements related to future, not past, events. In this context, forward-looking statements often address expected future business and financial performance and financial condition, and often contain words such as "expect," "anticipate," "intend," "plan," "believe," "seek," "see," "will," "would," or "target." Forward-looking statements by their nature address matters that are, to different degrees, uncertain, such as statements about the consummation of the proposed transaction and the expected performance of the combined companies. Uncertainties that could cause actual results to be materially different than those expressed in Sonus' and the GENBAND Parties' forward-looking statements include the failure to consummate the proposed transaction or to make or take any filing or other action required to consummate such transaction in a timely matter or at all. These or other uncertainties may cause actual future results to be materially different from those expressed in Sonus' and the GENBAND Parties' forward-looking statements. These risks and uncertainties include, but are not limited to, economic, competitive, legal, governmental and technological factors. Accordingly, there is no assurance that the expectations of Sonus or any GENBAND Party will be realized. Many factors could cause actual results to differ materially from these forward-looking statements with respect to the proposed transaction, including risks relating to the completion of the proposed transaction on anticipated terms and timing, including obtaining equity holder and regulatory approvals, anticipated tax treatment, unforeseen liabilities, future capital expenditures, revenues, expenses, earnings, synergies, economic performance, indebtedness, financial condition, losses, future prospects, business and management strategies for the management, expansion and growth of the new combined company's operations, the ability of Sonus and the GENBAND Parties to integrate the business successfully and to achieve anticipated synergies, potential litigation relating to the proposed transaction, and the risk that disruptions from the proposed transaction will harm Sonus' or the GENBAND Parties' business. While the list of factors presented here is considered representative, no such list should be considered to be a complete statement of all potential risks and uncertainties. Unlisted factors may present significant additional obstacles to the realization of forward looking statements. Consequences of

2

material differences in results as compared with those anticipated in the forward-looking statements could include, among other things, business disruption, operational problems, financial loss, legal liability to third parties and similar risks, any of which could have a material adverse effect on Sonus' or the GENBAND Parties' consolidated financial condition, results of operations or liquidity. Neither Sonus nor any GENBAND Party assumes any obligation to provide revisions to any forward-looking statements should circumstances change, except as otherwise required by securities and other applicable laws.

#### Sonus Customer/Partner note from R. Dolan to be sent with PDF attachment of release

Dear Valued Customer/Partner,

I am pleased to share the exciting news that moments ago, we announced that Sonus will combine with GENBAND to create what we believe will be a new leader in real-time communications (RTC), with increased size and market reach across products, customers and geographies. The joint press release we issued this morning is attached.

We are very excited about this combination, which brings together two complementary businesses and world-class teams that will be well positioned to offer even more value to customers and partners when it comes to quickly and securely launching new RTC offerings. Together, we will have greater resources across the globe to support your needs, through a broader and deeper sales footprint, increased ability to invest in growth, more efficient and effective R&D, and smoother migration paths to the virtualized all-IP world.

Importantly, it is business as usual at Sonus. The transaction is expected to be finalized in the second half of this year and until then, Sonus and GENBAND will continue to operate as independent companies. Your point of contact at Sonus will remain the same, and you can be sure that providing outstanding customer service remains our top priority.

We will keep you informed of the promising combination milestones as appropriate. In the meantime, please feel free to get in touch with your normal point of contact at Sonus with any questions. You can also visit www.rtcleaders.com for more information and we will provide updates using this site going forward.

As always, thank you for your continued partnership.

Sincerely,

Ray Dolan President and CEO Sonus Networks

#### IMPORTANT INFORMATION ABOUT THE TRANSACTION WILL BE FILED WITH THE SEC

In connection with the proposed transaction, Sonus will cause Solstice Sapphire Investments, Inc., a wholly-owned subsidiary of Sonus formed to act as a holding company in connection with the transaction ("NewCo"), to file with the SEC a Registration Statement on Form S-4 that will include a joint proxy statement of Sonus and GENBAND and certain of its affiliates (the "GENBAND Parties") and a prospectus of NewCo and Sonus and the parties may file with the SEC other relevant documents concerning the proposed transaction. Sonus will mail the joint proxy statement/prospectus to the Sonus stockholders and the GENBAND Party equity holders. SONUS STOCKHOLDERS AND GENBAND PARTY EQUITY HOLDERS ARE URGED TO READ THE REGISTRATION STATEMENT AND JOINT PROXY STATEMENT/PROSPECTUS REGARDING THE PROPOSED TRANSACTION WHEN IT BECOMES AVAILABLE AND ANY OTHER RELEVANT DOCUMENTS FILED WITH THE SEC, AS WELL AS ANY AMENDMENTS OR SUPPLEMENTS TO THOSE DOCUMENTS, BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION. Investors and security holders may obtain a free copy of the joint proxy statement/prospectus (when available) and other filings containing information about Sonus at the SEC's website at www.sec.gov. The joint proxy statement/prospectus (when available) and the other filings may also be obtained free of charge from Sonus' Investor Relations website (http://investors.sonusnet.com/) or by requesting them from Sonus' corporate secretary at Sonus

Networks, Inc., 4 Technology Park Drive, Westford, Massachusetts 01886, Attention: Corporate Secretary.

Sonus, NewCo, the GENBAND Parties and certain of their respective directors and executive officers, under the SEC's rules, may be deemed to be participants in the solicitation of proxies of Sonus stockholders in connection with the proposed transaction. Information about the directors and executive officers of Sonus and their ownership of Sonus common stock is set forth in the proxy statement for Sonus' 2017 annual meeting of stockholders, as filed with the SEC on Schedule 14A on April 28, 2017.

Additional information regarding the interests of those participants and other persons who may be deemed participants in the transaction may be obtained by reading the joint proxy statement/prospectus regarding the proposed transaction when it becomes available. Free copies of this document may be obtained as described in the preceding paragraphs.

# NO OFFERS OR SOLICITATIONS

This communication shall not constitute an offer to sell or the solicitation of an offer to sell or the solicitation of an offer to buy any securities, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offer of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the Securities Act of 1933, as amended.

### CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS

This communication contains "forward-looking statements"—that is, statements related to future, not past, events. In this context, forward-looking statements often address expected future business and financial performance and financial condition, and often contain words such as "expect," "anticipate," "intend," "plan," "believe," "seek," "see," "will," "would," or "target." Forward-looking statements by their nature address matters that are, to different degrees, uncertain, such as statements about the consummation of the proposed transaction and the expected performance of the combined companies. Uncertainties that could cause actual results to be materially different than those expressed in Sonus' and the GENBAND Parties' forward-looking statements include the failure to consummate the proposed transaction or to make or take any filing or other action required to consummate such transaction in a timely matter or at all. These or other uncertainties may cause actual future results to be materially different from those expressed in Sonus' and the GENBAND Parties'

forward-looking statements. These risks and uncertainties include, but are not limited to, economic, competitive, legal, governmental and technological factors. Accordingly, there is no assurance that the expectations of Sonus or any GENBAND Party will be realized. Many factors could cause actual results to differ materially from these forward-looking statements with respect to the proposed transaction, including risks relating to the completion of the proposed transaction on anticipated terms and timing, including obtaining equity holder and regulatory approvals, anticipated tax treatment, unforeseen liabilities, future capital expenditures, revenues, expenses, earnings, synergies, economic performance, indebtedness, financial condition, losses, future prospects, business and management strategies for the management, expansion and growth of the new combined company's operations, the ability of Sonus and the GENBAND Parties to integrate the business successfully and to achieve anticipated synergies, potential litigation relating to the proposed transaction, and the risk that disruptions from the proposed transaction will harm Sonus' or the GENBAND Parties' business. While the list of factors presented here is considered representative, no such list should be considered to be a complete statement of all potential risks and uncertainties. Unlisted factors may present significant additional obstacles to the realization of forward looking statements. Consequences of material differences in results as compared with those anticipated in the forward-looking statements could include, among other things, business disruption, operational problems, financial loss, legal liability to third parties and similar risks, any of which could have a material adverse effect on Sonus' or the GENBAND Parties' consolidated financial condition, results of operations or liquidity. Neither Sonus nor any GENBAND Party assumes any obligation to provide revisions to any forward-looking statements should circumstances change, except as other

#### Managers' Guide

For Internal Use Only - HIGHLY CONFIDENTIAL TO: Sonus Managers

SUBJECT: Managers Communications Guide — Sonus & GENBAND Combination

Dear Colleagues,

Today we are excited to announce that Sonus and GENBAND have entered into a definitive agreement to combine and create a leader in real-time communications. There are a lot of reasons why combining with GENBAND will accelerate our strategic plans, now that we've successfully transformed to a software-based company and are continuing our investment in new cloud architecture. The bottom line is that bringing our two companies together will create greater growth opportunities for both Sonus and GENBAND, and we'll be even better positioned for success as a combined organization.

It is important that we communicate effectively and accurately to all employees about this transaction, and express our confidence that this combination is great news for them, along with our shareholders/investors and customers.

As a manager, you will be asked by your team about the transaction, and we all need to speak with a consistent voice. Therefore, we ask that you reinforce the messages delivered in the enclosed Press Release and resist the temptation to speculate beyond what has been announced today. In addition to the Press Release, we are providing you with a series of Frequently Asked Questions (FAQ's) and the answers that can be relayed to your teams at this time. Many of the detailed questions that you and your team may have will be addressed once the transaction has closed. However, if you have any urgent questions that are not addressed within, please send them to your HR leader and they will work with the communications team to provide you with further information.

The key message for you to share with your teams is that while this is an exciting announcement, it is critical that we all stay focused on meeting our objectives and continue to serve our customers with the highest quality and professionalism possible. In other words: business as usual.

Thank you for your hard work and dedication to Sonus, which has helped us get to where we are today. Speaking on behalf of the entire Management team and Board, I'm proud to say that we have the best team in the business, and we're as ready as ever to embark on this next chapter together with GENBAND.

Best regards and many thanks,

Ray





# SONUS AND GENBAND TO COMBINE TO CREATE A GLOBAL LEADER IN REAL-TIME COMMUNICATIONS SOFTWARE SOLUTIONS

Companies' Combined 2016 Revenue was approximately \$680 Million and EBITDA was \$50 Million

Expected to be Substantially Accretive to Sonus' Non-GAAP EPS in 2018 and Generate Significant Cash Flow from Operations Post-Close

Projected Annual Cost Synergies of \$40-\$50 Million by the End of 2018

At Least \$100 Million in Annual EBITDA Expected Following Full Impact of Annualized Synergies; Fiscal Year 2020 Projected EBITDA to be approximately \$140 Million

Enhanced Scale and Geographic Reach and Highly Complementary Product Portfolios Expected to Enable Expanded Market Opportunities to Accelerate

Growth and Drive Shareholder Value

For Immediate Release: May 23, 2017

WESTFORD, MA and PLANO, TX — Sonus Networks, Inc. (Nasdaq: SONS) ("Sonus"), a global leader in securing cloud and real-time communications, and GENBAND<sup>TM</sup>, a leading provider of carrier and enterprise network transformation and real-time communications solutions, today announced a definitive agreement under which the two companies will combine to create a leader in next-generation communications networking, with increased scale and market reach across products, customers and geographies. Sonus and GENBAND shareholders will each own approximately 50% of the combined company. Based on the closing price of Sonus' common stock on May 22, 2017 of \$7.79 and estimated net cash at the time of closing, the transaction values the combined company at an enterprise value of approximately \$745 million.

The transaction combines Sonus' software-based leadership position in real-time communication virtualization, cloud-based session initiation protocol (SIP) and 4G/voice-over LTE solutions and security initiatives with GENBAND's network modernization, unified communications, and mobility and embedded communications solutions. Together, Sonus and GENBAND will be better positioned to enable network transformations to IP and cloud-based networks for communication service providers and enterprise customers worldwide, with a broader and deeper global sales footprint, increased ability to invest in growth, more efficient and effective R&D, and a comprehensive real-time communications product offering.

The two companies' combined 2016 revenue and EBITDA would have been approximately \$680 million and \$50 million, respectively, on a combined basis which excludes synergies and is prior to any impact from purchase accounting. The transaction is expected to be substantially accretive to Sonus' earnings per share in 2018. The combined company is expected to realize annual cost synergies of \$40 million to \$50 million by the end of 2018 and to drive solid cash flow from operations in the first year after closing. Following the full impact of expected annualized synergies, the combined company is expected to generate at least \$100 million in annual EBITDA, with fiscal year 2020 EBITDA projected to be approximately \$140 million.

#### **Management Comments**

"Together, Sonus and GENBAND create a market leader in real-time communications with enhanced capabilities to support our customers' move to cloud-based solutions," said Raymond Dolan, President and Chief Executive Officer of Sonus. "The transaction is expected to generate significant near- and

long-term value for shareholders, who we believe will benefit from their ownership in a combined company with increased scale and resources to invest in and accelerate each company's growth initiatives. This is a strategically and financially compelling transaction for Sonus and we are confident that together with GENBAND we will achieve our growth initiatives faster and more fully than either company could do on its own."

Mr. Dolan continued, "GENBAND has an outstanding reputation in the market with a talented team and products that are highly complementary to Sonus. We are excited to combine forces to provide enhanced reliability, performance and functionality for customers' real-time communications needs."

David Walsh, Chief Executive Officer and Chairman of GENBAND, said, "We are delighted to combine with Sonus at a time when the world's largest service providers and enterprises accelerate the modernization and transformation of their networks. Like Sonus, GENBAND has transitioned its business to support this industry shift and we have seen improving profitability over the last couple of years and into 2017. With this combination, we believe our complementary product portfolios and expanded global footprint will allow us to even better respond to the evolving needs of customers. Our heritage and culture fit perfectly together and the combined talent going forward is second to none. This is truly an ideal combination."

#### **Transaction Details**

Under the terms of the agreement, Sonus and GENBAND will combine under a newly formed holding company. Each Sonus shareholder will receive one share of common stock in the combined company for each existing Sonus share they own. The combined company will issue approximately 50 million shares of common stock to GENBAND's equity owners as well as \$22.5 million of consideration in the form of an unsecured note. Upon closing of the transaction, Sonus and GENBAND shareholders are each expected to own approximately 50 percent of the combined company on a fully diluted basis. The combined company will have an estimated net cash position of \$40 million to \$45 million at the time of close, which is expected to increase meaningfully in fiscal 2018 as the company realizes synergies.

The transaction has been unanimously approved by the Boards of Directors of both companies, and is expected to close in the second half of 2017, subject to Sonus and GENBAND shareholder approval, listing of the combined company's common stock on Nasdaq, Hart-Scott-Rodino review, and other customary closing conditions.

#### **Management and Board**

The Chief Executive Officer of the combined company will be Mr. Dolan. Mr. Walsh will oversee the Kandy business, a division of the combined company and currently GENBAND's cloud communications platform as a service (CPaaS) for global service providers and enterprises. Mr. Walsh will also assist with the integration of the businesses. Daryl Raiford, Chief Financial Officer of GENBAND, will serve as Chief Financial Officer of the combined company. The remainder of the leadership team will be named from the current leadership of both companies, and will be announced prior to closing.

The Board of Directors of the combined company will have five representatives designated by GENBAND and four representatives designated by Sonus. Sonus' designees will include Richard Lynch, current Chairman of Sonus and former Chief Technology Officer of Verizon, who will serve as Chairman, and Mr. Dolan. The full board composition will be announced in the near future.

#### **Strategic Benefits**

Complementary Strengths and Strategies: Both companies have pursued complementary strategies to enable network transformation for their customers. Sonus has concentrated on providing next generation cloud-based SIP and 4G/VoLTE solutions by providing network layer control, security, interworking, policy, and session management for the delivery of real-time communications. GENBAND's strategy is centered on seamlessly supporting its robust and long-term customer base of carriers and enterprises to modernize their network to IP with a broad product portfolio that includes soft switches, media gateways, and application servers, and leveraging that IP infrastructure with a rapid service creation capability. The combined company will have little product

overlap and will be well-positioned to deliver comprehensive solutions to service providers and enterprises migrating to a virtualized all-IP world.

- **Expanded Reach and Geographic Footprint:** Together, the combined company will have a global sales footprint in 27 countries. 67% of the combined 2016 revenues from the two companies were generated in the U.S. and Canada, 18% in EMEA, 11% in APAC and 4% in CALA. Together, Sonus and GENBAND will have a global presence, serving nearly every major Tier-1 communications service provider.
- Complementary Growth Initiatives: Both companies have established growth initiatives that are highly complementary. GENBAND's Kandy initiative enables service providers and other market participants to embed voice, video, messaging, chat, presence and more into any business process or application, enhancing outcomes for business, consumer and Internet of Things deployments. The Kandy CPaaS allows customers to rapidly create new revenue streams for customers/partners. Sonus' new offering is a cloud security and analytics platform for real time flows and digital services. This security offering provides end-to-end visibility and analytics enabling network-wide flow control, threat detection and mitigation. These combined growth initiatives aim to create the path for customers/partners to quickly and securely launch new real-time communications offerings.

# Sonus Reconfirms its Q2 2017 and Full Year 2017 Outlook

Sonus reconfirms its 2017 second quarter and full year outlook, as previously provided on April 26, 2017. Diluted earnings per share and loss per share are presented on both a GAAP and non-GAAP basis. A reconciliation of the non-GAAP to GAAP outlook and a statement on the use of non-GAAP financial measures are included at the end of this press release.

- · Revenue of approximately \$54 million
- · GAAP loss per share of approximately \$0.25
- · Non-GAAP loss per share of approximately \$0.10

#### First Half 2017

Revenue of approximately \$107 million

#### Fiscal 2017

- · Flat to low single digit revenue growth versus prior year
- · Non-GAAP diluted earnings per share of \$0.26, which equates to a GAAP loss per share of \$0.25

Sonus intends to update its guidance to include GENBAND following the close of the transaction, which is expected to occur in the second half of 2017.

#### **GENBAND Historical Financial Results:**

The following table summarizes the consolidated annual financial results of GENBAND (in millions):

		Fiscal Year Ended		
	Dec	ember 31, 2016	De	ecember 31, 2015
Total revenue	\$	427.1	\$	417.0
GAAP gross margin		50.2%		45.9%
Non-GAAP gross margin		50.8%		47.5%
GAAP operating expenses	\$	232.1	\$	282.2
Non-GAAP operating expenses	\$	205.7	\$	241.3
GAAP net loss	\$	(27.2)	\$	(91.7)
Non-GAAP net income (loss)	\$	2.1	\$	(43.5)
EBITDA(1)	\$	23.6	\$	(25.2)

(1) EBITDA represents earnings before interest, taxes, depreciation, amortization and certain other expenses.

#### Advisors

Evercore served as Sonus' financial advisor and Wilmer Cutler Pickering Hale and Dorr LLP served as Sonus' legal advisor. Guggenheim Securities and J.P. Morgan Securities LLC served as GENBAND's financial advisors and Latham & Watkins LLP served as GENBAND's legal advisor.

# **Conference Call Details**

There will be a conference call hosted by Mr. Dolan and Mr. Walsh.

Date: Today, May 23, 2017 Time: 8:30 a.m. (ET)

Dial-in number: 800-786-7015 International Callers: +1-303-223-0117

There will be a live, listen-only webcast of the conference call via the Sonus Investor website at http://investors.sonusnet.com/events.cfm where supporting materials, including a presentation, have been posted.

#### **Replay Information**

An archived version of the broadcast will be available on the Sonus Investor website shortly after the conclusion of the live event. A telephone playback of the call will be available following the conference call and can be accessed by calling 800-633-8284 or +1-402-977-9140 for international callers. The reservation number for the replay is 21852768.

# **About Sonus Networks**

Sonus brings intelligence and security to real-time communications. By helping the world embrace the next generation of Cloud-based SIP and 4G/LTE solutions, Sonus enables and secures latency-sensitive, mission critical traffic for VoIP, video, instant messaging and online collaboration. With Sonus, enterprises can give priority to real-time communications based on smart business rules while service providers can offer reliable, comprehensive and secure on-demand network services to their customers. With solutions deployed in more than 100 countries and nearly two decades of experience, Sonus offers a complete portfolio of hardware-based and virtualized session border controllers (SBCs), diameter signaling controllers (DSCs), policy/routing servers, network intelligence applications, media and signaling gateways and network analytics tools. For more information, visit www.Sonus.net or call 1-855-GO-Sonus.

## **About GENBAND**

GENBAND is a global leader in real-time communications software solutions for service providers, enterprises, independent software vendors, systems integrators and developers in over 80 countries. The company's Network Modernization, Unified Communications, Mobility and Embedded Communications solutions enable its customers to quickly capitalize on growing market segments and introduce differentiating products, applications and services.

GENBAND's market-leading solutions, which are deployable in the network, on premise or through the cloud, help its customers connect people to each other and address the growing demands of today's consumers and businesses for real-time communications wherever they happen to be. GENBAND's award-winning Marketing Advantage Program offers partners free resources, training and campaigns to fuel their success in implementing GENBAND powered services. To learn more visit GENBAND.com.

#### IMPORTANT INFORMATION ABOUT THE TRANSACTION WILL BE FILED WITH THE SEC

In connection with the proposed transaction, Sonus will cause Solstice Sapphire Investments, Inc., a wholly-owned subsidiary of Sonus formed to act as a holding company in connection with the transaction ("NewCo"), to file with the SEC a Registration Statement on Form S-4 that will include a joint proxy statement of Sonus and GENBAND and certain of its affiliates (the "GENBAND Parties") and a prospectus of NewCo and Sonus and the parties may file with the SEC other relevant documents concerning the proposed transaction. Sonus will mail the joint proxy statement/prospectus to the Sonus stockholders and the GENBAND Party equity holders. SONUS STOCKHOLDERS AND GENBAND PARTY EQUITY HOLDERS ARE URGED TO READ THE REGISTRATION STATEMENT AND JOINT PROXY STATEMENT/PROSPECTUS REGARDING THE PROPOSED TRANSACTION WHEN IT BECOMES AVAILABLE AND ANY OTHER RELEVANT DOCUMENTS FILED WITH THE SEC, AS WELL AS ANY AMENDMENTS OR SUPPLEMENTS TO THOSE DOCUMENTS, BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION. Investors and security holders may obtain a free copy of the joint proxy statement/prospectus (when available) and other filings containing information about Sonus at the SEC's website at www.sec.gov. The joint proxy statement/prospectus (when available) and the other filings may also be obtained free of charge from Sonus' Investor Relations website (http://investors.sonusnet.com/) or by requesting them from Sonus' corporate secretary at Sonus Networks, Inc., 4 Technology Park Drive, Westford, Massachusetts 01886, Attention: Corporate Secretary.

Sonus, NewCo, the GENBAND Parties and certain of their respective directors and executive officers, under the SEC's rules, may be deemed to be participants in the solicitation of proxies of Sonus stockholders in connection with the proposed transaction. Information about the directors and executive officers of Sonus and their ownership of Sonus common stock is set forth in the proxy statement for Sonus' 2017 annual meeting of stockholders, as filed with the SEC on Schedule 14A on April 28, 2017.

Additional information regarding the interests of those participants and other persons who may be deemed participants in the transaction may be obtained by reading the joint proxy statement/prospectus regarding the proposed transaction when it becomes available. Free copies of this document may be obtained as described in the preceding paragraphs.

## NO OFFERS OR SOLICITATIONS

This communication shall not constitute an offer to sell or the solicitation of an offer to sell or the solicitation of an offer to buy any securities, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offer of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the Securities Act of 1933, as amended.

# CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS

This communication contains "forward-looking statements"—that is, statements related to future, not past, events. In this context, forward-looking statements often address expected future business and financial performance and financial condition, and often contain words such as "expect," "anticipate," "intend," "plan," "believe," "seek," "see," "will," "would," or "target." Forward-looking statements by their nature address matters that are, to different degrees, uncertain, such as statements about the consummation of the proposed transaction and the expected performance of the combined companies. Uncertainties that could cause actual results to be materially different than those expressed in Sonus' and the GENBAND Parties' forward-looking statements include the failure to consummate the proposed transaction or to make or take any filing or other action required to consummate such transaction in a timely matter or at all. These or other uncertainties may cause actual future results to be materially different from those expressed in Sonus' and the GENBAND Parties' forward-looking statements. These risks and uncertainties include, but are not limited to, economic, competitive, legal, governmental and technological factors. Accordingly, there is no assurance that the expectations of Sonus or any GENBAND Party will be realized. Many factors could cause actual results to differ materially from these forward-looking

statements with respect to the proposed transaction, including risks relating to the completion of the proposed transaction on anticipated terms and timing, including obtaining equity holder and regulatory approvals, anticipated tax treatment, unforeseen liabilities, future capital expenditures, revenues, expenses, earnings, synergies, economic performance, indebtedness, financial condition, losses, future prospects, business and management strategies for the management, expansion and growth of the new combined company's operations, the ability of Sonus and the GENBAND Parties to integrate the business successfully and to achieve anticipated synergies, potential litigation relating to the proposed transaction, and the risk that disruptions from the proposed transaction will harm Sonus' or the GENBAND Parties' business. While the list of factors presented here is considered representative, no such list should be considered to be a complete statement of all potential risks and uncertainties. Unlisted factors may present significant additional obstacles to the realization of forward looking statements. Consequences of material differences in results as compared with those anticipated in the forward-looking statements could include, among other things, business disruption, operational problems, financial loss, legal liability to third parties and similar risks, any of which could have a material adverse effect on Sonus' or the GENBAND Parties' consolidated financial condition, results of operations or liquidity. Neither Sonus nor any GENBAND Party assumes any obligation to provide revisions to any forward-looking statements should circumstances change, except as otherwise required by securities and other applicable laws.

# DISCUSSION OF NON-GAAP FINANCIAL MEASURES

Sonus management uses several different financial measures, both GAAP and non-GAAP, in analyzing and assessing the overall performance of the business, making operating decisions, planning and forecasting future periods, and determining payments under compensation programs. Our annual financial plan is prepared both on a GAAP and non-GAAP basis, and the non-GAAP annual financial plan is approved by our board of directors. Continuous budgeting and forecasting for revenue and expenses are conducted on a non-GAAP basis (in addition to GAAP) and actual results on a non-GAAP basis are assessed against the annual financial plan. We consider the use of non-GAAP financial measures helpful in assessing the core performance of our continuing operations and liquidity, and when planning and forecasting future periods. By continuing operations, we mean the ongoing results of the business excluding certain

expenses and credits, including, but not limited to: stock-based compensation, amortization of intangible assets, acquisition-related expense, restructuring, certain gains included in other income (expense) and a deferred tax asset adjustment. We consider the use of non-GAAP earnings (loss) per share helpful in assessing the performance of the continuing operations of our business. While our management uses non-GAAP financial measures as a tool to enhance their understanding of certain aspects of our financial performance, our management does not consider these measures to be a substitute for, or superior to, GAAP measures. In addition, our presentations of these measures may not be comparable to similarly titled measures used by other companies. These non-GAAP financial measures should not be considered alternatives for, or in isolation from, the financial information prepared and presented in accordance with GAAP.

Investors are cautioned that there are material limitations associated with the use of non-GAAP financial measures as an analytical tool. In particular, many of the adjustments to Sonus' financial measures reflect the exclusion of items that are recurring and will be reflected in our financial results for the foreseeable future.

Stock-based compensation is different from other forms of compensation, as it is a non-cash expense. For example, a cash salary generally has a fixed and unvarying cash cost. In contrast, the expense associated with an equity-based award is generally unrelated to the amount of cash ultimately received by the employee, and the cost to us is based on a stock-based compensation valuation methodology and underlying assumptions that may vary over time. We believe that excluding non-cash stock-based compensation expense from our operating results facilitates the comparison of our financial statements to our historical operating results and to other companies in our industry.

We exclude the amortization of acquired intangible assets from non-GAAP expense and income measures. These amortization amounts are inconsistent in frequency and amount and are significantly

impacted by the timing and size of acquisitions. Although we exclude amortization of acquired intangible assets from our non-GAAP expenses, we believe that it is important for investors to understand that intangible assets contribute to revenue generation. We believe that excluding the non-cash amortization of intangible assets facilitates the comparison of our financial results to our historical operating results and to other companies in our industry as if the acquired intangible assets had been developed internally rather than acquired.

We consider certain transition, integration and other acquisition-related costs to be unpredictable and dependent on a significant number of factors that may be outside of our control. We do not consider these acquisition-related costs to be related to the continuing operations of the acquired business or the Company. In addition, the size, complexity and/or volume of an acquisition, which often drives the magnitude of acquisition-related costs, may not be indicative of such future costs. We believe that excluding acquisition-related costs facilitates the comparison of our financial results to our historical operating results and to other companies in our industry.

We have recorded restructuring expense to streamline operations and reduce operating costs by closing and consolidating certain facilities and reducing our worldwide workforce. Additionally, as previously announced, we expect to record restructuring expense in connection with new restructuring initiatives over the next twelve months. We review our restructuring accruals regularly and record adjustments (both expense and credits) to these estimates as required. We believe that excluding restructuring expense and credits facilitates the comparison of our financial results to our historical operating results and to other companies in our industry, as there are no future revenue streams or other benefits associated with these costs.

We expect to complete the sale of IP address blocks acquired in connection with our 2012 acquisition of Network Equipment Technologies, Inc. in the second half of 2017, and accordingly, have included a gain of \$0.6 million in our outlook for the full year 2017. This gain will be included as a component of Other Income, net. We believe that such gains are not part of our core business or ongoing operations. Accordingly, we believe that excluding the other income arising from this sale facilitates the comparison of our financial results to our historical results and to other companies in our industry.

We anticipate that we will reverse \$0.7 million of deferred tax assets related to net operating loss carryforwards for our subsidiary in Canada based on positive earnings evidence in the subsidiary over a consecutive three-year period. This adjustment will result in an income tax credit and reduce our provision in the reversal period. We believe that such adjustments are not part of our core business or ongoing operations. Accordingly, we believe that excluding the income tax credit arising from the reversal of the deferred tax assets facilitates the comparison of our financial results to our historical results and to other companies in our industry.

We believe that providing non-GAAP information to investors, in addition to the GAAP presentation, will allow investors to view the financial results in the way management views the operating results. We further believe that providing this information helps investors to better understand our financial performance and evaluate the efficacy of the methodology and information used by our management to evaluate and measure such performance.

#### **Contact Information**

Sonus

Investor Relations: Sara Leggat sleggat@sonusnet.com (978) 614-8841

Media Relations:

Wendy Tullo wtullo@sonusnet.com (978) 614-8167

GENBAND

Bita Milanian

# SONUS NETWORKS, INC. Reconciliation of Non-GAAP to GAAP Financial Measures Outlook - Q2 and FY 2017 \* (unaudited)

	onths ending 30, 2017
Earnings (loss) per share	
GAAP loss per share	\$ (0.25)
Stock-based compensation expense	0.08
Amortization of intangible assets	0.05
Restructuring	0.03
Sale of IP address blocks	(0.01)
Non-GAAP outlook	\$ (0.10)

	1	Year ending December 31, 2017	
Earnings (loss) per share			
GAAP loss per share	\$	(0.25)	
Stock-based compensation expense		0.30	
Amortization of intangible assets		0.19	
Acquisition-related expense		**	
Restructuring		0.04	
Sale of IP address blocks		(0.01)	
Deferred tax asset adjustment		(0.01)	
Non-GAAP outlook	\$	0.26	

<sup>\*</sup> As provided on April 26, 2017

#### Frequently Asked Questions (FAQ) — MANAGERS (INTERNAL)

# Q: Why is this transaction a good thing for employees?

A: From both a cultural and business perspective, Sonus and GENBAND are a great fit. Together, our two companies are well positioned to continue serving our customers and developing exciting and competitive real-time communications solutions that optimize the massive benefits of an all-IP and cloud services world, as service providers accelerate the modernization of their networks. The increased size and footprint of the combined company will allow us to grow revenues and profitability and increase our investment in new technology. This will benefit all stakeholders and create exciting opportunities for employees.

#### Q: How are we explaining this to customers?

A: While more details will be shared following the close of the transaction, both companies are committed to ensuring that the best solutions are provided to customers and partners. At this time, it is not appropriate for the two companies to work together on specific customer contracts or opportunities until the transaction closes, and additional information will be available after the close of the transaction.

# Q: What changes can we expect?

A: It is business as usual and Sonus and GENBAND will continue to operate independently until the close of the transaction. After the close we will communicate more on how we will integrate but we believe there will be a smooth transition.

### Q: Will our medical coverage/benefit plans change? What about compensation?

A: You will continue to receive pay and benefits from Sonus as usual, and there will be no changes at this time. Once the transaction is closed, any updates will be communicated as appropriate in normal course.

# Q: What are your plans for post-closing product support?

A: We will be reviewing the roadmaps for all products in both companies, and we will share decisions on these roadmaps when they are available after the close of the transaction. For now, it is business as usual.

# Q: When will we see the new organization structure?

A: Each company will run its business independently through closing of the transaction. Any changes to be implemented to the organizational structure of the combined business will be shared and made after the transaction closes.

### Q: Where can I go if I have more questions?

A: Please do not hesitate to contact your direct manager with any questions you may have.

<sup>\*\*</sup> Less than \$0.01 impact on earnings (loss) per share

- Q: What if I have a question my manager can't answer?
- A: As always you should feel free to reach out to your HR contact.
- **Q:** Will we have one intranet?
- A: Yes, upon close of the transaction.
- Q: How will our systems be affected (email, phones, CRM, applications, intranet etc.)?
- A: Both companies have excellent systems in place. As part of the integration planning process, we will be assessing all of our systems to ensure they can provide the greatest benefit to the combined company, and further details will be shared after closing.
- Q: Will the management team be sharing information regularly on our strategies and plans?
- A: For now, it is business as usual. We will of course share any updates along the way as appropriate. Once we have closed, the Management team will establish a regular cadence for communications.
- Q: Can I start talking to my peers at the other company?
- A: Please refrain from talking to peers at the other company until after closing. At that time, we'll provide more details about communication plans.
- Q: Where can I go for more information?
- A: Please visit www.rtcleaders.com. We will use this site to share information common to Sonus and GENBAND.

#### IMPORTANT INFORMATION ABOUT THE TRANSACTION WILL BE FILED WITH THE SEC

In connection with the proposed transaction, Sonus will cause Solstice Sapphire Investments, Inc., a wholly-owned subsidiary of Sonus formed to act as a holding company in connection with the transaction ("NewCo"), to file with the SEC a Registration Statement on Form S-4 that will include a joint proxy statement of Sonus and GENBAND and certain of its affiliates (the "GENBAND Parties") and a prospectus of NewCo and Sonus and the parties may file with the SEC other relevant documents concerning the proposed transaction. Sonus will mail the joint proxy statement/prospectus to the Sonus stockholders and the GENBAND Party equity holders. SONUS STOCKHOLDERS AND GENBAND PARTY EQUITY HOLDERS ARE URGED TO READ THE REGISTRATION STATEMENT AND JOINT PROXY STATEMENT/PROSPECTUS REGARDING THE PROPOSED TRANSACTION WHEN IT BECOMES AVAILABLE AND ANY OTHER RELEVANT DOCUMENTS FILED WITH THE SEC, AS WELL AS ANY AMENDMENTS OR SUPPLEMENTS TO THOSE DOCUMENTS, BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION. Investors and security holders may obtain a free copy of the joint proxy statement/prospectus (when available) and other filings containing information about Sonus at the SEC's website at www.sec.gov. The joint proxy statement/prospectus (when available) and the other filings may also be obtained free of charge from Sonus' Investor Relations website (http://investors.sonusnet.com/) or by requesting them from Sonus' corporate secretary at Sonus Networks, Inc., 4 Technology Park Drive, Westford, Massachusetts 01886, Attention: Corporate Secretary.

Sonus, NewCo, the GENBAND Parties and certain of their respective directors and executive officers, under the SEC's rules, may be deemed to be participants in the solicitation of proxies of Sonus stockholders in connection with the proposed transaction. Information about the directors and executive officers of Sonus and their ownership of Sonus common stock is set forth in the proxy statement for Sonus' 2017 annual meeting of stockholders, as filed with the SEC on Schedule 14A on April 28, 2017.

Additional information regarding the interests of those participants and other persons who may be deemed participants in the transaction may be obtained by reading the joint proxy statement/prospectus regarding the proposed transaction when it becomes available. Free copies of this document may be obtained as described in the preceding paragraphs.

#### NO OFFERS OR SOLICITATIONS

This communication shall not constitute an offer to sell or the solicitation of an offer to sell or the solicitation of an offer to buy any securities, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offer of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the Securities Act of 1933, as amended.

#### CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS

This communication contains "forward-looking statements"—that is, statements related to future, not past, events. In this context, forward-looking statements often address expected future business and financial performance and financial condition, and often contain words such as "expect," "anticipate," "intend," "plan," "believe," "seek," "see," "will," "would," or "target." Forward-looking statements by their nature address matters that are, to different degrees, uncertain, such as statements about the consummation of the proposed transaction and the expected performance of the combined companies. Uncertainties that could cause actual results to be materially different than those expressed in Sonus' and the GENBAND Parties' forward-looking statements include the failure to consummate the proposed transaction or to make or take any filing or other action required to consummate such transaction in a timely matter or at all. These or other uncertainties may cause actual future results to be materially different from those expressed in Sonus' and the GENBAND Parties' forward-looking statements. These risks and uncertainties include, but are not limited to, economic, competitive, legal, governmental and technological factors. Accordingly, there is no assurance that the expectations of Sonus or any GENBAND Party will be realized. Many factors could cause actual results to differ materially from these forward-looking statements with respect to the proposed transaction, including risks relating to the completion of the proposed transaction on anticipated terms and timing, including obtaining equity holder and regulatory approvals, anticipated tax treatment, unforeseen liabilities,

future capital expenditures, revenues, expenses, earnings, synergies, economic performance, indebtedness, financial condition, losses, future prospects, business and management strategies for the management, expansion and growth of the new combined company's operations, the ability of Sonus and the GENBAND Parties to integrate the business successfully and to achieve anticipated synergies, potential litigation relating to the proposed transaction, and the risk that disruptions from the proposed transaction will harm Sonus' or the GENBAND Parties' business. While the list of factors presented here is considered representative, no such list should be considered to be a complete statement of all potential risks and uncertainties. Unlisted factors may present significant additional obstacles to the realization of forward looking statements. Consequences of material differences in results as compared with those anticipated in the forward-looking statements could include, among other things, business disruption, operational problems, financial loss, legal liability to third parties and similar risks, any of which could have a material adverse effect on Sonus' or the GENBAND Parties' consolidated financial condition, results of operations or liquidity. Neither Sonus nor any GENBAND Party assumes any obligation to provide revisions to any forward-looking statements should circumstances change, except as otherwise required by securities and other applicable laws.

#### **Customer Communication Guide**

For Internal Use Only - HIGHLY CONFIDENTIAL

NOT FOR DISTRIBUTION

TO: Sonus Sales Team

SUBJECT: Customer Communications Guide — Sonus & GENBAND Combination

Sonus Sales Teams,

Today we are excited to announce that Sonus and GENBAND have entered into a definitive agreement to combine and create a leader in real-time communications. There are a lot of reasons why combining with GENBAND will accelerate our strategic plans, now that we've successfully transformed to a software-based company and are continuing our investment in new cloud architecture. The bottom line is that bringing our two companies together will create greater growth opportunities for both Sonus and GENBAND, and we'll be even better positioned for success as a combined organization.

It is important that we communicate effectively and accurately with our customers about this transaction, and express our confidence that this combination is great news for them, along with our shareholders/investors and employees.

As a customer-facing representative, you will be asked by your customers about the transaction, and we all need to speak with a consistent voice. Therefore, we ask that you reinforce the messages delivered in the enclosed Press Release and resist the temptation to speculate beyond what has been announced today. In addition to the Press Release, we are providing you with a series of Frequently Asked Questions (FAQ's) and the appropriate answers to relay to your teams at this time. Many of the detailed questions that you and your team may naturally have will be addressed once the transaction closes. However, if you have any urgent questions that are not addressed within, please send them to your sales leader and they will provide you with information as needed.

The key message for you to share with your teams is that while this is an exciting announcement, it is critical that we all stay focused on meeting our objectives and continue to serve our customers with the highest level of quality and professionalism possible. In other words: business as usual.

Thank you for your hard work and dedication to Sonus, which has helped us get to where we are today. Speaking on behalf of the entire Management team and Board, I'm proud to say that we have the best team in the business, and we're as ready as ever to embark on this next chapter together with GENBAND.

Best regards and many thanks,

Ray





#### SONUS AND GENBAND TO COMBINE TO CREATE A GLOBAL LEADER IN REAL-TIME COMMUNICATIONS SOFTWARE SOLUTIONS

Companies' Combined 2016 Revenue was approximately \$680 Million and EBITDA was \$50 Million

Expected to be Substantially Accretive to Sonus' Non-GAAP EPS in 2018 and Generate Significant Cash Flow from Operations Post-Close

Projected Annual Cost Synergies of \$40-\$50 Million by the End of 2018

At Least \$100 Million in Annual EBITDA Expected Following Full Impact of Annualized Synergies; Fiscal Year 2020 Projected EBITDA to be approximately \$140 Million

Enhanced Scale and Geographic Reach and Highly Complementary Product Portfolios Expected to Enable Expanded Market Opportunities to Accelerate

Growth and Drive Shareholder Value

For Immediate Release: May 23, 2017

WESTFORD, MA and PLANO, TX — Sonus Networks, Inc. (Nasdaq: SONS) ("Sonus"), a global leader in securing cloud and real-time communications, and GENBAND™, a leading provider of carrier and enterprise network transformation and real-time communications solutions, today announced a definitive agreement under which the two companies will combine to create a leader in next-generation communications networking, with increased scale and market reach across products, customers and geographies. Sonus and GENBAND shareholders will each own approximately 50% of the combined company. Based on the closing price of Sonus' common stock on May 22, 2017 of \$7.79 and estimated net cash at the time of closing, the transaction values the combined company at an enterprise value of approximately \$745 million.

The transaction combines Sonus' software-based leadership position in real-time communication virtualization, cloud-based session initiation protocol (SIP) and 4G/voice-over LTE solutions and security initiatives with GENBAND's network modernization, unified communications, and mobility and embedded communications solutions. Together, Sonus and GENBAND will be better positioned to enable network transformations to IP and cloud-based networks for communication service providers and enterprise customers worldwide, with a broader and deeper global sales footprint, increased ability to invest in growth, more efficient and effective R&D, and a comprehensive real-time communications product offering.

The two companies' combined 2016 revenue and EBITDA would have been approximately \$680 million and \$50 million, respectively, on a combined basis which excludes synergies and is prior to any impact from purchase accounting. The transaction is expected to be substantially accretive to Sonus' earnings per share in 2018. The combined company is expected to realize annual cost synergies of \$40 million to \$50 million by the end of 2018 and to drive solid cash flow from operations in the first year after closing. Following the full impact of expected annualized synergies, the combined company is expected to generate at least \$100 million in annual EBITDA, with fiscal year 2020 EBITDA projected to be approximately \$140 million.

#### **Management Comments**

"Together, Sonus and GENBAND create a market leader in real-time communications with enhanced capabilities to support our customers' move to cloud-based solutions," said Raymond Dolan, President and Chief Executive Officer of Sonus. "The transaction is expected to generate significant near- and

long-term value for shareholders, who we believe will benefit from their ownership in a combined company with increased scale and resources to invest in and accelerate each company's growth initiatives. This is a strategically and financially compelling transaction for Sonus and we are confident that together with GENBAND we will achieve our growth initiatives faster and more fully than either company could do on its own."

Mr. Dolan continued, "GENBAND has an outstanding reputation in the market with a talented team and products that are highly complementary to Sonus. We are excited to combine forces to provide enhanced reliability, performance and functionality for customers' real-time communications needs."

David Walsh, Chief Executive Officer and Chairman of GENBAND, said, "We are delighted to combine with Sonus at a time when the world's largest service providers and enterprises accelerate the modernization and transformation of their networks. Like Sonus, GENBAND has transitioned its business to support this industry shift and we have seen improving profitability over the last couple of years and into 2017. With this combination, we believe our complementary product portfolios and expanded global footprint will allow us to even better respond to the evolving needs of customers. Our heritage and culture fit perfectly together and the combined talent going forward is second to none. This is truly an ideal combination."

#### **Transaction Details**

Under the terms of the agreement, Sonus and GENBAND will combine under a newly formed holding company. Each Sonus shareholder will receive one share of common stock in the combined company for each existing Sonus share they own. The combined company will issue approximately 50 million shares of common stock to GENBAND's equity owners as well as \$22.5 million of consideration in the form of an unsecured note. Upon closing of the transaction, Sonus and GENBAND shareholders are each expected to own approximately 50 percent of the combined company on a fully diluted basis. The combined company will have an estimated net cash position of \$40 million to \$45 million at the time of close, which is expected to increase meaningfully in fiscal 2018 as the company realizes synergies.

The transaction has been unanimously approved by the Boards of Directors of both companies, and is expected to close in the second half of 2017, subject to Sonus and GENBAND shareholder approval, listing of the combined company's common stock on Nasdaq, Hart-Scott-Rodino review, and other customary closing conditions.

# **Management and Board**

The Chief Executive Officer of the combined company will be Mr. Dolan. Mr. Walsh will oversee the Kandy business, a division of the combined company and currently GENBAND's cloud communications platform as a service (CPaaS) for global service providers and enterprises. Mr. Walsh will also assist with the integration of the businesses. Daryl Raiford, Chief Financial Officer of GENBAND, will serve as Chief Financial Officer of the combined company. The remainder of the leadership team will be named from the current leadership of both companies, and will be announced prior to closing.

The Board of Directors of the combined company will have five representatives designated by GENBAND and four representatives designated by Sonus. Sonus' designees will include Richard Lynch, current Chairman of Sonus and former Chief Technology Officer of Verizon, who will serve as Chairman, and Mr. Dolan. The full board composition will be announced in the near future.

### **Strategic Benefits**

Complementary Strengths and Strategies: Both companies have pursued complementary strategies to enable network transformation for their customers. Sonus has concentrated on providing next generation cloud-based SIP and 4G/VoLTE solutions by providing network layer control, security, interworking, policy, and session management for the delivery of real-time communications. GENBAND's strategy is centered on seamlessly supporting its robust and long-term customer base of carriers and enterprises to modernize their network to IP with a broad product portfolio that includes soft switches, media gateways, and application servers, and leveraging that IP infrastructure with a rapid service creation capability. The combined company will have little product

overlap and will be well-positioned to deliver comprehensive solutions to service providers and enterprises migrating to a virtualized all-IP world.

- Expanded Reach and Geographic Footprint: Together, the combined company will have a global sales footprint in 27 countries. 67% of the combined 2016 revenues from the two companies were generated in the U.S. and Canada, 18% in EMEA, 11% in APAC and 4% in CALA. Together, Sonus and GENBAND will have a global presence, serving nearly every major Tier-1 communications service provider.
- Complementary Growth Initiatives: Both companies have established growth initiatives that are highly complementary. GENBAND's Kandy initiative enables service providers and other market participants to embed voice, video, messaging, chat, presence and more into any business process or application, enhancing outcomes for business, consumer and Internet of Things deployments. The Kandy CPaaS allows customers to rapidly create new revenue streams for customers/partners. Sonus' new offering is a cloud security and analytics platform for real time flows and digital services. This security offering provides end-to-end visibility and analytics enabling network-wide flow control, threat detection and mitigation. These combined growth initiatives aim to create the path for customers/partners to quickly and securely launch new real-time communications offerings.

## Sonus Reconfirms its Q2 2017 and Full Year 2017 Outlook

Sonus reconfirms its 2017 second quarter and full year outlook, as previously provided on April 26, 2017. Diluted earnings per share and loss per share are presented on both a GAAP and non-GAAP basis. A reconciliation of the non-GAAP to GAAP outlook and a statement on the use of non-GAAP financial measures are included at the end of this press release.

#### Q217

- · Revenue of approximately \$54 million
- · GAAP loss per share of approximately \$0.25
- · Non-GAAP loss per share of approximately \$0.10

#### First Half 2017

Revenue of approximately \$107 million

#### Fiscal 2017

- · Flat to low single digit revenue growth versus prior year
- · Non-GAAP diluted earnings per share of \$0.26, which equates to a GAAP loss per share of \$0.25

Sonus intends to update its guidance to include GENBAND following the close of the transaction, which is expected to occur in the second half of 2017.

#### **GENBAND Historical Financial Results:**

The following table summarizes the consolidated annual financial results of GENBAND (in millions):

	Fiscal Year Ended		
	mber 31, 2016	De	cember 31, 2015
Total revenue	\$ 427.1	\$	417.0
GAAP gross margin	50.2%		45.9%
Non-GAAP gross margin	50.8%		47.5%
GAAP operating expenses	\$ 232.1	\$	282.2
Non-GAAP operating expenses	\$ 205.7	\$	241.3
GAAP net loss	\$ (27.2)	\$	(91.7)
Non-GAAP net income (loss)	\$ 2.1	\$	(43.5)
EBITDA(1)	\$ 23.6	\$	(25.2)

(1) EBITDA represents earnings before interest, taxes, depreciation, amortization and certain other expenses.

# Advisors

Evercore served as Sonus' financial advisor and Wilmer Cutler Pickering Hale and Dorr LLP served as Sonus' legal advisor. Guggenheim Securities and J.P. Morgan Securities LLC served as GENBAND's financial advisors and Latham & Watkins LLP served as GENBAND's legal advisor.

#### **Conference Call Details**

There will be a conference call hosted by Mr. Dolan and Mr. Walsh.

Date: Today, May 23, 2017 Time: 8:30 a.m. (ET) Dial-in number: 800-786-7015

International Callers: +1-303-223-0117

There will be a live, listen-only webcast of the conference call via the Sonus Investor website at http://investors.sonusnet.com/events.cfm where supporting materials, including a presentation, have been posted.

# **Replay Information**

An archived version of the broadcast will be available on the Sonus Investor website shortly after the conclusion of the live event. A telephone playback of the call will be available following the conference call and can be accessed by calling 800-633-8284 or +1-402-977-9140 for international callers. The reservation number for the replay is 21852768.

#### **About Sonus Networks**

Sonus brings intelligence and security to real-time communications. By helping the world embrace the next generation of Cloud-based SIP and 4G/LTE solutions, Sonus enables and secures latency-sensitive, mission critical traffic for VoIP, video, instant messaging and online collaboration. With Sonus, enterprises can give priority to real-time communications based on smart business rules while service providers can offer reliable, comprehensive and secure on-demand network services to their customers. With solutions deployed in more than 100 countries and nearly two decades of experience, Sonus offers a complete portfolio of hardware-based and virtualized session border controllers (SBCs), diameter signaling controllers (DSCs), policy/routing servers, network intelligence applications, media and signaling gateways and network analytics tools. For more information, visit www.Sonus.net or call 1-855-GO-Sonus

#### **About GENBAND**

GENBAND is a global leader in real-time communications software solutions for service providers, enterprises, independent software vendors, systems integrators and developers in over 80 countries. The company's Network Modernization, Unified Communications, Mobility and Embedded Communications

solutions enable its customers to quickly capitalize on growing market segments and introduce differentiating products, applications and services. GENBAND's market-leading solutions, which are deployable in the network, on premise or through the cloud, help its customers connect people to each other and address the growing demands of today's consumers and businesses for real-time communications wherever they happen to be. GENBAND's award-winning Marketing Advantage Program offers partners free resources, training and campaigns to fuel their success in implementing GENBAND powered services. To learn more visit GENBAND.com.

#### IMPORTANT INFORMATION ABOUT THE TRANSACTION WILL BE FILED WITH THE SEC

In connection with the proposed transaction, Sonus will cause Solstice Sapphire Investments, Inc., a wholly-owned subsidiary of Sonus formed to act as a holding company in connection with the transaction ("NewCo"), to file with the SEC a Registration Statement on Form S-4 that will include a joint proxy statement of Sonus and GENBAND and certain of its affiliates (the "GENBAND Parties") and a prospectus of NewCo and Sonus and the parties may file with the SEC other relevant documents concerning the proposed transaction. Sonus will mail the joint proxy statement/prospectus to the Sonus stockholders and the GENBAND Party equity holders. SONUS STOCKHOLDERS AND GENBAND PARTY EQUITY HOLDERS ARE URGED TO READ THE REGISTRATION STATEMENT AND JOINT PROXY STATEMENT/PROSPECTUS REGARDING THE PROPOSED TRANSACTION WHEN IT BECOMES AVAILABLE AND ANY OTHER RELEVANT DOCUMENTS FILED WITH THE SEC, AS WELL AS ANY AMENDMENTS OR SUPPLEMENTS TO THOSE DOCUMENTS, BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION. Investors and security holders may obtain a free copy of the joint proxy statement/prospectus (when available) and other filings containing information about Sonus at the SEC's website at www.sec.gov. The joint proxy statement/prospectus (when available) and the other filings may also be obtained free of charge from Sonus' Investor Relations website (http://investors.sonusnet.com/) or by requesting them from Sonus' corporate secretary at Sonus Networks, Inc., 4 Technology Park Drive, Westford, Massachusetts 01886, Attention: Corporate Secretary.

Sonus, NewCo, the GENBAND Parties and certain of their respective directors and executive officers, under the SEC's rules, may be deemed to be participants in the solicitation of proxies of Sonus stockholders in connection with the proposed transaction. Information about the directors and executive officers of Sonus and their ownership of Sonus common stock is set forth in the proxy statement for Sonus' 2017 annual meeting of stockholders, as filed with the SEC on Schedule 14A on April 28, 2017.

Additional information regarding the interests of those participants and other persons who may be deemed participants in the transaction may be obtained by reading the joint proxy statement/prospectus regarding the proposed transaction when it becomes available. Free copies of this document may be obtained as described in the preceding paragraphs.

#### NO OFFERS OR SOLICITATIONS

This communication shall not constitute an offer to sell or the solicitation of an offer to sell or the solicitation of an offer to buy any securities, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offer of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the Securities Act of 1933, as amended.

# CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS

This communication contains "forward-looking statements"—that is, statements related to future, not past, events. In this context, forward-looking statements often address expected future business and financial performance and financial condition, and often contain words such as "expect," "anticipate," "intend," "plan," "believe," "seek," "see," "will," "would," or "target." Forward-looking statements by their nature address matters that are, to different degrees, uncertain, such as statements about the consummation of the proposed transaction and the expected performance of the combined companies. Uncertainties that could cause actual results to be materially different than those expressed in Sonus' and the GENBAND Parties' forward-looking statements include the failure to consummate the proposed transaction or to make or take any filing or other action required to consummate such transaction in a timely matter or at all. These or other uncertainties may cause actual future results to be materially different from those expressed in Sonus' and the GENBAND Parties' forward-looking statements. These risks and uncertainties include, but are not limited to, economic, competitive, legal, governmental and technological factors. Accordingly, there is no assurance that the expectations of Sonus or any GENBAND Party will be realized. Many factors could cause actual results to differ materially from these forward-looking

statements with respect to the proposed transaction, including risks relating to the completion of the proposed transaction on anticipated terms and timing, including obtaining equity holder and regulatory approvals, anticipated tax treatment, unforeseen liabilities, future capital expenditures, revenues, expenses, earnings, synergies, economic performance, indebtedness, financial condition, losses, future prospects, business and management strategies for the management, expansion and growth of the new combined company's operations, the ability of Sonus and the GENBAND Parties to integrate the business successfully and to achieve anticipated synergies, potential litigation relating to the proposed transaction, and the risk that disruptions from the proposed transaction will harm Sonus' or the GENBAND Parties' business. While the list of factors presented here is considered representative, no such list should be considered to be a complete statement of all potential risks and uncertainties. Unlisted factors may present significant additional obstacles to the realization of forward looking statements. Consequences of material differences in results as compared with those anticipated in the forward-looking statements could include, among other things, business disruption, operational problems, financial loss, legal liability to third parties and similar risks, any of which could have a material adverse effect on Sonus' or the GENBAND Parties' consolidated financial condition, results of operations or liquidity. Neither Sonus nor any GENBAND Party assumes any obligation to provide revisions to any forward-looking statements should circumstances change, except as otherwise required by securities and other applicable laws.

#### DISCUSSION OF NON-GAAP FINANCIAL MEASURES

Sonus management uses several different financial measures, both GAAP and non-GAAP, in analyzing and assessing the overall performance of the business, making operating decisions, planning and forecasting future periods, and determining payments under compensation programs. Our annual financial plan is prepared both on a GAAP and non-GAAP basis, and the non-GAAP annual financial plan is approved by our board of directors. Continuous budgeting and forecasting for revenue and expenses are conducted on a non-GAAP basis (in addition to GAAP) and actual results on a non-GAAP basis are assessed against the annual financial plan. We consider the use of non-GAAP financial measures helpful in assessing the core performance of our continuing operations and

liquidity, and when planning and forecasting future periods. By continuing operations, we mean the ongoing results of the business excluding certain expenses and credits, including, but not limited to: stock-based compensation, amortization of intangible assets, acquisition-related expense, restructuring, certain gains included in other income (expense) and a deferred tax asset adjustment. We consider the use of non-GAAP earnings (loss) per share helpful in assessing the performance of the continuing operations of our business. While our management uses non-GAAP financial measures as a tool to enhance their understanding of certain aspects of our financial performance, our management does not consider these measures to be a substitute for, or superior to, GAAP measures. In addition, our presentations of these measures may not be comparable to similarly titled measures used by other companies. These non-GAAP financial measures should not be considered alternatives for, or in isolation from, the financial information prepared and presented in accordance with GAAP.

Investors are cautioned that there are material limitations associated with the use of non-GAAP financial measures as an analytical tool. In particular, many of the adjustments to Sonus' financial measures reflect the exclusion of items that are recurring and will be reflected in our financial results for the foreseeable future.

Stock-based compensation is different from other forms of compensation, as it is a non-cash expense. For example, a cash salary generally has a fixed and unvarying cash cost. In contrast, the expense associated with an equity-based award is generally unrelated to the amount of cash ultimately received by the employee, and the cost to us is based on a stock-based compensation valuation methodology and underlying assumptions that may vary over time. We believe that excluding non-cash stock-based compensation expense from our operating results facilitates the comparison of our financial statements to our historical operating results and to other companies in our industry.

We exclude the amortization of acquired intangible assets from non-GAAP expense and income measures. These amortization amounts are inconsistent in frequency and amount and are significantly

impacted by the timing and size of acquisitions. Although we exclude amortization of acquired intangible assets from our non-GAAP expenses, we believe that it is important for investors to understand that intangible assets contribute to revenue generation. We believe that excluding the non-cash amortization of intangible assets facilitates the comparison of our financial results to our historical operating results and to other companies in our industry as if the acquired intangible assets had been developed internally rather than acquired.

We consider certain transition, integration and other acquisition-related costs to be unpredictable and dependent on a significant number of factors that may be outside of our control. We do not consider these acquisition-related costs to be related to the continuing operations of the acquired business or the Company. In addition, the size, complexity and/or volume of an acquisition, which often drives the magnitude of acquisition-related costs, may not be indicative of such future costs. We believe that excluding acquisition-related costs facilitates the comparison of our financial results to our historical operating results and to other companies in our industry.

We have recorded restructuring expense to streamline operations and reduce operating costs by closing and consolidating certain facilities and reducing our worldwide workforce. Additionally, as previously announced, we expect to record restructuring expense in connection with new restructuring initiatives over the next twelve months. We review our restructuring accruals regularly and record adjustments (both expense and credits) to these estimates as required. We believe that excluding restructuring expense and credits facilitates the comparison of our financial results to our historical operating results and to other companies in our industry, as there are no future revenue streams or other benefits associated with these costs.

We expect to complete the sale of IP address blocks acquired in connection with our 2012 acquisition of Network Equipment Technologies, Inc. in the second half of 2017, and accordingly, have included a gain of \$0.6 million in our outlook for the full year 2017. This gain will be included as a component of Other Income, net. We believe that such gains are not part of our core business or ongoing operations. Accordingly, we believe that excluding the other income arising from this sale facilitates the comparison of our financial results to our historical results and to other companies in our industry.

We anticipate that we will reverse \$0.7 million of deferred tax assets related to net operating loss carryforwards for our subsidiary in Canada based on positive earnings evidence in the subsidiary over a consecutive three-year period. This adjustment will result in an income tax credit and reduce our provision in the reversal period. We believe that such adjustments are not part of our core business or ongoing operations. Accordingly, we believe that excluding the income tax credit arising from the reversal of the deferred tax assets facilitates the comparison of our financial results to our historical results and to other companies in our industry.

We believe that providing non-GAAP information to investors, in addition to the GAAP presentation, will allow investors to view the financial results in the way management views the operating results. We further believe that providing this information helps investors to better understand our financial performance and evaluate the efficacy of the methodology and information used by our management to evaluate and measure such performance.

# **Contact Information**

Sonus

Investor Relations: Sara Leggat sleggat@sonusnet.com (978) 614-8841

Media Relations:

Wendy Tullo wtullo@sonusnet.com (978) 614-8167

GENBAND

# SONUS NETWORKS, INC. Reconciliation of Non-GAAP to GAAP Financial Measures Outlook - Q2 and FY 2017 \* (unaudited)

Year ending

		Three months ending June 30, 2017	
Earnings (loss) per share			
GAAP loss per share	\$	(0.25)	
Stock-based compensation expense		0.08	
Amortization of intangible assets		0.05	
Restructuring		0.03	
Sale of IP address blocks		(0.01)	
Non-GAAP outlook	\$	(0.10)	
	<del></del>		

	December 31, 2017	
Earnings (loss) per share		_
GAAP loss per share	\$	(0.25)
Stock-based compensation expense		0.30
Amortization of intangible assets		0.19
Acquisition-related expense		**
Restructuring		0.04
Sale of IP address blocks		(0.01)
Deferred tax asset adjustment		(0.01)
Non-GAAP outlook	\$	0.26

<sup>\*</sup> As provided on April 26, 2017

#### Frequently Asked Questions (FAQ) — CUSTOMERS

#### Q: Why is this transaction a positive thing for my company?

A: While both companies offer best-in-breed solutions in many categories, our combination will enable us to offer increasing value for you: better solutions, better services, better coverage, and complementary growth initiatives. In the interim we will continue to support new and existing customers with current needs and future plans. Please note that there is no change to any current business relationships. We look forward to sharing our plans after the close of the transaction, which is expected to be in the second half of 2017.

#### Q: What are your plans for post-closing product support?

A: We will be reviewing the roadmaps for all products in both companies, and we will share decisions on these roadmaps when they are available after the close of the transaction. For now, it is business as usual.

# Q: How is this going to impact the software, systems and product road maps?

A: It will take some time upon the close of the transaction to detail any specific impacts to solutions and services, but we look forward to sharing more detailed information following the close. This combination joins two companies with differing strengths and will result in new capabilities and solutions resulting in a company well positioned to capitalize on the fastest growing areas in the industry.

# Q: Who will my new sales person be?

A: Until the close, your team stays the same — business as usual. We will provide more information at that time.

### Q: Who will my service team be?

A: Until the close, your team stays the same — business as usual. We will provide more information at that time.

### Q: We have an upcoming bid, will the two companies have a combined offering?

A: No. Until the close, it is business as usual and the two companies continue to be independent.

# Q: Where can I go for more information?

A: Please visit www.rtcleaders.com

# IMPORTANT INFORMATION ABOUT THE TRANSACTION WILL BE FILED WITH THE SEC

<sup>\*\*</sup> Less than \$0.01 impact on earnings (loss) per share

In connection with the proposed transaction, Sonus will cause Solstice Sapphire Investments, Inc., a wholly-owned subsidiary of Sonus formed to act as a holding company in connection with the transaction ("NewCo"), to file with the SEC a Registration Statement on Form S-4 that will include a joint proxy statement of Sonus and GENBAND and certain of its affiliates (the "GENBAND Parties") and a prospectus of NewCo and Sonus and the parties may file with the SEC other relevant documents concerning the proposed transaction. Sonus will mail the joint proxy statement/prospectus to the Sonus stockholders and the GENBAND Party equity holders. SONUS STOCKHOLDERS AND GENBAND PARTY EQUITY HOLDERS ARE URGED TO READ THE REGISTRATION STATEMENT AND JOINT PROXY STATEMENT/PROSPECTUS REGARDING THE PROPOSED TRANSACTION WHEN IT BECOMES AVAILABLE AND ANY OTHER RELEVANT DOCUMENTS FILED WITH THE SEC, AS

WELL AS ANY AMENDMENTS OR SUPPLEMENTS TO THOSE DOCUMENTS, BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION. Investors and security holders may obtain a free copy of the joint proxy statement/prospectus (when available) and other filings containing information about Sonus at the SEC's website at www.sec.gov. The joint proxy statement/prospectus (when available) and the other filings may also be obtained free of charge from Sonus' Investor Relations website (http://investors.sonusnet.com/) or by requesting them from Sonus' corporate secretary at Sonus Networks, Inc., 4 Technology Park Drive, Westford, Massachusetts 01886, Attention: Corporate Secretary.

Sonus, NewCo, the GENBAND Parties and certain of their respective directors and executive officers, under the SEC's rules, may be deemed to be participants in the solicitation of proxies of Sonus stockholders in connection with the proposed transaction. Information about the directors and executive officers of Sonus and their ownership of Sonus common stock is set forth in the proxy statement for Sonus' 2017 annual meeting of stockholders, as filed with the SEC on Schedule 14A on April 28, 2017.

Additional information regarding the interests of those participants and other persons who may be deemed participants in the transaction may be obtained by reading the joint proxy statement/prospectus regarding the proposed transaction when it becomes available. Free copies of this document may be obtained as described in the preceding paragraphs.

#### NO OFFERS OR SOLICITATIONS

This communication shall not constitute an offer to sell or the solicitation of an offer to sell or the solicitation of an offer to buy any securities, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offer of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the Securities Act of 1933, as amended.

#### CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS

This communication contains "forward-looking statements"—that is, statements related to future, not past, events. In this context, forward-looking statements often address expected future business and financial performance and financial condition, and often contain words such as "expect," "anticipate," "intend," "plan," "believe," "seek," "see," "will," "would," or "target." Forward-looking statements by their nature address matters that are, to different degrees, uncertain, such as statements about the consummation of the proposed transaction and the expected performance of the combined companies. Uncertainties that could cause actual results to be materially different than those expressed in Sonus' and the GENBAND Parties' forward-looking statements include the failure to consummate the proposed transaction or to make or take any filing or other action required to consummate such transaction in a timely matter or at all. These or other uncertainties may cause actual future results to be materially different from those expressed in Sonus' and the GENBAND Parties' forward-looking statements. These risks and uncertainties include, but are not limited to, economic, competitive, legal, governmental and technological factors. Accordingly, there is no assurance that the expectations of Sonus or any GENBAND Party will be realized. Many factors could cause actual results to differ materially from these forward-looking statements with respect to the proposed transaction, including risks relating to the completion of the proposed transaction on anticipated terms and timing, including obtaining equity holder and regulatory approvals, anticipated tax treatment, unforeseen liabilities, future capital expenditures, revenues, expenses, earnings, synergies, economic performance, indebtedness, financial condition, losses, future prospects, business and management strategies for the management, expansion and growth of the new combined company's operations, the ability of Sonus and the GENBAND Parties to integrate the business successfully and to achieve anticipated synergies, potential litigation relating to the proposed transaction, and the risk that disruptions from the proposed transaction will harm Sonus' or the GENBAND Parties' business. While the list of factors presented here is considered representative, no such list should be considered to be a complete statement of all potential risks and uncertainties. Unlisted factors may present significant additional obstacles to the realization of forward looking statements. Consequences of material differences in results as compared with those anticipated in the forward-looking statements could

include, among other things, business disruption, operational problems, financial loss, legal liability to third parties and similar risks, any of which could have a material adverse effect on Sonus' or the GENBAND Parties' consolidated financial condition, results of operations or liquidity. Neither Sonus nor any GENBAND Party assumes any obligation to provide revisions to any forward-looking statements should circumstances change, except as otherwise required by securities and other applicable laws.





#### Merger FAQs for Sales, Marketing, and PLM

The merger will be subject to approval from the US Government from an anti-trust perspective, and subject to the approval of the shareholders of both companies. Even though the merger has been announced, Sonus and Genband remain separate entities—each subject to the US antitrust laws—until the merger officially closes and *each company must continue to do business as usual*. There should be no exchange of competitively sensitive information (including but not limited to prices, costs, sales strategies, roadmaps, or other customer-specific or product-specific information) between Sonus and Genband. Further, Sonus and Genband may not represent themselves as a single entity, especially to current or potential customers or partners.

Violation of these principles can have serious negative consequences for both companies, for the merger, and potentially for yourself. Please discuss with your supervisor or with company counsel whenever you are in doubt about the scope of these do's and don'ts.

It will be permissible to share certain information about post-merger plans and product support with customers, pursuant to approved talking points and guidelines that will be distributed in the coming weeks.

#### Communicating with your peers at the other company:

- 1. Can I reach out to my peers on the sales, marketing, or PLM teams at the other company to talk about accounts, GoTo Market, product planning, or roadmaps?
  - No. Continue to do business as usual until the merger closes. Since you would not have contacted your peers at the other company absent the merger, you should not contact them prior to close.
- 2. Can I talk to my peers if we don't discuss topics such as accounts, GTM, roadmaps?
  - To avoid any appearance of impropriety, you should not contact your peers at the other company until the merger closes, unless you have a reason to do so in the ordinary course of business and the contact has been approved by counsel.
- 3. What if my peers at the other company contact me?
  - The same guidelines from directly above apply.
- 4. Can I talk to people in the other company who are not in sales?
  - The guidelines detailed above apply, whether the other employee is in sales or serves in another capacity.
- 5. Can I accept a LinkedIn invitation from someone at the other company?
  - Yes. Just be sure to observe the guidelines above regarding communications. You should follow the Sonus or Genband corporate policy regarding social media. You should avoid making statements about the merger itself or its perceived effects on the market on social media. "Likes" and sharing of articles may be done if consistent with each company's social media policy.

# SONUS/GENBAND CONFIDENTIAL

1

#### **Communicating with customers:**

- 6. If both Genband and Sonus are bidding on the same project at the same customer, what can I say and not say to the customer?
  - You should inform the customer that the Sonus and Genband will conduct business as usual until the merger closes. Until then, Sonus and Genband are separate entities running separate businesses. We will provide additional detail and talking points about post-closing product support and other information that can be provided to customers about the merger.
- 7. If we are in a competitive situation with the other company at an account, should we back off? Alternately should we get more aggressive?
  - Until the merger closes, you should continue to do business as usual. You should not back off, and you should not get more aggressive either. The business imperative that each company should be a standalone entity, as communicated by your supervisor, should dictate how you respond to the competitive situation.
- 8. When will the restrictions on communications be removed?
  - When the merger officially closes. For now, the merger is still in a "proposed" state, and cannot close until the required regulatory and shareholder approvals are received.

# Integration of the sales, marketing, and PLM teams:

9. Do our comp plans or account assignments change?

For the remainder of 2017 it is expected we will stick with the existing comp plans and quotas. We will unite under one comp plan starting Jan 1, 2018. The new comp plan will take the best of each company's plans and merge them together.

10. Will we have one combined global sales meeting?

Yes, that is the plan, meeting to occur in early Q1 2018, details will be forthcoming

11. Do the companies use the same back office systems?

Both companies use SFDC. We plan to combine our SFDC instances together and go live with a united SFDC by Jan 1, 2018. There might be two separate SFDC quote configurators until we can merge them together sometime later in 2018.

Both companies also use Oracle and the plan is those instances will be merged. Same for all the installed base data. Timing TBD.

We plan to move to one system for T&E and one system commissions by Jan 1, 2018.

We will be bringing in a lot of external consultants who are experts in system consolidation to minimize the integration time.

12. After the merger closes, will all sales people sell the entire product portfolio?

We will evaluate the most optimal sales structure over the coming months and be ready with a plan when the merger is complete.

2

13. Should each company start learning about the other company's products now?

You are welcome to view publicly-available information about the other company's products. Do not seek information directly from employees at the other company. Sonus and Genband will make training materials available at the appropriate time.

14. Where does each company get their revenue from?

As an approximation, Genband generates about 40% of its revenue from maintenance, 17% from professional services, 30% from Softswitch/Media Gateways, 7% from SBC, 6% from UC/Kandy/other.

As an approximation, Sonus generates about 35% of its revenue from maintenance, 9% from professional services (including RENO & training), 41% of revenue from SBC, and 15% from trunking.

15. Anything else I need to avoid?

No joint sales calls, no joint marketing, no joint product roadmaps

Do not create your own messaging about the merger or modify approved messaging

Do not speculate or offer opinions, especially in regards to questions from customers or partners such as product roadmaps, end of life announcements, or work force integration

The competition will likely spread FUD about this merger. If you hear any of this, bring those messages back to the sales/marketing leadership instead of addressing them directly.

Do not assume the deal will close.

# IMPORTANT INFORMATION ABOUT THE TRANSACTION WILL BE FILED WITH THE SEC

In connection with the proposed transaction, Sonus will cause Solstice Sapphire Investments, Inc., a wholly-owned subsidiary of Sonus formed to act as a holding company in connection with the transaction ("NewCo"), to file with the SEC a Registration Statement on Form S-4 that will include a joint proxy statement of Sonus and GENBAND and certain of its affiliates (the "GENBAND Parties") and a prospectus of NewCo and Sonus and the parties may file with the SEC other relevant documents concerning the proposed transaction. Sonus will mail the joint proxy statement/prospectus to the Sonus stockholders and the GENBAND Party equity holders. SONUS STOCKHOLDERS AND GENBAND PARTY EQUITY HOLDERS ARE URGED TO READ THE REGISTRATION STATEMENT AND JOINT PROXY STATEMENT/PROSPECTUS REGARDING THE PROPOSED TRANSACTION WHEN IT BECOMES AVAILABLE AND ANY OTHER RELEVANT DOCUMENTS FILED WITH THE SEC, AS WELL AS ANY AMENDMENTS OR SUPPLEMENTS TO THOSE DOCUMENTS, BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION. Investors and security holders may obtain a free copy of the joint proxy statement/prospectus (when available) and other filings containing information about Sonus at the SEC's website at www.sec.gov. The joint proxy statement/prospectus (when available) and the other filings may also be obtained free of charge from Sonus' Investor Relations website (http://investors.sonusnet.com/) or by requesting them from Sonus' corporate secretary at Sonus Networks, Inc., 4 Technology Park Drive, Westford, Massachusetts 01886, Attention: Corporate Secretary.

Sonus, NewCo, the GENBAND Parties and certain of their respective directors and executive officers, under the SEC's rules, may be deemed to be participants in the solicitation of proxies of Sonus stockholders in connection with the proposed transaction. Information about the directors and executive

officers of Sonus and their ownership of Sonus common stock is set forth in the proxy statement for Sonus' 2017 annual meeting of stockholders, as filed with the SEC on Schedule 14A on April 28, 2017.

Additional information regarding the interests of those participants and other persons who may be deemed participants in the transaction may be obtained by reading the joint proxy statement/prospectus regarding the proposed transaction when it becomes available. Free copies of this document may be obtained as described in the preceding paragraphs.

#### NO OFFERS OR SOLICITATIONS

This communication shall not constitute an offer to sell or the solicitation of an offer to sell or the solicitation of an offer to buy any securities, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offer of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the Securities Act of 1933, as amended.

#### CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS

This communication contains "forward-looking statements"—that is, statements related to future, not past, events. In this context, forward-looking statements often address expected future business and financial performance and financial condition, and often contain words such as "expect," "anticipate," "intend," "plan," "believe," "seek," "see," "will," "would," or "target." Forward-looking statements by their nature address matters that are, to different degrees, uncertain, such as statements about the consummation of the proposed transaction and the expected performance of the combined companies. Uncertainties that could cause actual results to be materially different than those expressed in Sonus' and the GENBAND Parties' forward-looking statements include the failure to consummate the proposed transaction or to make or take any filing or other action required to consummate such transaction in a timely matter or at all. These or other uncertainties may cause actual future results to be materially different from those expressed in Sonus' and the GENBAND Parties' forward-looking statements. These risks and uncertainties include, but are not limited to, economic, competitive, legal, governmental and technological factors. Accordingly, there is no assurance that the expectations of Sonus or any GENBAND Party will be realized. Many factors could cause actual results to differ materially from these forward-looking statements with respect to the proposed transaction, including risks relating to the completion of the proposed transaction on anticipated terms and timing, including obtaining equity holder and regulatory approvals, anticipated tax treatment, unforeseen liabilities, future capital expenditures, revenues, expenses, earnings, synergies, economic performance, indebtedness, financial condition, losses, future prospects, business and management strategies for the management, expansion and growth of the new combined company's operations, the ability of Sonus and the GENBAND Parties to integrate the business successfully and to achieve anticipated synergies, potential litigation relating to the proposed transaction, and the risk that disruptions from the proposed transaction will harm Sonus' or the GENBAND Parties' business. While the list of factors presented here is considered representative, no such list should be considered to be a complete statement of all potential risks and uncertainties. Unlisted factors may present significant additional obstacles to the realization of forward looking statements. Consequences of material differences in results as compared with those anticipated in the forward-looking statements could include, among other things, business disruption, operational problems, financial loss, legal liability to third parties and similar risks, any of which could have a material adverse effect on Sonus' or the GENBAND Parties' consolidated financial condition. results of operations or liquidity. Neither Sonus nor any GENBAND Party assumes any obligation to provide revisions to any forward-looking statements should circumstances change, except as otherwise required by securities and other applicable laws.